# **UNITED STATES**

# SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# **FORM 10-Q**

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

(Mark One)

For the quarterly period ended June 30, 2025			
		OR	
☐ TRANSITION REPORT PURSUANT TO S	SECTION 13 OI	R 15(d) OF THE SECURITIES EXCHANG	E ACT OF 1934
For the transition period from to			
		Commission file number 001-04321	
	For	ge Global Holdings, Inc	
		me of registrant as specified in its cha	
Delaware			99-4383083
(State or other jurisdiction of incorporation o	r organization)		(I.R.S. Employer Identification No.)
	(Address	4 Embarcadero Center Floor 15 San Francisco, CA 94111 of principal executive offices, including zip co	de)
	(11441455	(415) 881-1612	
	(Regis	strant's telephone number, including area code)	
securities registered pursuant to Section 12(b) of the Act	, -	······································	
Title of each class		Trading Symbol(s)	Name of each exchange on which registered
Common stock, par value \$0.0001 per sha	are	FRGE	The New York Stock Exchange
or for such shorter period that the registrant was required indicate by check mark whether the registrant has submit hapter) during the preceding 12 months (or for such shows No	ted electronically	orts); and (2) has been subject to such filing recovery every Interactive Data File required to be submet registrant was required to submit such files.	mitted pursuant to Rule 405 of Regulation S-T (§232.405 of this
efinitions of "large accelerated filer," "accelerated filer,			naller reporting company, or an emerging growth company. See the "" in Rule 12b-2 of the Exchange Act:
Large accelerated filer		Accelerated filer	
Non-accelerated filer	$\boxtimes$	Smaller reporting company	
		Emerging growth company	
f an emerging growth company, indicate by check mark tandards provided pursuant to Section 13(a) of the Exch		has elected not to use the extended transition pe	eriod for complying with any new or revised financial accounting
ndicate by check mark whether the registrant is a shell c	ompany (as defir	ned in Rule 12b-2 of the Exchange Act). Yes	□ No ⊠
as of August 6, 2025, the registrant had 13,601,362 share	es of common sto	ock, \$0.0001 par value per share, outstanding, v	which reflects the registrant's 1-for-15 reverse stock split that

### CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

Unless the context otherwise requires, references in this Quarterly Report on Form 10-Q (this "Report") to "Forge," the "Company," "us," "we," "our," and any related terms are intended to mean Forge Global Holdings, Inc. and its consolidated subsidiaries.

Certain statements in this Report may constitute "forward-looking statements" for purposes of the federal securities laws. Our forward-looking statements include, but are not limited to, statements regarding our or our management team's expectations, hopes, beliefs, intentions, or strategies regarding the future. In addition, any statements that refer to projections, forecasts, or other characterizations of future events or circumstances, including any underlying assumptions, are forward-looking statements. The words "anticipate," "believe," "contemplate," "continue," "could," "estimate," "expect," "intends," "may," "might," "plan," "possible," "potential," "predict," "project," "should," "will," "would," and similar expressions may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking. Forward-looking statements in this Report may include, for example, statements about our ability to:

- effectively respond to general macroeconomic and business conditions;
- execute our business strategy, including monetization of services provided;
- anticipate the uncertainties inherent in the development of new business lines, strategies, products, and services;
- anticipate rapid technological changes and competitive threats;
- respond to uncertainties associated with product and service development and market acceptance;
- increase brand awareness;
- attract, train, and retain effective officers, employees, directors, and other key personnel;
- acquire, develop, and protect intellectual property;
- maintain key strategic relationships with partners;
- anticipate the significance and timing of contractual obligations;
- enhance future operating and financial results;
- integrate, operate, and manage any acquired businesses;
- respond to fluctuations in interest rates and foreign currency exchange rates;
- finance operations on an economically viable basis;
- meet future capital adequacy and liquidity requirements;
- obtain additional capital, including use of the debt market;
- comply with laws and regulations applicable to our business;
- stay abreast of modified or new laws and regulations that would apply to our business;
- · manage cybersecurity and technology risk management processes, including incident management processes;
- upgrade and maintain information technology systems;
- maintain disaster recovery and business continuity planning controls;
- manage vendor and third party processes;
- adequately support data governance and data privacy controls related to personal information and consumer data;
- maintain the listing of our securities on the NYSE or another national securities exchange;
- anticipate the impact of, and response to, new accounting standards;
- anticipate the impact of new tax laws that would apply to our business; and
- successfully defend litigation.

We caution you that the foregoing list may not contain all of the forward-looking statements made in this Report.

You should not rely upon forward-looking statements as predictions of future events. We have based the forward-looking statements contained in this Report primarily on our current expectations and projections about future events and trends that we believe may affect our business, financial condition, results of operations, and prospects. The outcome of the events described in these forward-looking statements is subject to risks, uncertainties, and other factors, including those described in the section titled "Risk Factors" and elsewhere in this Report. Moreover, we operate in a very competitive and rapidly changing environment. New risks and uncertainties emerge from time to time and it is not possible for us to predict all risks and uncertainties that could have an impact on the forward-looking statements contained in this Report. We cannot assure you that the results, events, and circumstances reflected in the forward-looking statements will be achieved or occur, and actual results, events, or circumstances could differ materially from those described in the forward-looking statements.

Neither we nor any other person assumes responsibility for the accuracy and completeness of any of these forward-looking statements. Moreover, the forward-looking statements made in this Report relate only to events as of the date on which the statements are made. We undertake no obligation to update any after the date of this Report or to reflect new information or the occurrence of unanticipated events, except as required by law. We may not actually achieve the plans, intentions, or

expectations disclosed in our forward-looking statements and you should not place undue reliance on our forward-looking statements. Our forward-looking statements do not reflect the potential impact of any future acquisitions, partnerships, mergers, dispositions, joint ventures, or investments we may make.

In addition, statements that "we believe" and similar statements reflect our beliefs and opinions on the relevant subject. These statements are based upon information available to us as of the date of this Report, and while we believe such information forms a reasonable basis for such statements, such information may be limited or incomplete, and our statements should not be read to indicate that we have conducted an exhaustive inquiry into, or review of, all potentially available relevant information. These statements are inherently uncertain and investors are cautioned not to unduly rely upon these statements.

# **Table of Contents**

	Page
Part I - Financial Information	
Item 1. Condensed Consolidated Financial Statements (Unaudited)	3
Notes to Condensed Consolidated Financial Statements (Unaudited)	11
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	26
Item 3. Quantitative and Qualitative Disclosures About Market Risk	37
Item 4. Controls and Procedures	38
Part II - Other Information	
Item 1. Legal Proceedings	40
Item 1A. Risk Factors	40
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	40
Item 3. Defaults Upon Senior Securities	40
Item 4. Mine Safety Disclosures	40
Item 5. Other Information	40
Item 6. Exhibits	40
<u>Signatures</u>	42
Part I- Financial Information	
3	

# FORGE GLOBAL HOLDINGS, INC. Unaudited Condensed Consolidated Balance Sheets (In thousands, except per share data)

Goodwill and other intangible assets, net         126,055         126,456           Operating lease right-of-use assets         3,985         5,107           Payment-dependent notes receivable         9,604         7,412           Other assets, noncurrent         1,664         2,444           Total assets         2 24,285         2 263,506           Liabilities and stockbolders' equity         8         24,249         1,941           Accounts payable         2,744         1,941           Accrued compensation and benefits         13,600         13,430           Accrued expenses and other current liabilities         6,765         6,310           Operating lease liabilities, current         2,032         3,463           Operating lease liabilities, noncurrent         3,231         3,694           Payment-dependent notes payable         9,604         7,412           Warrant liabilities         2,962         1,964           Warrant liabilities, noncurrent         3,231         3,694           Payment-dependent notes payable         9,604         7,412           Warrant liabilities         2,962         3,231         3,694           Other liabilities, noncurrent         3,23         3,23         3,23         3,23         3,23 <td< th=""><th></th><th></th><th>June 30, 2025</th><th colspan="3">December 31, 2024</th></td<>			June 30, 2025	December 31, 2024		
Cash and cash equivalents         \$ 4,310         \$ 1,118           Restricted cash         1,138         1,116           Accounts receivable, net of allowances of \$1.5 million and \$1.3 million, respectively         8,119         4,706           Prepaid expenses and other current assets         10,00         8,205           Investments         2,930         19,107           Internal-uses offware, property and equipment, net         15,05         19,000           Goodwill and other intangible assets, net         126,05         1,005           Operating lease right-of-use assets, net         9,604         7,412           Operating lease labelities assets, net         9,604         7,412           Total assets         2,604         9,604         7,412           Total assets         2,744         1,941         1,941         1,941         1,941         1,941         1,941         1,941         1,941         1,941         1,941         1,941         1,941 </th <th>Assets</th> <th></th> <th></th> <th></th>	Assets					
Restricted cain         1,18         1,116           Accounts receivable, not allowances of \$1.5 million and \$1.3 million, respectively         8,119         4,000           Prepaid expenses and other current assets         10,000         8,205           Investmens         2,030         1-10,100           Total current assets         1,557         2,920           Goodwill and other intangible assets, seed         1,558         2,930         1,510           Operating lease right-Gues assets         1,500         1,500         2,010           Opperating lease right-Gues assets         9,000         1,512         2,010           Opperating lease right-Gues assets         1,500         2,012         2,012         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014         2,014	Current assets:					
Accounts receivable, net of allowances of \$1.5 million and \$1.3 million, respectively Prepaid expenses and other current assets Investments Investment	Cash and cash equivalents	\$	54,310	\$ 105,140		
Prepaid expenses and other current assets         10,002         8,205           Investments         25,333         1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-	Restricted cash		1,138	1,116		
Investments         26,39         9,908         1,916           Total current assets         9,908         1,916         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,926         2,92	Accounts receivable, net of allowances of \$1.5 million and \$1.3 million, respectively		8,119	4,706		
Total current assets         \$ 99,980         \$ 19,167           Internal-use software, property and equipment, net         15,57         2,920           Goodwill and other intangible assets, net         126,055         126,456           Operating lease right-of-use assets         9,804         7,412           Payment-dependent notes receivable         1,604         2,244           Total assets         1,604         2,244           Total tassets         2,744         1,94           Tabilities and stockholders' equity         2,744         1,94           Accounts payable         2,744         1,94           Accounts payable         3,605         2,744         1,94           Accrued expenses and other current liabilities         3,605         3,603         3,603         3,603           Operating lease liabilities, current         3,203         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604         3,604	Prepaid expenses and other current assets		10,020	8,205		
Internal-use software, properly and equipment, net         1,557         2,920           Goodwill and other intangible assets, net         126,55         126,456           Operating lease right-of-use assets         3,985         5,107           Payment-dependent notes receivable         9,044         7,412           Other assets, noncurrent         166         2,444           Total assets         2,248.05         20,356           Liabilities and stockholder's equity         2,744         1,941           Accrued asset and other current liabilities         13,600         13,430           Accrued compensation and benefits         13,600         3,343           Accrued expenses and other current liabilities         6,765         6,310           Operating lease liabilities, current         3,23         3,694           Payment-dependent notes payable         9,204         9,204           Operating lease liabilities, noncurrent         3,23         3,694           Payment-dependent notes payable         9,00         9,00           Warrant liabilities         3,80         9,00         9,00           Other liabilities, noncurrent         3,20         3,60           Other liabilities, noncurrent         8,36         3,0         3,60           Oth	Investments		26,393	_		
Godwill and other intangible assets, net         126,055         126,456           Operating lease right-of-use assets         3,985         5,107           Payment-dependent notes recivable         9,604         7,412           Other assets, noncurrent         1,606         2,444           Total assets         1,607         2,243         2,243           Ishilities and stockholders'equity         8         22,744         1,941           Accounts payable         2,744         1,941           Accrued compensation and benefits         6,765         6,310           Accrued expenses and other current liabilities         6,765         6,310           Operating lease liabilities, current         9,032         3,403           Total current liabilities         9,042         7,412           Operating lease liabilities, noncurrent         9,042         7,412           Operating lease liabilities, noncurrent         9,064         7,412           Command to entripe payable         9,064         7,412           Operating lease liabilities, noncurrent         9,064         3,042           Total labilities         9,084         9,042         3,042           Total current liabilities         1,062         3,042         3,042           T	Total current assets	\$	99,980	\$ 119,167		
Operating lease right-of-use assets         3,057         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000         7,000	Internal-use software, property and equipment, net		1,557	2,920		
Payment-dependent notes receivable         9,044         7,412           Other ases, noncurrent         1,606         2,424         2,424           Total assets         5         242,825         263,506           Libilities and stockholders' equity         20,244         1,908           Chronic spayable         2,744         1,904           Accounds payable         2,744         1,914           Accured compensation and benefits         1,606         6,310           Operating lesse liabilities, current         2,024         1,616           Operating lesse liabilities, nonurrent         3,231         3,636           Operating lesse liabilities, nonurrent         9,604         7,412           Quarrent liabilities         9,604         7,412           Quarrent liabilities         9,604         7,412           Quarrent liabilities, noncurrent         9,604         7,412           Quarrent liabilities, noncurrent         9,604         7,412           Quarrent liabilities         1,904         7,412           Quarrent liabilities         1,904         7,412           Quarrent liabilities         1,904         7,412           Quarrent liabilities         1,904         7,412           Quarrent liabilit	Goodwill and other intangible assets, net		126,055	126,456		
Other assets, noncurrent         1,644         2,444           Total assets         2,243,809         2,630,600           Lishilities         2,243         3,243,200           Current liabilities         2,244         1,941           Accounts payable         2,134         1,941           Accounde compensation and benefits         13,600         1,343           Accounde compensation and benefits         6,765         6,310           Operating lease liabilities, current         2,023         3,643           Operating lease liabilities, noncurrent         3,231         3,604           Operating lease liabilities, noncurrent         3,231         3,604           Operating lease liabilities, noncurrent         9,604         3,204           Operating lease liabilities, noncurrent         3,231         3,604           Operating lease liabilities, noncurrent         3,201         3,604           Appendiculation of the contract of table interest of the contract of table interest of table intere	Operating lease right-of-use assets		3,985	5,107		
Total sisted to Mischilders' equity         Value of the Mischilders	Payment-dependent notes receivable		9,604	7,412		
Current liabilities and stockholders' equity   Current liabilities   Current liabiliti	Other assets, noncurrent		1,664	2,444		
Current liabilities:         2,744         1,941           Accounts payable         13,600         13,400           Accrued compensation and benefits         13,600         13,400           Accrued expenses and other current liabilities         6,765         6,310           Operating lease liabilities, current         2,032         3,603           Total current liabilities         3,214         5,214           Operating lease liabilities, noncurrent         3,201         3,604           Payment-dependent notes payable         9,604         7,412           Warrant liabilities         296         192           Other liabilities, noncurrent         296         192           Other liabilities, noncurrent         329         36,044           Warrant liabilities         38,04         3,040           Other liabilities         38,04         3,040           Other liabilities and cuttingencies (Note 6)         38,04         3,040           Total liabilities         5,000         3,040         3,040           Other liabilities and cuttingencies (Note 6)         5         3,040         3,040           Total liabilities and cuttingencies (Note 6)         5         3,040         3,040         3,040           Total pain liabilities an	Total assets	\$	242,845	\$ 263,506		
Accounts payable         2,744         1,941           Accrued compensation and benefits         13,600         13,430           Accrued expenses and other current liabilities         6,765         6,310           Operating lease liabilities, current         2,032         3,463           Total current liabilities         8,25,141         9,25,144           Operating lease liabilities, noncurrent         3,231         3,694           Payment-dependent notes payable         9,604         7,412           Warrant liabilities         296         1922           Other liabilities, noncurrent         329         3,231           Other liabilities, noncurrent         329         3,202           Total liabilities         8,360         3,364           Payment dependent notes payable         329         3,22           Other liabilities, noncurrent         329         3,22           Total liabilities         8,360         3,364           Payment dependent notes payable         3,20         3,22           Total liabilities, noncurrent         329         3,22           Total liabilities, noncurrent         3,20         3,26           Total liabilities, noncurrent         3,20         3,26           Total liabilities, noncurren	ž. V					
Accrued compensation and benefits         13,600         13,430           Accrued expenses and other current liabilities         6,765         6,310           Operating lease liabilities, current         2,032         3,463           Total current liabilities         \$ 25,141         \$ 25,144           Operating lease liabilities, noncurrent         3,231         3,694           Payment-dependent notes payable         9,604         7,412           Warrant liabilities         296         192           Other liabilities, noncurrent         329         329           Other liabilities, noncurrent         38,001         \$ 36,004           Warrant liabilities         296         192           Other liabilities, noncurrent         329         322           Other liabilities         296         192           Other liabilities, noncurrent         38,001         \$ 36,004           Other liabilities         38,001         \$ 36,004           Other liabilities         38,001         \$ 36,004           Other liabilities         38,000         \$ 36,004         \$ 36,004           Commitments and contingencies (Note 6)         1         1         1         1         1         1         1         1         1         1<	Current liabilities:					
Accrued expenses and other current liabilities         6,765         6,310           Operating lease liabilities, current         2,032         3,463           Total current liabilities         \$ 25,144         \$ 25,144           Operating lease liabilities, noncurrent         3,231         3,694           Payment-dependent notes payable         9,604         7,412           Warrant liabilities         296         192           Other liabilities, noncurrent         329         322           Total liabilities         \$ 38,01         \$ 36,764           Commitments and contingencies (Note 6)         329         322           Stockholders' equity:         7         2           Preferred stock, \$0,0001 par value; 10,000 shares authorized; 12,411 and 12,427 shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively         1         1           Treasury stock, at cost, 10 shares as of both June 30, 2025 and December 31, 2024, respectively         575,676         570,606           Accumulated other comprehensive income         1,193         572           Accumulated deficit         3,723         3,469           Total Forge Global Holdings, Inc. stockholders' equity         5         200,521         3,249,249           Noncontrolling Interest         3,223         3,160         3,242 <td>Accounts payable</td> <td></td> <td>2,744</td> <td>1,941</td>	Accounts payable		2,744	1,941		
Operating lease liabilities, current         2,032         3,463           Total current liabilities         5,514         2,514           Operating lease liabilities, noncurrent         3,231         3,694           Payment-dependent notes payable         9,604         7,412           Warrant liabilities         296         192           Other liabilities, noncurrent         39,202         3,604           Other liabilities, noncurrent         39,002         3,604           Commitments and contingencies (Note 6)         3,800         3,604           Stockholders' equity:         7         4           Preferred stock, \$0,0001 par value; 100,000 shares authorized; no shares issued and outstanding         5         5         3,604           Common stock, \$0,0001 par value; 133,333 shares authorized; 12,411 and 12,427 shares issued and outstanding as of Jun. 30,2025 and December 31, 2024, respectively         1         1         1           Treasury stock, at cost; 10 shares as of both June 30, 2025 and December 31, 2024, respectively         575,676         570,606           Accumulated other comprehensive income         1,193         572,606         570,606           Accumulated deficit         2,005,212         3,203,802         3,406,702         3,406,702           Total Forge Global Holdings, Inc. stockholders' equity         2,00	Accrued compensation and benefits		13,600	13,430		
Total current liabilities         25,144         3,25,144           Operating lease liabilities, noncurrent         3,231         3,694           Payment-dependent notes payable         9,604         7,412           Warrant liabilities         296         192           Other liabilities, noncurrent         329         322           Total liabilities         3 8,601         3 8,601         3 6,764           Commitments and contingencies (Note 6)         5         38,601         3 6,764           Commitments and contingencies (Note 6)         5         3,702         5           Stockholders' equity:         -         -         -           Preferred stock, \$0,0001 par value; 133,333 shares authorized; 12,411 and 12,427 shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         1         2         2         2         2<	Accrued expenses and other current liabilities		6,765	6,310		
Operating lease liabilities, noncurrent         3,231         3,694           Payment-dependent notes payable         9,604         7,412           Warrant liabilities         296         192           Other liabilities, noncurrent         329         322           Other liabilities, noncurrent         3 38,601         3 36,764           Committents and contingencies (Note 6)         38,601         3 36,764           Committents and contingencies (Note 6)         5         5         5           Stockholders' equity:         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -         -	Operating lease liabilities, current		2,032	3,463		
Payment-dependent notes payable         9,604         7,412           Warrant liabilities         296         192           Other liabilities, noncurrent         329         322           Total liabilities         \$ 38,601         \$ 36,764           Commitments and contingencies (Note 6)         5         38,601         \$ 36,764           Stockholders' equity:         Freferred stock, \$0,0001 par value; 100,000 shares authorized; no shares issued and outstanding         —         —         —           Common stock, \$0,0001 par value; 133,333 shares authorized; 12,411 and 12,427 shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively         (625)         (625)           Treasury stock, at cost; 10 shares as of both June 30, 2025 and December 31, 2024, respectively         575,676         570,606           Accumulated other comprehensive income         1,193         572           Accumulated deficit         (375,724)         (346,972)           Total Forge Global Holdings, Inc. stockholders' equity         \$ 200,521         \$ 223,582           Noncontrolling Interest         3,723         3,160           Total stockholders' equity         \$ 204,244         \$ 226,742	Total current liabilities	\$	25,141	\$ 25,144		
Warrant liabilities         296         192           Other liabilities, noncurrent         329         322           Total liabilities         \$ 38,001         \$ 36,764           Commitments and contingencies (Note 6)         Stockholders' equity:         Stockholders' equity:         \$ 2         \$ 2           Preferred stock, \$0.0001 par value; 100,000 shares authorized; no shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively         \$ 2         \$ 2           Common stock, \$0.0001 par value; 133,333 shares authorized; 12,411 and 12,427 shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively         \$ 1         \$ 1           Treasury stock, at cost; 10 shares as of both June 30, 2025 and December 31, 2024, respectively         \$ 575,676         \$ 570,606           Accumulated other comprehensive income         \$ 1,193         \$ 572           Accumulated deficit         \$ 200,521         \$ 223,582           Total Forge Global Holdings, Inc. stockholders' equity         \$ 200,521         \$ 223,582           Noncontrolling Interest         \$ 204,244         \$ 226,742	Operating lease liabilities, noncurrent		3,231	3,694		
Other liabilities, noncurrent         329         322           Total liabilities         \$ 38,601         \$ 36,764           Commitments and contingencies (Note 6)         \$ 38,601         \$ 36,764           Stockholders' equity:         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001         \$ 50,0001	Payment-dependent notes payable		9,604	7,412		
Total liabilities         \$ 38,601         \$ 36,764           Commitments and contingencies (Note 6)         \$ 38,601         \$ 36,764           Stockholders' equity:         \$ 50,000         \$ 36,764           Preferred stock, \$0.0001 par value; 100,000 shares authorized; no shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively         \$ 1         1           Common stock, \$0.0001 par value; 133,333 shares authorized; 12,411 and 12,427 shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively         \$ 625         \$ 625           Additional paid-in capital         \$ 575,676         \$ 70,606           Accumulated other comprehensive income         \$ 1,193         \$ 572           Accumulated deficit         (375,724)         (346,972)           Total Forge Global Holdings, Inc. stockholders' equity         \$ 200,521         \$ 223,582           Noncontrolling Interest         3,723         3,160           Total stockholders' equity         \$ 204,244         \$ 226,742	Warrant liabilities		296	192		
Commitments and contingencies (Note 6)         Stockholders' equity:       Preferred stock, \$0.0001 par value; 100,000 shares authorized; no shares issued and outstanding       —       —         Common stock, \$0.0001 par value; 133,333 shares authorized; 12,411 and 12,427 shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively       1       1       1         Treasury stock, at cost; 10 shares as of both June 30, 2025 and December 31, 2024, respectively       (625)       625)         Additional paid-in capital       575,676       570,606         Accumulated other comprehensive income       1,193       572         Accumulated deficit       (375,724)       (346,972)         Total Forge Global Holdings, Inc. stockholders' equity       \$ 200,521       \$ 223,582         Noncontrolling Interest       3,723       3,160         Total stockholders' equity       \$ 204,244       \$ 226,742	Other liabilities, noncurrent		329	322		
Stockholders' equity:           Preferred stock, \$0.0001 par value; 100,000 shares authorized; no shares issued and outstanding         —         —           Common stock, \$0.0001 par value; 133,333 shares authorized; 12,411 and 12,427 shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively         1         1           Treasury stock, at cost; 10 shares as of both June 30, 2025 and December 31, 2024, respectively         (625)         (625)           Additional paid-in capital         575,676         570,606           Accumulated other comprehensive income         1,193         572           Accumulated deficit         (375,724)         (346,972)           Total Forge Global Holdings, Inc. stockholders' equity         \$         200,521         \$         223,582           Noncontrolling Interest         3,723         3,160           Total stockholders' equity         \$         204,244         \$         226,742	Total liabilities	\$	38,601	\$ 36,764		
Preferred stock, \$0.0001 par value; 100,000 shares authorized; no shares issued and outstanding         —         —           Common stock, \$0.0001 par value; 133,333 shares authorized; 12,411 and 12,427 shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively         1         1           Treasury stock, at cost; 10 shares as of both June 30, 2025 and December 31, 2024, respectively         (625)         (625)           Additional paid-in capital         575,676         570,606           Accumulated other comprehensive income         1,193         572           Accumulated deficit         (375,724)         (346,972)           Total Forge Global Holdings, Inc. stockholders' equity         \$ 200,521         \$ 223,582           Noncontrolling Interest         3,723         3,160           Total stockholders' equity         \$ 204,244         \$ 226,742	Commitments and contingencies (Note 6)					
Common stock, \$0.0001 par value; 133,333 shares authorized; 12,411 and 12,427 shares issued and outstanding as of June 30, 2025 and December 31, 2024, respectively       1       1         Treasury stock, at cost; 10 shares as of both June 30, 2025 and December 31, 2024, respectively       (625)       (625)         Additional paid-in capital       575,676       570,606         Accumulated other comprehensive income       1,193       572         Accumulated deficit       (375,724)       (346,972)         Total Forge Global Holdings, Inc. stockholders' equity       \$ 200,521       \$ 223,582         Noncontrolling Interest       3,723       3,160         Total stockholders' equity       \$ 204,244       \$ 226,742	Stockholders' equity:					
30, 2025 and December 31, 2024, respectively       1       1         Treasury stock, at cost; 10 shares as of both June 30, 2025 and December 31, 2024, respectively       (625)       (625)         Additional paid-in capital       575,676       570,606         Accumulated other comprehensive income       1,193       572         Accumulated deficit       (375,724)       (346,972)         Total Forge Global Holdings, Inc. stockholders' equity       \$ 200,521       \$ 223,582         Noncontrolling Interest       3,723       3,160         Total stockholders' equity       \$ 204,244       \$ 226,742	Preferred stock, \$0.0001 par value; 100,000 shares authorized; no shares issued and outstanding		_	_		
Additional paid-in capital         575,676         570,606           Accumulated other comprehensive income         1,193         572           Accumulated deficit         (375,724)         (346,972)           Total Forge Global Holdings, Inc. stockholders' equity         \$ 200,521         \$ 223,582           Noncontrolling Interest         3,723         3,160           Total stockholders' equity         \$ 204,244         \$ 226,742		;	1	1		
Accumulated other comprehensive income         1,193         572           Accumulated deficit         (375,724)         (346,972)           Total Forge Global Holdings, Inc. stockholders' equity         \$ 200,521         \$ 223,582           Noncontrolling Interest         3,723         3,160           Total stockholders' equity         \$ 204,244         \$ 226,742	Treasury stock, at cost; 10 shares as of both June 30, 2025 and December 31, 2024, respectively		(625)	(625)		
Accumulated deficit         (375,724)         (346,972)           Total Forge Global Holdings, Inc. stockholders' equity         \$ 200,521         \$ 223,582           Noncontrolling Interest         3,723         3,160           Total stockholders' equity         \$ 204,244         \$ 226,742	Additional paid-in capital		575,676	570,606		
Total Forge Global Holdings, Inc. stockholders' equity         \$ 200,521         \$ 223,582           Noncontrolling Interest         3,723         3,160           Total stockholders' equity         \$ 204,244         \$ 226,742	Accumulated other comprehensive income		1,193	572		
Noncontrolling Interest         3,723         3,160           Total stockholders' equity         \$ 204,244         \$ 226,742	Accumulated deficit		(375,724)	(346,972)		
Total stockholders' equity \$ 204,244 \$ 226,742	Total Forge Global Holdings, Inc. stockholders' equity	\$	200,521	\$ 223,582		
<u> </u>	Noncontrolling Interest		3,723	 3,160		
Total liabilities and stockholders' equity \$ 242,845 \ \$ 263,506	• •	_	- ,	 		
	Total liabilities and stockholders' equity	\$	242,845	\$ 263,506		

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

# FORGE GLOBAL HOLDINGS, INC. Unaudited Condensed Consolidated Statements of Operations (In thousands, except per share data)

	<b>Three Months Ended June 30</b>				Six Months Ended June 30					
		2025		2024		2025		2024		
Revenues:										
Marketplace revenue	\$	18,597	\$	11,679	\$	34,594	\$	20,199		
Custodial administration fees		9,142		10,603		18,441		21,325		
Total revenues	\$	27,739	\$	22,282	\$	53,035	\$	41,524		
Transaction-based expenses:										
Transaction-based expenses		(155)		(256)		(347)		(285)		
Total revenues, less transaction-based expenses	\$	27,584	\$	22,026	\$	52,688	\$	41,239		
Operating expenses:		· · · · · · · · · · · · · · · · · · ·						·		
Compensation and benefits		27,193		28,784		56,684		58,627		
Technology and communications		4,667		2,649		9,016		5,709		
General and administrative		2,144		2,508		4,398		7,570		
Professional services		1,204		1,605		3,536		3,822		
Advertising and market development		1,528		1,243		2,743		2,333		
Acquisition-related transaction costs		1,988		_		1,988		_		
Depreciation and amortization		909		1,781		1,895		3,597		
Rent and occupancy		786		1,107		1,732		2,242		
Total operating expenses	\$	40,419	\$	39,677	\$	81,992	\$	83,900		
Operating loss	\$	(12,835)	\$	(17,651)	\$	(29,304)	\$	(42,661)		
Interest and other income:										
Interest income		803		1,495		1,845		3,204		
Change in fair value of warrant liabilities		(294)		2,280		(103)		6,727		
Other income, net		76		94		130		170		
Total interest and other (expense) income	\$	585	\$	3,869	\$	1,872	\$	10,101		
Loss before provision for income taxes	\$	(12,250)	\$	(13,782)	\$	(27,432)	\$	(32,560)		
Provision for income taxes		189		258		1,205		474		
Net loss	\$	(12,439)	\$	(14,040)	\$	(28,637)	\$	(33,034)		
Net income (loss) attributable to noncontrolling interest		141		(316)		115		(686)		
Net loss attributable to Forge Global Holdings, Inc.	\$	(12,580)	\$	(13,724)	\$	(28,752)	\$	(32,348)		
Net loss per share attributable to Forge Global Holdings, Inc. common stockholders:										
Basic	\$	(1.01)	\$	(1.13)	\$	(2.30)	\$	(2.67)		
Diluted	\$	(1.01)	\$	(1.13)	\$	(2.30)	\$	(2.67)		
Weighted-average shares used in computing net loss per share attributable to Forge Global Holdings, Inc. common stockholders:										
Basic		12,474		12,179		12,503		12,112		
Diluted		12,474		12,179		12,503		12,112		

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

# FORGE GLOBAL HOLDINGS, INC. Unaudited Condensed Consolidated Statements of Comprehensive Loss (In thousands)

	Three Months	Ended June 30,	5	Six Months Ended June 30,			
	 2025 2024			2025	2024		
Net loss	\$ (12,439)	\$ (14,040)	\$	(28,637)	\$	(33,034)	
Foreign currency translation adjustment	 711	(76)		1,069		(331)	
Comprehensive loss	(11,728)	(14,116)		(27,568)		(33,365)	
Less: Comprehensive loss attributable to noncontrolling interest	 453	(348)		563		(827)	
Comprehensive loss attributable to Forge Global Holdings, Inc.	\$ (12,181)	\$ (13,768)	\$	(28,131)	\$	(32,538)	

# FORGE GLOBAL HOLDINGS, INC. Unaudited Condensed Consolidated Statements of Changes in Stockholders' Equity (In thousands)

_		Common Stock		_	Additional	Accumulated Other							
_	Shares <sup>(1)</sup>	Amount <sup>(1)</sup>	Treasury Stock		Paid-In Capital <sup>(1)</sup>	A	ccumulated Deficit	Co	mprehensive Income	No	oncontrolling Interest		Total
Balance as of December 31, 2024	12,427	\$ 1	\$ (625)	\$	570,606	\$	(346,972)	\$	572	\$	3,160	\$	226,742
Issuance of common stock upon release of restricted stock units	270	(*)	_		(*)		_		_		_		_
Tax withholding related to vesting of restricted stock units	(62)	(*)	_		(679)		_		_		_		(679)
Issuance of common stock upon exercise of vested options	3	(*)	_		25		_		_		_		25
Vesting of early exercised stock options and restricted stock awards	_	_	_		18		_		_		_		18
Stock-based compensation expense	_	_	_		6,519		_		_		_		6,519
Net loss	_	_	_		_		(16,172)		_		(26)		(16,198)
Foreign-currency translation adjustment	_				_		_		222		136		358
Balance as of March 31, 2025	12,638	\$ 1	(625)	\$	576,489	\$	(363,144)	\$	794	\$	3,270	\$	216,785
Issuance of common stock upon release of restricted stock units	93	(*)	_		(*)		_		_		_		_
Tax withholding related to vesting of restricted stock units	(12)	(*)	_		(170)		_		_		_		(170)
Issuance of common stock upon exercise of vested options	7	(*)	_		48		_		_		_		48
Share buyback	(315)	(*)	_		(4,139)		_		_		_		(4,139)
Cash in lieu of fractional shares	_	_	_		(4)		_		_		_		(4)
Vesting of early exercised stock options and restricted stock awards	_	_	_		16		_		_		_		16
Stock-based compensation expense	_	_	_		3,436		_		_		_		3,436
Net loss	_	_	_		_		(12,580)		_		141		(12,439)
Foreign-currency translation adjustment	_	_	_		_		_		399		312		711
Balance as of June 30, 2025	12,411	\$ 1	\$ (625)	\$	575,676	\$	(375,724)	\$	1,193	\$	3,723	\$	204,244

		Common Stock		Additional			N d III	
<del>-</del>	Shares(1)	Amount <sup>(1)</sup>	Treasury Stock	Capital <sup>(1)</sup>	Accumulated Deficit	Comprehensive Loss	Noncontrolling Interest	Total
Balance as of December 31, 2023	11,793	\$ 1	\$ (625)	\$ 543,863	\$ (280,638)	\$ 911	\$ 4,906	\$ 268,418
Issuance of common stock upon release of restricted stock units	258	(*)	_	(*)	_	_	_	_
Tax withholding related to vesting of restricted stock units	(72)	(*)	_	(2,302)	_	_	_	(2,302)
Issuance of common stock upon exercise of vested options	21	(*)	_	227	_	_	_	227
Repurchase of early exercised stock options	_	_	_	_	_	_	_	_
Vesting of early exercised stock options and restricted stock awards	_	_	_	36	_	_	_	36
Stock-based compensation expense	_	_	_	9,467	_	_	_	9,467
Net loss	_	_	_	_	(18,624)	_	(370)	(18,994)
Foreign-currency translation adjustment	_	_	_	_	_	(146)	(109)	(255)
Balance as of March 31, 2024	12,001	\$ 1	\$ (625)	\$ 551,291	\$ (299,262)	\$ 765	\$ 4,427	\$ 256,597
Issuance of common stock upon release of restricted stock units	204	(*)	_	(*)	_	_	_	_
Tax withholding related to vesting of restricted stock units	(38)	(*)	_	(1,135)	_	_	_	(1,135)
Issuance of common stock upon exercise of vested options	12	(*)	_	234	_	_	_	234
Repurchase of early exercised stock options	(1)	_	_	_	_	_	_	_
Vesting of early exercised stock options and restricted stock awards	_	(*)	_	35	_	_	_	35
Stock-based compensation expense	_	_	_	7,859	_	_	_	7,859
Net loss	_	_	_	_	(13,724)	_	(316)	(14,040)
Foreign-currency translation adjustment						(44)	(32)	(76)
Balance as of June 30, 2024	12,178	\$ 1	\$ (625)	\$ 558,284	\$ (312,986)	\$ 721	\$ 4,079	\$ 249,474

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

<sup>(\*)</sup> amount less than 1 (1) Amounts have been adjusted to reflect the reverse stock split that became effective on April 14, 2025. Refer to Note 2, "Summary of Significant Accounting Policies" for further information about the reverse stock split.

# FORGE GLOBAL HOLDINGS, INC. Unaudited Condensed Consolidated Statements of Cash Flows (In thousands of U.S. dollars)

	Six Months Ended June 30,				
		2025		2024	
Cash flows from operating activities:					
Net loss	\$	(28,637)	\$	(33,034)	
Adjustments to reconcile net loss to net cash used in operations:					
Share-based compensation		9,955		17,326	
Depreciation and amortization		1,687		3,597	
Amortization of right-of-use assets		1,122		1,305	
Loss on impairment of long lived assets		_		186	
Allowance for doubtful accounts		269		216	
Change in fair value of warrant liabilities		103		(6,727)	
Other		(2)		(10)	
Changes in operating assets and liabilities:					
Accounts receivable		(3,682)		(673)	
Prepaid expenses and other assets		(1,017)		(4,228)	
Accounts payable		824		62	
Accrued expenses and other liabilities		496		(1,854)	
Accrued compensation and benefits		171		(1,926)	
Operating lease liabilities		(1,894)		(1,046)	
Net cash used in operating activities	\$	(20,605)	\$	(26,806)	
Cash flows from investing activities:					
Maturity of investments and term deposits		15,207		6,559	
Purchases of investments and term deposits		(41,409)		_	
Purchases of property and equipment		(151)		(667)	
Net cash provided by (used in) investing activities	\$	(26,353)	\$	5,892	
Cash flows from financing activities:					
Proceeds from exercise of options		73		461	
Taxes withheld and paid related to net share settlement of equity awards		(849)		(3,437)	
Share buyback		(4,139)		_	
Cash paid for fractional shares related to stock split		(4)		_	
Net cash used in financing activities	\$	(4,919)	\$	(2,976)	
Effect of changes in currency exchange rates on cash and cash equivalents		1,069		(331)	
Net decrease in cash and cash equivalents		(50,808)		(24,221)	
Cash, cash equivalents and restricted cash, beginning of the period		106,256		145,785	
Cash, cash equivalents and restricted cash, end of the period	\$	55,448	\$	121,564	
Cash, cash equivalents and restricted cash, end of the period	Ψ	33,440	•	121,304	
Reconciliation of cash, cash equivalents and restricted cash to the amounts reported within the consolidated balance sheets					
Cash and cash equivalents	\$	54,310	\$	120,475	
Restricted cash		1,138		1,089	
Total cash, cash equivalents and restricted cash, end of the period	\$	55,448	\$	121,564	
-					

Supplemental disclosures of cash flow information:		
Cash paid for taxes	\$ 669 \$	323
Supplemental disclosure of non-cash investing and financing activities:		
Lease liabilities arising from obtaining right-of-use assets	\$ — \$	4,506
Vesting of early exercised stock options and restricted stock awards	34	71
Purchase of property and equipment accrued and not yet paid	_	13
Issuance of common stock upon release of restricted stock units	_	1

The accompanying notes are an integral part of these unaudited condensed consolidated financial statements.

# FORGE GLOBAL HOLDINGS, INC.

### NOTES TO UNAUDITED CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

# 1 Organization and Description of Business

Forge Global Holdings, Inc. (the "Company" and f/k/a Motive Capital Corp) is a financial services platform headquartered in San Francisco, California. The Company is committed to democratizing access to private markets through a purpose-built technology-driven platform. To serve the distinct needs of investors, shareholders, and companies, the Company offers an integrated platform of complementary solutions to support client engagement with the private market from beginning to end. The Company believes this holistic approach yields strong platform-based network effects, fueling participation in the private market and the Company's growth.

### 2 Summary of Significant Accounting Policies

### Basis of Presentation and Consolidation

The accompanying unaudited condensed consolidated financial statements include the accounts of the Company and its subsidiaries and have been prepared in accordance with generally accepted accounting principles in the United States of America ("GAAP") and in accordance with the rules and regulations of the Securities and Exchange Commission ("SEC"). All intercompany balances and transactions have been eliminated in consolidation. Certain prior period amounts on our unaudited condensed consolidated balance sheet have been reclassified to conform with our current period presentation.

In the normal course of business, the Company has transactions with various investment entities. In certain instances, the Company provides investment advisory services to pooled investment vehicles, each, a Single Asset Fund ("SAF"). The Company does not have discretion to make any investment, except for the specific investment for which a SAF was formed. The Company performs an assessment to determine (a) whether the Company's investments or other interests will absorb portions of a variable interest entity's expected losses or receive portions of the entity's expected residual returns and (b) whether the Company's involvement, through holding interests directly or indirectly in the entity would give it a controlling financial interest. The Company consolidates entities in which it, directly or indirectly, is determined to have a controlling financial interest. Consolidation conclusions are reviewed quarterly to identify whether any reconsideration events have occurred.

The Company has a majority ownership interest in Forge Europe GmbH ("Forge Europe") and accounts for Forge Europe as a fully consolidated subsidiary. The remaining interest, held by Deutsche Börse Aktiengesellschaft ("DBAG"), is reported as a noncontrolling interest in the unaudited condensed consolidated financial statements. DBAG is a related party of the Company.

There have been no changes to the Company's significant accounting policies described in the audited consolidated financial statements for the year ended December 31, 2024, that have had a material impact on these unaudited condensed consolidated financial statements and related notes.

### Unaudited Interim Condensed Consolidated Financial Information

These financial statements are presented in accordance with the rules and regulations of the SEC and do not include all disclosures normally required in annual consolidated financial statements prepared in accordance with GAAP. As such, the information included herein should be read in conjunction with the consolidated financial statements and accompanying notes as of and for the year ended December 31, 2024 (the "audited consolidated financial statements") that was included in the Company's Annual Report on Form 10-K filed on March 6, 2025, which provides a more complete discussion of the Company's accounting policies and certain other information. In management's opinion, the interim unaudited condensed consolidated financial statements have been prepared on the same basis as the annual financial statements, which include only normal recurring adjustments, necessary for a fair statement of the Company's financial position as of June 30, 2025 and its unaudited condensed consolidated results of operations and cash flows for the three and six months ended June 30, 2025 and 2024. The results of operations for the three and six months ended June 30, 2025 are not necessarily indicative of the results expected for the year ending December 31, 2025 or any other future interim or annual periods.

# Change in Capital Structure

Effective April 14, 2025, the Company effected a 1-for-15 reverse stock split of all of its authorized, issued, and outstanding shares of common stock. All share and per share amounts presented in the unaudited condensed consolidated financial statements and accompanying notes, including, but not limited to, shares authorized, issued and outstanding, dollar amounts of common stock, additional paid-in capital, earnings/(loss) per share, options and other equity securities under the Company's equity incentive plans, and warrants have been retroactively adjusted for all periods presented in order to reflect this change in capital structure. The reverse stock split decreased the number of authorized shares of common stock but did not affect the number of authorized shares of the Company's preferred stock or the par value of the common stock.

# Segment Information

The Company operates as a single operating segment and reportable segment. The Company's chief operating decision maker ("CODM") is its Chief Executive Officer, who reviews financial information presented on a consolidated basis for purposes of making operating decisions, allocating resources and evaluating the Company's financial performance. Accordingly, we have concluded that we consist of a single operating segment and reportable segment for accounting and financial reporting purposes. The CODM uses net income as the measure of profit or loss for purposes of assessing segment performance and deciding how to allocate resources, primarily by monitoring actual results against the forecast. As of June 30, 2025 and December 31, 2024, long-lived assets located outside of the United States were not material.

### Use of Estimates

The preparation of consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses during the reporting period. Such management estimates include, but are not limited to, collectability of accounts receivable, the fair value of financial assets and liabilities, the useful lives of acquired intangible assets and property and equipment, the impairment of long-lived assets and goodwill, the fair value of warrants, equity awards and share-based compensation expenses, including the derived service period for the awards containing market-based vesting conditions, and the valuation of deferred tax assets. These estimates are inherently subjective in nature and, therefore, actual results may differ from the Company's estimates and assumptions. The Company bases its estimates on historical experience and also on assumptions that it believes are reasonable. Further, the Company applies judgment in determining whether, directly or indirectly, it has a controlling financial interest in the SAFs, in order to conclude whether any of the SAFs must be consolidated.

The Company believes the estimates and assumptions underlying the unaudited condensed consolidated financial statements are reasonable and supportable based on the information available as of June 30, 2025. These estimates may change as new events occur and additional information is obtained, and related financial impacts will be recognized in the Company's unaudited condensed consolidated financial statements as soon as those events become known.

# Cash, Cash Equivalents and Investments

The Company considers all highly liquid investments with original maturities of three months or less from the date of purchase to be cash equivalents. Cash and cash equivalents consist primarily of bank deposit accounts, term deposits (less than 90 days), investments in money market mutual funds, and U.S. government treasury bills.

The Company's investments consist primarily of corporate debt securities and U.S. government and agency securities. These securities are classified as trading and are reported at fair value, with unrealized gains and losses reported in earnings. Purchase premiums and discounts are amortized or accreted using the effective interest method over the life of the related security and included in interest income in the unaudited condensed consolidated statements of operations. The Company also holds term deposits in excess of 90 days, which serve as backstops for certain lease arrangements.

Forge Trust Co., a non-depository South Dakota trust company and wholly-owned subsidiary, offers custodial services for individual retirement accounts ("IRA accounts") for a variety of assets, including at times, uninvested cash. Forge Trust Co. does not have any beneficial interest in client assets outside of custodial administration fees and does not use client assets for its own purposes. Total assets under custody as of June 30, 2025 and December 31, 2024 are \$18.1 billion and \$16.9 billion, respectively. Uninvested custodial client cash was \$440.3 million and \$482.9 million as of June 30, 2025 and December 31, 2024, respectively.

### Goodwill and Other Intangible Assets, Net

Goodwill represents the excess of the aggregate fair value of the consideration transferred in a business combination over the fair value of the assets acquired, net of liabilities assumed. Goodwill is not amortized but tested for impairment annually on October 1, or more frequently if events or changes in circumstances indicate the goodwill may be impaired. The Company has one operating segment and reporting unit, and therefore goodwill is tested for impairment at the entity level. The Company may elect to utilize a qualitative assessment to evaluate whether it is more likely than not that the fair value of

the reporting unit is less than its carrying value and if so, perform a quantitative test. Events or circumstances that could indicate an impairment include a significant change in the business climate, regulatory environment, established business plans, operating performance indicators, or competition. Potential impairment indicators may also include, but are not limited to, (i) the results of the Company's most recent annual or interim impairment testing, (ii) downward revisions to internal forecasts, (iii) declines in the Company's market capitalization below its book value, and the magnitude and duration of those declines, (iv) a reorganization resulting in a change to the Company's operating segments, and (v) other macroeconomic factors, such as increases in interest rates that may affect the weighted average cost of capital or volatility in the equity and debt markets. If the Company's market capitalization continues to decline or future performance varies from current expectations, assumptions, or estimates, including assumptions related to current macroeconomic uncertainties, this may trigger a future impairment charge. No impairment charges were recognized during the three and six months ended June 30, 2025 and June 30, 2024, respectively.

# Concentration of Credit Risks

Financial instruments that potentially subject the Company to concentrations of credit risk primarily comprise cash and cash equivalents and restricted cash, investments, accounts receivable, term deposits, payment-dependent notes receivable, and payment-dependent notes payable. Cash and cash equivalents and restricted cash may, at times, exceed amounts insured by the Federal Deposit Insurance Corporation and the Securities Investor Protection Corporation. The Company performs periodic evaluations of the relative credit standing of these financial institutions to limit the amount of credit exposure. The Company's investments consist of corporate debt securities and U.S. government and agency securities. The issuers of these debt securities are financially sound and subject to minimal credit risk.

The Company's exposure to credit risk associated with its contracts with holders of private company securities ("sellers") and investors ("buyers") (sellers and buyers collectively "counterparty" or "client") related to the transfer of private securities is measured on an individual counterparty basis. Concentrations of credit risk can be affected by changes in political, industry, or economic factors. To reduce the potential for risk concentration, the Company's exposure is monitored in light of changing counterparty and market conditions. As of June 30, 2025 and December 31, 2024, the Company did not have any material concentrations of credit risk outside the ordinary course of business.

As of June 30, 2025 and December 31, 2024, no clients accounted for more than 10% of the Company's accounts receivable. No client accounted for more than 10% of total revenue, less transaction-based expenses, for the three and six months ended June 30, 2025 and June 30, 2024.

# Revenue by Geographic Location

For the three and six months ended June 30, 2025 and 2024, revenue outside of the United States (including U.S. territories), based on client billing address, was \$2.1 million, \$4.3 million, \$1.8 million and \$3.3 million, respectively.

# Warrant Liabilities

The Company recognizes certain warrant instruments as derivative liabilities in accordance with ASC 815, Derivatives and Hedging. See Note 3 "Fair Value Measurements". The Company recognizes the warrant instruments as liabilities at fair value and adjusts the carrying value of the instruments to fair value at each reporting period until they are redeemed, exchanged, expired, or exercised. The Company will continue to adjust the warrant liabilities for changes in the fair value until the earlier of a) the exercise, exchange, or expiration of the warrants or b) the redemption of the warrants, at which time the warrants will be reclassified to additional paid-in-capital.

### Comprehensive Loss

Comprehensive loss consists of Net loss and Other comprehensive income or loss. The Company's Other comprehensive income or loss is comprised of foreign currency translation gains and losses. Accumulated other comprehensive loss, as presented in the unaudited condensed consolidated financial statements consists of changes in unrealized gains and losses on foreign currency translation.

### Recent Accounting Pronouncements

In November 2024, the FASB issued ASU 2024-03 *Income Statement- Reporting Comprehensive Income- Expense Disaggregation Disclosures* (Subtopic 220-40): Disaggregation of Income Statement Expenses, to improve financial reporting by requiring public entities to disclose additional information about specific expense categories in the notes to the financial statements of interim and annual reporting periods, including amounts and qualitative descriptions of employee compensation, depreciation, and intangible asset amortization. The standard is effective for fiscal years beginning after December 15, 2026, and interim periods within annual reporting periods beginning after December 15, 2027, with early adoption permitted. The standard should be applied prospectively, although retrospective application is permitted. The Company is currently evaluating the impact of adopting ASU 2024-03.

In December 2023, the FASB issued ASU 2023-09, *Income Taxes (Topic 740): Improvements to Income Tax Disclosures*, which requires public entities, on an annual basis, to provide disclosure of specific categories in the rate reconciliation, as well as disclosure of income taxes paid disaggregated by jurisdiction. ASU 2023-09 is effective for fiscal years beginning after December 15, 2024, with early adoption permitted. The Company is currently evaluating the impact of adopting ASU 2023-09 on its financial statements and has not yet determined its transition approach.

### 3 Fair Value Measurements

Cash equivalents, term deposits, corporate debt securities, U.S. government and agency securities, payment-dependent notes receivable, payment-dependent notes payable, and warrant liabilities are stated at fair value on a recurring basis. Cash, restricted cash, accounts receivable, accounts payable, and accrued liabilities are stated at their carrying value, which approximates fair value due to the short time these financial instruments are held to the expected receipt or payment date.

The Company classifies cash equivalents, including money market funds and U.S. government treasury bills, within Level 1 of the fair value hierarchy because the Company values these instruments using quoted market prices. The Company classifies term deposits, corporate debt securities, and U.S. government and agency securities as Level 2 of the fair value hierarchy because these investments are valued using observable market inputs without quoted market prices. The Company classifies the December 2023 Warrants (as defined herein) within Level 2 of the fair value hierarchy as these warrants are valued using a Black-Scholes option-pricing model which uses observable inputs. Assumptions in the model include, but are not limited to, risk-free interest rate, expected volatility of the Company's stock price, expected term and expected dividend yield. The Company classifies Payment-dependent notes receivable and payable and its Private Placement Warrants (as defined herein) as Level 3 of the fair value hierarchy as the fair value measurements are based on valuation techniques that use significant inputs that are unobservable which are described in more detail below.

The following tables present the fair value hierarchy for assets and liabilities measured at fair value on a recurring basis (in thousands):

	As of June 30, 2025							
		Level 1	Level 2		Level 3		Total	
Cash and cash equivalents:								
Money market funds	\$	15,451	\$		\$		\$	15,451
Corporate debt securities		_		2,601		_		2,601
U.S. government agency bonds		_		2,305		_		2,305
Term deposits (less than 90 days)		_		7,325		_		7,325
Payment-dependent notes receivable		_		_		9,604		9,604
Investments:								
Corporate debt securities		_		13,430		_		13,430
U.S. government securities		_		497		_		497
U.S. government agency bonds		_		12,466		_		12,466
Term deposits <sup>(1)</sup>		_		1,083				1,083
Total financial assets	\$	15,451	\$	39,707	\$	9,604	\$	64,762
Payment-dependent notes payable	\$	_	\$		\$	9,604	\$	9,604
December 2023 Warrants		_		30		_		30
Private Placement Warrants						266		266
Total financial liabilities	\$		\$	30	\$	9,870	\$	9,900

	As of December 31, 2024							
		Level 1		Level 2		Level 3		Total
Cash and cash equivalents:								
Money market funds	\$	56,300	\$	_	\$		\$	56,300
U.S. government securities		21,482		_		_		21,482
Term deposits (less than 90 days)		_		7,306		_		7,306
Payment-dependent notes receivable		_		_		7,412		7,412
Term deposits <sup>(1)</sup>		_		1,068				1,068
Total financial assets	\$	77,782	\$	8,374	\$	7,412	\$	93,568
						_		
Payment-dependent notes payable	\$	_	\$	_	\$	7,412	\$	7,412
December 2023 Warrants		_		45		_		45
Private Placement Warrants				_		147		147
Total financial liabilities	\$	_	\$	45	\$	7,559	\$	7,604

<sup>(1)</sup> Included in prepaid expenses and other current assets on the unaudited condensed consolidated balance sheets as of June 30, 2025 and December 31, 2024.

# Payment-Dependent Notes Receivable and Payment-Dependent Notes Payable

The Company classifies payment-dependent notes receivable and payment-dependent notes payable within Level 3 of the fair value hierarchy if the underlying securities are equity of private companies whose regular financial and nonfinancial information is generally not available other than when it is publicly disclosed, or significant unobservable inputs are used to estimate fair value.

The Company estimates the fair value of payment-dependent notes receivable and payment-dependent notes payable utilizing market data and completed transactions made through the Company's platform for the relevant private securities as well as mutual fund valuations of private companies as relevant data inputs.

### **Private Placement Warrants**

The Company classifies the Private Placement Warrants within Level 3 due to the valuation technique used to estimate fair value. A Monte Carlo simulation model was used to estimate fair value as of June 30, 2025 and December 31, 2024, respectively.

The Company estimated the fair value of the Private Placement Warrant liabilities as of June 30, 2025 and December 31, 2024, respectively, using the following key assumptions:

	As of	June 30, 2025	A	s of December 31, 2024
Fair value of underlying securities	\$	19.04	\$	13.95
Expected term (years)		1.7		2.2
Expected volatility		95.0 %		82.5 %
Risk-free interest rate		3.8 %		4.3 %
Expected dividend yield		<u> </u>		<u> </u>
Fair value per warrant	\$	0.54	\$	0.30

The Company recorded changes in the fair value of the Private Placement Warrants as follows (in thousands):

	For the three months ended June 30				For the six months ended June 30				
		2025		2024		2025		2024	
Beginning Balance	\$	_	\$	3,102	\$	147	\$	4,727	
Change in fair value of warrant liability <sup>(1)</sup>		266		(1,255)		118		(2,880)	
Balance, June 30	\$	266	\$	1,847	\$	266	\$	1,847	

(1) The change in fair value of warrant liability is recorded in the unaudited condensed consolidated statement of operations within change in fair value of warrant liabilities.

# Transfers Into and Out of Level 3

The Company transfers financial instruments out of Level 3 on the date when underlying input parameters are readily observable from active markets with or without quoted market prices.

For Payment-dependent notes receivable and payable, transfers from Level 3 to Level 1 generally relate to a company going public and listing on a national securities exchange. During the periods presented, there were no transfers of payment-dependent notes receivable and payable into or out of Level 3.

The following table provides a reconciliation for all financial assets and liabilities measured at fair value using significant unobservable inputs (Level 3) for the six months ended June 30, 2025 and 2024 (in thousands):

	evel 3 Financial Assets	Total Level 3 Financial Liabilities
Balance as of December 31, 2024	\$ 7,412	7,559
Change in fair value of payment-dependent notes receivable	2,192	_
Change in fair value of Private Placement Warrants	_	118
Change in fair value of payment-dependent notes payable	 	2,192
Balance as of June 30, 2025	\$ 9,604	\$ 9,869
	 evel 3 Financial Assets	Total Level 3 Financial Liabilities

Balance as of December 31, 2023 \$ 5,593 \$ 10,320
Change in fair value of payment-dependent notes receivable \$ 1,165 \$ —
Change in fair value of Private Placement Warrants \$ — (2,880)
Change in fair value of payment-dependent notes payable \$ = 1,165
Balance as of June 30, 2024 \$ \$ 6,758 \$ 8,605

# 4 Goodwill and Intangible Assets, Net

The components of goodwill and intangible assets and accumulated amortization are as follows (in thousands):

As of June 30, 2025						
Weighted Average Remaining Amortization Period	(	Gross Carrying Amount		Accumulated Amortization		Net Carrying Amount
Indefinite	\$	120,948	\$	_	\$	120,948
0.0 years		13,200		(13,200)		
4.3 years		7,507		(4,864)		2,643
1.3 years		960		(720)		240
		21,667		(18,784)		2,883
Indefinite		2,224		_		2,224
		2,224		_		2,224
	\$	144,839	\$	(18,784)	\$	126,055
	Indefinite  0.0 years 4.3 years 1.3 years	Indefinite \$  0.0 years 4.3 years 1.3 years	Weighted Average Remaining Amortization Period         Gross Carrying Amount           Indefinite         \$ 120,948           0.0 years         13,200           4.3 years         7,507           1.3 years         960           Indefinite         2,224           2,224	Weighted Average Remaining Amount         Gross Carrying Amount           Indefinite         \$ 120,948           0.0 years         13,200           4.3 years         7,507           1.3 years         960           Indefinite         21,667           Indefinite         2,224           2,224         2,224	Weighted Average Remaining Amortization Period         Gross Carrying Amount         Accumulated Amortization           Indefinite         \$ 120,948         \$ —           0.0 years         13,200         (13,200)           4.3 years         7,507         (4,864)           1.3 years         960         (720)           21,667         (18,784)           Indefinite         2,224         —           2,224         —	Weighted Average Remaining Amortization Period         Gross Carrying Amount         Accumulated Amortization           Indefinite         \$ 120,948         \$ — \$           0.0 years         \$ 13,200         (13,200)           4.3 years         7,507         (4,864)           1.3 years         960         (720)           21,667         (18,784)           Indefinite         2,224         —           2,224         —

	As of December 31, 2024																				
	Weighted Average Remaining Amortization Period	(	Gross Carrying Amount																Accumulated Amortization		Net Carrying Amount
Goodwill:																					
Goodwill from acquisitions	Indefinite	\$	120,948	\$	_	\$	120,948														
Finite-lived intangible assets:																					
Developed technology	0.0 years		13,200		(13,200)		_														
Client relationships	4.8 years		7,507		(4,559)		2,948														
Launched IPR&D assets	1.8 years		960		(624)		336														
Total finite-lived intangible assets			21,667		(18,383)		3,284														
Indefinite-lived intangible assets:																					
Trade name - website domain	Indefinite		2,224		_		2,224														
Total infinite-lived intangible assets			2,224				2,224														
Total goodwill and intangible assets		\$	144,839	\$	(18,383)	\$	126,456														

Amortization expense related to finite-lived intangible assets for the three and six months ended June 30, 2025 was \$0.2 million and \$0.4 million, respectively, and is included in depreciation and amortization in the accompanying unaudited condensed consolidated statements of operations. Amortization expense related to finite-lived intangible assets for the three and six months ended June 30, 2024 was \$1.0 million and \$2.0 million, respectively, and is included in the depreciation and amortization expense in the accompanying unaudited condensed consolidated statements of operations.

The table below presents estimated future amortization expense for finite-lived intangible assets as of June 30, 2025 (in thousands):

	Amount
Remainder of 2025	\$ 401
2026	754
2027	610
2028	610
2029	508
Total	\$ 2,883

# 5 Leases

The Company leases real estate for office space under operating leases.

The Company has an option to extend the lease term for a period of 0.1 years to 4.3 years. Renewal options are not considered in the remaining lease term because it is not reasonably certain that the Company will exercise such option.

Operating lease expense, included in rent and occupancy in the unaudited condensed consolidated statements of operations, was as follows (in thousands):

		Three Months Ended June 30				Six Months Ended June 30,			
		2025		2024		2025		2024	
Operating lease expense	\$	594	\$	923	\$	1,379	\$	1,706	
Variable lease expense		126		97		223		201	
Total operating lease expenses	\$	720	\$	1,020	\$	1,602	\$	1,907	
	-							<del></del>	
Sublease income <sup>(1)</sup>	\$	137	\$	95	\$	242	\$	191	

(1) Sublease income is included in other income in the unaudited condensed consolidated statements of operations.

The table below presents additional information related to the Company's operating leases (in thousands):

	As of June 30, 2025	As of December 31, 2024
Weighted-average remaining lease term (in years)	4.1	4.3
Weighted-average discount rate	7.2 %	7.2 %

There were no right-of-use assets impairments recognized during the three and six months ended June 30, 2025. During the six months ended June 30, 2024, it was determined that office space under an existing lease would no longer be

used and the associated right-of-use asset was reduced to \$0 and an impairment of \$0.2 million was recognized in rent and occupancy expense in the unaudited condensed consolidated statements of operations.

Future undiscounted lease payments under operating leases as of June 30, 2025 were as follows (in thousands):

	Lease Payment Obligation		Sublease Income		Ne	et Lease Obligation
Remaining 2025	\$	1,796	\$	(87)	\$	1,709
2026		1,071		_		1,071
2027		1,103		_		1,103
2028		1,136		_		1,136
2029		774		_		774
Total undiscounted lease payments	\$	5,880	\$	(87)	\$	5,793
Less: imputed interest		(617)				
Present value of future lease payments		5,263				
Less: operating lease liabilities, current		2,032				
Operating lease liabilities, noncurrent	\$	3,231				

### 6 Commitments and Contingencies

The Company is subject to claims and lawsuits in the ordinary course of business, including arbitration, class actions and other litigation, some of which include claims for substantial or unspecified damages. The Company may also be the subject of inquiries, investigations, and proceedings by regulatory and other governmental agencies. The Company reviews these matters on an ongoing basis and provides disclosures and records loss contingencies in accordance with the loss contingencies accounting guidance. The Company establishes an accrual for losses at management's best estimate when the Company assesses that it is probable that a loss has been incurred and the amount of the loss can be reasonably estimated. If no amount within the range is considered a better estimate than any other amount, an accrual for losses is recorded based on the bottom amount of the range. The Company did not have any accrual for loss contingencies as of June 30, 2025 and December 31, 2024. The Company monitors these matters for developments that would affect the likelihood of a loss and the accrued amount, if any, and adjusts the amount as appropriate.

# 401(k) Plan

The Company has established a tax-qualified retirement plan under Section 401(k) of the Internal Revenue Code for all of its U.S. employees, including executive officers, who satisfy certain eligibility requirements, including requirements relating to age and length of service. The Company provides a discretionary match on 100% of employee contributions up to 2% of eligible earnings. During the three and six months ended June 30, 2025, the Company recorded 401(k) contribution expense related to the defined contribution plan of \$0.3 million and \$0.6 million, respectively, in compensation and benefits in the Company's unaudited condensed consolidated statements of operations. During the three and six months ended June 30, 2024, the Company recorded 401(k) contribution expense related to the defined contribution plan of \$0.3 million and \$0.5 million, respectively, in compensation and benefits in the unaudited condensed consolidated statements of operations.

# Non-Cancelable Purchase Obligations

In the normal course of business, the Company enters into non-cancelable purchase commitments with various parties mainly for its operating leases, software products, and services. As of June 30, 2025, the Company had outstanding

non-cancelable purchase obligations with a term of 12 months or longer, excluding operating lease obligations (see Note 5, "Leases," for additional information) as follows (in thousands):

	A	Amount
Remainder of 2025	\$	2,853
2026		1,323
2027		727
2028		325
2029		325
Thereafter		162
Total	\$	5,715

### Other

The Company is required to file extensive informational returns and forms to various tax authorities in connection with its custodial and trading solutions. If such filings are incomplete or untimely, the Company may be subjected to compliance fees, fines or penalties if the Company does not meet the requirements of reasonable cause, safe harbor, or other relief as may be provided by the relevant tax authorities. In May 2025, the Company resolved the self-identified informational filing matter with the Internal Revenue Service and agreed to pay an immaterial compliance fee.

# 7 Share-Based Compensation

The Company effected a 1-for-15 reverse stock split of all of its authorized, issued and outstanding shares of common stock on April 14, 2025. Proportional adjustments were made to the number of shares of common stock awarded and available for issuance under the Company's equity incentive plans, as well as the exercise price and the number of shares issuable upon the exercise or conversion of the Company's outstanding stock options and other equity securities under the Company's equity incentive plans. See Note 2, "Summary of Significant Accounting Policies," for additional information.

#### 2025 Inducement Plan

In March 2025, the Company adopted the 2025 Inducement Plan (the "Inducement Plan") pursuant to which the Company reserved 100,000 shares of its common stock to be used exclusively for grants of equity-based awards to individuals who were not previously employees or directors of the Company, as an inducement material to the individual's entry into employment with the Company. The Inducement Plan was adopted by the Company's board of directors without stockholder approval pursuant to NYSE Listed Company Manual Rule 303A.08. In July 2025, the Company amended and restated the Inducement Plan to reserve an additional 83.330 shares.

### Share Repurchase Program

In March 2025, the Company's board of directors approved a share repurchase program of up to \$10.0 million. The program does not obligate the Company to acquire any particular amount of its common stock, and may be modified, suspended, or terminated at any time at the Company's discretion. The program has no expiration date. As of June 30, 2025, the Company has repurchased 314,701 shares at an average price of \$13.15 per share and approximately \$5.9 million remains available for repurchase under the share repurchase program.

### Stock Compensation

Stock compensation for the periods indicated below are as follows (in thousands):

	Three Months Ended June 30,				Six Months Ended June 30,			
		2025		2024		2025		2024
RSUs	\$	3,225	\$	7,083	\$	9,298	\$	15,750
Stock options		211		776		657		1,576
Total share-based compensation	\$	3,436	\$	7,859	\$	9,955	\$	17,326

# **RSUs**

The Company's RSUs are convertible into shares of the Company's common stock upon vesting on a one-to-one basis, and generally contain time-based vesting conditions. RSUs granted to certain executives also contain market-based vesting conditions or performance-based vesting conditions. The RSUs generally vest over the service period of one to three years.

RSU activity during the six months ended June 30, 2025 was as follows:

	Total RSUs	Time-based	Performance-based	Market-based	Gra	hted-Average nt Date Fair ie Per Share
Unvested as of December 31, 2024	839,958	539,379	113,378	187,201	\$	48.91
Granted	123,278	123,278	_	_		15.88
Vested <sup>(1)</sup>	(371,428)	(274,646)	(65,516)	(31,266)		67.56
Forfeited	(65,938)	(54,330)	(11,608)			35.73
Unvested as of June 30, 2025	525,870	333,681	36,254	155,935	\$	29.41

(1) Common stock has not been issued in connection with 25,323 vested RSUs because such RSUs were unsettled as of June 30, 2025.

Future share-based compensation expense for unvested RSUs as of June 30, 2025 was \$9.7 million, which will be recognized over a weighted-average period of 1.17 years.

# 8 Income Taxes

The Company's effective tax rate from continuing operations was 1.5% and 4.4% for the three and six months ended June 30, 2025, respectively. The Company's effective tax rate from continuing operations was 2.1% and 1.6% for the three and six months ended June 30, 2024, respectively. The Company's effective tax rate for the three and six months ended June 30, 2025 was impacted by an \$0.8 million reserve against a prior year federal net operating loss carryback claim that may not be realized. The Company's full valuation allowance in the United States caused the year-to-date effective tax rate to be different from the U.S. federal statutory tax rate.

On July 4, 2025, the United States enacted into law new tax legislation, the One Big Beautiful Bill Act, ("OBBBA"). The OBBBA includes provisions modifying the corporate income tax code, including the immediate expensing of domestic research and development expenditures for tax purposes, 100% bonus depreciation for qualified assets, and reinstating favorable treatment for certain business tax items. The legislation is retroactive and effective for the 2025 tax year. The Company is still assessing the impact of the OBBBA on its federal net operating loss deferred tax assets as well as its results of operations and cash flows.

# 9 Net Loss per Share

The Company has one class of common stock. The diluted net loss per share attributable to common stockholders is calculated by giving effect to all potentially dilutive common stock equivalents during the period using the treasury stock method or the if-converted method, if applicable. The Company's stock options, warrants, and early exercised stock options are considered to be potential common stock equivalents but have been excluded from the calculation of diluted net loss per share attributable to common stockholders because the holders of these securities do not have a contractual right to share in the Company's losses, and their effect would be antidilutive. Therefore, the net loss for the three and six months ended June 30, 2025 and 2024 was attributed to common stockholders only.

The following table sets forth the computation of basic and diluted net loss per share attributable to common stockholders for the periods presented (in thousands, except for per share data):

	Three Months	Ended June 30,	Six Months Ended June 30,			
	2025	2024	2025	2024		
Numerator:						
Net loss attributable to common stockholders, basic	\$ (12,580)	\$ (13,724)	\$ (28,752)	\$ (32,348)		
Net loss attributable to common stockholders, diluted	\$ (12,580)	\$ (13,724)	\$ (28,752)	\$ (32,348)		
Denominator:						
Weighted-average number of shares used to compute net loss per share attributable to common stockholders, basic	12,474	12,179	12,503	12,112		
Weighted-average number of shares used to compute net loss per share attributable to common stockholders, diluted	12,474	12,179	12,503	12,112		
Net loss per share attributed to common stockholders:						
Basic	\$ (1.01)	\$ (1.13)	\$ (2.30)	\$ (2.67)		
Diluted	\$ (1.01)	\$ (1.13)	\$ (2.30)	\$ (2.67)		

The following potentially dilutive shares (in thousands) were excluded from the calculation of diluted shares outstanding as the effect would have been anti-dilutive:

	June 30, 2025	June 30, 2024
Warrants to purchase common stock	194	219
Private Placement Warrants	492	492
Common stock subject to repurchase	5	19
Outstanding options	462	481
Restricted stock units	526	1,041
Total	1,679	2,252

# **10 Related Party Transactions**

On September 7, 2022 the Company and DBAG formed Forge Europe GmbH. DBAG is a stockholder of the Company. See Note 2, "Summary of Significant Accounting Policies" for additional information.

Forge Global Advisors LLC ("FGA"), a wholly-owned subsidiary of the Company and an investment adviser registered under the Investment Advisers Act of 1940, as amended, advises investment funds, each of which are organized as

a series of Forge Investments LLC and segregated portfolio companies of Forge Investments SPC and Forge Investments II SPC (such investment funds and portfolio companies are individually and collectively referred to as "Single Asset Funds" or "SAFs"). The SAFs are each formed for the purpose of investing in securities relating to a single private company and are owned by different investors. FGA serves as the manager of the Forge Investments LLC series. The Company utilizes a third-party fund administrator to manage the Forge Investments SPC and Forge Investments II SPC SAFs. The Company has no ownership interest nor participation in the gains or losses of the SAFs. The Company does not consolidate Forge Investments LLC, Forge Investments SPC, Forge Investments II SPC, or any of the SAFs, because the Company has no direct or indirect interest in the SAFs and the amount of expenses the Company pays on behalf of the SAFs are not significant to the entities. Investors in the SAFs do not have any recourse to the assets of the Company.

While not contractually required, FGA may, at its sole discretion, absorb certain expenses on behalf of the SAFs. Audit and accounting related services are recorded in professional services in the unaudited condensed consolidated statements of operations. Professional services expenses of \$0.4 million and \$0.7 million were recognized in the unaudited condensed consolidated statements of operations during the three and six months ended June 30, 2025. Professional services expenses of \$0.3 million and \$0.7 million were recognized in the unaudited condensed consolidated statements of operations during the three and six months ended June 30, 2024.

A family member of one of the Company's former executive officers is a portfolio manager for investment funds that engage in secondary transactions with the Company in the ordinary course of business. Such transactions became related party transactions upon the employee's appointment to executive officer in April 2023. One of such funds is also a client of the Company's data and related products. No revenue was recognized from the transactions, however the Company recorded commissions of \$0.2 million for services provided for the three and six months ended June 30, 2025.

The Company also has an equity method investment in EQUIAM, LLC ("Equiam"), a data-powered venture capital manager. Equiam also engages in secondary transactions with the Company through its affiliate private investment fund in the ordinary course of business. The Company holds a 25.8% ownership of Equiam's issued and outstanding Class A units. On May 2, 2024, the Company invested \$0.5 million in an unsecured convertible note and warrant offering by Equiam. The unsecured convertible note shall become due and payable after the third anniversary of the issuance date, unless converted or prepaid. The investment in the unsecured convertible note and warrant is accounted for as consideration payable to a customer under ASC 606-10-32-25. As of June 30, 2025, the unsecured convertible note had a carrying value of \$0.4 million which is recorded in prepaid expenses and other current assets, as well as other assets, noncurrent, on the unaudited condensed consolidated balance sheets.

# 11 Subsequent Events

On July 1, 2025, the Company acquired Accuidity, LLC ("Accuidity"), a U.S. privately held asset management firm. The total consideration is comprised of \$10.0 million cash and 1.2 million shares of newly issued common stock. The Company will provide the required disclosures for the acquisition in its Quarterly Report on Form 10-Q for the quarter ending September 30, 2025.

On July 1, 2025, the Company amended and restated the Inducement Plan and subsequently filed a Form S-8 with the Securities and Exchange Commission to register and make available for issuance 83,330 additional shares of the Company's common stock under the Inducement Plan.

# MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis provide information that our management believes is relevant to an assessment and understanding of our consolidated results of operations and financial condition. This discussion and analysis should be read together with the unaudited condensed consolidated financial statements and related notes to those statements in this Report, as well as our audited consolidated financial statements and related notes to those statements included in our Annual Report on Form 10-K filed with the SEC on March 6, 2025 (the "Annual Report"). This discussion and analysis contains forward-looking statements based upon current expectations that involve risks, uncertainties, and assumptions, as described under the heading "Forward-Looking Statements." Actual results and timing of selected events may differ materially from those anticipated in these forward-looking statements as a result of various factors, including those set forth under "Risk Factors" or elsewhere in this Report.

Unless the context otherwise requires, references in this section to "Forge," the "Company," "we," "us," and "our," refer to Forge Global Holdings, Inc. and its subsidiaries.

### **Business Overview**

Forge is committed to democratizing access to private markets through our comprehensive suite of private market solutions, including a purpose-built technology-driven platform connecting buyers and sellers of private shares with investment and custody opportunities. Our marketplace offers data, insights, and tools to help institutions and individuals confidently navigate the private market. Our Private Company Solutions ("PCS") offerings deliver tailored solutions to private companies for their liquidity and capital formation needs. Our asset management solution enhances investment efficiency through a suite of single asset funds that invest in private companies. Our data solutions provide clients with key insights into a traditionally opaque market and drive additional participation in our marketplace and platform generally. We also provide custodial services for self-directed retirement accounts, specializing in the custody of alternative assets for our clients.

Forge's platform attracts a broad spectrum of market participants, which fall into three general categories: investors, shareholders, and companies. To serve the distinct needs of each of these categories, we have strategically invested in complementary solutions to offer our clients the most critical tools to participate in the private market ecosystem through an integrated platform that allows them to engage in various investment opportunities and supports the process from beginning to end. We believe this holistic approach yields strong platform-based network effects, fueling participation in the private market and our growth.

Our revenue is primarily driven by fees generated from our flagship marketplace and custody solutions. Revenue includes fees charged for private market transactions on the Forge marketplace and fees charged for account and asset management solutions from our custody solutions.

# **Key Factors Affecting our Financial Performance**

In addition to growing and maintaining our client base and continuing to invest in our platform, we consider the following to be the key factors affecting our financial performance. These and other factors are discussed in more detail in the Risk Factors in our Annual Report.

### Market Trends

*Private Market Trends* — Supply of and demand for private company shares fluctuates with various factors, including but not limited to, anticipated or planned IPOs, mergers and acquisitions activity in the public and private space, private company funding activity, exits by private equity or investment firms, participation in the private market by private companies generally, and demand from individual and institutional investors.

Consumer Behavior — Buyers' and sellers' behaviors vary over time and are affected by numerous conditions. For example, behavior may be impacted by social or economic factors such as changes in disposable income levels and the need for liquidity, employee tenure, general interest in investing, interest rate levels, and reaction to stock market volatility. There may also be high profile IPOs, SPACs, or idiosyncratic events impacting single companies that impact consumer behavior. These shifts in consumer behavior may influence interest in our products and solutions over time.

Macroeconomic Environment — Behavior and risk appetites by individual and institutional accredited investors, as well as businesses, are impacted by various factors in the overall macroeconomic environment, including but not limited to, the interest rate environment, volatility and liquidity risks to private equity valuations, and uncertainty around settlement prices for illiquid assets. These factors could all, individually or together, impact investor appetite and investment preferences across the alternative investment and private markets space. In particular, cash administration fees are based on prevailing interest rates and custodial client cash balances, and currently makes up the largest portion of our custodial administration fee revenue.

### Revenue and Take Rate

*Types of Products* — We may adjust fees to account for different operational costs and transaction-based costs that may be incurred for different products. The mix of transactions in different products will impact overall revenues and take rate.

Types of Clients — The type of client may influence our revenue. Institutional and individual clients may be charged fee rates depending on different factors, including but not limited to, the size of the transaction, the demand for the underlying private company shares, or the type of custody services provided. Clients that come to our platform through third-party brokers, PCS, or partnerships may also impact revenue. The mix of clients in any given period will impact our overall revenues and take rate.

## **Segment Information**

The Company operates as a single operating segment and reportable segment. The Company's chief operating decision maker ("CODM") is its Chief Executive Officer, who reviews financial information presented on a consolidated basis for purposes of making operating decisions, allocating resources and evaluating the Company's financial performance. Accordingly, we have concluded that we consist of a single operating segment and reportable segment for accounting and financial reporting purposes. The CODM uses net income as the measure of profit or loss for purposes of assessing segment performance and deciding how to allocate resources, primarily by monitoring actual results against the forecast.

### **Key Business Metrics**

We monitor the following key business metrics to help us evaluate our business, identify trends affecting our business, formulate business plans, and make strategic decisions. The tables below reflect period-over-period changes in our key business metrics, along with the percentage change between such periods. Percentages may not be replicated based on the rounded figures presented. We believe the following business metrics are useful in evaluating our business.

		Three Mo	onths	Ended		QoQ	)		Six Mon	ths Eı	ıded	YoY	<i>r</i>										
Dollars in thousands	Jı	June 30, 2025		June 30, 2025		June 30, 2025		June 30, 2025		larch 31, 2025		March 31, 2025		March 31, 2025		Change	% Change	J	June 30, 2025	J	une 30, 2024	 Change	% Change
MARKETPLACE SOLUTIONS																							
Trades		927		963		(36)	(4)%		1,890		1,436	454	32 %										
Volume	\$	756,110	\$	692,391	\$	63,719	9 %	\$	1,448,501	\$	688,856	\$ 759,645	110 %										
Net Take Rate		2.4 %	)	2.3 %	Ó	0.1 %	4 %		2.4 %		2.9 %	(0.5)%	(17)%										
Marketplace revenues, less transaction-based expenses	\$	18,490	\$	15,831	\$	2,659	17 %	\$	34,321	\$	19,914	\$ 14,407	72 %										

• Trades are defined as the total number of orders executed by us on behalf of private investors and stockholders. Increasing the number of orders is critical to increasing our revenue and, in turn, to achieving profitability.

- Volume is defined as the total sales value for all securities traded through our Forge marketplace, which is the aggregate value of the issuer company's equity attributed to both the buyer and seller in a trade and as such a \$100 trade of equity between buyer and seller would be captured as \$200 of volume for us. Although we typically capture a commission on each side of a trade, we may not in certain cases due to factors such as the use of a third-party broker by one of the parties or supply factors that would not allow us to attract sellers of shares of certain issuers. Volume is influenced by, among other things, the pricing and quality of our services as well as market conditions that affect private company valuations, such as increases in valuations of comparable companies at IPO.
- Net Take Rates are defined as our marketplace revenues, less markets-related transaction-based expenses, divided by Volume. These represent the
  percentage of fees earned by our marketplace on any transactions executed from the commission we charged on such transactions less transaction-based
  expenses, which is a determining factor in our revenue. The Net Take Rate can vary based upon the service or product offering and is also affected by
  the average order size and transaction frequency.

	As	of or for the T	hree	Months Ended	QoQ	A	s of or for the S	Six N	Ionths Ended		YoY			
Dollars in thousands	J	June 30, 2025	M	larch 31, 2025	Change	% Change	June 30, 2025		June 30, 2024		Change		% Change	
<b>CUSTODY SOLUTIONS</b>														
Total Custodial Accounts		2,598,846		2,508,443	90,403	4 %		2,598,846		2,211,108		387,738	18 %	
Assets Under Custody	\$	18,132,637	\$	17,635,034	\$ 497,603	3 %	\$	18,132,637	\$	16,600,408	\$	1,532,229	9 %	
Custodial Client Cash	\$	440,278	\$	459,685	\$ (19,407)	(4)%	\$	440,278	\$	494,972	\$	(54,694)	(11)%	
Custodial administration fees less transaction-based expens		9,094	\$	9,273	\$ (179)	(2)%	\$	18,367	\$	21,325	\$	(2,958)	(14)%	

- Total Custodial Accounts are defined as our clients' custodial accounts that are established on our platform and billable. These relate to our Custodial
  Administration fees revenue stream and are an important measure of our business as the number of Total Custodial Accounts is an indicator of our
  future revenues from certain account maintenance and transaction fees.
- Assets Under Custody is the reported value of all client holdings held under our agreements, including cash submitted to us by the responsible party.
  These assets can be held at various financial institutions, issuers, and in our vault. As the custodian of the accounts, we collect all interest and dividends, handle all fees and transactions, and any other considerations for the assets concerned. Our fees are earned from the overall maintenance activities of all assets and are not charged on the basis of the dollar value of Assets Under Custody, but we believe that Assets Under Custody is a useful metric for assessing the relative size and scope of our business.
- Custodial Client Cash is a component of Assets Under Custody representing the value of cash held on behalf of clients held under our agreements. These assets are held at various financial institutions. Our fees are earned from the administration activities we perform with respect to these balances. The amount of custodial client cash is a determining factor in our revenue.

### **Non-GAAP Financial Measures**

In addition to Forge's financial results determined in accordance with generally accepted accounting principles in the United States of America ("GAAP"), Forge presents Adjusted EBITDA and Adjusted EPS, non-GAAP financial measures. Forge uses these non-GAAP financial measures to evaluate its ongoing operations and for internal planning and forecasting purposes. Forge believes these non-GAAP financial measures, when taken together with the corresponding GAAP financial measures, provide meaningful supplemental information regarding its performance by excluding specific financial items that have less bearing on its core operating performance. Forge considers Adjusted EBITDA and Adjusted EPS to be important measures because they help illustrate underlying trends in its business and historical operating performance on a more consistent basis.

However, non-GAAP financial information is presented for supplemental informational purposes only, has limitations as an analytical tool, and should not be considered in isolation or as a substitute for financial information presented in accordance with GAAP. In addition, other companies, including companies in Forge's industry, may calculate similarly titled non-GAAP financial measures differently or may use other measures to evaluate their performance, all of which could reduce the usefulness as a tool for comparison. A reconciliation is provided below for Adjusted EBITDA to net loss attributable to common stockholders, the most directly comparable financial measure stated in accordance with GAAP and Adjusted EPS to EPS. Investors are encouraged to review Adjusted EBITDA and Adjusted EPS and the respective reconciliations and not to rely on any single financial measure to evaluate Forge's business.

Forge defines Adjusted EBITDA as net loss attributable to Forge Global Holdings, Inc., adjusted to exclude: (i) net loss attributable to noncontrolling interest, (ii) provision for income taxes, (iii) depreciation and amortization, (iv) share-based compensation expense, (v) interest income, (vi) change in fair value of warrant liabilities, and (vii) other significant gains, losses, and expenses such as impairments, acquisition-related transaction and reorganization costs that Forge believes are not indicative of its ongoing results.

Forge defines Adjusted EPS as net loss attributable to Forge Global Holdings, Inc., adjusted to exclude: (i) net change in fair value of warrant liabilities and (ii) the tax effect of the adjustment at Forge's effective tax rate from continuing operations divided by the weighted average shares outstanding for the respective periods.

The following table reconciles net loss attributable to Forge Global Holdings, Inc. to Adjusted EBITDA for the periods presented below:

		Three Mon	nths E	Ended	Six Mon	ths Ended			
(in thousands)	Ju	ne 30, 2025	M	arch 31, 2025	June 30, 2025	J	une 30, 2024		
Net loss attributable to Forge Global Holdings, Inc.	\$	(12,580)	\$	(16,172)	\$ (28,752)	\$	(32,348)		
Add:									
Interest income		(803)		(1,042)	(1,845)		(3,204)		
Provision for income taxes		189		1,016	1,205		474		
Depreciation and amortization		909		986	1,895		3,597		
Net loss attributable to noncontrolling interest		141		(26)	115		(686)		
Loss on impairment of long lived assets		_		_	_		186		
Share-based compensation expense		3,436		6,519	9,955		17,326		
Change in fair value of warrant liabilities		294		(191)	103		(6,727)		
Acquisition-related transaction costs		1,988		_	1,988		_		
Other		993		_	993		_		
Adjusted EBITDA	\$	(5,433)	\$	(8,910)	\$ (14,343)	\$	(21,382)		

Some of the limitations of Adjusted EBITDA include: (i) Adjusted EBITDA does not properly reflect capital commitments to be paid in the future, and (ii) although depreciation and amortization are non-cash charges, the underlying assets may need to be replaced and Adjusted EBITDA does not reflect these capital expenditures. In evaluating Adjusted EBITDA, be aware that in the future we will incur expenses similar to the adjustments in this presentation. Our presentation

of Adjusted EBITDA should not be construed as an inference that our future results will be unaffected by these expenses or any unusual or non-recurring items. We compensate for these limitations by providing specific information regarding the GAAP items excluded from Adjusted EBITDA. When evaluating our performance, consider Adjusted EBITDA in addition to, and not a substitute for, other financial performance measures, including our net loss and other GAAP results.

The following table reconciles EPS to Adjusted EPS for the periods presented below:

		Three Mo	nths I		Six Mont	ths Ended			
(in thousands, except share and per share data)	Ju	ne 30, 2025	M	arch 31, 2025	J	June 30, 2025	J	une 30, 2024	
Net loss attributable to common stockholders, basic and diluted	\$	(12,580)	\$	(16,172)	\$	(28,752)	\$	(32,348)	
Add:									
Change in fair value of warrant liabilities		294		(191)		103		(6,727)	
Income tax (expense) benefit of adjustment		(4)		13		(4)		108	
Adjusted net loss attributable to common stockholders, basic and diluted	\$	(12,290)	\$	(16,350)	\$	(28,653)	\$	(38,967)	
Weighted average shares - basic and diluted		12,474		12,534		12,503		12,112	
EPS - basic and diluted	\$	(1.01)	\$	(1.29)	\$	(2.30)	\$	(2.67)	
Adjusted EPS - basic and diluted	\$	(0.99)	\$	(1.30)	\$	(2.30)	\$	(3.22)	

Amounts may not recalculate due to rounding.

### **Basis of Presentation**

The unaudited condensed consolidated financial statements and accompanying notes included elsewhere in this Report include our accounts and accounts of our consolidated subsidiaries and were prepared in accordance with GAAP.

### **Components of Results of Operations**

#### Revenue

Marketplace revenue — Our marketplace revenue consists of fees earned by us in connection with our marketplace, PCS, asset management, and data solutions. We earn marketplace solution revenue from non-underwritten transactions, such as private placements of equity securities. These fees vary depending on multiple factors, including the size of the transaction, the demand for the underlying equity security and the form of the transaction. PCS earns program and transaction fees from companies through tender offer and structured liquidity programs. Our asset management revenue consists of set-up fees charged in connection with direct investments in member interests in Forge-managed SAFs. Revenues generated from our data solutions include subscription fees for our data products and license fees in connection with our segment and sector private market indices. The number of trades, dollar volume of trades, and Net Take Rate are the key business metrics we monitor to evaluate the financial performance of our marketplace solutions business.

Custodial administration fees — We generate revenue from cash administration fees, account maintenance fees, asset fees, and transaction fees. Cash administration fees are based on prevailing interest rates and client cash balances and currently make up the majority of custodial administration fee revenue. With respect to the account maintenance fees, we assess a flat quarterly fee per account, with additional fees based on the number and types of assets held and the number and type of transactions executed. The account revenues depend on the number of total custodial accounts, which include accounts clients opened directly with us and the activity within these accounts, as well as accounts we custody on behalf of partners.

# Transaction-based expenses

Transaction-based expenses represent third-party fees incurred to support our marketplace and custody solutions. These include, but are not limited to, third-party broker fees and transfer fees related to placement provided to brokerage clients to facilitate transactions and to a lesser extent those for fund management, and fund settlement expenses that relate to services provided to the Forge-managed SAFs. Custodial transaction fees include third-party vault storage fees for precious metals we custody. We generally expect these expenses to increase in absolute dollars as our revenue grows.

### Compensation and benefits

Compensation and benefits expense is our most significant operating expense and includes employee wages, bonuses, share-based compensation, severance costs, benefits, and employer taxes. The incentive component of our compensation and benefits expense consists of amounts paid on the achievement of sales targets and discretionary bonuses, which are based on both our financial performance and individual employee performance. While we expect our compensation and benefits expense to increase as our revenue grows and we hire additional personnel to support new products and services, in the near term, we are focused on aligning our headcount with current business needs. The share-based compensation component of compensation and benefits expense may or may not increase as we continue to align our headcount with current and future business needs.

### Professional services

Professional services expense includes fees for accounting, tax, auditing, legal, and regulatory services, as well as consulting services received in connection with strategic and technology initiatives. We have and may continue to incur additional professional services expenses relating to public company regulatory requirements and customary practices.

# Advertising and market development

Advertising and market development is an important driver of our value and we intend to continue making meaningful investments in the Forge brand and growth marketing. This includes brand advertising, thought leadership, content marketing, public relations, partnerships, and other strategies that amplify our brand. We have a rigorous approach to measuring client lifetime value and optimizing our client acquisition investments according to market dynamics and effective return on investment ("ROI"). We manage our discretionary expenses in growth marketing in real-time, as audience-specific dynamics show positive ROI. We generally expect our marketing expenses to increase in the long term in absolute dollars but manage our spend judiciously and adapt as market conditions evolve.

### Rent and occupancy

Rent and occupancy expense is related to our leased property and includes rent, maintenance, real estate taxes, utilities, impairment and other related costs.

### Technology and communications

Technology and communications consist of costs for our hosting fees paid to third-party data centers, software development engineers, and maintenance of our computer hardware and software required to support our technology and cybersecurity. Technology and communications also include costs for network connections for our electronic platforms and telecommunications. We generally expect our technology and communications expense to increase over the long term as we continue to innovate on our offerings and services and increase headcount.

# General and administrative

General and administrative includes insurance, travel and entertainment, allowances for bad debt, reserves for contingent losses including legal proceedings, and other general and administrative costs.

# Depreciation and amortization

Depreciation and amortization is attributable to property and equipment, intangible assets, and capitalized internal-use software.

### Interest income

Interest income primarily includes interest income earned on our cash, cash equivalents, investments and term deposits.

### Change in fair value of warrant liabilities

Changes in the fair value of warrant liabilities are related to warrant liabilities that are marked-to-market each reporting period with the change in fair value recorded in the accompanying unaudited condensed consolidated statements of operations until the warrants are exercised, expire, or other facts and circumstances that could lead the warrant liabilities to be reclassified to stockholders' equity occur.

### Other income, net

Other income, net, includes realized and unrealized gains and losses in connection with investments, sublease income, and other non-operating income and expenditures.

# Provision for income taxes

Provision for income taxes consists of federal, state, and foreign income taxes. We maintain a valuation allowance against deferred tax assets net of deferred tax liabilities, with the exception of certain indefinite-lived liabilities, as we have concluded it is not more likely than not that we will realize our net deferred tax assets.

# **Results of Operations**

The following table sets forth our unaudited condensed consolidated statements of operations for the interim periods presented. Percentages may not be replicated based on the rounded figures presented.

	Three Mor	Ended		Six Months E	nded June 30			
(in thousands)	June 30, 2025		March 31, 2025	June 30, 2025			June 30, 2024	
Total revenues, less transaction-based expenses	\$ 27,584	\$	25,104	\$	52,688	\$	41,239	
Operating expenses:								
Compensation and benefits	27,193		29,491		56,684		58,627	
Other	13,226		12,082		25,308		25,273	
Total operating expenses	40,419		41,573		81,992		83,900	
Operating loss	(12,835)		(16,469)		(29,304)		(42,661)	
Total interest and other (expense) income	585		1,287		1,872		10,101	
Loss before provision for income taxes	(12,250)		(15,182)		(27,432)		(32,560)	
Provision for income taxes	189		1,016		1,205		474	
Net loss	(12,439)		(16,198)		(28,637)		(33,034)	
Net income (loss) attributable to noncontrolling interest	141		(26)		115		(686)	
Net loss attributable to Forge Global Holdings, Inc.	\$ (12,580)	\$	(16,172)	\$	(28,752)	\$	(32,348)	

## Revenue

		Three Mo	nths F	Ended	Q	οQ		Six Months H	Inde	d June 30	YoY			
(in thousands)	June 30, 2025		March 31, 2025		Change	% Change	June 30, 2025		June 30, 2024		Change		% Change	
Marketplace revenue	\$	18,597	\$	15,997	\$ 2,600	16 %	\$	34,594	\$	20,199	\$	14,395	71 %	
Custodial administration fees		9,142		9,299	(157)	(2)%		18,441		21,325		(2,884)	(14)%	
Total revenues		27,739		25,296	2,443	10 %		53,035		41,524		11,511	28 %	
Transaction-based expenses:														
Transaction-based expenses		(155)		(192)	37	(19)%		(347)		(285)		(62)	22 %	
Total revenues, less transaction-based expenses	\$	27,584	\$	25,104	\$ 2,480	10 %	\$	52,688	\$	41,239	\$	11,449	28 %	

### Comparison of the Three Months Ended June 30, 2025 and March 31, 2025

Total revenues, less transaction-based expenses increased \$2.5 million, or 10%.

Marketplace revenue increased by \$2.6 million, or 16%, driven by a 9% increase in trading volume and a 15 basis point increase in net take rate. We attribute higher trading volumes to improved market dynamics that drove a diversity of new and re-engaged interest in our platform and larger average trade sizes.

Custodial administration fees decreased by \$0.2 million, or 2%, driven by lower cash administration fees attributable to a 4% decline in custodial client cash balances.

# Comparison of the Six Months Ended June 30, 2025 and June 30, 2024

Total revenues, less transaction-based expenses increased \$11.4 million, or 28%.

Marketplace revenue increased by \$14.4 million, or 71%, driven by a 110% increase in trading volume offset, in part, by a 52 basis point decrease in net take rate. We attribute higher trading volumes to improved market dynamics that drove a diversity of new and re-engaged interest in our platform and larger average trade sizes. Net take rate in the six months ended June 30, 2025 was lower due to large block transactions executed at lower net take rates and generally declining net take rates as trading volume increases.

Custodial administration fees decreased by \$2.9 million, or 14%, driven by lower cash administration fees attributable to lower interest rates and an 11% decline in custodial client cash balances.

### **Operating Expenses**

### Compensation and benefits

		Three Mo	nths	Ended	Qo		Six Mont	hs E	nded		YoY			
(in thousands)	June 30, 2025		]	March 31, 2025	Change	% Change	J	June 30, 2025		June 30, 2024		Change	% Change	
Salary	\$	13,288	\$	13,334	\$ (46)	(0.3)%	\$	26,622	\$	28,335	\$	(1,713)	(6)%	
Incentive compensation and other bonus		7,381		6,633	748	11 %		14,014		8,778		5,236	60 %	
Share-based compensation		3,436		6,519	(3,083)	(47)%		9,955		17,326		(7,371)	(43)%	
Severance		1,212		1,323	(111)	(8)%		2,535		1,008		1,527	151 %	
Benefits and other		1,876		1,682	194	12 %		3,558		3,180		378	12 %	
Total compensation and benefits	\$	27,193	\$	29,491	\$ (2,298)	(8)%	\$	56,684	\$	58,627	\$	(1,943)	(3)%	

### Comparison of the Three Months Ended June 30, 2025 and March 31, 2025

Compensation and benefits expense decreased \$2.3 million, or 8%.

Salary expense was flat, in line with the continuing focus on cost management and hiring. Severance expense in the current quarter relates primarily to a reorganization in connection with the launch of the Company's next generation marketplace platform, while costs in the prior quarter relate primarily to the CFO transition.

Incentive compensation and other bonus expense increased \$0.7 million, primarily driven by the increase in marketplace revenue.

Share-based compensation expense decreased \$3.1 million. Share-based compensation expense in the current quarter includes accelerated amortization in connection with the reorganization, while costs in the prior quarter relate primarily to the CFO transition and immediately vested one-time grants to certain employees.

# Comparison of the Six Months Ended June 30, 2025 and June 30, 2024

Compensation and benefits expense decreased \$1.9 million, or 3%.

Salary expense decreased \$1.7 million, or 6%, due to the realization of cost initiatives taken by the Company in the second half of 2024. The \$1.5 million increase in severance expense was primarily driven by costs recognized in the six months ended June 30, 2025 in connection with the Company's marketplace reorganization and CFO transition.

Incentive compensation and other bonus expense increased \$5.2 million primarily driven by the increase in marketplace revenue.

Share-based compensation expense decreased \$7.4 million primarily related to lower amortization in connection with prior years' grants and delays in finalizing 2025 annual grants for certain executives and employees offset in part by accelerated amortization in connection with the marketplace reorganization and CFO transition. The Company expects share-based compensation to continue to decline as high grant date value awards from 2022 completely vest.

# Other Operating Expenses

		Three Mo	nths	Ended		Qe	οQ		Six Mont	hs En	ded	 YoY			
(in thousands)	June 30, 2025 N		Mai	March 31, 2025		Change	% Change	Jι	une 30, 2025	June 30, 2024		Change	% Change		
Technology and communications	\$	4,667	\$	4,349	\$	318	7 %	\$	9,016	\$	5,709	\$ 3,307	58 %		
General and administrative		2,144		2,254		(110)	(5)%		4,398		7,570	(3,172)	(42)%		
Professional services		1,204		2,332		(1,128)	(48)%		3,536		3,822	(286)	(7)%		
Advertising and market development		1,528		1,215		313	26 %		2,743		2,333	410	18 %		
Acquisition-related transaction costs		1,988		_		1,988	n/m		1,988		_	1,988	n/m		
Depreciation and amortization	ı	909		986		(77)	(8)%		1,895		3,597	(1,702)	(47)%		
Rent and occupancy		786		946		(160)	(17)%		1,732		2,242	(510)	(23)%		
Total other operating expenses	\$	13,226	\$	12,082	\$	1,144	9 %	\$	25,308	\$	25,273	\$ 35	0.1 %		

n/m not meaningful

### Comparison of the Three Months Ended June 30, 2025 and March 31, 2025

Other operating expenses increased \$1.1 million, or 9%.

The primary contributors to the quarter over quarter increase relate to acquisition-related expenses in connection with the Accuidity acquisition, which closed on July 1, 2025, higher advertising and marketing spend for client acquisition and new product launch, and increased third-party software engineer expense recorded in technology and communication expense reflecting the Company's continuing investment in its flagship marketplace and custody solutions through the use of off-shore resources. These increases were offset by other cost reductions in line with the Company's continued focus on expense management and control.

# Comparison of the Six Months Ended June 30, 2025 and June 30, 2024

Other operating expenses were flat for the six months ended June 30, 2025 compared to the six months ended June 30, 2024.

Acquisition-related costs in connection with the Accuidity acquisition that closed on July 1, 2025, higher advertising and market development expenses and higher technology and communication expense for offshore third party software engineers were offset by lower depreciation and amortization, rent and occupancy and professional services expenses.

Other operating expenses in the six months ended June 30, 2024, also included certain non-recurring charges, including \$2.8 million in legal settlement costs recorded in general and miscellaneous expense and a \$0.2 million right-of-use asset impairment charge recorded in rent and occupancy expense.

#### Total interest and other income

	<b>Three Months Ended</b>			QoQ		Six Months Ended			nded	YoY		
(in thousands)	June	30, 2025	Mar	ch 31, 2025	Change	% Change	Ju	ine 30, 2025	Ju	ne 30, 2024	Change	% Change
Interest income	\$	803	\$	1,042	\$ (239)	(23)%	\$	1,845	\$	3,204 \$	(1,359)	(42)%
Change in fair value of warrant liabilities		(294)		191	(485)	(254)%		(103)		6,727	(6,830)	(102)%
Other income, net		76		54	22	41 %		130		170	(40)	(23)%
Total interest and other (expense) income	\$	585	\$	1,287	\$ (702)	(55)%	\$	1,872	\$	10,101 \$	(8,229)	(81)%

#### Comparison of the Three Months Ended June 30, 2025 and March 31, 2025

Interest income declined \$0.2 million driven by lower investment yields and declining cash and cash equivalent balances.

Changes in the fair value of warrant liabilities were driven by changes in key valuation assumptions including the Company's share price and share price volatility as of the valuation dates. For the three months ended June 30, 2025, the Company recognized a \$0.3 million loss from warrant revaluations, in comparison to a \$0.2 million gain for the three months ended March 31, 2025. See Note 3, "Fair Value Measurements" of the notes to our unaudited condensed consolidated financial statements.

## Comparison of the Six Months Ended June 30, 2025 and June 30, 2024

Interest income decreased \$1.4 million driven by lower investment yields and declining cash and cash equivalent balances.

Changes in the fair value of warrant liabilities were driven by changes in key valuation assumptions including the Company's share price and share price volatility as of the valuation dates. The unfavorable change of \$0.1 million in the fair value of warrant liabilities in the current year compares to the favorable change of \$6.7 million in the prior year.

#### **Liquidity and Capital Resources**

We have financed our operations primarily through revenue from operations and issuances of securities. Our primary requirements for liquidity and capital are to finance working capital and capital expenditures.

As of June 30, 2025, our principal sources of liquidity are our cash and cash equivalents balance of \$54.3 million and investments of \$26.4 million.

We believe our existing liquidity as of June 30, 2025 will be sufficient to meet our operating working capital and capital expenditure requirements for the next twelve months and the foreseeable future. Our future equity and financing requirements will depend on many factors including our growth rate, the timing and extent of spending to support development of our platform, and the expansion of sales and marketing activities. Although we currently are not a party to any financing agreement and do not have any understanding with any third parties with respect to potential investments in, or acquisitions of, businesses or technologies, we may enter into these types of arrangements in the future, which could also require us to seek additional equity or debt financing. In the event that additional financing is required from outside sources, we may not be able to raise it on terms acceptable to us or at all. If we are unable to raise additional capital when desired, our business, results of operations, and financial condition would be materially and adversely affected.

We intend to continue to make investments in product development, sales efforts, and additional general and administrative costs in connection with operating as a public company. We expect to continue to maintain financing flexibility in the current market conditions. As a result, we may require additional capital resources to execute strategic initiatives to grow our business.

#### Investments

#### **Table of Contents**

Our investment policy and strategy are focused on the preservation of capital and supporting our liquidity requirements. We invest in highly-rated debt securities that are considered held-for-trading investments with average duration in the portfolio of less than 3 months and a maximum maturity of 6 months. Based on our investment positions as of June 30, 2025, a hypothetical 100 basis point increase in interest rates across all maturities would not be significant. Losses would only be realized if we sold the investments prior to maturity or in the event of a default by the issuer.

We seek to reduce credit risk through prudent asset selection, actively monitoring our investment portfolio and the underlying credit quality of our holdings. Our pre-acquisition due diligence and processes for monitoring performance include the evaluation of, among other things, credit and risk ratings.

#### Share Repurchase Program

In March 2025, our board of directors approved a share repurchase program of up to \$10.0 million. During the three and six months ended June 30, 2025, we have repurchased 314,701 shares at an average price of approximately \$13.15 per share and approximately \$5.9 million remains available for repurchase under the share repurchase program.

## Cash Flow Summary

The following table summarizes our cash flows for the periods presented:

		Six Months Ended June 30,			
(in thousands)	2025		2024		
Net cash used in:					
Operating activities	\$	(20,605)	\$	(26,806)	
Investing activities	\$	(26,353)	\$	5,892	
Financing activities	\$	(4,919)	\$	(2,976)	

## Operating Activities

Cash used in operating activities for the six months ended June 30, 2025 of \$20.6 million was primarily driven by our net loss of \$28.6 million, adjusted for non-cash charges of \$13.1 million and net cash outflows of \$5.1 million in connection with changes in our operating assets and liabilities. Non-cash charges primarily consist of share-based compensation, depreciation and amortization, amortization of right-of-use assets, and changes in fair value of warrant liabilities. Cash outflows in connection with changes in our operating assets and liabilities are primarily driven by payment of annual incentive compensation and higher accounts receivable, net as a result of higher revenue.

Cash used in operating activities for the six months ended June 30, 2024 of \$26.8 million was primarily driven by our net loss of \$33.0 million, adjusted for non-cash charges of \$15.9 million and net cash outflows of \$9.7 million in connection with changes in our operating assets and liabilities. Non-cash charges primarily consist of share-based compensation, depreciation and amortization, amortization of right-of-use assets, and changes in fair value of warrant liabilities. Cash outflows in connection with changes in our operating assets and liabilities were primarily driven by payment of annual incentive compensation and payments in connection with legal settlements.

## Investing Activities

Cash used in investing activities for the six months ended June 30, 2025 of \$26.4 million was primarily driven by investment of excess corporate cash balances into short-term investments.

Cash provided by investing activities was \$5.9 million for the six months ended June 30, 2024, which consisted primarily of cash received for the maturity of term deposits.

# Financing Activities

Cash used in financing activities was \$4.9 million and \$3.0 million for the six months ended June 30, 2025 and June 30, 2024, respectively, and relates to share repurchases and stock option and other equity award activities and settlements.

#### **Contractual Obligations**

Our contractual obligations have not changed materially outside of the normal course of business as disclosed in our Annual Report, except as described in Note 6, "Commitments and Contingencies," to our unaudited condensed consolidated financial statements included elsewhere in this Report.

## **Critical Accounting Policies and Estimates**

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in our unaudited condensed consolidated financial statements and accompanying notes. We base our estimates on historical experience, current business factors, and various other assumptions that we believe are necessary to consider forming a basis for making judgments about the carrying values of assets and liabilities, the recorded amounts of revenue and expenses, and the disclosure of contingent assets and liabilities. We are subject to uncertainties such as the impact of future events, economic and political factors, and changes in our business environment; therefore, actual results could differ from these estimates. Accordingly, the accounting estimates used in the preparation of our unaudited condensed consolidated financial statements will change as new events occur, as more experience is acquired, as additional information is obtained and as our operating environment changes. Changes in estimates are made when circumstances warrant. Such changes in estimates and refinements in estimation methodologies are reflected in reported results of operations; if material, the effects of changes in estimates are disclosed in the notes to our unaudited condensed consolidated financial statements.

On an ongoing basis, we evaluate our estimates and assumptions. Our actual results may differ from these estimates under different assumptions or conditions.

#### Goodwill and Other Intangible Assets, Net

Goodwill represents the excess of the aggregate fair value of the consideration transferred in a business combination over the fair value of the assets acquired, net of liabilities assumed. Goodwill is not amortized but is tested for impairment annually on October 1, or more frequently if events or changes in circumstances indicate the goodwill may be impaired. These events or circumstances could include a significant change in the business climate, regulatory environment, established business plans, operating performance indicators, or competition. Potential impairment indicators may also include, but are not limited to, (i) the results of the Company's most recent annual or interim impairment testing, (ii) downward revisions to internal forecasts, (iii) declines in the Company's market capitalization below its book value, and the magnitude and duration of those declines, (iv) a reorganization resulting in a change to the Company's operating segments, and (v) other macroeconomic factors, such as increases in interest rates that may affect the weighted average cost of capital or volatility in the equity and debt markets. If the Company's market capitalization continues to decline or future performance varies from current expectations, assumptions, or estimates, including assumptions related to current macroeconomic uncertainties, this may trigger a future impairment charge.

We performed our annual goodwill impairment test as of October 1, 2024 and a quantitative assessment as of December 31, 2024. The Company used a qualitative assessment as of June 30, 2025. The qualitative analysis evaluated factors, including, but not limited to, economic, market and industry conditions and the overall financial performance of the reporting unit. Based on the results of our qualitative impairment assessment, we concluded that it is more likely than not that the fair value of our reporting unit sufficiently exceeded its carrying value and as such, did not require further assessment. No impairment charges were recognized during the six months ended June 30, 2025 and June 30, 2024.

There have been no other material changes to our critical accounting policies and estimates as compared to those described in the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations" set forth in our Annual Report.

#### **Recent Accounting Pronouncements**

See the section titled "Summary of Significant Accounting Policies" in Note 2 of the notes to our unaudited condensed consolidated financial statements in this Report.

# Item 3. Quantitative and Qualitative Disclosures About Market Risk

#### **Table of Contents**

We are a smaller reporting company as defined by Rule 12b-2 of the Exchange Act and are not required to provide the information otherwise required under this item.

#### **Item 4. Controls and Procedures**

# **Evaluation of Disclosure Controls and Procedures**

Our management, with the participation of our Chief Executive Officer and Chief Financial Officer, has evaluated the effectiveness of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act). Based on that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that, because of the material weakness in internal control over financial reporting disclosed below, our disclosure controls and procedures were not effective as of June 30, 2025.

After giving full consideration to the material weakness, and the additional analyses and other procedures the Company performed to ensure the unaudited condensed consolidated financial statements included in this Quarterly Report on Form 10-Q were prepared in accordance with GAAP, management has concluded that the unaudited condensed consolidated financial statements present fairly, in all material respects, our financial position, results of operations and cash flows for the periods disclosed in conformity with GAAP. The Company is in the process of developing and implementing a remediation plan for the material weakness, which is described below.

#### Material Weakness in Internal Control over Financial Reporting

Management is responsible for establishing and maintaining adequate internal control over financial reporting as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act. Our management, including our Chief Executive Officer and Chief Financial Officer, under the oversight of the Audit Committee of our Board of Directors, conducted an evaluation of the effectiveness of our internal control over financial reporting as of June 30, 2025. In making this assessment, our management used the criteria established in Internal Control-Integrated Framework (2013) issued by the Committee of Sponsoring Organizations of the Treadway Commission ("COSO"). Based on that evaluation, management concluded that the Company's internal control over financial reporting was not effective as of June 30, 2025 due to the material weakness in internal control over financial reporting, described below.

Our internal control system is designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with U.S. GAAP. Because of inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

A material weakness is a deficiency, or combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of the annual or interim financial statements will not be prevented or detected on a timely basis.

Subsequent to the issuance of our earnings press release for the fiscal quarter ended June 30, 2025, management identified a material weakness in internal control over the calculation of fair value of warrant liabilities. Our controls over valuation of warrant liabilities were not sufficiently designed to capture the effect of changes in the number of warrants outstanding when calculating the fair value of warrant liabilities as of the end of the reporting period and our analysis of fluctuations during the period were not sufficiently designed to identify the error. These control deficiencies were the result of our risk assessment not identifying the risks associated with new, unusual or significant transactions and designing controls to address such risks.

Management concluded that the item noted above constituted a material weakness in its internal control over financial reporting as of June 30, 2025.

The error resulting from this material weakness did not cause material misstatements in previously issued annual or interim consolidated financial statements. The error was corrected prior to the issuance of the Company's unaudited condensed consolidated financial statements as of June 30, 2025, and for the three and six months then ended. If not remediated timely, the deficiency described above could result in a material misstatement to the future annual or interim consolidated financial statements.

#### **Table of Contents**

#### Management's Plan to Remediate the Material Weakness

The Company is committed to remediating the material weakness in a timely manner. We plan to design and implement controls to ensure new, unusual or significant transaction risks are addressed, including the following remediation actions:

- Reconciliation and review control requirements over the number of outstanding warrants at each valuation date;
- Controls related to its monthly and quarterly analysis of valuation inputs and assumptions to enhance the precision of its analytical reviews to assess whether the impact of new, significant or unusual transactions are properly reflected in the financial statements; and
- Incorporating an accounting and disclosure risk assessment process and control for the Company's review of new, unusual or significant transactions and their related accounting and disclosure impacts.

While the Company believes these efforts will improve its internal controls and address the underlying cause of the material weakness, the material weakness will not be remediated until the remediation plan has been fully implemented and Management has concluded that the improvements added to the current control environment are operating effectively for a sufficient period of time. The Company cannot be certain that the steps being taken will be sufficient to remediate the control deficiencies that led to the material weakness in internal control over financial reporting or prevent future material weaknesses or control deficiencies from occurring. In addition, the Company cannot be certain that all material weaknesses in our internal control over financial reporting have been identified, or that in the future it will not have additional material weaknesses in its internal control over financial reporting.

## Changes in Internal Control over Financial Reporting

Other than the material weakness discussed above, there have not been any changes in our internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) during the period covered by this Report that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

#### Inherent Limitations on Effectiveness of Controls

Our management, including our Chief Executive Officer and Chief Financial Officer, do not expect that our disclosure controls or our internal control over financial reporting will prevent all errors and all fraud. A control system, no matter how well conceived and operated, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. Further, the design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, have been detected. These inherent limitations include the realities that judgments in decision-making can be faulty, and that breakdowns can occur because of a simple error or mistake. Additionally, controls can be circumvented by the individual acts of some persons, by collusion of two or more people or by management override of the controls. The design of any system of controls is also based in part upon certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions; over time, controls may become inadequate because of changes in conditions, or the degree of compliance with policies or procedures may deteriorate. Because of the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected.

#### **Part II- Other Information**

# **Item 1. Legal Proceedings**

Information regarding legal proceedings is available in Note 6, "Commitments and Contingencies" to the unaudited condensed consolidated financial statements in this Report.

## Item 1A. Risk Factors

There have been no material changes from the risk factors previously described in the section titled "Item 1A. Risk Factors" in our Annual Report.

## Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Issuer Purchases of Equity Securities

The following table summarizes the share repurchase activity for the three months ended June 30, 2025:

	Total Number of Shares Purchased	Average Price Paid Per Share (1)	Total Number of Shares Purchased as Part of Publicly Announced Program	Ŝ	proximate Dollar Value of Shares That May Yet Be Purchased Under the Program (2) (in millions)
April 1, 2025 to April 30, 2025	264,701	\$13.11	264,701	\$	6,529,770
May 1, 2025 to May 31, 2025	50,000	\$13.36	50,000	\$	5,861,770
June 1, 2025 to June 30, 2025	_	_	_	\$	_
Total	314,701	\$13.15	314,701		

(1) Average price paid per share excludes broker commissions and fees.

#### **Item 3. Defaults Upon Senior Securities**

None.

# **Item 4. Mine Safety Disclosures**

None.

#### **Item 5. Other Information**

# **Securities Trading Plans of Directors and Executive Officers**

During the three months ended June 30, 2025, no director or officer, as defined in Rule 16a-1(f) under the Exchange Act, adopted or terminated a "Rule 10b5-1 trading arrangement" or a "non-Rule 10b5-1 trading arrangement".

## Item 6. Exhibits

The exhibits listed in the accompanying Exhibit Index are filed or incorporated by reference as part of this Report.

#### **Exhibit Index**

<sup>(2)</sup> In March 2025, the Company's board of directors authorized the repurchase of up to \$10.0 million in shares of our common stock. For additional information, refer to Note 2, "Summary of Significant Accounting Policies" to the unaudited condensed consolidated financial statements in this Report.

Exhibit Number	Description	Incorporated by Reference From Form	Incorporated by Reference From Exhibit Number	Date Filed
2.1*§	Agreement and Plan of Merger, dated July 1, 2025, among the Company, Accuidity, LLC, and the other parties thereto.	Filed herewith		
3.1	Certificate of Amendment to the Certificate of Incorporation of Forge Global Holdings, Inc.	8-K	3.1	April 14, 2025
10.1	Consulting Agreement, effective April 25, 2025, by and between Larry Leibowitz and the registrant.	Filed herewith		
10.2	Separation and General Release Agreement, effective May 2, 2025, by and between Jennifer Phillips and the registrant.	Filed herewith		
10.3	Consulting Agreement, dated June 10, 2025, by and between Jennifer Phillips and the registrant.	Filed herewith		
10.4	Outside Director Compensation Policy, as amended on June 19, 2025	Filed herewith		
10.5	Amended and Restated Forge Global Holdings, Inc. 2025 Inducement Plan.	S-8	99.1	July 15, 2025
31.1	Certification of the Chief Executive Officer pursuant to Section 302 of the Sarbanes- Oxley Act.  Certification of the Chief Financial Officer pursuant to Section 302 of the Sarbanes-	Filed herewith		
31.2	Oxley Act.	Filed herewith		
32.1*	Certifications of the Chief Executive Officer and Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act.	Furnished herewith		
	Inline XBRL Instance Document (the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL			
101.INS	document).	Filed herewith		
101.SCH	Inline XBRL Taxonomy Extension Schema Document.	Filed herewith		
101.CAL	Inline XBRL Taxonomy Extension Calculation Linkbase Document.	Filed herewith		
101.DEF	Inline XBRL Taxonomy Extension Definition Linkbase Document.	Filed herewith		
101.LAB	Inline XBRL Taxonomy Extension Label Linkbase Document.	Filed herewith		
101.PRE	Inline XBRL Taxonomy Extension Presentation Linkbase Document.	Filed herewith		
104	Cover Page Interactive Data File (formatted as Inline XBRL and contained in Exhibit 101).	Filed herewith		

<sup>\*</sup> The certifications furnished in Exhibit 32.1 hereto are deemed to accompany this Report and will not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended. Such certifications will not be deemed to be incorporated by reference into any filings under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, except to the extent that the registrant specifically incorporates it by reference.

 $<sup>\</sup>S$  Schedules and exhibits have been omitted pursuant to Item 601(a)(5) of Regulation S-K. Certain portions of this exhibit have been omitted in accordance with Item 601(b)(10)(iv) of Regulation S-K. The Company agrees to furnish supplementally to the SEC a copy of any omitted information upon request by the SEC.

## **Signatures**

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this Report to be signed on its behalf by the undersigned thereunto duly authorized.

Forge Global Holdings, Inc.

Date: August 7, 2025 By: /s/ Kelly Rodriques

Kelly Rodriques

Chief Executive Officer (Principal Executive

Officer)

Date: August 7, 2025 By: /s/ James Nevin

James Nevin

Chief Financial Officer (Principal Financial

Officer)

Exhibit 2.1

Agreement and Plan of Merger

by and among

Forge Global Holdings, Inc.,

Margo Merger Sub I, LLC,

Margo Merger Sub II, LLC,

Accuidity, LLC, and

KOSTKA LLC, as the Securityholder Representative

July 1, 2025

# ARTICLE 1 THE MERGERS 2

1.1	The Closing 2
1.2	The Mergers 2
1.3	Charter Documents; Managers and Officers 3
1.4	Effect of the First Merger on Company Units and Merger Sub I Units 4
1.5	Effect of the Second Merger on Membership Units of the Surviving Company and Membership Interests of Merger Sub II 5
1.6	Adjustment Holdback Amount; Indemnity Escrow Amount; Expense Fund; Closing Payments 5
1.7	Payment of Closing Consideration; Exchange 6
1.8	Post-Closing Reconciliation 7
1.9	Fractional Shares 11
1.10	Equitable Adjustments 11
1.11	Tax Withholding 12
1.12	Further Assurances 12
1.13	Tax Consequences 12
1.14	Earn-Out 13
1.15	Legends 22
	ARTICLE 2 REPRESENTATIONS AND WARRANTIES OF THE COMPANY 23
2.1	Organization and Good Standing 23
2.2	Subsidiaries 24
2.3	Power, Authorization and Validity 24
2.4	Capitalization of the Company 25
2.5	No Conflict; Required Consents 27
2.6	Litigation 27
2.7	Taxes 28
2.8	Related Party Transactions 32
2.9	Company Financial Statements 33
2.10	Title to Properties 34
2.11	Absence of Certain Changes 35
2.12	Company Material Contracts 37
2.13	Intellectual Property 40
2.14	Privacy and Data Protection 47

2.15	Compliance with Laws 49
2.16	Employees, ERISA and Other Compliance 52
2.17	Books and Records; Bank Accounts 57
2.18	Insurance 57
2.19	Environmental Matters 58
2.20	Customers and Suppliers 58
2.21	Anti-Money Laundering Laws 59
2.22	Anti-Corruption and Anti-Bribery Laws 59
2.23	Trade Compliance 60
2.24	Outbound Investment 60
2.25	Brokers; Transaction Expenses 61
2.26	Social Media 61
2.27	Capitalization of the Aggregator 61
2.28	No Other Representations and Warranties; Non-Reliance 62
A	RTICLE 3 REPRESENTATIONS AND WARRANTIES OF PARENT AND THE MERGER SUBS 62
3.1	Organization and Good Standing 62
3.2	Power, Authorization and Validity 63
3.3	No Conflicts; Required Consents 63
3.4	Capitalization 64
3.5	Parent Shares 65
3.6	Merger Subs 65
3.7	SEC Reports; Internal Control 65
3.8	Cash Resources 67
3.9	Compliance with Laws 67
3.10	Litigation 68
3.11	Certain Restraints 68
3.12	No Other Representations and Warranties; Non-Reliance 68
	ARTICLE 4 COVENANTS 68
4.1	Employment Arrangements; Termination of Certain Company Benefit Arrangements 69
4.2	Indemnification of Company Managers and Officers 70

4.3 Public Information 71

4.4	Listing 71
	ARTICLE 5 ESCROW AGREEMENT AND EXCHANGE AGENT AGREEMENT
5.1	Escrow Agreement 71
5.2	Exchange Agent Agreement 71
	ARTICLE 6 CLOSING DELIVERABLES 71
6.1	Member Deliverables 72
6.2	Employment Matters 72
6.3	Other Deliverables 72
	ARTICLE 7 SURVIVAL; INDEMNIFICATION 73
7.1	Survival 73
7.2	Indemnification of the Indemnified Parties 74
7.3	Limitations 75
7.4	Claim Notice 77
7.5	Resolution of Claim Notice 77
7.6	Defense and Settlement of Third-Party Claims 78
7.7	Indemnity Escrow Arrangements 79
7.8	Payment of Indemnity Escrow Amount 80
7.9	Right to Satisfy Indemnification Claims by Reducing Earnout Amounts 80
7.10	Tax Consequences of Indemnification Payments 81
7.11	No Right of Contribution 81
7.12	Exclusive Remedy 82
7.13	Appointment of Securityholder Representative 82
	ARTICLE 8 TAX MATTERS 84
8.1	Tax Returns 84
8.2	Cooperation 86
8.3	Tax Audits 86
8.4	Transfer Taxes 87
8.5	Refunds 87
8.6	Post-Closing Actions 88
	ARTICLE 9 MISCELLANEOUS 88
9.1	Governing Law; Jurisdiction; Venue 88
9.2	Assignment 89

9.3 Severability 89

71

- 9.4 Counterparts 89
- 9.5 Other Remedies 89
- 9.6 Amendments and Waivers 90
- 9.7 Expenses 90
- 9.8 Notices 90
- 9.9 WAIVER OF JURY TRIAL 92
- 9.10 Interpretation; Rules of Construction 92
- 9.11 Disclosure Schedule 93
- 9.12 Agreement Binding on the Parties; Third-Party Beneficiary Rights 93
- 9.13 Public Announcement 93
- 9.14 Confidentiality 94
- 9.15 Entire Agreement 94
- 9.16 Acknowledgement 94
- 9.17 Waiver of Conflicts Regarding Representation; Non-Assertion of Attorney-Client Privilege 94

# TABLE OF ANNEXES, EXHIBITS AND SCHEDULES

# ANNEXES AND EXHIBITS

## Annex A Certain Definitions

Exhibit A Form of Joinder Agreement

Exhibit B Form of Investor Questionnaire

Exhibit C-1 Form of Founder Restricted Stock Agreement

Exhibit C-2 Form of Aggregator Acknowledgement Agreement

Exhibit C-3 Form of Holder Restricted Stock Agreement

Exhibit D Form of First Certificate of Merger

Exhibit E Form of Second Certificate of Merger

Exhibit F Form of Letter of Transmittal

Exhibit G Closing Statement

Exhibit I Escrow Agreement

# **SCHEDULES**

Schedule 1.1(a) Signing Members

Schedule 1.1(b) Company Entity Representatives

Schedule 1.1(c) Parent Entity Representatives

Schedule 1.8 Closing Statement Information

Schedule 1.8(a) Agreed Accounting Principles

Schedule 1.14(a) Earn Out Targets and Thresholds

Schedule 1.14(a)(ix) Business Management and Governance

Schedule 7.2(a)(vii) Additional Indemnifiable Matters

Schedule A Expense Fund Sharing Percentage

## AGREEMENT AND PLAN OF MERGER

This Agreement and Plan of Merger (this "Agreement") is made and entered into as of July 1, 2025 (the "Agreement Date") by and among Forge Global Holdings, Inc., a Delaware corporation ("Parent"), Margo Merger Sub I, LLC, a Delaware limited liability company and a wholly owned indirect subsidiary of Parent ("Merger Sub I"), Margo Merger Sub II, LLC, a Delaware limited liability company and a wholly owned indirect subsidiary of Parent ("Merger Sub II" and, together with Merger Sub I, the "Merger Subs"), Accuidity, LLC, a Delaware limited liability company (the "Company"), and Kostka LLC, a New York limited liability company, solely in such Person's capacity as representative of the securityholders of the Company for certain purposes described in this Agreement (the "Securityholder Representative"). Unless the context otherwise requires, references herein to the "parties" means Parent, the Merger Subs, the Company and the Securityholder Representative. Certain capitalized terms used herein have the meanings set forth in Annex A to this Agreement.

#### Recitals

- A. The Parties intend that Merger Sub I shall merge with and into the Company (the "First Merger"), with the Company to be the surviving company of the First Merger, upon the terms and subject to the conditions set forth in this Agreement and in accordance with the Delaware Limited Liability Company Act (the "DLLCA"), and, as part of the same overall transaction, the Company would then merge with and into Merger Sub II (the "Second Merger" and, together with the First Merger, the "Mergers"), with Merger Sub II to be the surviving company of the Second Merger, on the terms and subject to the conditions set forth in this Agreement and pursuant to the DLLCA.
- B. The Governing Bodies of Parent, the Merger Subs and the Company have determined that the Mergers are in the best interests of their respective companies and stockholders or members, as applicable, and have approved and declared advisable the Mergers in accordance with their respective Charter Documents on the terms and subject to the conditions set forth in this Agreement.
- C. Concurrently with the execution and delivery of this Agreement, and as a condition and inducement to Parent's willingness to enter into this Agreement, the Persons identified on <u>Schedule 1.1(a)</u> (the "**Signing Members**") are executing and delivering to Parent (i) joinder agreements in substantially the form attached hereto as <u>Exhibit A</u> (the "**Joinder Agreement**"), and (ii) investor suitability questionnaires substantially in the form attached hereto as <u>Exhibit B</u> (the "**Investor Questionnaires**").
- D. Prior to the execution and delivery of this Agreement, members sufficient to deliver the Member Approval, including all Signing Members, have executed and delivered to the Company a member written consent approving this Agreement and the other transactions and agreements contemplated hereby (the "Written Consent"), and the Company has delivered the Written Consent to Parent.

- E. Concurrently with the execution and delivery of this Agreement, and as a condition and inducement to Parent's willingness to enter into this Agreement, (i) each Founder is executing and delivering to Parent (a) an employment agreement providing for employment with Parent and/or one of its Subsidiaries (an "Employment Agreement"), a proprietary information and additional covenants agreement (a "PICA") (together, the "Employment Documents"), and (b) a non-competition and nonsolicitation agreement (a "Non-Competition Agreement"), (ii) Kostka LLC is executing and delivering to Parent a restricted stock agreement in substantially the form attached hereto as Exhibit C-1 (the "Founder Restricted Stock Agreement"), pursuant to which Kostka LLC has agreed that a specified portion of the Closing Stock Consideration issued to it with respect to its Company Class A Units in connection with the Mergers (the "Founder Restricted Stock Amount") will be subject to certain restrictions on transfer and a possibility of forfeiture, (iii) the Aggregator is executing and delivering to Parent an acknowledgement agreement in substantially the form attached hereto as Exhibit C-2 (the "Aggregator Acknowledgement Agreement") pursuant to which the Aggregator has agreed that a specified portion of the Closing Stock Consideration issued to it with respect to its Company Profits Interest Units in connection with the Mergers (the "Aggregator Restricted Stock Amount") will be subject to certain restrictions on transfer in accordance with the terms thereof and (iv) each of [\*\*\*] is executing and delivering to Parent a restricted stock agreement in substantially the form attached hereto as Exhibit C-3 (together with the Founder Restricted Stock Agreement and the Aggregator Acknowledgment Agreement, the "Restricted Stock Agreements") pursuant to which it has agreed that a specified portion of the Closing Stock Consideration issued to it with respect to its Company Class A Units in connection with the Mergers (together with the Founder Restricted Stock Amount and the Aggregator Restricted Stock Amount, the "Restricted Stock Amounts") will be subject to certain restrictions on transfer in accordance with the terms thereof.
- F. Parent and the Company intend that the Mergers be treated as an integrated transaction that qualifies as a reorganization within the meaning of Section 368(a) of the Code.

# Agreement

Now, Therefore, in consideration of the foregoing and the representations, warranties, covenants and agreements contained herein, and intending to be legally bound hereby, the parties hereby agree as follows:

#### **ARTICLE 1**

## THE MERGERS

- 1.1 <u>The Closing</u>. The closing of the Mergers (the "Closing") shall take place through the electronic exchange of documentation simultaneously with the execution and delivery of this Agreement. The date on which the Closing occurs is referred to herein as the "Closing Date."
- 1.2 <u>The Mergers</u>. At the Closing, the Company and Merger Sub I shall cause the First Merger to be consummated by filing a certificate of merger, in substantially the form attached hereto as <u>Exhibit D</u> (the "**First Certificate of Merger**"), with the Secretary of State of

the State of Delaware in accordance with the DLLCA. The time of such filing and acceptance by the Secretary of State of the State of Delaware, or such later time as may be agreed in writing by Parent and the Company prior to the Closing and specified in the First Certificate of Merger, shall be referred to herein as the "Effective Time." At the Effective Time, Merger Sub I shall be merged with and into the Company, the separate existence of Merger Sub I shall cease, and the Company shall continue as the surviving limited liability company of the First Merger (the "Surviving Company"). As soon as practicable after the Effective Time, but in all cases within one (1) Business Day thereafter, the Surviving Company and Merger Sub II shall cause the Second Merger to be consummated by filing a certificate of merger, in substantially the form attached hereto as Exhibit E (the "Second Certificate of Merger"), with the Secretary of State of the State of Delaware in accordance with the DLLCA. The surviving entity of the Second Merger after the Second Effective Time is sometimes referred to herein as the "Surviving Entity". The time of such filing and acceptance by the Secretary of State of the State of Delaware, or such later time as may be agreed in writing by Parent and the Company prior to the Closing and specified in the Second Certificate of Merger, shall be referred to herein as the "Second Effective Time." The effects of the Mergers shall be as provided in this Agreement and the DLLCA.

1.3 <u>Charter Documents; Managers and Officers</u>. Unless otherwise agreed in writing by Parent and the Company prior to the Effective Time:

# (a) <u>The First Merger</u>.

- (i) at the Effective Time, by virtue of the First Merger and without any further action on the part of the Company, Merger Sub I or any other Person, (i) the certificate of formation of the Company shall be amended to conform to the certificate of formation of Merger Sub I as in effect immediately prior to the Effective Time (except that the name of the Surviving Company shall not be changed), and (ii) subject to Section 4.2(a), the Company LLC Agreement shall be amended to conform to the limited liability company agreement of Merger Sub I as in effect immediately prior to the Effective Time (except that the name of the Surviving Company shall not be changed) and, as so amended, shall become the limited liability company agreement of the Surviving Company; and
- (ii) at the Effective Time, (i) OpCo shall be the sole and managing member of the Surviving Company, and (ii) the officers of the Company as of immediately prior to the Effective Time shall be the initial officers of the Surviving Company immediately after the Effective Time, each to hold office until their respective successors are duly appointed or elected and qualified, as applicable, or their earlier death, resignation or removal.

# (b) <u>The Second Merger</u>.

(i) at the Second Effective Time, by virtue of the Second Merger and without any further action on the part of the Surviving Company, Merger Sub II or any other Person, (i) the certificate of formation of Merger Sub II shall remain as in effect immediately prior to the Second Effective Time and become the certificate of formation of the Surviving Entity (except that the name of the Surviving Entity shall be changed to the name of the Surviving Company), and (ii) subject to Section 4.2(a), the Merger Sub II LLC Agreement shall be amended to reflect the name of the Surviving Company as the name of the Surviving Entity and, as so amended, shall become the limited liability company agreement of the Surviving Entity; and

- (ii) at the Second Effective Time, (i) OpCo shall be the sole and managing member of the Surviving Entity, and (ii) the officers of the Surviving Company as of immediately prior to the Second Effective Time (and an additional representative, who will be the Chief Financial Officer of Parent) shall be the initial officers of the Surviving Entity immediately after the Second Effective Time, each to hold office until their respective successors are duly appointed or elected and qualified, as applicable, or their earlier death, resignation or removal.
- 1.4 <u>Effect of the First Merger on Company Units and Merger Sub I Units</u>. At the Effective Time, by virtue of the First Merger and without any further action on the part of the parties, any Company Unitholder or any other Person:
- (a) each Merger Sub I Unit that is issued and outstanding immediately prior to the Effective Time shall be converted into one newly and validly issued, fully paid and nonassessable unit in the Surviving Company, which shall be the only unit of the Surviving Company that is issued and outstanding immediately after the Effective Time;
- (b) each Company Unit held in the Company's treasury or owned by the Company, Parent or any Subsidiary of the Company or Parent immediately prior to the Effective Time shall be cancelled and extinguished without consideration or conversion (the "Cancelled Units");
- (c) subject to the terms and conditions of this Agreement, each Company Class A Unit or Company Profits Interest Unit that is issued and outstanding immediately prior to the Effective Time shall be cancelled and extinguished and, other than the Cancelled Units, automatically converted into the right of the Company Securityholder holding such Company Class A Unit or Company Profits Interest Unit immediately prior to the Effective Time to receive in the aggregate in respect of all such Company Class A Units and Company Profits Interest Units:
  - (i) an amount in cash set forth opposite the name of such Company Securityholder under the column "Closing Cash Payment" on the Closing Statement;
  - (ii) a number of shares of Parent Common Stock set forth opposite the name of such Company Securityholder under the column "Closing Stock Payment" on the Closing Statement (<u>provided</u> that certain of such shares will be subject to restrictions on transfer and/or the possibility of forfeiture in accordance with the applicable Restricted Stock Agreement);
  - (iii) such Company Securityholder's Pro Rata Share of the portion of the Adjustment Holdback Amount, if any, required to be delivered to the Company Securityholders in accordance with Section 1.8 as and when any such delivery is required to be made;
  - (iv) an amount in cash and a number of shares of Parent Common Stock equal to such Company Securityholder's Pro Rata Share of the Post-Closing Excess Amount with respect to any disbursements required to be made in connection with the Post-Closing Excess Amount (if any) in accordance with <u>Section 1.8(i)</u>;
  - (v) such Company Securityholder's Pro Rata Share of the portion of the Indemnity Escrow Amount, if any, required to be delivered to the Company

Securityholders in accordance with Section 7.7 as and when any such delivery is required to be made;

- (vi) in the case of each Company Unitholder and the Aggregator, such Company Securityholder's Expense Fund Sharing Percentage of the Expense Fund, if any, required to be delivered to such Company Securityholders in accordance with Section 7.13(e) as and when any such delivery is required to be made; and
- (vii) such Company Securityholder's Pro Rata Share of the Earnout Stock Consideration (or to the extent such Company Securityholder is an Unaccredited Investor, a cash payment in lieu thereof), if any, required to be issued (or paid with respect to Unaccredited Investors) to the Company Securityholders in accordance with Sections 1.14(e)(i), 1.14(e) (ii), and 1.14(e)(iii) as and when any such issuance (or payment) is required to be made.
- 1.5 Effect of the Second Merger on Membership Units of the Surviving Company and Membership Interests of Merger Sub II. At the Second Effective Time, by virtue of the Second Merger and without any action on the part of any holder of any membership units of the Surviving Company or any membership interests of Merger Sub II, (i) units in the Surviving Company shall be cancelled and extinguished without payment of any consideration therefor and (ii) each membership interest of Merger Sub II shall remain outstanding and represent membership interests of the Surviving Entity.
  - 1.6 Adjustment Holdback Amount; Indemnity Escrow Amount; Expense Fund; Closing Payments.
- (a) At the Closing, Parent shall withhold (or cause to be withheld) the Adjustment Holdback Amount as partial security for the satisfaction of any Post-Closing Deficit Amount under <u>Section 1.8</u>.
- (b) At the Closing or, if the Closing occurs after the applicable wire deadline on the Closing Date, the next Business Day, Parent shall deposit, or cause to be deposited, with the Escrow Agent the Indemnity Escrow Amount as partial security (i) for the satisfaction of any Post-Closing Deficit Amount under Section 1.8, (ii) for the indemnification, compensation and reimbursement obligations of the Indemnifying Parties under Article 7 and (iii) for the payment of Pre-Closing Taxes under Article 8.
- (c) At the Closing or, if the Closing occurs after the applicable wire deadline on the Closing Date, the next Business Day, Parent shall deposit, or cause to be deposited, with the Securityholder Representative the Expense Fund, by wire transfer of immediately available funds to an account designated in writing by the Securityholder Representative, to be held by the Securityholder Representative until used by the Securityholder Representative or delivered to the Company Securityholders in accordance with the terms of this Agreement.
- (d) Each Company Securityholder's right, if any, to receive cash from the Adjustment Holdback Amount, the Indemnity Escrow Amount or the Expense Fund, as applicable, are non-transferable and non-assignable, except that each Company Securityholder shall be entitled to assign such Company Securityholder's rights to such amounts by will, by the Laws of intestacy, divorce, community property or by other similar operation of law; <u>provided</u>, that any such assignee shall be subject to the terms hereof or any other document contemplated by the terms hereof, in each case that is applicable to the Company Securityholder.

- (e) At the Closing or, if the Closing occurs after the applicable wire deadline on the Closing Date, the next Business Day, Parent shall pay (or cause to be paid) the Closing Indebtedness of the Company with respect to borrowed money that remains unpaid as of the Closing Date to the Persons for whom the Company has delivered to Parent duly executed copies of bank pay-off letters with respect to such Closing Indebtedness.
- (f) At the Closing or, if the Closing occurs after the applicable wire deadline on the Closing Date, the next Business Day, Parent shall pay (or cause to be paid) the Transaction Expenses to the Persons for whom the Company has delivered to Parent invoices in respect of such Transaction Expenses.

## 1.7 <u>Payment of Closing Consideration; Exchange.</u>

- (a) PNC Bank, N.A. shall serve as the Exchange Agent (the "Exchange Agent") for the Mergers.
- (b) At the Closing or, if the Closing occurs after the applicable wire deadline on the Closing Date, the next Business Day, Parent shall deposit (or cause to be deposited) cash to the Exchange Agent in an aggregate amount equal to \$7,257,573.99, to be held by the Exchange Agent in accordance with the terms of the exchange agent agreement (the "Exchange Agent Agreement") to be executed at the Closing by Parent and the Exchange Agent; provided, that Parent shall not deposit with the Exchange Agent any portion of the Adjustment Holdback Amount, the Indemnity Escrow Amount or the Expense Fund.
- (c) As promptly as practicable after a Company Securityholder delivers (which may be done electronically) to the Exchange Agent a duly completed and validly executed letter of transmittal in substantially the form attached hereto as Exhibit F (a "Letter of Transmittal") and any other documents Parent or the Exchange Agent may reasonably require in order to effect such exchange, the Exchange Agent shall notify Parent of such delivery, and as promptly as practicable after such time, Parent shall (i) cause the Transfer Agent to issue to such Company Securityholder the number of shares of Parent Common Stock in an individual restricted account for such Company Securityholder at the Transfer Agent that such Company Securityholder is entitled to receive at the Closing pursuant to Section 1.4 and (ii) cause the Exchange Agent to pay to such Company Securityholder the amount of cash that such Company Securityholder is entitled to receive at the Closing pursuant to Section 1.4 (including any Fractional Share Consideration). For the avoidance of doubt, subject to the terms of the applicable Restricted Stock Agreements, the Parent Common Stock held by the Transfer Agent pursuant to this Section 1.7(c) will be shown as issued and outstanding on Parent's financial statements and the applicable Company Securityholder will be shown as the registered owner of the Parent Common Stock on the books and records of Parent and shall have all rights with respect to such Parent Common Stock (including the right to vote such shares, the right to receive on a current basis any cash dividends or other distributions made with respect to such Parent Common Stock, which dividends or other distributions shall be promptly paid to the applicable owner of such Parent Common Stock, and the right of possession or transfer of such Parent Common Stock in accordance with applicable securities laws).
- (d) At the Effective Time, the Company Securityholders shall cease to have any rights as equity holders of the Company, and the unit transfer books of the Company shall be closed with respect to all Company Units outstanding immediately prior to the Effective Time (<u>provided</u>, for the avoidance of doubt and notwithstanding anything to the contrary, this sentence shall not limit any rights under this Agreement, including the right to receive the applicable

Merger Consideration). No further transfer of any such Company Units shall be made on such unit transfer books after the Effective Time.

(e) Any portion of the Merger Consideration that was deposited with the Exchange Agent and remains undistributed to Company Securityholders following the first anniversary of the Closing Date shall be delivered to Parent upon demand, and Company Securityholders shall thereafter (if Parent has made such demand) look only to Parent for satisfaction of their claims for any Merger Consideration payable with respect to the Company Securities previously held by such Company Securityholder without any interest thereon. Notwithstanding anything to the contrary contained herein, none of Parent, the Merger Subs, the Company, the Surviving Company, the Surviving Entity, the Securityholder Representative or the Exchange Agent shall be liable to any Company Securityholder for Merger Consideration delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law. Any Merger Consideration remaining undistributed to Company Securityholders immediately prior to such time as such Merger Consideration would otherwise escheat to or become the property of any Governmental Authority shall, to the extent permitted by Law, become the property of the Parent free and clear of all claims or interest of any Person previously entitled thereto.

# 1.8 <u>Post-Closing Reconciliation</u>.

- (a) The Company has delivered to Parent (a) the spreadsheet attached as Exhibit G containing the information set forth on Schedule 1.8 (the "Closing Statement") and (b) a certificate, duly executed on behalf of the Company by the chief executive officer of the Company, certifying that all of calculations and information included therein reflect the Company's good faith estimate thereof. The Company has also delivered to Parent supporting calculations and documentation of such calculations, in reasonable detail, concurrently with the delivery of the Closing Statement. The Company has prepared the Closing Statement in accordance with income tax accounting principles using the accrual method of accounting as set forth in more detail on Schedule 1.8(a) (the "Agreed Accounting Principles"). Nothing in this Section 1.8(a) shall in any way limit the right of Parent to arrive at a different calculation of the Adjusted Cash Consideration pursuant to Section 1.8(c) or any Indemnified Party to be indemnified for inaccuracies in such Closing Statement pursuant to Article 7 in accordance with the terms (and subject to the limitations) thereof.
- (b) The Estimated Closing Cash, the Estimated Closing Net Working Capital Surplus, if any, the Estimated Closing Indebtedness, the Estimated Unpaid Transaction Expenses, the Estimated Unpaid Pre-Closing Taxes and the Estimated Closing Net Working Capital Shortfall, if any, set forth in the Closing Statement will be used for purposes of calculating the Adjusted Cash Consideration at the Closing (which calculation shall be subject to adjustment following the Closing pursuant to, and in accordance with, the terms of this Section 1.8).
- (c) As soon as reasonably practicable after the Closing Date, and in any event within ninety (90) days of the Closing Date, Parent shall prepare and deliver to the Securityholder Representative a statement (the "Post-Closing Statement") that shall set forth a calculation of (i) the Closing Cash, (ii) the Closing Net Working Capital Surplus, if any, (iii) the Closing Indebtedness, (iv) the Unpaid Transaction Expenses, (v) the Unpaid Pre-Closing Taxes, (vi) the Closing Net Working Capital Shortfall, if any, and (vii) the Adjusted Cash Consideration calculated based on such estimated amounts, in each case accompanied by supporting calculations and documentation of such calculations, in reasonable detail. Parent shall prepare

the Post-Closing Statement in accordance with the Agreed Accounting Principles, consistent with the Closing Statement.

- (d) After the Securityholder Representative's receipt of the Post-Closing Statement, Parent shall (i) make available to the Securityholder Representative and its Representatives the books and records used in preparing the Closing Statement and (ii) use commercially reasonable efforts to assist the Securityholder Representative (including by providing reasonable access to appropriate personnel of Parent and its Subsidiaries) in its reasonable review of the Closing Statement.
- (e) The Securityholder Representative shall notify Parent in writing (the "Notice of Adjustment Disagreement") within forty-five (45) days of the Securityholder Representative's receipt of the Post-Closing Statement (the "Adjustment Review Period") if the Securityholder Representative disagrees with any portion of the Post-Closing Statement. The Notice of Adjustment Disagreement shall set forth in reasonable detail the basis for such disagreement, the amounts involved and the Securityholder Representative's proposed adjustments to the Post-Closing Statement with reasonably detailed supporting documentation. If no Notice of Adjustment Disagreement is received by Parent on or prior to the expiration date of the Adjustment Review Period, then the Post-Closing Statement and all amounts set forth therein shall be deemed to have been accepted by the Securityholder Representative and shall become final and binding upon the parties hereto and the Company Securityholders for all purposes under this Agreement.
- During the thirty (30) days immediately following the delivery of a Notice of Adjustment Disagreement (the "Adjustment Resolution Period"), if any, the Securityholder Representative and Parent shall seek in good faith to resolve any disagreement that they may have with respect to the matters specified in the Notice of Adjustment Disagreement. Any items agreed to by the Securityholder Representative and Parent in a written agreement executed and delivered by each of the Securityholder Representative and Parent, together with any items not disputed or objected to by the Securityholder Representative in the Notice of Adjustment Disagreement, are collectively referred to herein as the "Resolved Matters." If at the end of the Adjustment Resolution Period, the parties have been unable to resolve any differences they may have with respect to the matters specified in the Notice of Adjustment Disagreement, the Securityholder Representative and Parent, or either of them, shall refer all matters in the Notice of Adjustment Disagreement other than the Resolved Matters (the "Unresolved Matters") to BDO LLP (the "Independent Accountant"). In the event that BDO LLP refuses or is otherwise unable to act as the Independent Accountant, the Securityholder Representative and Parent shall cooperate in good faith to appoint an independent certified public accounting firm in the United States of national recognition mutually agreeable to the Securityholder Representative and Parent, in which event "Independent Accountant" shall mean such firm. Within thirty (30) days after the submission of such matters to the Independent Accountant, the Independent Accountant, acting as an expert and not as an arbitrator, will make a final determination, binding on the parties hereto for all purposes under this Agreement, of the appropriate amount of each of the Unresolved Matters. With respect to each Unresolved Matter, such determination, if not in accordance with the position of either the Securityholder Representative or Parent, shall not be in excess of the higher, nor less than the lower, of the amounts advocated by the Securityholder Representative in the Notice of Adjustment Disagreement or Parent in the Post-Closing Statement with respect to such Unresolved Matter. For the avoidance of doubt, the Independent Accountant shall not review any line items in the Post-Closing Statement or make any determination with respect to any matter other than the Unresolved Matters. During the review by the Independent Accountant, Parent and the Securityholder Representative shall each make available to the Independent Accountant such individuals and such information, books, records

and work papers, as may be reasonably required by the Independent Accountant to fulfill its obligations under this Section 1.8(f); provided, that the external accountants of Parent or the Company shall not be obligated to make any working papers available to the Independent Accountant unless and until the Independent Accountant has signed a customary confidentiality and hold harmless agreement relating to such access to working papers in form and substance reasonably acceptable to such independent accountants. The fees and expenses of the Independent Accountant shall be borne by Parent and the Securityholder Representative (on behalf of the Company Securityholders) based on the inverse of the percentage that the Independent Accountant's resolution of the disputed items covered by such Notice of Adjustment Disagreement (before such allocation) bears to the total amount of such disputed items as originally submitted to the Independent Accountant (for example, if the total amount of such disputed items as originally submitted to the Independent Accountant equals \$1,000 and the Independent Accountant awards \$600 in favor of the Securityholder Representative's position, sixty percent (60%) of the fees and expenses of the Independent Accountant would be borne by Parent and forty percent (40%) of the fees and expenses of the Independent Accountant would be borne by the Securityholder Representative (on behalf of the Company Securityholders)).

- (g) The "Final Closing Statement" shall be (i) in the event that no Notice of Adjustment Disagreement is delivered by the Securityholder Representative to Parent prior to the expiration of the Adjustment Review Period, the Post-Closing Statement delivered by Parent to the Securityholder Representative pursuant to Section 1.8(c), (ii) in the event that a Notice of Adjustment Disagreement is delivered by the Securityholder Representative to Parent prior to the expiration of the Adjustment Review Period and Parent and the Securityholder Representative are able to agree on all matters set forth in such Notice of Adjustment Disagreement, the Post-Closing Statement delivered by Parent to the Securityholder Representative pursuant to Section 1.8(c) as adjusted pursuant to the written agreement executed and delivered by Parent and the Securityholder Representative or (iii) in the event that a Notice of Adjustment Disagreement is delivered by the Securityholder Representative to Parent prior to the expiration of the Adjustment Review Period and Parent and the Securityholder Representative are unable to agree on all matters set forth in such Notice of Adjustment Disagreement, the Post-Closing Statement delivered by Parent to the Securityholder Representative pursuant to Section 1.8(c) as adjusted to be consistent with the Resolved Matters and the final determination of the Independent Accountant of the Unresolved Matters in accordance with Section 1.8(f). The date on which the Final Closing Statement is finally determined in accordance with this Section 1.8(g) is hereinafter referred to as the "Determination Date." The Final Closing Statement and all amounts set forth therein shall be final and binding upon the parties hereto and the Company Securityholders for all purposes under this Agreement.
- (h) If the Adjusted Cash Consideration set forth in the Final Closing Statement (as finally determined in accordance with this Section 1.8) is less than the Adjusted Cash Consideration set forth in the Closing Statement (after giving effect to any revisions thereto in accordance with Section 1.8(a)) (such difference, the "Post-Closing Deficit Amount"), then Parent shall be entitled to, first, permanently retain the Adjustment Holdback Amount (or a portion thereof) equal to the absolute value of the Post-Closing Deficit Amount, and if insufficient to satisfy the Post-Closing Deficit Amount, second, to permanently retain an amount of cash from the Indemnity Escrow Amount equal to the absolute value of the remaining Post-Closing Deficit Amount (without any need to make a claim for indemnification in accordance with the procedures set forth in Article 7). Each delivery of any portion of the Adjustment Holdback Amount to Parent pursuant to this Section 1.8(h) shall be made in proportion to the Company Securityholders' respective Pro Rata Shares of the Adjustment Holdback Amount being delivered. Any portion of the Adjustment Holdback Amount to be so delivered shall be rounded down to the nearest cent.

- (i) If the Adjusted Cash Consideration set forth in the Final Closing Statement (as finally determined in accordance with this Section 1.8) is more than the Adjusted Cash Consideration set forth in the Closing Statement (after giving effect to any revisions thereto in accordance with Section 1.8(a)) (such difference, the "Post-Closing Excess Amount"), then Parent shall promptly (but in all events within five (5) Business Days after the Determination Date), pay in cash (including any Fractional Share Consideration) and issue in shares of Parent Common Stock the Post-Closing Excess Amount to the Company Securityholders in accordance with Section 1.4(c)(iv), it being agreed that each Company Securityholder shall be entitled to receive an amount of cash and number of shares of Common Stock that is proportional to the amount of cash and number of shares of Parent Common Stock that such Company Securityholder received at Closing pursuant to Section 1.4(c); provided, that as a condition to Parent's obligation to make such payments or issuances, the Securityholder Representative shall first deliver to Parent an updated Closing Statement setting forth the portion of the Post-Closing Excess Amount payable and/or issuable to each Company Securityholder.
- (j) Following the payment of any Post-Closing Deficit Amount to Parent in accordance with <u>Section 1.8(h)</u> or, if there is a Post-Closing Excess Amount, the payment by Parent of such amount to the Company Securityholders in accordance with <u>Section 1.8(i)</u>, Parent shall release the remainder of the Adjustment Holdback Amount to the Company Securityholders, which distribution shall be made to a Company Securityholder in accordance with such Company Securityholder's Pro Rata Share of such amount (as set forth on the Closing Statement); <u>provided</u>, that as a condition to Parent's obligation to make such distributions, the Securityholder Representative shall first deliver to Parent an updated Closing Statement setting forth the portion of the Adjustment Holdback Amount to be distributed to each Company Securityholder. Each delivery of any portion of the Adjustment Holdback Amount to Company Securityholders pursuant to this <u>Section 1.8(j)</u> shall be made in proportion to the Company Securityholders respective Pro Rata Shares of the Adjustment Holdback Amount being delivered. Any portion of the Adjustment Holdback Amount to be so delivered to a particular Company Securityholder shall be rounded down to the nearest cent.
- (k) Parent, the Surviving Company and the Surviving Entity shall be entitled to conclusively rely upon the updated Closing Statement delivered by the Securityholder Representative, including with respect to whether any individual Company Securityholder received the appropriate portion of any such distribution, and in no event will Parent, the Surviving Company, the Surviving Entity or any of their Affiliates have any liability to any Person on account of payments or distributions made in accordance with the updated Closing Statement delivered by the Securityholder Representative.
- 1.9 <u>Fractional Shares.</u> No fraction of a share of Parent Common Stock will be issued or delivered by virtue of the Merger. The aggregate number of shares of Parent Common Stock issuable or deliverable to any Company Securityholder shall be rounded down to the nearest whole share. Notwithstanding any other provision of this Agreement, any holder of Company Class A Units or Company Profits Interests Units converted pursuant to the Merger who would otherwise have been entitled to receive a fraction of a share of Parent Common Stock shall receive, in lieu thereof, cash in an amount equal to such fractional part of a share of Parent Common Stock multiplied by (a) in the case of any portion of a share of Parent Common Stock otherwise issuable or deliverable (i) as part of the Closing Stock Consideration or (ii) as part of the Post-Closing Excess Amount, the Reference Price, (b) in the case of any portion of a share of Parent Common Stock otherwise issuable or deliverable as part of the First Earnout Stock Consideration, the First Earnout VWAP, (c) in the case of any portion of a share of Parent Common Stock otherwise issuable or deliverable as part of the Second Earnout Stock Consideration, the Second Earnout VWAP or (d) in the case of any portion of a share of Parent

Common Stock otherwise issuable or deliverable as part of the Third Earnout Stock Consideration, the Third Earnout VWAP (the "Fractional Share Consideration").

- 1.10 Equitable Adjustments. If at any time during the period from and after the date of this Agreement, any change in the outstanding shares of Parent Common Stock shall occur by reason of any reclassification, recapitalization, stock split or combination, exchange or readjustment of shares, or any stock dividend thereon with a record date during such period, the number of shares of Parent Common Stock (including the number of shares of Parent Common Stock to be issued or delivered as part of the Merger Consideration), the Reference Price and any number or amount contained in this Agreement which is based on the price of Parent Common Stock (including the First Earnout VWAP, the Second Earnout VWAP and the Third Earnout VWAP) or the number of shares of Parent Common Stock shall be equitably adjusted to the extent necessary to provide the parties the same economic effect as contemplated by this Agreement prior to such reclassification, recapitalization, stock split or combination, exchange or readjustment of shares, or stock dividend thereon.
- 1.11 Tax Withholding. Each of Parent, the Surviving Company, the Surviving Entity and their Affiliates shall be entitled to deduct and withhold, or cause to be deducted and withheld, from the Merger Consideration or any other payment otherwise payable pursuant to this Agreement (including the Adjustment Holdback Amount and the Indemnity Escrow Amount), the amounts required to be deducted and withheld under the Code, or any provision of state, local or foreign Tax Law, with respect to the making of such payment and, to the extent that amounts are so deducted and withheld, such amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of whom such deduction and withholding was made. Each Person entitled to payment under this Agreement shall provide Parent with a valid IRS Form W-9 or applicable IRS Form W-8. The parties agree to cooperate to allow Parent, at its election, to effectuate such withholding by means acceptable to Parent, including by paying the applicable Merger Consideration for which such withholding is required, or its cash value, to the Surviving Entity and causing the Surviving Entity to withhold the applicable amounts through the Surviving Entity's payroll system.
- 1.12 <u>Further Assurances</u>. If, at any time after the Effective Time, any further instruments, deeds, assignments or assurances are reasonably necessary to consummate the Mergers, then the Securityholder Representative, Parent, the Surviving Company, the Surviving Entity and their respective officers and directors are hereby authorized to execute and deliver all such proper deeds, assignments, instruments and assurances to consummate the Mergers.

# 1.13 <u>Tax Consequences</u>.

- (a) For U.S. federal and applicable state and local income Tax purposes, the parties hereto intend that (i) the First Merger and the Second Merger, be considered together as integrated steps in a single transaction for U.S. federal income Tax purposes, and shall together qualify as a tax-free reorganization within the meaning of Section 368(a) of the Code and the Treasury Regulations promulgated thereunder, and (ii) this Agreement is hereby adopted as a "plan of reorganization" within the meaning of Section 368 of the Code and Treasury Regulations Sections 1.368-2(g) and 1.368-3.
- (b) The parties intend that, for applicable Tax purposes, the shares of Parent Common Stock issuable to the Founders hereunder in respect of Company Units held by the Founders, including any such shares subject to the Founder Restricted Stock Agreement, shall be treated as consideration for such shares of Company Units and not as compensation for services

(together with the Tax consequences set forth in Section 1.13(a), the "Intended Tax Treatment").

(c) The parties agree to file their respective Tax Returns in a manner consistent with the Intended Tax Treatment, and shall not take a position on any Tax Return or any Action with respect to Taxes inconsistent with the Intended Tax Treatment, unless otherwise required by applicable Law. Except for the representations in Section 2.7, none of the parties to this Agreement makes any representation as to the Intended Tax Treatment. The Company acknowledges that the Company and the Company Unitholders are relying solely on their own Tax advisors in connection with this Agreement and the other transactions and agreements contemplated hereby.

#### 1.14 Earn-Out.

- (a) <u>Earnout Definitions</u>. For purposes of this Agreement, the following terms shall have the following meanings:
- (i) "2026 Earnout Period" means a period of time that commences at 12:01 AM Pacific Time on January 1, 2026 and ends at 11:59 PM Pacific Time on December 31, 2026.
- (ii) "2027 Earnout Period" means a period of time that commences at 12:01 AM Pacific Time on January 1, 2027 and ends at 11:59 PM Pacific Time on December 31, 2027.
  - (iii) "2026 Earnout Target" has the meaning set forth in Schedule 1.14(a).
  - (iv) "2027 Earnout Target" has the meaning set forth in Schedule 1.14(a).
  - (v) "2026 Earnout Threshold Ceiling" has the meaning set forth in Schedule 1.14(a).
  - (vi) "2026 Earnout Threshold Floor" has the meaning set forth in Schedule 1.14(a).
  - (vii) "2027 Earnout Threshold Ceiling" has the meaning set forth in Schedule 1.14(a).
  - (viii) "2027 Earnout Threshold Floor" has the meaning set forth in Schedule 1.14(a).
- (ix) "Budget Schedule" has the meaning set forth in <u>Schedule 1.14(a)(ix)</u> contemplating the resources to be made available to the Earnout Business following the Closing during the Earnout Period.
- (x) "Change in Control Transaction" means the occurrence of the closing of (i) the acquisition, directly or indirectly, by any Person or "group" (as defined in Section 13(d) of the Exchange Act), whether as a single transaction or series of related transactions, including by way of a merger, consolidation or other business combination, of 50% or more of the combined voting power of the then outstanding voting power of

Parent or the Surviving Entity, (ii) a "de-Spac" transaction of Parent or the Surviving Entity, or (iii) the sale, transfer or other disposition (whether directly or indirectly through a subsidiary, and whether by merger, consolidation other business combination, exclusive license or otherwise) of all or substantially all of the assets of Parent and its Subsidiaries (on a consolidated basis) or the Surviving Entity and its Subsidiaries (on a consolidated basis).

- (xi) "Earnout Amount" means any of the First Earnout Amount, the Second Earnout Amount, or the Third Earnout Amount, as applicable.
  - (xii) "Earnout Business" has the meaning set forth in Schedule 1.14(a)(ix).
- (xiii) "Earnout Default Event" means a Triggering Event that is not cured prior to the Triggering Event Consideration Period applicable to such Triggering Event in accordance with, and subject to, Section 1.14(b).
- (xiv) "Earnout Period" means any of the SEC Approval Earnout Period, the 2026 Earnout Period or the 2027 Earnout Period, as applicable.
- (xv) "Earnout Stock Consideration" means any of the First Earnout Stock Consideration, Second Earnout Stock Consideration, or Third Earnout Stock Consideration, as applicable.
  - (xvi) "Earnout Target" means the 2026 Earnout Target or the 2027 Earnout Target, as applicable.
- (xvii) "Expense Cap" means the limit on the total annual operating expenses the fund can charge as fees to investors as set forth in the "Fund Prospectus," "Private Placement Memorandum," "Operating Agreement," "Limited Partnership Agreement," or similar agreement or document.

# (xviii) "First Earnout Amount" has the following meaning:

- (A) If (1) (x) Parent or any of its Affiliates (including the Surviving Entity) receives approval from the SEC for the Megacorn Fund (1940 Act file No. 811-24060) and (y) the Megacorn Fund has validly issued registered fund securities to a New Limited Partner or New Equityholder, in each case before the expiration of the SEC Approval Earnout Period (the satisfaction of clauses (x) and (y), the "First Earnout Achievement"), or (2) an Earnout Default Event is finally determined pursuant to this Section 1.14 to have occurred during the SEC Approval Earnout Period (regardless of when such final determination occurs), then the "First Earnout Amount" shall be an amount equal to \$4,166,666.66;
- (B) except as otherwise provided in clause (A) above, the "First Earnout Amount" shall be an amount equal to zero.
- (xix) "First Earnout Stock Consideration" means an amount equal to the quotient of (i) the First Earnout Amount <u>divided by</u> (ii) the volume weighted average closing sale price of the Parent Common Stock as reported on the New York Stock Exchange (or such other national securities exchange on which shares of Parent Common

Stock may then be traded) for the thirty (30) consecutive trading days ending on (and including) the date that is two (2) trading days immediately preceding the date that the First Earnout Amount is determined to be conclusive, final and binding on the parties hereto (the "First Earnout VWAP", as adjusted and appropriate to reflect any stock splits, stock dividends, reverse stock splits, combinations, reorganizations, reclassifications, or similar events affecting Parent Common Stock); provided, however, that if the First Earnout VWAP is less than \$15.00, the First Earnout VWAP shall be deemed to be \$15.00, and if the First Earnout VWAP is greater than \$18.75, the First Earnout VWAP shall be deemed to be \$18.75.

- (xx) "New Limited Partner or New Equityholder" means a limited partner or equityholder of the Megacorn Fund that was not prior to the Closing a limited partner or equityholder of the Megacorn Fund.
  - (xxi) "Recurring Net Revenue" [\*\*\*]
  - (xxii) "SEC Approval Earnout Period" means [\*\*\*]
  - (xxiii) "Second Earnout Amount" has the following meaning:
  - (A) If the Recurring Net Revenue for the 2026 Earnout Period is less than the 2026 Earnout Threshold Floor, then the "Second Earnout Amount" shall mean an amount equal to zero.
  - (B) If the Recurring Net Revenue for the 2026 Earnout Period is greater than the 2026 Earnout Threshold Floor and less than the 2026 Earnout Threshold Ceiling, then the "Second Earnout Amount" shall mean an amount equal to \$4,166,666.67, <u>multiplied by</u> a fraction, (1) the numerator of which the Recurring Net Revenue for the 2026 Earnout Period, and (2) the denominator of which is the 2026 Earnout Target.
  - (C) If (1) the Recurring Net Revenue for the 2026 Earnout Period is greater than the 2026 Earnout Threshold Ceiling, or (2) an Earnout Default Event is finally determined pursuant to this <u>Section 1.14</u> to have occurred during the 2026 Earnout Period (regardless of when such final determination occurs), then the "Second Earnout Amount" shall mean an amount equal to \$5,416,666.67.
- (xxiv) "Second Earnout Stock Consideration" means an amount equal to the quotient of (i) the Second Earnout Amount divided by (ii) the volume weighted average closing sale price of the Parent Common Stock as reported on the New York Stock Exchange (or such other national securities exchange on which shares of Parent Common Stock may then be traded) for the 30 consecutive trading days ending on (and including) the date that is two (2) trading days immediately preceding the date that the Second Earnout Amount is determined to be conclusive, final and binding on the parties hereto (the "Second Earnout VWAP", as adjusted and appropriate to reflect any stock splits, stock dividends, reverse stock splits, combinations, reorganizations, reclassifications, or similar events affecting Parent Common Stock); provided, however, that if the Second Earnout VWAP is less than \$15.00, the Second Earnout VWAP shall be deemed to be \$15.00, and if the Second Earnout VWAP is greater than \$18.75, the Second Earnout VWAP shall be deemed to be \$18.75.

# (xxv) "Third Earnout Amount" has the following meaning:

- (A) If the Recurring Net Revenue for the 2027 Earnout Period is less than the 2027 Earnout Threshold Floor, then the "Third Earnout Amount" shall mean an amount equal to zero.
- (B) If the Recurring Net Revenue for the 2027 Earnout Period is greater than the 2027 Earnout Threshold Floor and less than the 2027 Earnout Threshold Ceiling, then the "Third Earnout Amount" shall mean an amount equal to \$4,166,666.67, <u>multiplied by</u> a fraction, (1) the numerator of which the Recurring Net Revenue for the 2027 Earnout Period, and (2) the denominator of which is the 2027 Earnout Target.
- (C) If (1) the Recurring Net Revenue for the 2027 Earnout Period is greater than the 2027 Earnout Threshold Ceiling, or (2) an Earnout Default Event is finally determined pursuant to Section 1.14 to have occurred during the 2027 Earnout Period (regardless of when such final determination occurs), then the "Third Earnout Amount" shall mean an amount equal to \$5,416,666.67.
- (D) Notwithstanding anything to the contrary, if (1) the Recurring Net Revenue for the 2026 Earnout Period is less than the 2026 Earnout Threshold Floor and (2) the Recurring Net Revenue for the 2027 Earnout Period is more than the 2027 Earnout Threshold Ceiling, then the "Third Earnout Amount" otherwise calculated pursuant to the foregoing clauses (A)-(C) shall be increased by \$2,916,666.67.
- (xxvi) "Third Earnout Stock Consideration" means an amount equal to the quotient of (i) the Third Earnout Amount divided by (ii) the volume weighted average closing sale price of the Parent Common Stock as reported on the New York Stock Exchange (or such other national securities exchange on which shares of Parent Common Stock may then be traded) for the 30 consecutive trading days ending on (and including) the date that is two (2) trading days immediately preceding the date that the Third Earnout Amount is determined to be conclusive, final and binding on the parties hereto (the "Third Earnout VWAP", as adjusted and appropriate to reflect any stock splits, stock dividends, reverse stock splits, combinations, reorganizations, reclassifications, or similar events affecting Parent Common Stock); provided, however, that if the Third Earnout VWAP is less than \$15.00, the Third Earnout VWAP shall be deemed to be \$15.00, and if the Third Earnout VWAP is greater than \$18.75, the Third Earnout VWAP shall be deemed to be \$15.75.
- (xxvii) "**Triggering Event**" means (i) Parent's failure to provide funds contemplated in the budget for the Earnout Business (x) for the fical year 2025 as set forth in <u>Schedule 1.14(a)(ix)</u> and (y) for the fiscal years 2026 and 2027 once such budgets are established in accordance with the annual planning and budgeting processes of Parent described in <u>Schedule 1.14(a)(ix)</u> or (ii) Parent's breach of the covenant applicable to Parent in <u>Section 1.14(g)(i)(A)</u> or Parent's material breach of any covenants to be performed by Parent with respect to the governance and management of the Earnout Business as contemplated by <u>Schedule 1.14(a)(ix)</u>.
  - (b) <u>Triggering Event Notice</u>.

- (i) If, following the Closing and during an Earnout Period, the Securityholder Representative believes that a Triggering Event has occurred, then the Securityholder Representative shall promptly provide written notice to Parent of such Triggering Event (such notice, a "**Triggering Event Notice**"), which notice shall provide a reasonably detailed description of the Triggering Event.
- (ii) Following Parent's receipt of the Triggering Event Notice, Parent shall have 20 calendar days to cure the deficiency causing the alleged Triggering Event.
- (iii) If, by 11:59 PM Pacific Time on the date that is the 20<sup>th</sup> days after the date that Parent receives the Triggering Event Notice (such 20-day period, the "**Triggering Event Consideration Period**"), Parent notifies the Securityholder Representative that Parent has cured the deficiency causing such Triggering Event, or Parent notifies the Securityholder Representative in writing that Parent disputes the Securityholder Representative's determination that a Triggering Event has occurred (each such notice, the "**Triggering Event Dispute Notice**"), then representatives of Parent and the Securityholder Representative shall meet at least once on a date no later than 10 calendar days after the Triggering Event Consideration Period to confirm the matter giving rise to the Triggering Event has been cured or attempt to resolve the dispute.
- (iv) If Parent and the Securityholder Representative are unable to resolve the dispute, or if Parent does not deliver a Triggering Event Dispute Notice prior to the expiration of the Triggering Event Consideration Period, then the determination of whether a Triggering Event occurred and whether it resulted in an Earnout Default Event will be determined pursuant to the dispute resolution process set forth in Section 1.14(d).

# (c) <u>Determination of Earnout Amounts</u>.

No later than 60 calendar days after the last day of the applicable Earnout Period, Parent shall prepare and deliver to the Securityholder Representative a written statement (the "Earnout Calculation Statement") setting forth in reasonable detail Parent's determination of whether the First Earnout Achievement has been satisfied (for the SEC Approval Earnout Period) or of Recurring Net Revenue (for the 2026 Earnout Period or 2027 Earnout Period, as applicable), and Parent's calculation of the applicable Earnout Amount; provided, however, that (x) solely with respect to First Earnout Amount, should the Securityholder Representative provide written notification to Parent of the Securityholder Representative's good faith belief that the First Earnout Achievement has been satisfied, and Parent does not in good faith dispute such achievement no later than 15 Business Days after the Securityholder Representative delivers such notification, then Parent shall pay (or cause to be paid) the First Earnout Amount in the manner provided in Section 1.14(e)(i) (it being understood that the expiration of time for Parent to assert such a dispute shall constitute a final determination of the First Earnout Amount) and (y) should the Securityholder Representative provide written notification to Parent of the Securityholder Representative's good faith belief that an Earnout Default Event has occurred, and Parent does not in good faith dispute such occurrence no later than 15 Business Days after Parent's receipt of the Securityholder's written notification, then Parent shall pay (or cause to be paid) the First Earnout Amount, Second Earnout Amount or Third Earnout Amount, as applicable, in the manner provided in Section 1.14(e) (it being understood that the expiration of time for Parent to assert such a dispute shall constitute a final determination of the First Earnout Amount, Second Earnout Amount or Third Earnout Amount, as applicable); provided further, that if Parent does dispute (within the required time period) the Securityholder Representative's good

faith belief that the First Earnout Achievement has been satisfied or that an Earnout Default Event has occurred, and Parent and the Securityholder Representative are unable to resolve such dispute within 30 calendar days of Parent's notification to the Securityholder Representative of such dispute, the parties shall resolve the dispute in accordance with Section 1.14(d).

- Calculation Statement, the Securityholder Representative shall notify Parent in writing by 5:00 PM Pacific Time on the 20<sup>th</sup> Business Day following the receipt of such Earnout Calculation Statement of such dispute (such date, with respect to such Earnout Calculation Statement, the "Earnout Dispute Deadline" and such notice, the "Earnout Dispute Notice"), which Earnout Dispute Notice shall provide a reasonably detailed description of such dispute, including whether the Securityholder Representative believes an Earnout Default Event has occurred and the Securityholder Representative's calculation of the applicable Earnout Amount. If the Securityholder Representative does not deliver an Earnout Dispute Notice on or before the Earnout Dispute Deadline with respect to such Earnout Calculation Statement, then Parent's calculation of the Earnout Amount shall be deemed conclusive, final and binding on the parties hereto, and none of the Company Securityholders or the Securityholder Representative shall be permitted to dispute such determination.
- (iii) If the Securityholder Representative timely delivers an Earnout Dispute Notice with respect to such Earnout Calculation Statement to Parent, and Parent and the Securityholder Representative are unable to mutually agree on the applicable Earnout Amount within 30 calendar days following receipt by Parent of the Earnout Dispute Notice, then the determination of the Earnout Amount will be settled pursuant to the dispute resolution process in Section 1.14(d).
- (d) Arbitration. Either Parent or the Securityholder Representative may demand arbitration of any dispute in this Section 1.14, which arbitration shall be conducted in the Borough of Manhattan, New York, New York under the commercial arbitration rules then in effect of the American Arbitration Association. The arbitrator shall determine how all expenses relating to the arbitration shall be paid, including the respective expenses of each party, the fees of the arbitrator and the administrative fee of the American Arbitration Association. The arbitrator shall set a limited time period and establish procedures designed to reduce the cost and time for discovery while allowing Parent and the Securityholder Representative an opportunity, adequate in the judgment of the arbitrator, to discover relevant information from the other party about the subject matter in dispute. The arbitrator shall rule upon motions to compel or limit discovery and shall have the authority to impose sanctions, including attorneys' fees and costs, to the same extent as a competent court of law or equity, should the arbitrator determine that discovery was sought without substantial justification or that discovery was refused or objected to without substantial justification. Parent and the Securityholder Representative shall provide the arbitrator with all requested information and documentation in such party's possession or control related to the subject matter of the applicable dispute. The decision of the arbitrator with respect to the maters in dispute shall be final, conclusive and binding upon the parties to this Agreement. Such decision shall be written and shall be supported by written findings of fact and conclusions which shall set forth the award, judgment, decree or order awarded by the arbitrator. Judgment upon an award rendered by the arbitrator may be entered in any court having jurisdiction.
  - (e) <u>Distribution of Earnout Amounts</u>.

- (i) Following the final determination of the First Earnout Amount pursuant to this Section 1.14, subject to the other terms and conditions of this Agreement, including Section 1.14(f) and Section 7.9, Parent shall, within ten (10) calendar days of such final determination, (x) cause the Transfer Agent to issue a number of shares of Parent Common Stock in an individual restricted account to each Company Securityholder that is an Accredited Investor and that has previously delivered a Letter of Transmittal to the Exchange Agent in accordance with Section 1.7(c) equal to such Company Securityholder's Pro Rata Share of the First Earnout Stock Consideration, (y) deliver an amount of cash to the Exchange Agent equal to aggregate Pro Rata Share of all Company Securityholders that are Unaccredited Investors of the First Earnout Amount and promptly thereafter cause the Exchange Agent to distribute such cash to each such Company Securityholder that has previously delivered a Letter of Transmittal to the Exchange Agent in accordance with Section 1.7(c) equal to such Company Securityholder's Pro Rata Share of the First Earnout Amount, and (z) deliver an amount of cash to the Exchange Agent equal to the aggregate Fractional Share Consideration that all Company Securityholders that are Accredited Investors are entitled to in respect of the First Earnout Stock Consideration, and promptly thereafter cause the Exchange Agent to distribute such cash to each such Company Securityholder that has previously delivered a Letter of Transmittal to the Exchange Agent in accordance with Section 1.7(c) equal to the portion of such Fractional Share Consideration attributable to Company Securities held by such Company Securityholder.
- Following the final determination of the Second Earnout Amount pursuant to this Section 1.14, subject to the other terms and conditions of this Agreement, including Sections 1.14(f) and Section 7.9, Parent shall, within the later of (i) 90 calendar days after the end of the 2026 Earnout Period or (ii) ten (10) calendar days of such final determination (provided that if an Earnout Default Event occurs during the 2026 Earnout Period, Parent shall do so within ten (10) calendar days of the final determination of such occurrence), (x) cause the Transfer Agent to issue a number of shares of Parent Common Stock in an individual restricted account to each Company Securityholder that is an Accredited Investor and that has previously delivered a Letter of Transmittal to the Exchange Agent in accordance with Section 1.7(c) equal to such Company Securityholder's Pro Rata Share of the Second Earnout Stock Consideration, (y) deliver an amount of cash to the Exchange Agent equal to the aggregate Pro Rata Share of all Company Securityholders that are Unaccredited Investors of the Second Earnout Amount and promptly thereafter cause the Exchange Agent to distribute such cash to each such Company Securityholder that has previously delivered a Letter of Transmittal to the Exchange Agent in accordance with Section 1.7(c) equal to such Company Securityholder's Pro Rata Share of the Second Earnout Amount, and (z) deliver an amount of cash to the Exchange Agent equal to the aggregate Fractional Share Consideration that all Company Securityholders that are Accredited Investors are entitled to in respect of the Second Earnout Stock Consideration, and promptly thereafter cause the Exchange Agent to distribute such cash to each such Company Securityholder that has previously delivered a Letter of Transmittal to the Exchange Agent in accordance with Section 1.7(c) the portion of such Fractional Share Consideration attributable to Company Securities held by such Company Securityholder.
- (iii) Following the final determination of the Third Earnout Amount pursuant to this Section 1.14, subject to the other terms and conditions of this Agreement, including Section 1.14(f) and Section 7.9, Parent shall, within the later of (i) 90 calendar days after the end of the 2027 Earnout Period or (ii) ten (10) calendar days of such final determination (provided that if an Earnout Default Event occurs during the 2027 Earnout

Period, Parent shall do so within ten (10) calendar days of the final determination of such occurrence), (x) cause the Transfer Agent to issue a number of shares of Parent Common Stock in an individual restricted account to each Company Securityholder that is an Accredited Investor and that has previously delivered a Letter of Transmittal to the Exchange Agent in accordance with Section 1.7(c) equal to such Company Securityholder's Pro Rata Share of the Third Earnout Stock Consideration, (y) deliver an amount of cash to the Exchange Agent equal to the aggregate Pro Rata Share of all Company Securityholders that are Unaccredited Investors of the Third Earnout Amount and promptly thereafter cause the Exchange Agent to distribute such cash to each such Company Securityholder that has previously delivered a Letter of Transmittal to the Exchange Agent in accordance with Section 1.7(c) equal to such Company Securityholder's Pro Rata Share of the Third Earnout Amount, and (z) deliver an amount of cash to the Exchange Agent equal to the aggregate Fractional Share Consideration that all Company Securityholders that are Accredited Investors are entitled to in respect of the Third Earnout Stock Consideration, and Parent shall promptly thereafter cause the Exchange Agent to distribute such cash to each such Company Securityholder that has previously delivered a Letter of Transmittal to the Exchange Agent in accordance with Section 1.7(c) the portion of such Fractional Share Consideration attributable to Company Securities held by such Company Securityholder.

(iv) The right of any Company Securityholder to receive an Earnout Amount: (A) shall not be evidenced by a certificate of other instrument; (B) shall not be assignable or otherwise transferable by such Company Securityholder (except that each Company Securityholder shall be entitled to assign such rights by will, by the Laws of intestacy, divorce, community property or by other similar operation of law); (C) shall not accrue interest; and (D) does not represent any right other than the right to receive the consideration set forth in this Section 1.14. Any attempted transfer of the right to the Earnout Stock Consideration by any holder thereof in violation of clause (B) of the preceding sentence shall be null and void.

(v) [\*\*\*]

(f) <u>Right of Setoff</u>. Notwithstanding anything set forth in this Agreement to the contrary, the Company Securityholders' rights to receive their respective portion of the Earnout Stock Consideration (if any) shall be subject to the set-off rights provided for in <u>Section 7.9</u>.

# (g) <u>Covenants</u>.

- (i) During the Earnout Periods, Parent (A) shall not intentionally take any action (or omit to take any action) that has the primary purpose of impairing the achievement of the Earnout Amounts, and (B) shall cause the Earnout Business to be operated in compliance with <u>Schedule 1.14(a)(ix)</u>.
- (ii) To the extent a Change in Control Transaction occurs during an Earnout Period, Parent shall use commercially reasonable efforts to cause (A) the applicable purchaser entity to assume the obligations of Parent set forth in this <u>Section 1.14</u>, and (B) the Company Securityholders to be third-party beneficiaries of the applicable agreement pursuant to which such purchaser entity assumes such obligations.

## 1.15 <u>Legends</u>.

(a) Each certificate or book-entry security entitlement representing any shares of Parent Common Stock (or any other securities issued in respect of such shares upon any stock split, stock dividend, recapitalization, merger, consolidation or similar event) issued to or held by any Company Securityholder that is an Accredited Investor in accordance with the terms hereof shall bear the following legends (in addition any other legends required by any applicable Laws, the Parent's Charter Documents or any other agreement to which Company Securityholder is a party):

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AND HAVE BEEN ACQUIRED FOR INVESTMENT AND NOT WITH A VIEW TO, OR IN CONNECTION WITH, THE SALE OR DISTRIBUTION THEREOF. NO SUCH SALE OR DISTRIBUTION MAY BE EFFECTED WITHOUT AN EFFECTIVE REGISTRATION STATEMENT RELATED THERETO OR AN OPINION OF COUNSEL IN A FORM SATISFACTORY TO THE COMPANY THAT SUCH REGISTRATION IS NOT REQUIRED UNDER THE SECURITIES ACT OF 1933.

At any time following the first anniversary of the Closing Date, assuming such holder is not an affiliate (as defined in Rule 405 under the Securities Act) of Parent, and subject to the receipt of customary documentation required from the holder of the applicable Parent Shares in connection therewith and compliance with applicable Laws, upon the written request of such holder, Parent shall use its commercially reasonable efforts to (i) promptly instruct the Transfer Agent to remove any restrictive legends applicable to the Parent Shares held by such holder and (ii) cause its legal counsel to deliver the necessary legal opinions, if any, to the Transfer Agent in connection with the instruction under subclause (i). At any time and from time to time, on or prior to first anniversary of the Closing Date, in connection with a bona-fide sale of Parent Shares effected in compliance with the requirements of Rule 144 under the Securities Act or through any broker-dealer sale transactions described in the plan of distribution set forth within any prospectus and pursuant to an effective Registration Statement of which such prospectus forms a part, Parent shall use its commercially reasonable efforts, subject to the receipt of customary documentation required from the holder of the applicable Parent Shares and broker in connection therewith and compliance with applicable Laws, to (x) promptly instruct the Transfer Agent to remove any restrictive legends applicable to the Parent Shares being sold and (y) cause its legal counsel to deliver the necessary legal opinions, if any, to the Transfer Agent in connection with the instruction under subclause (x).

## **ARTICLE 2**

## REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Subject to <u>Section 9.11(a)</u>, except as set forth in the Company Disclosure Schedule, the Company represents and warrants to Parent as follows:

2.1 Organization and Good Standing. The Company is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware. The Company and each of its Subsidiaries has all requisite limited liability company or, as applicable, corporate or comparable power and authority to own, operate and lease its assets and properties and to carry on the Company Business. The Company and each of its Subsidiaries is duly qualified or licensed to do business, and is in good standing, in each jurisdiction where the character of the properties owned, leased or operated by it or the character or location of its assets or properties (whether owned, leased or licensed) or nature of its activities makes such qualification or licensing necessary, in each case except where the failure to be so duly qualified, licensed and in good standing would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company. The Company has made available to Parent complete and correct copies of the Charter Documents, each as amended to date, of the Company and each of its Subsidiaries. The Charter Documents of the Company and its Subsidiaries are in full force and effect with respect to the Company or any of its Subsidiaries party thereto and, to the Company's Knowledge, with respect to each other Person party thereto. Neither the Company nor any of its Subsidiaries are or have been in violation of the provisions of any current or past Charter Documents in any material respect.

#### 2.2 Subsidiaries.

- (a) <u>Section 2.2(a)</u> of the Company Disclosure Schedule sets forth a list of each Subsidiary of the Company, in each case together with the jurisdiction of incorporation, organization or formation, the nature of its organization (corporation, partnership, limited liability company, etc.), authorized, issued and outstanding equity securities of each such Subsidiary of the Company and the ownership thereof as of immediately prior to the Closing. The Company owns, directly or indirectly, all equity securities in each of its Subsidiaries, free and clear of all Encumbrances (other than Encumbrances imposed by applicable securities Laws), and all such equity securities are duly authorized, validly issued, fully paid and non-assessable (to the extent such concept is applicable to such equity securities), were issued in compliance in all material respects with all applicable Laws and the Charter Documents of the Company's Subsidiaries and were not issued in violation of any Contract or preemptive rights, purchase options, rights of first refusal or similar rights.
- (b) Neither the Company nor any of its Subsidiaries (i) owns or controls, directly or indirectly, any equity securities in any other Person other than the Subsidiaries set forth in <u>Section 2.2(a)</u> of the Company Disclosure Schedule, or (ii) has any right or obligation to acquire any equity securities of any other Person, or make any investment (in the form of a loan, capital contribution or otherwise) in, or provide any guarantee with respect to the obligations of, any other Person.

## 2.3 Power, Authorization and Validity.

(a) Power and Authority. The Company has all requisite limited liability company power and authority to execute and deliver this Agreement and each of the Company Ancillary Agreements and to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution and delivery and performance by the Company of this Agreement and each of the Company Ancillary Agreements and the consummation of the transactions contemplated hereby or thereby have been duly and validly approved and authorized by all requisite limited liability company action on the part of the Company.

- (b) <u>Enforceability</u>. This Agreement has been duly executed and delivered by the Company. This Agreement and each of the Company Ancillary Agreements are, or when executed and delivered by the Company shall be, assuming the due authorization, execution and delivery by Parent, the Merger Subs and the other Persons party hereto or thereto, valid and binding obligations of the Company, enforceable against the Company in accordance with their respective terms, subject to the Enforceability Exceptions.
- (c) <u>Manager Approval</u>. The Managers (as defined in the Company LLC Agreement) have, at a meeting duly called and held, by a unanimous vote of the Managers, or by written consent in lieu thereof: (i) approved and declared advisable this Agreement, (ii) determined that the Mergers and other transactions contemplated by this Agreement are advisable, fair to, and in the best interests of the Company and its members and approved the same, (iii) approved the Company Ancillary Agreements and the transactions contemplated thereby, (iv) resolved to recommend to the Members of the Company the adoption of this Agreement, and (v) directed that this Agreement be submitted to the Members of the Company for adoption (such approval in clauses (i) through (v), the "Manager Approval").
- (d) <u>Required Vote of Company Members</u>. The Member Approval is the only votes or consents of the holders of any class or series of Company Units necessary to adopt or approve this Agreement, the Mergers and the other matters set forth in the Written Consent, and, to the extent such approval is required, the Company Ancillary Agreements and the other transactions contemplated hereby and thereby.

# 2.4 <u>Capitalization of the Company.</u>

- (a) <u>Authorized and Outstanding Capitalization of the Company</u>. The issued and outstanding equity securities of the Company consists solely of (i) 473,460 Class A Company Units, and (ii) 175,745 Company Profits Interest Units.
- (i) the number and series of issued and outstanding Company Units held by each Company Unitholder as of the Agreement Date and (ii) the date of issuance of the Company Units held by such Company Unitholder. No Company Units are issued or outstanding as of the Agreement Date that are not set forth on Section 2.4(b) of the Company Disclosure Schedule. The Company does not hold any Company Units in its treasury and does not otherwise own any Company Units. All issued and outstanding Company Units (w) have been duly authorized and validly issued, (x) were offered, issued, sold and delivered by the Company in compliance in all material respects with applicable Law, the Charter Documents, and all requirements set forth in applicable Contracts, (y) were not issued in violation of any preemptive rights, purchase options, rights of first refusal or similar rights under any provision of the DLLCA, the Company's Charter Document or any Contract to which the Company or any of its Subsidiaries is a party and (z) except as set forth on Section 2.4(b) of the Company Disclosure Schedule, are not subject to vesting, repurchase, forfeiture, any right of rescission, right of first refusal or preemptive right under applicable Law, the Company's Charter Documents or any Contract to which the Company or any of its Subsidiaries is a party. There is no Liability for distributions accrued and unpaid by the Company. The allocation of the Merger Consideration among the Company Securityholders pursuant to the terms hereof complies with the Charter Documents of the Company. All of the outstanding Company Units are uncertificated, and the Company has never issued physical certificates in respect of Company Units.
- (c) <u>Company Profits Interest Units</u>. <u>Section 2.4(c)</u> of the Company Disclosure Schedule sets forth, as of the Agreement Date, for each Company Profits Interest

Unit: (i) the name of the holder of such Company Profits Interest Unit, (ii) the threshold amount of such Company Profits Interest Unit, (iii) the number of Company Profits Interest Units held by such holder, and (iv) the date of grant. Except for the Company Profits Interest Units set forth on Section 2.4(c) of the Company Disclosure Schedule, there are no Company Profits Interest Units that are issued and outstanding. True, complete and correct copies of each agreement for Company Profits Interest Units have been made available to Parent, and such agreements have not been amended, modified or supplemented since being made available to Parent, and there are no agreements, understandings or commitments to amend, modify or supplement such agreements in any case from those made available to Parent. The terms of each Company Profits Interest Units permit the treatment of Company Profits Interest Units as provided herein, without the consent or approval of any Company Profits Interest Unit Holders, the Company Unitholders or any other Person other than the Governing Body of the Company, approval from which was obtained prior to the execution and delivery hereof by the Company. No Company Profits Interest Units are subject to vesting, any right of rescission, right of first refusal or preemptive right under applicable Law, the Company's Charter Documents or any Contract to which the Company or any of its Subsidiaries is a party, or to the Knowledge of the Company, any other Contract. All Company Profits Interest Units have been issued in compliance in all material respects under applicable Law, the Company's Charter Documents or any Contract to which the Company or any of its Subsidiaries is a party.

- (d) No Other Rights. Except as set forth on Section 2.4(b) and Section 2.4(c) of the Company Disclosure Schedule, there is no outstanding (x) equity appreciation right, option, restricted equity, restricted equity unit, "phantom" equity or any similar security or right to which the Company or any of its Subsidiaries is a party that is derivative or provides any economic benefit based, directly or indirectly, on the value or price of any equity security of the Company or (y) warrant, call, right, commitment, conversion privilege or preemptive or other right or Contract to which the Company or any of its Subsidiaries is a party to purchase or otherwise acquire any Company Units or any security or debt convertible into or exchangeable for Company Units or obligating the Company to grant, extend or enter into any such equity appreciation right, option, restricted equity, restricted equity unit, "phantom" equity, warrant, call, right, commitment, conversion privilege or preemptive or other right or Contract. There is no voting agreement, registration right, rights of first refusal, preemptive right, co-sale right or other similar right or restriction applicable to any outstanding security of the Company to which the Company or any of its Subsidiaries is a party.
- (e) <u>Equity-Related Agreements</u>. <u>Section 2.4(e)</u> of the Company Disclosure Schedule sets forth a complete and correct list of all Equity-Related Agreements. The Company has made available to Parent complete and correct copies of all Equity-Related Agreements.
- (f) Promised Securities. Section 2.4(f) of the Company Disclosure Schedule (i) identifies each Person with an offer letter or other Contract that contemplates a grant of Company Profits Interest Units or the issuance of other Company Securities by the Company or any of its Subsidiaries, or who has otherwise been promised Company Profits Interest Units or other Company Securities by the Company or any of its Subsidiaries, which Company Profits Interest Units have not been granted, or other Company Securities have not been issued, as of the Agreement Date (each, a "Specified Person") and (ii) for each Specified Person, the number and type (including class and series, if applicable) of Company Profits Interest Units or Company Securities so contemplated or promised to be granted or issued. The Company has made available to Parent complete and correct copies of each such offer letter, Contract or promise or, if oral, written summaries thereof.
  - 2.5 <u>No Conflict; Required Consents.</u>

- (a) The execution, delivery and performance by the Company of this Agreement and each of the Company Ancillary Agreements, and the consummation by the Company of the transactions contemplated hereby and thereby, do not and will not: (i) conflict with or violate the Charter Documents of the Company, (ii) subject to compliance with Section 2.5(c), conflict with or violate any Laws or any judgment, decree or order to which the Company or any of its Subsidiaries is subject, or (iii) result (or would result, with or without notice or lapse of time, or both) in (A) any breach of or default under, (B) any Person's right to consent, notice, or right of termination, acceleration, cancellation, modification or amendment of, or right to any increased, additional, accelerated or guaranteed payment or performance under, (C) any Encumbrance (other than Permitted Encumbrances) on any properties or assets of the Company pursuant to, or (D) any adverse effect to the rights or obligations of the Company or loss of any benefit for the Company under, any Company Material Contract or government permit, except, in the case of clauses (ii) and (iii), as would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole.
- (b) The Company and the Mergers are not, and by the passage of time will not be, subject to a right of first negotiation, right of first offer or refusal, or any other similar right granted by the Company (or any of its Affiliates) to or in favor of a third party with respect to an Acquisition Proposal or a potential Acquisition Proposal or otherwise that could affect or cause any delay in the consummation of the Mergers.
- (c) No consent, approval, order or authorization of, or registration, declaration or filing with, or notice to any Governmental Authority is necessary or required to be made or obtained by the Company to enable the Company to lawfully execute, deliver and perform this Agreement and each of the Company Ancillary Agreements and to consummate the transactions contemplated hereby and thereby, except for (i) such consents, approvals, orders, authorizations, registrations, declarations and filings, if any, that if not made or obtained by the Company would not reasonably be expected to be material to the Company and its Subsidiaries, taken as a whole or would not be reasonably be expected to impede, interfere with, prevent or delay, the consummation of the Mergers or the other transactions contemplated hereby, and (ii) the filing of the First Certificate of Merger and the Second Certificate of Merger with the Secretary of State of the State of Delaware.
- (d) The Company has obtained all consents from the Funds, and provided all notices to the Funds, required under the Advisers Act describing the transactions contemplated by this Agreement.
- 2.6 <u>Litigation</u>. There is no Action pending or, to the Knowledge of the Company, threatened, against the Company or any of its Subsidiaries or any officer, director or employee or agent of the Company and each of its Subsidiaries in his or her capacity as such. There is no judgment, decree, injunction, rule or order of any Governmental Authority, arbitrator or mediator binding on the Company and each of its Subsidiaries or any of its assets or properties. Neither the Company nor any of its Subsidiaries has brought any Action that is currently pending against any Governmental Authority or any other Person.

#### 2.7 <u>Taxes</u>.

## (a) <u>Tax Returns, Taxes and Audits</u>.

(i) The Company and each of its Subsidiaries (A) has properly completed and timely filed all federal and state income and other material Tax Returns required to be filed by it, and all such Tax Returns are complete and correct in all

material respects, (B) has timely paid all Taxes required to be paid by it for which payment was due (whether or not shown on any Tax Return), (C) has established an adequate accrual or reserve for the payment of all Taxes payable in respect of the periods or portions thereof prior to the Balance Sheet Date (which accrual or reserve as of the Balance Sheet Date is fully reflected on the face of the Company Balance Sheet (rather than in any notes thereto)), (D) has made all estimated Tax payments (including accounting for franchise taxes and withholding taxes) required to be made sufficient to avoid any underpayment, penalties or interest, (E) has no Liability for Taxes in excess of the amount so paid or accruals or reserves so established, and (F) since the Balance Sheet Date, has not incurred any Liability for Taxes outside the Ordinary Course of Business or otherwise inconsistent with past custom and practice other than as a result of the transactions contemplated by this Agreement, and the election to be treated as a corporation for income tax purposes effective as of February 2, 2025. The Company and each of its Subsidiaries has made available to Parent complete and correct copies of all federal and state income Tax Returns and other material Tax Returns, examination reports, and statements of deficiencies assessed against or agreed to by the Company and each of its Subsidiaries filed or received for all taxable years remaining open under the applicable statute of limitations.

- (ii) No deficiency for any Tax has been threatened, claimed, proposed or assessed against the Company, each of its Subsidiaries or any of its officers, employees or agents in their capacity as such.
- (iii) The Company and each of its Subsidiaries has not received from the IRS or any other Governmental Authority (including any sales or use Tax authority) any written or, to the Company's Knowledge, any unwritten (A) notice indicating an intent to open an audit or other review related to any Tax matter, (B) request for information related to any Tax matter, or (C) notice of deficiency or proposed adjustment of any amount of Tax proposed, asserted, or assessed by any Governmental Authority against the Company. No Tax Return of the Company or each of its Subsidiaries is under audit by the IRS or any other Governmental Authority and any past audits (if any) have been completed and fully resolved to the satisfaction of the applicable Governmental Authority conducting such audit and all Taxes determined by such audit to be due from the Company have been paid in full to the applicable Governmental Authorities or adequate reserves therefor have been established and are reflected on the face of the Company Balance Sheet (rather than in any notes thereto). No claim has ever been made by a Governmental Authority in a jurisdiction where the Company or each of its Subsidiaries does not file Tax Returns that it is or may be subject to Tax or required to file any Tax Return in that jurisdiction.
- (iv) No Tax liens are currently in effect against any of the assets of the Company or each of its Subsidiaries other than liens for Taxes not yet due and payable. There is not in effect any waiver by the Company or each of its Subsidiaries of any statute of limitations with respect to any Taxes nor has the Company or each of its Subsidiaries agreed to any extension of time for filing any Tax Return that has not been filed. The Company and each of its Subsidiaries has not consented to extend the period in which any Tax may be assessed or collected by any Tax agency or authority which extension is still in effect.
- (v) No Company Units or Company Profits Interest Units are, or have ever been, subject to a "substantial risk of forfeiture" within the meaning of Section 83 of the Code.

- (vi) The Company and each of its Subsidiaries will not be required to include any item of income in, or exclude any item of deduction from, taxable income for any Tax period (or portion thereof) ending after the Closing Date as a result of: (A) any change in method of accounting made prior to the Closing Date for a taxable period ending on or prior to the Closing Date, including the application of Section 481 or Section 263A of the Code (or any corresponding or similar provisions of state, local or foreign Tax Laws) to transactions, events or accounting methods employed prior to the Closing, (B) any use of an improper method of accounting for a taxable period (or portion thereof) ending on or prior to the Closing Date; (C) any "closing agreement," as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign Tax Law) executed on or prior to the Closing Date, (D) any "intercompany transaction" or any "excess loss account" (within the meaning of Treasury Regulations Sections 1.1502-13 and 1502-19, respectively) (or any corresponding or similar provisions of state, local or foreign Tax Law) occurring or arising with respect to any transaction on or prior to the Closing Date, (E) any installment sale, open transaction or other transaction made on or prior to the Closing Date, (F) any prepaid amount received or paid on or prior to the Closing Date or (G) the application of Section 952(c)(2) of the Code or (H) the application of Section 951 or 951A of the Code with respect to income earned or recognized or payments received prior to the Closing.
- (vii) The Company and each of its Subsidiaries has collected and remitted all sales, use, value added, ad valorem, personal property and similar Taxes ("Sales Taxes") with respect to sales made or services provided and, for all sales or provision of services that are exempt from Sales Taxes and that were made without charging or remitting Sales Taxes, the Company and each of its Subsidiaries has received and retained any required Tax exemption certificates or other documentation qualifying such sale or provision of services as exempt.
- (b) Withholding. The Company and each of its Subsidiaries has complied with all Laws relating to the payment, collection and withholding of any Tax (including withholding of Taxes pursuant to Sections 1441, 1442, 1445 and 1446 of the Code or any corresponding or similar provisions of state, local or foreign Tax Law), and has, within the time and in the manner prescribed by Law, collected and withheld from employee wages and other amounts payable to or from third parties (including members, partners or other service providers) and paid over to the proper Governmental Authorities all amounts required to be so collected and withheld and paid over under all Laws (including the Federal Insurance Contribution Act, Medicare, Federal Unemployment Tax Act and relevant state income and employment Tax withholding Laws), including federal, state, local and foreign Taxes, and has timely filed or provided all withholding Tax Returns in accordance with Law. All IRS Forms W-2 and 1099 or other applicable forms required with respect thereto have been properly completed and timely filed.

#### (c) Tax Status and Indemnification Obligations.

- (i) The Company and each of its Subsidiaries is not a party to or bound by any Tax sharing, indemnity, allocation or similar Contract (other than Contracts entered into in the Ordinary Course of Business the primary purpose of which does not relate to Taxes), and the Company has no Liability to another party under any such Contract.
- (ii) The Company and each of its Subsidiaries is not now, nor has ever been, a member of a consolidated, combined, unitary, aggregate or affiliated group

(within the meaning of Code Section 1504(a)) of which the Company and each of its Subsidiaries was not the ultimate parent corporation. The Company and each of its Subsidiaries has no Liability for the Taxes of any Person under Treasury Regulations Section 1.1502-6 (or any corresponding or similar provision of state, local or foreign Tax Law), as a transferee or successor, by Contract (other than Contracts entered into in the Ordinary Course of Business the primary purpose of which does not relate to Taxes). None of the Company, each of its Subsidiaries or any "dual resident corporation" (within the meaning of Section 1503(d) of the Code) in which the Company or each of its Subsidiaries is considered to hold an interest, has incurred a dual consolidated loss within the meaning of Section 1503 of the Code.

- (iii) The Company and each of its Subsidiaries has never constituted either a "distributing corporation" or a "controlled corporation" in a distribution of stock intended to qualify in whole or in part for Tax-free treatment under Section 355 of the Code (or so much of Section 356 of the Code as relates to Section 355 of the Code).
- (iv) The Company and each of its Subsidiaries has not been a party to a transaction or Contract that is in conflict with the Tax rules on transfer pricing in any relevant jurisdiction. All applicable transfer pricing rules have been complied with, and all documentation required by all relevant transfer pricing Laws have been timely prepared.
- (v) Prior to February 2, 2025, the Company was properly treated as a partnership for U.S. federal income Tax purposes. On and after February 2, 2025, the Company has been classified as a corporation for U.S. federal and applicable state income tax purposes. Each Subsidiary of the Company is and has always been treated as either a partnership or disregarded entity for U.S. federal income Tax purposes. The Company uses, and has always used, the accrual method of accounting for income Tax purposes and the taxable year of the Company is the calendar year ending December 31.
- (vi) The Company and each of its Subsidiaries is not nor has it ever been the beneficiary of any Tax exemption or Tax holiday.
- (vii) The Company and each of its Subsidiaries has not ever participated in or is participating in an international boycott within the meaning of Section 999 of the Code.
- (viii) The Company and each of its Subsidiaries has not ever requested or received a ruling from any Tax authority or signed a closing or other agreement with any Tax authority.
- (ix) The Company and each of its Subsidiaries (A) has not, and has not ever had, a permanent establishment in any country other than the country in which it is organized and resident, (B) has not engaged in a trade or business in any country other than the country in which it is organized and resident that subjected it to Tax in such country, (C) is not, and has not ever been, subject to Tax in a jurisdiction outside the country in which it is organized or resident, or (D) is not required to register in any jurisdiction for VAT purposes pursuant to applicable Law.
- (x) The Company and each of its Subsidiaries is not, nor has it been, a "United States real property holding corporation" within the meaning of Section

- 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.
- (xi) The Company and each of its Subsidiaries have not been nor will be required to include in income any amount under Section 965 of the Code.
- (xii) The Company and each of its Subsidiaries have not requested nor received any letter ruling from the IRS (or any comparable ruling from any other Tax authority).
- (xiii) The Company and each of its Subsidiaries have not deferred or delayed the payment of any Taxes under the CARES Act or otherwise as a result of the effects of the COVID-19 Pandemic.
- (xiv) Each "nonqualified deferred compensation plan" within the meaning of Section 409A of the Code has been operated and maintained, in all material respects, in operational and documentary compliance with Section 409A of the Code and applicable guidance thereunder.
- (xv) No Company Benefit Arrangement provides for any tax "gross-up" or similar "make-whole" payments.
- (xvi) Neither the execution and delivery of this Agreement, the shareholder approval of this Agreement, nor the consummation of the transactions contemplated hereby will result (either alone or in conjunction with any other event) in any "excess parachute payment" as defined in Section 280G of the Code. Any solicitation of Company Securityholder approval of any payments in connection with the transactions contemplated by this Agreement that may constitute "parachute payments" within the meaning of Section 280G of the Code was solicited in accordance with Section 280G(b)(5)(B) of the Code and the Treasury Regulations thereunder.
- (xvii) The Company and each of its Subsidiaries have not made an election to be subject to the Partnership Audit Procedures with respect to a taxable period beginning before January 1, 2018. For this purpose, the "Partnership Audit Procedures" means Subchapter C of Chapter 63 of the Code, as modified by Section 1101 of the Bipartisan Budget Act of 2015, Pub. L. No. 114-74, any amended or successor version, Treasury Regulations promulgated thereunder, official interpretations thereof, related notices or other related administrative guidance.
- (xviii) The Company and each of its Subsidiaries have not taken or agreed to take any action nor to the Knowledge of the Company is there any fact or circumstance that would reasonably be expected to prevent or impede the Mergers from qualifying as a tax-free "reorganization" within the meaning of Section 368(a) of the Code.
- (xix) The Company is not and has never been liable for any amount due to a Governmental Authority under any applicable escheat, abandoned property or other similar law.
- (d) <u>No Tax Shelters</u>. The Company and each of its Subsidiaries has (i) no disclosure obligation under Section 6662 of the Code or comparable provisions of state, local or foreign Law (ii) not participated in any listed transaction within the meaning of Treasury Regulations Section 1.6011-4(b). The Company and each of its Subsidiaries have not

consummated, has not participated in, and is not currently participating in any transaction which was or is a "tax shelter" transaction as defined in Sections 6662, 6011, 6012 or 6111 of the Code or the Treasury Regulations promulgated thereunder.

- (e) <u>Certain Limitations and Qualifications</u>. Notwithstanding anything to the contrary in this Agreement, it is agreed and understood that (i) other than with respect to <u>Sections 2.7(a)(vi)</u>, <u>2.7(c)(i)</u>, <u>2.7(c)(ii)</u>, <u>2.7(c)(xii)</u>, or <u>2.7(c)(xiii)</u>, the representations and warranties of the Company in this <u>Section 2.7</u> refer only to activities prior to the Closing and shall not serve as representations and warranties regarding, or a guarantee of, nor can they be relied upon with respect to, Taxes attributable to any Tax period (or portion thereof) beginning, or Tax positions taken, after the Closing Date, and (ii) no representations or guarantees are made with respect to the amount or availability of Tax attributes of the Company or any of its Subsidiaries for any taxable period (or portion thereof) beginning after the Closing Date.
- 2.8 <u>Related Party Transactions</u>. There are no outstanding amounts payable to or receivable from, or advances by the Company and each of its Subsidiaries to, and the Company and each of its Subsidiaries is not otherwise a creditor or debtor to, or party to any Contract or transaction with, any Related Party, except (a) to the extent provided for by the terms and conditions of Company Benefit Arrangements listed in <u>Section 2.16(m)</u> of the Company Disclosure Schedule or for amounts due as salaries and bonuses under employment agreements or employee benefit plans and amounts payable in reimbursement of ordinary expenses and (b) transactions evidenced by the Equity-Related Agreements listed in <u>Section 2.4(e)</u> of the Company Disclosure Schedule.

# 2.9 <u>Company Financial Statements</u>.

- (a) <u>Section 2.9(a)</u> of the Company Disclosure Schedule sets forth the Company Financial Statements. The Company Financial Statements: (i) are derived from the Books and Records, (ii) fairly present in all material respects the financial condition of the Company and its Subsidiaries at the dates therein indicated and the results of operations and cash flows of the Company and its Subsidiaries for the periods therein specified, and (iii) have been prepared in accordance with the cash basis method of accounting applied on a basis consistent with prior periods and the Company Financial Statements do not have notes and are subject to normal year-end adjustments, the effect of which are not, individually or in the aggregate, material to the Company and its Subsidiaries, taken as a whole.
- (b) Any of the accounts receivable, whether billed or unbilled, of the Company and each of its Subsidiaries arose in the Ordinary Course of Business, are carried at values determined in accordance with the Agreed Accounting Principles consistently applied, are not subject to any valid set-off or counterclaim, do not represent obligations for goods sold on consignment, on approval or on a sale-or-return basis or subject to any other repurchase or return arrangement.
- (c) Any of the accounts payable of the Company and each of its Subsidiaries arose in the Ordinary Course of Business and are carried at values determined in accordance with the Agreed Accounting Principles consistently applied and are not more than 90 days' past due.
- (d) The Company and each of its Subsidiaries has no material Liabilities except for (i) those shown on the Company Balance Sheet, (ii) those that were incurred after the Balance Sheet Date in the Ordinary Course of Business, (iii) any executory obligations (that are not arising out of breach of contract or breach of warranty) arising under any Contracts to which the Company or any of its Subsidiaries is a party and (iv) Transaction Expenses. The Company

and each of its Subsidiaries has no binding "off-balance sheet arrangement" within the meaning of Item 303 of Regulation S-K promulgated under the Securities Act. No bankruptcy, insolvency, winding up, or similar proceedings have occurred or are pending or, to the Knowledge of the Company, threatened against the Company or any of its Subsidiaries or the Company's or any of its Subsidiaries' properties or assets.

- (e) <u>Section 2.9(e)</u> of the Company Disclosure Schedule sets forth a complete and correct list of each item of Indebtedness (other than Wage Obligations) of the Company and each of its Subsidiaries as of the Agreement Date, identifying the name and address of the creditor thereto, all Contracts evidencing such Indebtedness, the amount of such Indebtedness as of the close of business on the Agreement Date, and any restriction or penalty upon the prepayment of any such Indebtedness.
- (f) The Company and each of its Subsidiaries maintains a system of internal accounting controls reasonably designed for a Company and each of its Subsidiaries of its size and stage to provide reasonable assurances (i) that transactions, receipts and expenditures of the Company and each of its Subsidiaries are being executed and made only in accordance with appropriate authorizations of management and the managers of the Company, (ii) that transactions are recorded as necessary (A) to permit preparation of financial statements in conformity with the cash basis method of accounting applied on a basis consistent with prior periods and (B) to maintain accountability for assets, (iii) regarding prevention or timely detection of unauthorized acquisition, use or disposition of the assets of the Company and each of its Subsidiaries and (iv) that the amount recorded for assets on the books and records of the Company and each of its Subsidiaries is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences.
- (g) Neither the Company nor, to the Knowledge of the Company, any of its Subsidiaries or Representatives, has received or otherwise obtained any written, or to the Knowledge of the Company, oral, complaint, allegation, assertion or claim regarding the accounting or auditing practices, procedures, methodologies or methods of the Company and each of its Subsidiaries or its internal accounting controls, including any complaint, allegation, assertion or claim that the Company has engaged in questionable accounting or auditing practices. There has been no fraud, whether or not material, involving any member of the board of managers or any officer of the Company or any of its Subsidiaries or any employee of the Company or any of its Subsidiaries who has a significant role in the Company's and or any of its Subsidiaries' internal controls over financial reporting.
- 2.10 <u>Title to Properties</u>. The Company and each of its Subsidiaries has good and valid title to, or in the case of leased assets and properties, valid leasehold interests in, all of the material tangible assets and properties (including those shown on the Company Balance Sheet and not subsequently disposed as reflected in the books and records of the Company and each of its Subsidiaries) used or held for use by the Company or any of its Subsidiaries in the operation of the Company Business, free and clear of all Encumbrances, other than Permitted Encumbrances. All machinery, vehicles, equipment and other material tangible personal property owned or leased by the Company or any of its Subsidiaries or used by the Company or any of its Subsidiaries in the Company Business are in good operating condition and satisfactory repair, normal wear and tear excepted and are sufficient for the continued operation of the Company Business immediately following the Closing. <u>Section 2.10(a)</u> of the Company Disclosure Schedule sets forth a complete and correct list of all leases or licenses with respect to material tangible personal property used by the Company or any of its Subsidiaries. <u>Section 2.10(b)</u> of the Company Disclosure Schedule sets forth a complete and correct list of all real property leases or licenses to which the Company or any of its Subsidiaries is a party. All leases

of real or personal property to which the Company or any of its Subsidiaries is a party are in full force and effect with respect to the Company or its Subsidiary that is a party thereto and, to the Company's Knowledge, with respect to each other party thereto, and afford the Company or its Subsidiaries a valid leasehold interest in, or license to use, the real or personal property that is the subject of such lease or license. All rents, required deposits and additional rents which are due under the terms of such leases have been paid in full. The Company and each of its Subsidiaries has never owned any real property.

- 2.11 <u>Absence of Certain Changes</u>. Since the Balance Sheet Date and through the Agreement Date, the Company Business has been operated in the Ordinary Course of Business and there has not been with respect to the Company any Material Adverse Effect or with respect to the Company or any of its Subsidiaries any material damage, destruction or loss of any material property or material asset, whether or not covered by insurance, except ordinary wear and tear. Except as set forth in <u>Section 2.11</u> of the Company Disclosure Schedule, since the Balance Sheet Date, there has not been with respect to the Company or any of its Subsidiaries any:
- (a) amendment or change in its Charter Documents (except for any change to the information set forth on any schedule of members);
- (b) incurrence, creation or assumption of (i) any Encumbrance on any of its assets or properties (other than Permitted Encumbrances) or (ii) any indebtedness for borrowed money;
- (c) acceleration or release of any vesting condition to the right to purchase or otherwise acquire any Company Securities, or any acceleration or release of any right to repurchase Company Securities upon the securityholder's termination of employment or services with it or pursuant to any right of first refusal;
- (d) payment or discharge of any of its Liabilities except (i) in the Ordinary Course of Business or (ii) Transaction Expenses;
- (e) purchase, license, sale, grant, assignment or other disposition or transfer, of any of its material assets (including material Company Intellectual Property Rights and other material intangible assets) or properties, other than in the Ordinary Course of Business;
- (f) declaration, setting aside or payment of any dividend on, or the making of any other distribution in respect of, its equity securities, or any split, combination or recapitalization of its Company Securities or any direct or indirect redemption, purchase or other acquisition of any of its Company Securities or any change in any right, preference, privilege or restriction of any of its outstanding Company Securities (other than repurchases of Company Securities in accordance with applicable Contracts in connection with the termination of service of employees or other service providers, in each case in effect on the Agreement Date and disclosed on the Company Disclosure Schedule);
- (g) (i) grant or issuance of Company Securities or (ii) promise with respect to the granting of Company Profits Interest Units or other Company Securities;
- (h) creation of any Subsidiary or the acquisition of any shares or other securities of, or any direct or indirect equity interest in, any Person;

- (i) hiring, termination or resignation of any officer, employee, independent contractor or other service provider or director of the Company, except as required by applicable Law or a Company Benefit Arrangement in effect on the date of this Agreement and listed in <u>Section 2.16</u> of the Company Disclosure Schedule;
- (j) change or increase (or promise to change or increase) in the compensation or benefits payable or to become payable to any of its current or former officers, directors, employees, independent contractors, agents, or other service providers, or change or increase (or promise to change or increase) in any bonus, pension, severance, change-of-control, retention, insurance or other benefit payment or arrangement (including any equity awards) payable or to become payable to any of its current officers, directors, managers, employees, independent contractors, consultants, agents, or other service providers, except as required by applicable Law or a Company Benefit Arrangement in effect on the date of this Agreement and listed in Section 2.16 of the Company Disclosure Schedule;
- (k) change with respect to title, reporting or principal roles or responsibilities of any of its officers or management, supervisory or other key personnel, except as required by applicable Law or a Company Benefit Arrangement in effect on the date of this Agreement and listed in <u>Section 2.16</u> of the Company Disclosure Schedule;
- (l) loan, advance or capital contribution by the Company to, or any investment by the Company in, any Person (other than the advancement of travel expenses to employees in the Ordinary Course of Business);
- (m) entering into or negotiation of any collective bargaining agreement or similar agreement or relationship with any union, works council, or labor organization and/or labor association;
  - (n) change in the manner in which it extends discounts, credits or warranties;
  - (o) any acquisition, sale or transfer of any material asset of the Company;
- (p) adoption of a plan or agreement of complete or partial liquidation, dissolution, restructuring, consolidation or other reorganization;
- (q) change in accounting or Tax reporting methods or practices (including any change in depreciation or amortization policies or rates or revenue recognition policies) or any revaluation of any of its assets;
- (r) settlement or compromise of any claim, notice, audit report or assessment in respect of Taxes; amendment to or filing of any Tax Return; making of, change in, or revocation of any election in respect of Taxes; adoption, change in, or revocation of any accounting method in respect of Taxes; surrender of any right to claim a refund of Taxes; entering into of any Tax allocation, sharing or indemnity agreement or closing agreement relating to Taxes; consent to any extension or waiver of the limitation period applicable to any claim or assessment in respect of Taxes; or incurred any Tax outside the Ordinary Course of Business; or otherwise taking any action, omitting to take any action or entered into any other transaction outside the Ordinary Course of Business that would have the effect of increasing the Tax liability or reducing any Tax asset of Parent in respect of any Tax period ending after the Closing Date;

- (s) deferral of the payment of any accounts payable other than in the Ordinary Course of Business, or any discount, accommodation, customer credit or other concession made in order to accelerate or induce the collection of any receivable;
- (t) Action or, to the Knowledge of the Company, threatened Action, initiated by or against the Company, or any Action or, to the Knowledge of the Company, threatened Action, settled or otherwise resolved by the Company;
  - (u) capital expenditure in excess of \$25,000;
  - (v) entering into any new line of business; or
- (w) entry or commitment to enter into, any Contract to do any of the things described in the preceding clauses (a) through (v) other than negotiations and agreements with Parent and its representatives regarding the transactions contemplated by this Agreement.

# 2.12 <u>Company Material Contracts</u>.

- (a) <u>Section 2.12(a)</u> of the Company Disclosure Schedule sets forth a list of each Contract of the following types to which the Company or any of its Subsidiaries is a party or to which the Company or any of its Subsidiaries or any of their respective assets or properties is bound (excluding any Company Benefit Arrangement), including the applicable subsection(s) to which such Contract is responsive:
  - (i) any Contract pursuant to which the Company and its Subsidiaries (A) made or received in 2024 cash payments in excess of \$25,000 or (B) would reasonably be expected to make or receive in 2025, cash payments in excess of \$25,000;
  - (ii) any Contract with the Company's users, customers or clients other than pursuant to the Company's standard end user agreement(s) ("Standard EULAs");
  - (iii) any dealer, distributor, reseller, VAR (value added reseller), sales representative or similar Contract under which any third party is authorized to sell, license, sublicense, distribute, market or take orders for any Company Offering or Company Technology;
  - (iv) any Contract that (i) provides for the authorship, invention, creation, conception or other development of any Technology or Intellectual Property Rights (A) by the Company for any other Person or (B) for the Company by any other Person, including, in the case of each of clauses (A) and (B), any joint development, (ii) provides for the assignment or other transfer of any ownership interest in Technology or Intellectual Property Rights (1) to the Company from any other Person (other than Invention Assignment Agreements, as defined below) or (2) by the Company to any other Person, (iii) includes any grant of an Intellectual Property License to any other Person by the Company or any of its Subsidiaries ("Outbound Licenses") (other than, with respect to this subsection (iii) only, non-exclusive licenses granted to the Company's end users, customers or clients in the Ordinary Course of Business), or (iv) includes any grant of an Intellectual Property License to the Company or any of its Subsidiaries by any other Person ("Inbound Licenses") (other than, with respect to this subsection (iv) only, (x) Non-Negotiated Vendor Contracts, and (y) licenses for Open Source Software listed in Section 2.13(n) of the Company Disclosure Schedule);

- (v) any Contract (other than the Charter Documents of the Company or any of its Subsidiaries) providing for the establishment or operation of a partnership, joint venture, joint marketing, joint development or similar arrangement with any other Person;
- (vi) any Contract under which the Company has advanced or loaned any amount to any of its managers, officers, directors, or employees (other than under the Company's 401(k) plan) or to any other Person;
- (vii) any Contract creating or evidencing any Indebtedness (other than Wage Obligations) of the Company or otherwise placing an Encumbrance (other than a Permitted Encumbrance) on any asset of the Company;
- (viii) any Contract (A) that restricts the Company's or any of its Subsidiaries', or following the Effective Time will restrict Parent's or any of its Affiliates', right to (1) participate or compete in any line of business, market or geographic area during any period of time, or (2) solicit potential employees, independent contractors or other suppliers or customers or (B) pursuant to which the Company or any of its Subsidiaries has granted most favored nation pricing;
- (ix) any Contract under which the Company or any of its Subsidiaries grant or are bound by or, following the Effective Time, purports to have Parent or any of its Affiliates grant or be bound by, any exclusive rights, noncompetition rights, rights of first negotiation or similar rights;
- (x) any Contract that following the Effective Time would or would purport to: (i) require Parent or any of its Affiliates (other than the Company or any of its Subsidiaries) to grant any Intellectual Property License, or (ii) restrict Parent or any of its Affiliates (other than the Company or any of its Subsidiaries) from performing any of the activities listed in Section 2.12(a)(vii);
- (xi) any Contract providing for the sale, issuance, grant, exercise, award, purchase, repurchase or redemption of any Company Securities;
- (xii) any Contract with any labor union, works council, labor organization and/or labor association, or any collective bargaining agreement or similar Contract with the Company's employees;
- (xiii) any Contract relating to the settlement or other resolution of any Action or threatened Action (including any agreement under which any employment-related claim is settled);
- (xiv) (i) any Contract that includes an obligation by the Company to indemnify any other Person against any claim of infringement, misappropriation, misuse, dilution or violation of any Intellectual Property Rights or Technology, and (ii) any other Contract of guarantee, support, indemnification, assumption or endorsement of, or any similar commitment with respect to, the obligations, Liabilities or indebtedness of any other Person, other than, in the case of each of clauses (i) and (ii), Non-Negotiated Vendor Contracts and Standard EULAs;
  - (xy) any Contract with any Related Party (other than those subject to the exceptions in Section 2.8);

- (xvi) any Contract relating to the acquisition or disposition of a material portion of the assets of, or any equity interest in, any Person or business, whether by way of merger, consolidation, amalgamation, plan or scheme of arrangement, purchase or sale of Company Securities or assets, license or otherwise;
- (xvii) any Contract that involves the sharing of profits with other Persons or the payment of royalties or referral fees to any other Person, excluding Non-Negotiated Vendor Contracts;
- (xviii) any Contract imposing any support, maintenance or service obligations on the part of the Company, excluding any such Contract that has been entered into in the Ordinary Course of Business;
  - (xix) any Contract that contains an earn-out, escrow or other similar contingent payment or obligation;
- (xx) any Contract between the Company and any Governmental Authority, university, college other educational institution or research center;
- (xxi) all Contracts with any customer or other Person to develop or customize any Company product or services, or to provide support for, customize or develop any third-party product, service or platform;
- (xxii) all business associate agreements, which shall include whether the Company is acting as a subcontractor business associate on behalf of such Person party to the business associate agreement or if such Person is acting as a subcontractor business associate to the Company;
  - (xxiii) any lead generation Contracts or similar arrangements;
- (xxiv) any Contracts pursuant to which the Company or any of its Subsidiaries are parties to any agreement with an Advisory Client (including, without limitation, any side letters or other arrangements to allocate or bear expenses of any Fund or provide any nonstandard rights or privileges to any Advisory Client);
- (xxv) any Contract under which the Company or any of its Subsidiaries has triggered an obligation to provide a refund or rebate or any other payment related to volume-based pricing discounts or similar provisions;
- (xxvi) any Contract pursuant to which any of the Company or any of its Subsidiaries has undertaken or agreed to cap, waive, offset, reimburse or otherwise reduce any or all fees or charges payable by or with respect to any of the Fund Investors; or
- (xxvii) any Contract relating to the distribution, marketing, advertising or sales of any Company Offerings or Company Intellectual Property Rights.
- (b) All Company Material Contracts are in written form. The Company has made available to Parent correct and complete copies of each Company Material Contract, including all modifications, amendments, and supplements thereto (excluding non-exclusive licenses granted to the Company's end users, customers or clients in the Ordinary Course of Business, Non-Negotiated Vendor Contracts, and licenses for Open Source Software listed in

Section 2.13(n) of the Company Disclosure Schedule). Each of the Company Material Contracts constitutes a valid and binding obligation of the Company or any of its Subsidiary (and, to the Company's Knowledge, any other party thereto), as applicable, enforceable in accordance with its terms against the Company or any of its Subsidiaries (and, to the Company's Knowledge, any other party thereto), subject to the Enforceability Exceptions, and is in full force and effect in accordance with its terms with respect to Company or any of its Subsidiaries (and, to the Company's Knowledge, any other party thereto). Neither the Company nor any of its Subsidiaries nor, to the Company's Knowledge, any other party thereto, is in material breach or default under any Company Material Contract, no event has occurred that (with or without notice or lapse of time, or both) would reasonably be expected to or would constitute a material breach or default thereunder by the Company or any of its Subsidiaries or, to the Company's Knowledge, any other party thereto, and neither the Company nor any of its Subsidiaries has received (i) any written or, to the Knowledge of the Company, oral claim of any such material breach or default or (ii) written notice of cancellation by the other party to any Company Material Contract.

# 2.13 <u>Intellectual Property</u>.

- (a) Section 2.13(a) of the Company Disclosure Schedule sets forth a complete and correct list of all (i) Registered Company Intellectual Property Rights, and (ii) social media identifiers and material unregistered Marks owned or purported to be owned by, or exclusively licensed to, the Company or any of its Subsidiaries. For each item of Registered Company Intellectual Property Rights, Section 2.13(a) of the Company Disclosure Schedule lists (A) the record owner of such item, and, if different, the legal owner and beneficial owner of such item, (B) the jurisdiction in which such item is issued, registered or pending, (C) the issuance, registration or application date and number of such item, and (D) for each Domain Name registration, the applicable Domain Name registrar, the name of the registrant and the expiration date for the registration.
- (b) Each item of Registered Company Intellectual Property Rights is registered in the name of the Company or any of its Subsidiaries. All necessary fees and filings (including the payment of maintenance fees and taxes) with respect to any Registered Company Intellectual Property Rights have been timely submitted to the relevant Governmental Authorities and Domain Name registrars to maintain such Registered Company Intellectual Property Rights in full force and effect. Except as set forth in Section 2.13(b) of the Company Disclosure Schedule, there are no renewals, annuities, payments, fees, responses to office actions or other filings required to be made and having a due date with respect to any Registered Company Intellectual Property Rights within 120 days after the Agreement Date. No issuance or registration obtained and no application filed by the Company or any of its Subsidiaries for any Intellectual Property Rights has been cancelled, abandoned, allowed to lapse or not renewed, except where the Company or any of its Subsidiaries has, in its reasonable business judgment, decided to cancel, abandon, allow to lapse or not renew such issuance, registration or application. Neither the Company nor any of its Subsidiaries has claimed any status in the application for or registration of any Registered Company Intellectual Property Rights, including "small business status," that would not be applicable to Parent, the Surviving Company and/or the Surviving Entity. The Registered Company Intellectual Property Rights are, and, as of and immediately following the Effective Time, will be subsisting and enforceable, and, to the Company's Knowledge, there are no facts or circumstances that would render any Registered Company Intellectual Property Rights invalid or unenforceable.
- (c) The Company or one of its Subsidiaries is the sole and exclusive owner of all right, title and interest in and to (i) all Registered Company Intellectual Property Rights and

- (ii) all other Company Intellectual Property Rights and Company Technology owned or purported to be owned by, or subject to an obligation to be assigned to, the Company (clauses (i) and (ii) collectively, the "Owned Company IP"), free and clear of all Encumbrances other than licenses granted under non-exclusive licenses granted to end users, customers or clients in the Ordinary Course of Business. The Company or one of its Subsidiaries has the sole and exclusive right to bring a claim or suit against a third party for infringement or misappropriation of Owned Company IP. Neither the Company nor any of its Subsidiaries (i) transferred to any Person ownership of, or granted any exclusive license with respect to, any Intellectual Property Rights that are or would have been, but for such transfer or grant, Company Intellectual Property Rights or (ii) permitted the rights of the Company in any Intellectual Property Rights that are or were, at the time, Company Intellectual Property Rights to lapse or enter into the public domain. Neither the Company nor any of its Subsidiaries has undertaken any joint development or similar activities with any other Person. No Company Intellectual Property Rights or Company Technology or Company Offering is subject to any claim, proceeding or outstanding decree, order, judgment, stipulation or Contract restricting in any manner, the use, transfer, or, (except for nonexclusive licenses granted pursuant to Intellectual Property Licenses listed in Section 2.12(a)(iv)(3) of the Company Disclosure Schedule and Standard EULAs in each case that would restrict the Company's or any of its Subsidiaries' ability to grant exclusive licenses) licensing thereof by the Company or any of its Subsidiaries, or which may affect the validity, use or enforceability of such Company Intellectual Property Rights or Company Technology or Company Offering. All Company Intellectual Property Rights, Company Technology, and Company Offering that are not Owned Company IP ("Licensed IP") are validly licensed to the Company or any of its Subsidiaries pursuant to written Contracts. The Company has (and will continue to have immediately following the Effective Time) valid and continuing rights (under such Contracts) to use, sell, license and otherwise exploit, as the case may be, all Licensed IP as the same are currently used, sold, licensed and otherwise exploited by the Company.
- (d) All Owned Company IP is fully and freely transferable and assignable and may be transferred and assigned to Parent without restriction and without payment of any kind to any third Person.
- (e) The Owned Company IP and the Licensed IP constitute all of the Intellectual Property Rights and Technology that are used by the Company or any of its Subsidiaries to conduct the Company Business as it currently is conducted, including the design, development, manufacture, use, marketing, import for resale, distribution, licensing out and sale of any Company Offering.
- (f) None of the Registered Company Intellectual Property Rights have been or are subject to any interference, derivation, reexamination (including ex parte reexamination, inter partes reexamination, inter partes review, post grant review or covered business method (CBM) review), cancellation, or opposition proceeding.
- (g) Copies of the Company's or any of its Subsidiaries' standard form(s) of non-disclosure agreement, consulting or independent contractor agreement, the Standard EULA, and any other of the Company's standard form(s), including attachments, of agreements with content contributors or non-exclusive licenses of, or agreements to provide on a non-exclusive basis, Company Offerings to customers (collectively, the "Standard Form Agreements") have been made available to Parent.
- (h) Neither the conduct of the Company Business as it is currently conducted by the Company or any of its Subsidiaries, nor any Company Offering (including the design, development, use, practice, offering, licensing, provision, import, branding, advertising,

promotion, marketing, sale, distribution, making available, or other exploitation of any Company Offering) (i) has been or is infringing, misappropriating (or resulting from the misappropriation of), diluting, using or disclosing without authorization, or otherwise violating any Intellectual Property Rights of any third Person, (ii) has been or is contributing to or inducing any infringement, misappropriation, or other violation of any Intellectual Property Rights of any third Person, or (iii) has been or is constituting unfair competition or trade practices under the Laws of any relevant jurisdiction.

- (i) Neither the Company nor any of its Subsidiaries has received any notice or claim or, to the Company's Knowledge, threatened claim from any Person (i) alleging any infringement, misappropriation, misuse, dilution, violation, or unauthorized use or disclosure of any Intellectual Property Rights or Technology or unfair competition, (ii) inviting the Company to take a license under any Intellectual Property Rights or consider the applicability of any Intellectual Property Rights to any Company Offerings or the conduct of the Company Business or (iii) challenging the ownership, use, validity or enforceability of any Company Intellectual Property Rights or Company Technology. To the Company's Knowledge, there is no basis for any Person to make any such allegation, invitation, or challenge. Neither the Company nor any of its Subsidiaries is in violation in any material respect of the terms of any Intellectual Property License.
- (j) To the Company's Knowledge, no Person is infringing, misappropriating, misusing, diluting or violating any Company Intellectual Property Rights, Company Technology owned by or exclusively licensed to the Company, or Company Offering. Neither the Company nor any of its Subsidiaries has made any written or unwritten claim against any Person alleging any infringement, misappropriation, misuse, dilution or violation of any Company Intellectual Property Rights, Company Technology owned or purported to be owned by or exclusively licensed to the Company or any of its Subsidiaries, or Company Offering.
- (k) Neither this Agreement nor the transactions contemplated by this Agreement will result in Parent, the Company, any of the Company's Subsidiaries or any of their respective Affiliates being obligated to pay any royalties or other license fees with respect to Intellectual Property Rights of any third Person in excess of those payable by the Company or any of its Subsidiaries in the absence of this Agreement or the transactions contemplated hereby.
- (I) The Company and each of its Subsidiaries has taken commercially reasonable measures to protect all Company Data, Proprietary Information of the Company and all Proprietary Information of any third Person in the Company's or any of its Subsidiaries' possession or control, or to which the Company or any of its Subsidiaries has access, with respect to which the Company or any of its Subsidiaries has a confidentiality obligation. No such Proprietary Information has been authorized to be disclosed or has been actually disclosed to any Person other than pursuant to a written confidentiality Contract restricting the disclosure and use of such Proprietary Information. Each current and former employee, director, consultants, and independent contractor of the Company or any of its Subsidiaries that has been involved in the authorship, invention, creation, conception or other development of Intellectual Property Rights or any Technology in the scope of his, her or its employment or engagement with the Company or any of its Subsidiaries that effectively and validly assigns to the Company all Intellectual Property Rights and Technology authored, invented, created, conceived, or otherwise developed by such employee, consultant, independent contractor, or director in the scope of his, her or its employment or engagement with the Company in a form made available to Parent prior to the Agreement Date (an "Invention Assignment Agreement"). No current or former employee, consultant, independent contractor

or director of the Company or any of its Subsidiaries has (i) excluded any Technology (or any Intellectual Property Rights in or to any Technology) authored, invented, created, conceived, or otherwise developed prior to his or her or its employment or engagement with the Company from his or her or its assignment of inventions pursuant to such Person's Invention Assignment Agreement, (ii) failed to affirmatively indicate in such Invention Assignment Agreement that no Technology authored, invented, created, conceived, or otherwise developed by him or her or it prior to his or her or its employment or engagement with the Company exists, (iii) alleged, to the Company or any of its Subsidiaries or, to the Knowledge of the Company, any third Person, ownership or other exclusive rights by such employee, consultant, independent contractor or director, in any Technology authored, invented, created, conceived or otherwise developed by such employee, consultant, independent contractor or director in the scope of his, her or its employment or engagement with the Company or any of its Subsidiaries, or (iv) failed to effectively waive all moral rights held by such employee, consultant, independent contractor or director, in the scope of his, her or its employment or engagement with the Company in favor of the Company or any of its Subsidiaries. Without limiting the foregoing, all rights in, to and under all Intellectual Property Rights and Technology created by the Company's founders for or on behalf of or in contemplation of the Company or any of its Subsidiaries (or the Company Business) prior to their commencement of employment with the Company have been duly and validly assigned to the Company or any of its Subsidiaries.

- Software that is incorporated into, integrated or bundled with, linked with, or used in the Company Technology, except for Software licensed under Non-Negotiated Vendor Contracts or Open Source Software (and, for each item required to be listed in Section 2.13(m) of the Company Disclosure Schedule, the name of the licensor or owner of the Software and the Contract under which Software is licensed). Neither the Company nor any of its Subsidiaries has, nor has any other Person acting on the Company's or its Subsidiaries' behalf disclosed, delivered or licensed to any Person (other than employees, directors, consultants and independent contractors of the Company or any of its Subsidiaries who have entered into written confidentiality obligations with the Company or any of its Subsidiaries with respect to such source code or related materials), agreed to disclose, deliver or license to any Person (other than employees, directors, consultants and independent contractors of the Company and any of its Subsidiaries who have entered into written confidentiality obligations with the Company or any of its Subsidiaries with respect to such source code or related materials), or permitted the disclosure or delivery to any escrow agent or other Person of, any source code for any Company Offering or Company Intellectual Property Rights, and no Person has any right, contingent or otherwise, to obtain access to or use such source code. Neither this Agreement nor the transactions contemplated by this Agreement including the assignment or transfer to Parent, by operation of Law or otherwise, of any Company Intellectual Property Rights will result in, or entitle any Person to demand, the disclosure, delivery or license of any source code for any Company Offering to any Person.
- (n) Section 2.13(n) of the Company Disclosure Schedule sets forth a complete and correct list of all Open Source Software that it has embedded, linked, or distributed in connection with any Company Technology (whether generally available or in development), including the name of each such item and the license(s) applicable to each such item, and indicates whether such item has been distributed or modified by the Company or any of its Subsidiaries. Neither the Company nor any of its Subsidiaries has used Open Source Software in any manner that (i) subjects any Company Intellectual Property Rights to a Copyleft License or requires the disclosure or distribution in source code form of any Company Intellectual Property

Rights, including any portion of any Company Software or Company Offering other than such unmodified Open Source Software, (ii) requires the licensing of any Company Intellectual Property Rights, or any portion of any Company Software or Company Offering other than such unmodified Open Source Software, for the purpose of making derivative works, (iii) imposes any restriction on the consideration to be charged for the distribution of any Company Intellectual Property Rights, (iv) creates obligations for the Company or any of its Subsidiaries with respect to Company Intellectual Property Rights or grants to any third Person, any rights or immunities under Company Intellectual Property Rights, or (v) imposes any other limitation, restriction or condition on the right of the Company or any of its Subsidiaries to use or distribute any Company Intellectual Property Rights. With respect to any Open Source Software that is used by the Company or any of its Subsidiaries in the operation of its business, the Company and any applicable Subsidiaries are in compliance with all applicable licenses with respect thereto. There are and have been no claims asserted against the Company, any of its Subsidiaries or any of its customers or distributors related to Company Software, Company Offerings or Company Intellectual Property Rights.

- (o) The Company and each of its Subsidiaries conducts and has conducted the Company Business in such a manner as to take reasonable advantage, if and when applicable, of the safe harbors provided by Section 512 of the Digital Millennium Copyright Act (the "DMCA") and by any substantially similar applicable law in any other jurisdiction in which the Company or any of its Subsidiaries conducts the Company Business, including to the extent applicable by informing users of its products and services of such policy, designating an agent for notice of infringement, misappropriation or violation claims, registering such agent with the United States Copyright Office, and taking appropriate action expeditiously upon receiving notice of possible infringement, misappropriation or violation in accordance with the "notice and take-down" procedures of the DMCA or such other applicable law.
- (p) (i) No government funding and no facilities of any university, college, other educational institution or research center were used in the development of any Owned Company IP or any Company Offering or otherwise made available to the Company or any of its Subsidiaries for any purpose, and (ii) no Governmental Authority or any university, college, other educational institution or research center owns, purports to own, has any other rights in or to (including through any Intellectual Property License), or has any option to obtain any rights in or to, any Owned Company IP or Company Offering. No employee of the Company or any of its Subsidiaries who has been involved in the creation or development of any Technology or Intellectual Property Rights for the Company or any of its Subsidiaries, or have had access to such Technology or Intellectual Property Rights, has performed services for the government, university, college, or other educational institution or research center during a period of time during which such employee, consultant or independent contractor was also performing services for the Company or any of its Subsidiaries. Without limiting the foregoing, there are no current or contingent usage rights, march-in rights, manufacturing restrictions or other rights of any governmental entity in or to any Company Intellectual Property Rights, or in or to any other Intellectual Property Rights that is either used by or for the Company or any of its Subsidiaries or that is otherwise necessary to conduct the Company Business as currently conducted.
- (q) The consummation of the transactions contemplated by this Agreement will not result in the loss or impairment of any right of the Company or any of its Subsidiaries to own, use, practice or otherwise exploit any Company Intellectual Property Rights, Company Technology or Company Data. Neither the execution, delivery and performance of this Agreement or any Company Ancillary Agreement, nor the consummation of the transactions contemplated by this Agreement or any Company Ancillary Agreement will, pursuant to any Contract to which the Company or any of its Subsidiaries is a party or otherwise bound, result (or

purport to result) in the transfer or grant by the Company or Parent or any of their respective Affiliates to any Person (other than Parent and its Affiliates) of any ownership interest or Intellectual Property License with respect to any Company Intellectual Property Rights or Technology of Parent or any of its Affiliates.

- (r) There are no Contracts between the Company or any of its Subsidiaries on one hand and any other Person on the other hand with respect to any Intellectual Property Right or Technology under which there is any dispute regarding the scope of such Contract, or performance under such Contract, including with respect to any payments to be made or received by the Company or any of its Subsidiaries thereunder.
- (s) Each Company Offering performs in accordance with its documented specifications and as the Company or any of its Subsidiaries has warranted or guaranteed to its customers and clients in all material respects.
- (t) The Company has and enforces a policy to document all known bugs, errors and defects in Company Offerings, and such documentation is retained and is available internally at the Company. There are no bugs, errors or defects in Company Offerings or Company Software which do, or may reasonably be expected to, adversely affect the value, functionality or fitness for the intended purpose of such Company Offering or Company Software that would reasonably be expected to adversely affect the Company's ability or any of its Subsidiaries' abilities to perform any of their respective contractual obligations; nor has there been any claims asserted against the Company, any of its Subsidiaries or any of their respective customers or distributors related to Company Offerings or Company Intellectual Property Rights; and neither the Company nor any of its Subsidiaries has been nor are they required to recall any Company Offerings.
  - (u) No Company Offering or Company Intellectual Property Rights contains any Contaminants.
- (v) <u>Section 2.13(v)(i)</u> of the Company Disclosure Schedule identifies (A) all Company AI Products, and (B) all AI Technologies that are owned by or purported to be owned by the Company or any of its Subsidiaries. <u>Section 2.13(v)(ii)</u> of the Company Disclosure Schedule identifies each Third-Party AI Product used in and material to the operation of the Company's business and the manner in which the Company or any of its Subsidiaries uses each such Third-Party AI Product.
- (w) The Company and its Subsidiaries use and has used all AI Technologies in compliance in all material respects with all applicable terms. Neither the Company nor any of its Subsidiaries has received any (A) complaint from any Person, or (B) governmental inquiry related to its use of AI Technologies. All data and other information submitted to AI Technologies by or on behalf of the Company or any of its Subsidiaries is either proprietary to the Company or any its Subsidiaries or has been obtained by the Company and its applicable Subsidiaries in accordance with the applicable terms and Laws governing such use (including each end user license agreement, terms of use, privacy policies, consents, or other terms that govern the Company's collection and use of third party data) and in compliance with all required consent and notification obligations. Neither the Company nor any of its Subsidiaries has permitted any data (including Personal Information) or information Processed through the IT Systems to be used by any other Person to train any AI Technologies, nor has the Company or any of its Subsidiaries permitted the Company's or any of its Subsidiaries or of any third

Person, to be used by any other Person to train any AI Technologies. The Company and each of its applicable Subsidiaries has conducted impact assessments that evaluate ethical, bias, adequacy, privacy and security risks for all uses of AI Technologies to the extent required under applicable Law or Privacy Requirement. Neither the Company, any of its Subsidiaries nor any of its employees, consultants, independent contractors, or other service providers have used or use Third-Party AI Products to develop any Company Intellectual Property Rights that the Company or any of its Subsidiaries intended to maintain as proprietary in a manner that it believes would materially affect the Company or any its Subsidiaries' ownership or rights therein.

### 2.14 Privacy and Data Protection.

- (a) The Company, each of its Subsidiaries and to the Company's Knowledge, its vendors, consultants, agents suppliers and/or subcontractors who store, maintain, transmit or have access to Personal Information ("Company Service Providers") have at all times complied in all material respects with: (i) all applicable Privacy Laws; (ii) the Company's policies regarding privacy and data security, including all privacy policies and all public statements made by the Company about its privacy and/or data security practices; (iii) if applicable, any rules of self-regulatory organizations, codes of conduct, or other industry frameworks to which the Company or any of its Subsidiaries is bound, including the Payment Card Industry Data Security Standard and all other rules of the payment card brands; and (iv) all contractual obligations relating to the receipt, collection, compilation, use, storage, processing, sharing, safeguarding, security, disposal, destruction, disclosure, or transfer (including the transfer by or on behalf of the Company of Personal Information) (all of the foregoing, collectively, the "Privacy Requirements").
- (b) To the extent required by Privacy Requirements and except as otherwise disclosed in <u>Section 2.14(b)</u> of the Company Disclosure Schedule, the Company displays a privacy policy on each website owned, controlled or operated by the Company or any of its Subsidiaries, and each such privacy policy incorporates all material disclosures to data subjects required by the Privacy Requirements. None of the disclosures made or contained in any such privacy policy has been materially inaccurate, misleading or deceptive, or in violation of the Privacy Requirements, including containing any material omission.
- (c) Neither the Company nor any of its Subsidiaries has received any claim or complaint in writing regarding its or its service providers or agents' collection, use or disclosure of any data (including Personal Information) and/or failure to comply with any Privacy Requirements. Except as disclosed in <u>Section 2.14(b)</u> of the Company Disclosure Schedule, neither the Company, nor any of its Subsidiaries nor, to the Knowledge of the Company, any of the Company Service Providers collects, stores, uses, accesses, discloses or transfers Personal Information outside of the United States of America.
- (d) In connection with each third-party servicing, outsourcing, processing, or otherwise using Personal Information collected, held, or controlled by or on behalf of the Company or any of its Subsidiaries, the Company and its Subsidiaries has, since July 1, 2022, contractually obligated any such third party to (i) comply with applicable Privacy Laws with respect to Personal Information, (ii) act only in accordance with the instructions of the Company or such applicable Subsidiary, (iii) take appropriate steps to protect and secure Personal Information and from unauthorized disclosure, (iv) restrict use of Personal Information to those authorized or required under the servicing, outsourcing, processing, or similar arrangement, and (v) certify or guarantee the return or adequate disposal or destruction of Personal Information.

- (e) The Company has implemented commercially reasonable, though in no event less than industry-standard organizational, physical, administrative and technical measures required by Privacy Requirements to protect the integrity, security and operations of the IT Systems, transactions executed thereby, and data owned by the Company or any of its Subsidiaries, including protecting against loss and against damage, accidental loss or destruction, unauthorized or unlawful access, use, modification, disclosure or other misuse. The Company has implemented industry standard policies and procedures to detect and respond to data security breaches and unauthorized access or use of the IT Systems, Personal Information, and data owned or controlled by the Company or any of its Subsidiaries including by its employees and the Company Service Providers that have access to Personal Information. The Company provides its respective employees with regular privacy and data security training.
- (f) The Company has since July 1, 2022: (A) regularly conducted and regularly conducts vulnerability testing, risk assessments, and external audits of, and tracks security incidents related to the Company's and its Subsidiaries' systems and products (collectively, "Information Security Reviews"); (B) timely corrected any material exceptions or vulnerabilities identified in such Information Security Reviews; (C) made available complete and correct copies of all Information Security Reviews; and (D) timely installed software security patches and other fixes to identified technical information security vulnerabilities. The IT Systems are sufficient for the conduct of the Company Business. The Company has continuously operated in a manner to preserve and maintain the performance, security and integrity of the IT Systems. The Company maintains reasonable and appropriate back-up and disaster recovery procedures.
- (g) There have not been any actual or (alleged in writing to the Company) incidents of, or claims or Actions related to, data security breaches, ransomware incidents, other adverse events, unauthorized access or use of any of the IT Systems, or unauthorized acquisition, destruction, damage, disclosure, loss, corruption, alteration, or use of any Personal Information or data owned, held, controlled, licensed and/or otherwise processed by the Company or any of its Subsidiaries or provided by the Company or any of its Subsidiaries or Company Service Providers and, to the Company's Knowledge, there are no facts or circumstances which would reasonably serve as the basis for any such allegations or claims. Neither the Company nor any of its Subsidiaries has notified, nor been required to notify, any Person of any information security breach or incident. Neither the Company nor any of its Subsidiaries has received any correspondence relating to, or notice of any Actions, claims, investigations or alleged violations of, Laws with respect to Personal Information from any Person, there is no such ongoing Action, claim, investigation or allegation and, to the Company's Knowledge, there are no facts or circumstances which would form the basis for any such Action, claim, investigation or allegation.
- (h) To the Company's Knowledge, no investigation, inspection, audit or other proceeding of any nature involving allegations of any violation of Privacy Requirements is threatened in writing, pending, or contemplated by any Governmental Authority or other third party against the Company, any of its Subsidiaries or Company Service Providers with respect to such Company Service Providers' Processing on behalf of the Company or any of its Subsidiaries.
- (i) The consummation of any of the transactions contemplated by this Agreement will not violate any Privacy Requirements as they currently exist or as they existed at any time during which any of the Personal Information was collected or obtained. Neither the Company nor any of its Subsidiaries is subject to any Privacy Requirements that, following the Closing, would prohibit the Company or Parent from receiving or using Personal Information in

the manner in which the Company or any of its Subsidiaries receives and uses such Personal Information prior to the Closing.

(j) Neither the Company nor any of its Subsidiaries has shared, sold, rented or otherwise made available, and is not sharing, selling, renting or otherwise making available, to third parties any Personal Information for remuneration or other consideration.

## 2.15 <u>Compliance with Laws</u>.

- (a) The Company and each of its Subsidiaries are, and have at all times since July 1, 2022 been, in compliance in all material respects with all applicable Laws.
- (b) The Company and each of its Subsidiaries hold, and have at all times since July 1, 2022 held, all permits, licenses and approvals from, and have made all filings with, Governmental Authorities that are required to be held to conduct the Company Business in compliance with applicable Law and applicable Contracts ("Governmental Permits"), and all such Governmental Permits are valid and in full force and effect. Since July 1, 2022, neither the Company nor any of its Subsidiaries has received any written notice or other written communication, or to the Knowledge of the Company, any oral notice or other oral communication, from any Governmental Authority regarding (i) any actual or possible violation of applicable Law or any Governmental Permit or any failure to comply with any term or requirement of any Governmental Permit or (ii) any actual or possible revocation, withdrawal, suspension, cancellation, termination or modification of any Governmental Permit.
- (c) All materials, products and services distributed or marketed by the Company and each of its Subsidiaries have at all times since July 1, 2022 made all material disclosures to users or customers required by Law, and, since July 1, 2022 none of such disclosures made or contained in any such materials, products or services have been inaccurate, misleading or deceptive in any material respect.
- (d) Neither the Company nor any of its Subsidiaries are required to be registered as a broker-dealer with the SEC or any state authority and are not required to be a member of FINRA or any other self-regulatory organization. No Governmental Authority has alleged, to the Company's Knowledge, that such registration is required.
- (e) Neither the Company nor any of its Subsidiaries are, or are required to be, registered as an exchange with the SEC under Section 3 of the Exchange or Rule 3b-16 thereunder or to become an alternative trading system pursuant to Regulation ATS under the Exchange Act. No Governmental Authority has alleged, to the Company's Knowledge, that such registration is required.
- (f) Neither the Company nor any of its Subsidiaries has ever entered into or been subject to any judgment, consent decree, compliance order or administrative order with respect to any aspect of the business, affairs, properties or assets of the Company or any of its Subsidiaries or received civil investigative demand, subpoena, demand letter, administrative inquiry or formal or informal complaint (including a qui tam complaint), criminal information or claim from any Governmental Authority with respect to any aspect of the Company Business, affairs, properties or assets of the Company or any Subsidiary.
- (g) Neither the Company, any of its Subsidiaries, nor their respective control persons, directors, officers, agents or employees are subject to a "statutory disqualification" as defined in Section 3(a)(39) of the Exchange Act, nor would they be so subject if such persons

were "associated" with a broker-dealer, as that term is defined in Section 3(a)(18) of the Exchange Act.

- (h) The Company is an investment adviser duly filing with the SEC under the Investment Advisers Act of 1940 (the "Advisers Act") as an exempt reporting adviser and has adopted and implemented compliance policies and procedures to the extent required by applicable Laws, including applicable sections of the Advisers Act.
- (i) The Company and each of its Subsidiaries, as applicable, has, since July 1, 2022, filed all material registrations, reports, statements, notices and other filings required to be filed with the SEC, FINRA, and state securities author and any other Governmental Authority (the "Filings"). The Filings, when made, complied in all material respects with applicable Laws.
- (j) Except for routine examinations conducted by the SEC, FINRA or any other Governmental Authority in the Ordinary Course of Business of the Company, (i) no Governmental Authority has, to the Knowledge of the Company, initiated any administrative proceeding or investigation into the business or operations of the Company or any of its Subsidiaries and (ii) to the Knowledge of the Company, there are no unresolved material violations or material exceptions raised by any Governmental Authority with respect to the Company or any of its Subsidiaries.
- (k) The Company and each of its Subsidiaries has complied in all material respects with all applicable obligations, requirements and conditions of each investment advisory agreement to which the Company or any of its Subsidiaries are a party ("Advisory Agreement"). Each Advisory Agreement is a valid and binding agreement of the Company or its applicable Subsidiary and, to the Knowledge of the Company, each other party thereto. Each Advisory Agreement is in compliance in all material respects with the requirements of applicable Laws, including the Advisers Act to the extent applicable to such Advisory Agreement and, with respect to the Funds, the Investment Company Act to the extent applicable to such Funds. No Advisory Agreement provides for payment of compensation to the Company that is prohibited by Section 205(a)(1) of the Advisers Act. At no time has the Company or any of its Subsidiaries been in violation of any other applicable Laws with respect to the custody of client funds.
- (1) Neither the Company, any of its Subsidiaries, nor, to the Knowledge of the Company, any "affiliated persons" (as defined in Section 2(a)(3) of the Investment Company Act of 1940 (the "Investment Company Act") of the Company is (i) ineligible under Section 9(a) or 9(b) of the Investment Company Act to serve as an investment adviser to an investment company registered under the Investment Company Act or (ii) subject to any of the "Bad Actor" disqualifications described in Rule 506(d)(1) (i) to (viii) of Regulation D under the Securities Act. Neither the Company, any of its Subsidiaries, nor, to the Knowledge of the Company, any "person associated with an investment adviser" (as defined in the Advisers Act and applied to the Company) is ineligible pursuant to Section 203 of the Advisers Act to serve as an investment adviser or as a person associated with an investment adviser. There is no regulatory investigation, examination or other action or proceeding, to the Knowledge of the Company, pending or threatened, that would result in the Company or any of its Subsidiaries (or, to the Knowledge of the Company, any "affiliated person" of, or "associated person" of Company) becoming ineligible to serve in such positions or capacity or requiring disclosure to the clients of Company or any of its Subsidiaries.
- (m) Neither the Company, nor any of its Subsidiaries nor, to the Knowledge of the Company, any "covered associate" of the Company or its Subsidiaries, has made any

"contributions" or "payments" to any "government officials" that causes any compensation it accepts to be in violation of Rule 206(4)-5 under the Advisers Act.

- (n) None of the directors or officers of the Company or any of its Subsidiaries have been indicted for or convicted of any felony involving, or any crime involving, fraud, misrepresentation or insider trading. To the Knowledge of the Company, none of the directors or officers of the Company or any of its Subsidiaries are: (i) subject to any outstanding order barring, suspending or otherwise materially limiting the right of such officer or director to engage in any activity conducted as part of the Company's business as currently conducted, or (ii) a party to any consent agreement, memorandum of understanding, disciplinary agreement, commitment letter or similar undertaking to, or subject to any written order or written directive by or a recipient of any supervisory letter from, in each case, any Governmental Authority directly affecting the Company's or any of its Subsidiaries' businesses.
- (o) The information contained in the materials filed by the Company on Form ADV under the Advisers Act, was true and complete in all material respects at the time of filing (or if any portion thereof were amended, such portions were true and complete in all material respects as of the date of such amendment) and the Company has made all amendments to such forms as is required under the Advisers Act.
  - (p) The Company does not serve as an investment adviser to a registered investment company.
- (q) Except as set forth in <u>Section 2.15(q)</u> of the Company Disclosure Schedule, no exemptive orders, "no-action" letters or similar exemptions or regulatory relief have been obtained, nor are any requests pending therefor, by or with respect to (i) the Company, (ii) any Subsidiary of the Company, (iii) any Affiliate of the Company, (iv) any officer, manager, director, partner, equity holder, owner or employee, as applicable, of the Company in connection with the business of the Company or any of its Subsidiaries, (v) the Funds or (vi) any other Fund Investor.
- (r) Each of Megacorn Fund LP, MCF Spacex-I LLC, MCF Anduril-I LLC, MCF Epic Games LLC, MCF SpaceX LLC MCF Anduril LLC, MCF NL LLC and DoD Accelerator Fund L.P. (each a "Fund") was duly established, is validly existing and, to the extent such concept is relevant in the applicable jurisdiction, in good standing under the Laws of its jurisdiction of organization and has all requisite power and authority to conduct its business in the manner currently conducted. Neither the Company nor any of its Subsidiaries provides investment advisory services to any fund other than the Funds. All equity interests of each Fund issued and currently outstanding: (i) have been issued and sold by such Fund in compliance in all material respects with applicable Law; and (ii) have been duly authorized. Each Fund is operating, and since such Fund's inception has operated, in compliance with all applicable Laws in all material respects. The Company and each of its applicable Subsidiaries has performed investment advisory services with respect to each Fund in material compliance with the investment advisory agreement for such Fund and applicable Law, in each case in all material respects. No client to whom the Company or any of its Subsidiaries provides investment advisory services constitutes "plan assets" for purposes of Title I of ERISA or Section 4975 of the Internal Revenue Code.

# 2.16 <u>Employees, ERISA and Other Compliance</u>.

(a) <u>Section 2.16(a)(i)</u> of the Company Disclosure Schedule is a complete and correct list of all current employees of the Company or any of its Subsidiaries as of the

Agreement Date, and for each such employee, his or her: (1) job position or title; (2) base salary or hourly wage (as applicable); (3) annual commission opportunity or bonus target; (4) classification as full-time, part-time, temporary or seasonal; (5) classification as exempt or non-exempt under applicable state, federal or foreign overtime Laws and/or overtime regulations; (6) accrued but unused vacation and/or paid time off (PTO) balance; (7) visa type (if any); (8) commencement date of employment with the Company; (9) work location (e.g., U.S. city/state); (10) severance entitlements, if any; (11) if applicable, leave status (including anticipated return to work date); and (12) the total amount of bonus, severance, retention, change in control and/or other amounts to be paid to such employee at the Closing or otherwise in connection with the transactions contemplated hereby. Section 2.16(a)(ii) of the Company Disclosure Schedule is a complete and correct list of all independent contractors, consultants, advisors and leased workers of the Company or any of its Subsidiaries as of the Agreement Date and classified by the Company as other than employees, or compensated other than through Form W-2 wages "Contingent Workers," and for each such individual, his or her: (1) fee or compensation arrangements; (2) the specific entity that engages the individual; (3) commencement date with the Company or any Affiliate of the Company; (4) service location (e.g., U.S. state); (5) description of services provided; (6) notice required to terminate the relationship; and (7) confirmation as to whether a written independent contractor agreement exists.

- (b) The Company and its Subsidiaries have, at all times since July 1, 2022, in all material respects, correctly classified and paid their employees as exempt employees or non-exempt employees under the Fair Labor Standards Act and other applicable Laws. All employees of the Company or any of its Subsidiaries are, and have been since their respective start of employment by the Company or any of its Subsidiaries, legally permitted to be employed by the Company or any of its Subsidiaries in the jurisdiction in which such employee is employed. All independent contractors consultants and Contingent Workers providing services to the Company or any of its Subsidiaries are and have been, in all material respects, at all times since July 1, 2022, properly classified, treated and paid as independent contractors for purposes of federal and applicable state, local, and/or foreign Tax Laws, Laws applicable to employee benefits and other Laws. Neither the Company nor any of its Subsidiaries has any employment, contractor, consulting, or other similar Contracts currently in effect that are not terminable at will (other than agreements with the sole purpose of providing for the confidentiality of Proprietary Information or assignment of inventions).
- (c) The Company and each of its Subsidiaries have, at all times since July 1, 2022, (i) been in compliance in all material respects with all applicable Laws respecting employment, including without limitation, employment practices, terms and conditions of employment, discrimination, harassment, retaliation, employee health and safety, wages and hours, overtime pay, payroll documents, equal opportunity, immigration compliance, occupational health and safety, termination or discharge, plant closing and mass layoff requirements, reductions in force, affirmative action, workers' compensation, disability, unemployment compensation, whistleblower laws, collective bargaining, health care continuation requirements of COBRA, the requirements of the Family and Medical Leave Act of 1993, as amended, the requirements of the Health Insurance Portability and Accountability Act of 1996, as amended, and any similar provisions of state Law and (ii) withheld, paid and reported all amounts required by Law or by Contract to be withheld, paid and reported with respect to compensation, wages, salaries, commissions, severance payments, bonuses, and/or other payments to current and former employees, independent contractors, consultants, and Contingent Workers of the Company. Neither the Company nor any of its Subsidiaries is liable for any payment to any trust or other fund governed by or maintained by or on behalf of any Governmental Authority with respect to unemployment compensation benefits, social security or any other applicable social insurance, or other benefits or obligations for employees of the

Company or any of its Subsidiaries for failure to comply with any of the foregoing (other than routine payments to be made in the Ordinary Course of Business). There are no pending or, to the Knowledge of the Company, threatened Actions against the Company or any of its Subsidiaries under any worker's compensation policy or long-term disability policy.

- (d) Neither the Company nor any of its Subsidiaries is a party to, bound by, or currently negotiating any collective bargaining or similar agreement with any labor union, works council, labor association, or labor organization, nor are any organized groups of its employees represented by any labor union, works council, labor association, or labor organization. There is no pending, or to the Company's Knowledge, threatened, labor dispute, work slowdown, work stoppage, strike, lockouts, picketing, investigation by a Governmental Authority, involving the Company or any of its Subsidiaries. To the Knowledge of the Company, (i) no employee or group of employees or Contingent Workers of the Company or any of its Subsidiaries currently intends to terminate his or her employment or service arrangement with the Company or any of its Subsidiaries, and (ii) no employee of the Company or any of its Subsidiaries has an outstanding offer to join a business that may be competitive with the Company Business.
- (e) The Company and each of its Subsidiaries is in compliance in all material respects with the requirements of the Immigration Reform Control Act of 1986 and has a complete copy of U.S. Citizenship and Immigration Services Form I-9 for each of its U.S. employees.
- (f) Neither the Company nor any of its Subsidiaries is party to, and, has not been involved in any way in, any Action, or received written notice of any threatened Action, in which the Company or any of its Subsidiaries was, or is, alleged to have violated any Contract or Law relating to employment, including equal opportunity, discrimination, retaliation, harassment, immigration, wages, hours, unpaid compensation, classification of employees as exempt from overtime or minimum wage Laws, classification of workers as independent contractors, benefits, collective bargaining, the payment of social security and similar Taxes, occupational safety and health, and/or privacy rights of employees.
- (g) There is no pending or, to the Knowledge of the Company, threatened, and neither the Company nor any of its Subsidiaries has received notice of any, investigation or audit by a Governmental Authority responsible for the enforcement of labor, immigration or employment regulations and neither the Company nor any of its Subsidiaries has been found by any Governmental Authority to have engaged in any unfair labor practice, as defined in the National Labor Relations Act (29 U.S.C. § 151 et seq.) or other applicable Laws.
- (h) There has been no "mass layoff," "employment loss," or "plant closing" or similar group employment loss as defined by the WARN Act or any other and/or similar state, local, or foreign Law in respect of or affecting any site of employment of the Company or any of its Subsidiaries and neither the Company nor any of its Subsidiaries has been affected by any transaction or engaged in any lay-offs or employment terminations sufficient in number to trigger application of any such Law. During the ninety (90) day period preceding the date hereof, no employee Contingent Worker has suffered an "employment loss" as defined in the WARN Act with respect to the Company or any of its Subsidiaries.
- (i) Neither the Company nor any of its Subsidiaries is subject to any affirmative action obligations under any Law, including, without limitation, Executive Order 11246, nor is the Company or any of its Subsidiaries a government contractor or subcontractor for purposes of any Law with respect to the terms and conditions of employment, including the Service Contracts Act or prevailing wage Laws.

- (j) To the Company's Knowledge, no employee, independent contractor, consultant or other Contingent Worker of the Company or any of its Subsidiaries is in violation of (i) any material term of any employment or independent contractor Contract or (ii) any term of any other Contract or any restrictive covenant relating to the right of any such employee, independent contractor, consultant or other Contingent Worker to be employed by or to render services to the Company or any of its Subsidiaries or to use Proprietary Information of others. To the Company's Knowledge, the employment of any employee or engagement of any independent contractor, consultant or other Contingent Worker by the Company or any of its Subsidiaries does not subject them to Liability to any third party.
- (k) Neither the Company nor any of its Subsidiaries has been a party to or involved in a settlement agreement with a current or former employee or independent contractor that relates to claims or allegations of sexual harassment or sexual misconduct, or other unlawful harassment, discrimination or retaliation. No claims or allegations of sexual harassment or sexual misconduct, or other unlawful harassment, discrimination or retaliation have been made to, or to the Knowledge of the Company, against or involving the Company or any of its Subsidiaries with respect to any current or former officer, director, manager, employee, independent contractor of the Company or any of its Subsidiaries in his or her capacity as such, and no investigations of sexual harassment or sexual misconduct, or other unlawful harassment, discrimination, or retaliation have been conducted by or on behalf of or involved the Company or any of its Subsidiaries with respect to any current or former officer, director, manager, employee, independent contractor of the Company. To the Company's Knowledge, there are no facts that would reasonably be expected to give rise to a claim of sexual harassment or misconduct, other unlawful harassment or unlawful discrimination or retaliation against or involving, the Company or any of its Subsidiaries or any officer, director, manager, employee, or Contingent Worker thereof.
- (l) <u>Section 2.16(l)</u> of the Company's Disclosure Schedule identifies each employee of the Company who is subject to a non-competition, non-solicitation and confidentiality agreement with the Company. The Company has made available a form of each such agreement.
- (m) <u>Section 2.16(m)</u> of the Company Disclosure Schedule sets forth a complete and correct list of each material Company Benefit Arrangement. All Company Benefit Arrangements, other than employment agreements or bonuses or equity compensation, are and have been provided through a professional employer organization arrangement. The Company shall use commercially reasonable efforts to obtain such documents regarding the Company Benefit Arrangement from the professional employer organization as Parent may reasonably request.
- (n) Complete and correct copies of the following documents, with respect to each material Company Benefit Arrangement, where applicable, have been made available to Parent: (i) all documents embodying or other governing such Company Benefit Arrangement (or for unwritten Company Benefit Arrangement a written description of the material terms of such arrangement) and any current funding medium for the Company Benefit Arrangement; (ii) the most recent IRS determination, opinion, or advisory letter; (iii) the most recently filed IRS Form 5500; (iv) the most recent actuarial valuation report; (v) the most recent summary plan description (or other descriptions provided to employees) and all modifications thereto; (vi) non-discrimination testing results for the last three (3) plan years and (vii) all non-routine correspondence to and from any state or federal agency related to such Company Benefit Arrangement.

- (o) Each Company Benefit Arrangement that is intended to qualify under Section 401(a) of the Code is so qualified and has received a favorable determination or approval letter from the IRS with respect to such qualification, or may rely on an opinion letter issued by the IRS with respect to a pre-approved plan adopted in accordance with the requirements for such reliance, or has time remaining for application to the IRS for a determination of the qualified status of such Company Benefit Arrangement for any period for which such Company Benefit Arrangement would not otherwise be covered by an IRS determination and, to the Knowledge of the Company, no event or omission has occurred that would cause any Company Benefit Arrangement to lose such qualification.
- (p) Each Company Benefit Arrangement is and has been operated in material compliance with applicable Laws and regulations and with its terms including without limitation ERISA, and the Code. No Company Benefit Arrangement is, or within the past six (6) years has been, the subject of an application or filing under a government sponsored amnesty, voluntary compliance, or similar program, or been the subject of any self-correction under any such program. No litigation or governmental Action (other than those relating to routine claims for benefits) is pending or, to the Knowledge of the Company, threatened with respect to any Company Benefit Arrangement, and, to the Knowledge of the Company, there is no reasonable basis for any such Action. All payments and/or contributions required to have been timely made with respect to all Company Benefit Arrangements either have been made or have been accrued in accordance with the terms of the applicable Company Benefit Arrangement and applicable Law. Neither the Company nor any current or former employee or manager thereof, nor, to the Knowledge of the Company, any other Person, has been engaged in any non-exempt prohibited transaction (as defined in Section 406 of ERISA or Section 4975 of the Code) with respect to any Company Benefit Arrangement. The Company has not at any time attempted to maintain, for any Company Benefit Arrangement, a "grandfathered" status under the Affordable Care Act.
- (q) Neither the Company nor any of its ERISA Affiliates has in the past six years maintained, contributed to, or been required to contribute or had any Liability with respect to, including on account of any ERISA Affiliate (whether contingent or otherwise), to (i) any employee benefit plan that is or was subject to Title IV of ERISA, Section 412 of the Code, Section 302 of ERISA, (ii) a Multiemployer Plan, (iii) any funded welfare benefit plan within the meaning of Section 419 of the Code, (iv) any "multiple employer plan" (within the meaning of Section 210 of ERISA or Section 413(c) of the Code), or (v) any "multiple employer welfare arrangement" (as such term is defined in Section 3(40) of ERISA), excluding, with respect to clauses (iv) and (v), any such arrangements offered through a professional employer organization, and neither the Company nor any of its ERISA Affiliates has ever incurred any Liability for a "prohibited transaction" under Title IV of ERISA that has not been paid in full.
- (r) Neither the Company nor any ERISA Affiliate provides or has any obligation to provide health care or any other non-pension benefits to any employees after their employment is terminated and the Company has no outstanding but unfulfilled promises to provide such post-termination benefits (other than as required by Part 6 of Subtitle B of Title I of ERISA or similar state law or for a limited period of time following a termination of employment pursuant to the terms of an existing employment, severance or similar agreement in effect as of the date hereof).
- (s) Each Company Benefit Arrangement may be amended, terminated or otherwise modified (including cessation of participation) by the Company or any of its Subsidiaries to the greatest extent permitted by applicable Law, and no employee communications or provision of any Company Benefit Arrangement has failed to effectively

reserve the right of the Company or Subsidiary to so amend, terminate or otherwise modify such Company Benefit Arrangement.

- (t) No Company Benefit Arrangement is subject to the laws of any jurisdiction outside the United States.
- (u) Neither the execution and delivery of this Agreement, the member approval of this Agreement, nor the consummation of the transactions contemplated hereby would (either alone or in conjunction with any other event) (i) result in, or cause the accelerated vesting payment, any payment or benefit to any employee, officer, director or other service provider of the Company; or (ii) further restrict any rights of the Company or any of its Subsidiaries to amend or terminate any Company Benefit Arrangement.

#### 2.17 Books and Records; Bank Accounts.

- (a) The Company has made available to Parent complete and correct copies of the minute books, unit transfer ledgers, and other similar limited liability company records, each as amended to date, of the Company and its Subsidiaries. The Company and each of its Subsidiaries has maintained Books and Records that are complete and correct in all material respects and accurately and fairly reflect, in all material respects, the Company Business. At the Closing, the minute books and other Books and Records of the Company its Subsidiaries will be in the possession of the Company or its Subsidiaries. Section 2.17(a) of the Company Disclosure Schedule sets forth a list as of the Agreement Date of all of the current officers, managers, and directors of the Company and its Subsidiaries (including, for each such individual, the position held and period of service).
- (b) <u>Section 2.17(b)</u> of the Company Disclosure Schedule sets forth the names and locations of all banks, trust companies, savings and loan associations and other financial institutions at which the Company or any of its Subsidiaries maintains accounts of any nature and the names of all Persons authorized to draw thereon or make withdrawals therefrom.
- 2.18 <u>Insurance</u>. The Company and its Subsidiaries maintain the policies of insurance and bonds set forth in <u>Section 2.18</u> of the Company Disclosure Schedule (the "**Insurance Policies**"), which include all legally required workers' compensation and other insurance coverage. Complete and correct copies of the Insurance Policies have been made available to Parent. <u>Section 2.18</u> of the Company Disclosure Schedule sets forth (a) the name of the insurer under each such Insurance Policy, the type of Insurance Policy, policy number and the term and amount of coverage thereunder, and (b) all claims made under such Insurance Policy. There is no claim pending under any of such Insurance Policy as to which coverage has been questioned, denied or disputed by the underwriters of such policy or bond or for which its total value (inclusive of defense expenses) would reasonably be expected to exceed the applicable policy limits. All premiums due and payable under all such Insurance Policies and bonds have been timely paid, and the Company and its Subsidiaries are otherwise in compliance in all material respects with the terms of such policies and bonds. To the Company's Knowledge, no insurance provider has threatened to terminate or increase the premium with respect to, any of such Insurance Policies. Except as set forth in <u>Section 2.18</u> of the Company Disclosure Schedule, all Insurance Policies are occurrence-based.
- 2.19 <u>Environmental Matters</u>. The Company and each of its Subsidiaries is and at all times has been in compliance in all material respects with all Environmental Laws which compliance includes the possession of all Governmental Permits required under Environmental Laws and compliance with the terms and conditions thereof. Neither the Company nor any of its

Subsidiaries has received any written notice or, to the Knowledge of the Company, other oral communication, whether from a Governmental Authority, citizens groups, employee or otherwise, that alleges that the Company or any of its Subsidiaries is not in compliance in any material respects with any Environmental Law. All Governmental Permits held by the Company or any of its Subsidiaries pursuant to Environmental Law (if any) are identified in <u>Section 2.19</u> of the Company Disclosure Schedule.

## 2.20 <u>Customers and Suppliers</u>.

- (a) Section 2.20(a)(i) of the Company Disclosure Schedule sets forth a list of all limited partners or other equityholders in each Fund as of June 1, 2025 (each, a "Fund Investor"). Section 2.20(a)(ii) of the Company Disclosure Schedule sets forth the top twenty five (25) limited partners or other equityholders in the Funds (or group of affiliated limited partners or other investors) based on assets under management (for all Funds in the aggregate) as of June 1, 2025 (each a "Significant Fund Investor"). As of the Agreement Date, none of the Significant Fund Investors has terminated or cancelled its relationship with the Company. Since January 1, 2024 neither the Company nor any of its Subsidiaries has had any material disputes with any Significant Fund Investor. The Company has no Knowledge of any material dissatisfaction on the part of any Significant Fund Investor or any facts or circumstances that would reasonably be expected to lead to such material dissatisfaction. Since January 1, 2024, neither the Company nor any of its Subsidiaries has received any written or, to the Knowledge of the Company, oral notice from any Significant Fund Investor that such Significant Fund Investor will not continue as a Fund Investor of the Company or any of its Subsidiaries or, following the Effective Time, Parent or any of its Affiliates, or that such Significant Fund Investor intends to terminate, breach or request a material modification to existing Contracts with the Company or any of its Subsidiaries or any Fund.
- (b) Section 2.20(b) of the Company Disclosure Schedule sets forth the top five (5) vendors and suppliers of products and services (excluding, for the avoidance of doubt, independent contractors) to the Company or any of its Subsidiaries based on amounts paid or payable by the Company or any of its Subsidiaries to such vendors and suppliers during each of (i) the twelve-month period ending on December 31, 2024 and (ii) the year-to-date period ending on the last day of the calendar month immediately preceding the Agreement Date (each, a "Significant Supplier"). The Company and its Subsidiaries are current in their payments consistent with the payment schedule for such Significant Supplier established and agreed in the Ordinary Course of Business to all Significant Suppliers and neither the Company nor any of its Subsidiaries have, and since January 1, 2024 have not had, any material dispute concerning Contracts with or products and/or services provided by any Significant Supplier that arose or remain unresolved. The Company has no Knowledge of any material dissatisfaction on the part of any Significant Supplier or any facts or circumstances that would reasonably be expected to lead to such material dissatisfaction. Since January 1, 2024, neither the Company nor any of its Subsidiaries have received any written or, to the Knowledge of the Company, oral notice from any Significant Supplier that such Significant Supplier shall not continue as a supplier to the Company or, following the Effective Time, Parent or any of its Affiliates, or that such Significant Supplier intends to terminate, breach or not renew existing Contracts with the Company or its Subsidiaries.
- (c) <u>Section 2.20(c)</u> of the Company Disclosure Schedule sets forth the aggregate dollar amount of assets under management by the Company and its Subsidiaries as of June 1, 2025.

2.21 <u>Anti-Money Laundering Laws</u>. The Company and each of its Subsidiaries is and has always been, in compliance in all material respects with the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, the United Kingdom Proceeds of Crime Act 2002 and all other applicable anti-money laundering and counter terrorist financing Laws, and has established policies, procedures and internal controls designed to ensure compliance with such Laws.

# 2.22 <u>Anti-Corruption and Anti-Bribery Laws</u>.

- (a) Neither the Company nor any of its Subsidiaries, nor, to the Company's Knowledge, any of their respective directors, managers, officers, agents, employees, independent contractors or other representatives, or any other Person associated with or acting for or on behalf of the Company or any of its Subsidiaries, has, directly or indirectly, in connection with the conduct of any activity of the Company or any of its Subsidiaries in violation of the FCPA, the Bribery Act or any other applicable anti-corruption or anti-bribery Law:
  - (i) made, offered or promised to make or offer any payment, loan or transfer of anything of value, including any reward, advantage or benefit of any kind, to or for the benefit of any Foreign Government Official, candidate for public office, political party or political campaign, or any official of such party or campaign, for the purpose of (A) influencing any act or decision of such Foreign Government Official, candidate, party or campaign or any official of such party or campaign to do or omit to do any act in violation of a lawful duty, (C) obtaining or retaining business for or with any Person, (D) expediting or securing the performance of official acts of a routine nature, or (E) otherwise securing any improper advantage;
  - (ii) paid, offered or agreed or promised to make or offer any bribe, payoff, influence payment, kickback, unlawful rebate or other similar unlawful payment of any nature;
  - (iii) made, offered or agreed or promised to make or offer any unlawful contributions, gifts, entertainment or other unlawful expenditures;
    - (iv) established or maintained any unlawful fund of corporate monies or other properties;
  - (v) created or caused the creation of any false or inaccurate books and records related to any of the foregoing; or
  - (vi) otherwise violated any provision of the Foreign Corrupt Practices Act of 1977, as amended, 15 U.S.C. §§ 78dd-1, et seq. ("FCPA"), the United Kingdom Bribery Act 2010 (the "Bribery Act") or any other applicable anti-corruption or anti-bribery Law.
- (b) Neither the Company nor any of its Subsidiaries has undergone and, to the Knowledge of the Company, is not undergoing any audit, review, inspection, investigation, survey or examination by a Governmental Authority relating to the compliance of the Company with the FCPA, the Bribery Act, or anti-corruption, or anti-kickback Laws. To the Knowledge of the Company, there are no threatened Actions with respect to the compliance of the Company or any of its Subsidiaries with the FCPA or the Bribery Act, or any anti-corruption or anti-kickback Laws by the Company or any of its Subsidiaries.

## 2.23 <u>Trade Compliance</u>.

- (a) The Company and each of its Subsidiaries has at all times conducted its export, import and related transactions in accordance in all material respects with (i) all applicable U.S. export, re-export, import, anti-boycott, and economic sanctions Laws and regulations, including the Export Administration Regulations, the Arms Export Control Act, the International Traffic in Arms Regulations, the trade and economic sanctions regulations administered by the U.S. Department of Treasury's Office of Foreign Assets Control, and (ii) all other applicable import and export control Laws and regulations in the other countries in which the Company conducts business ("Export Control Laws").
- (b) The Company and each of its applicable Subsidiaries has obtained, and is in compliance in all material respects with, all material export licenses and other required consents, authorizations, waivers, approvals, and orders, and have made or filed any and all necessary notices, registrations, declarations and filings with any Governmental Authority, and has met the requirements of any license exceptions or exemptions, as required in connection with (i) the export and re-export of products, services, Software or other Technology, and (ii) releases of technical data, Software or other Technology to foreign nationals located in the United States and abroad ("Export Approvals").
- (c) The Company and each of its applicable Subsidiaries is in compliance in all material respects with the terms of all applicable Export Approvals.
- (d) There are no pending or, to the Knowledge of the Company, threatened inquiries, investigations, enforcement actions, voluntary disclosures or other claims against the Company or any of its Subsidiaries with respect to Export Control Laws or Export Approvals.
- 2.24 <u>Outbound Investment</u>. Neither the Company nor any of its Subsidiaries is (i) a "person of a country of concern"; or (ii) engaged in any "covered activity," as these terms are defined in 31 C.F.R. Part 850, as implemented or revised from time to time (the "**Outbound Investment Rules**"). Neither the Company nor any of its Subsidiaries has the intention of becoming a "person of a country of concern" that engages in any "covered activity." Neither the Company nor any of its Subsidiaries is a person that directly or indirectly holds a board seat on, a voting or equity interest in, or any contractual power to direct or cause the direction of the management of policies of any "covered foreign person" (as defined in the Outbound Investment Rules).
- 2.25 <u>Brokers; Transaction Expenses</u>. No investment banker, broker, finder or similar party is or shall be entitled to any payment of any fees or expenses in connection with the origin, negotiation or execution of this Agreement or in connection with the Mergers or any other transaction contemplated by this Agreement based upon arrangements made by or on behalf of the Company, any of its Subsidiaries or any of their respective Affiliates at or prior to the Effective Time.
- 2.26 <u>Social Media</u>. <u>Section 2.26</u> of the Company Disclosure Schedule sets forth a true, correct and complete list of all social media accounts that the Company or any of its Subsidiaries uses, operates or maintains, including in connection with marketing or promoting any Company products or services ("**Social Media Accounts**"). <u>Section 2.26</u> of the Company Disclosure Schedule also lists for each such Social Media Account, any account name(s), user name(s), nickname(s), display name(s), and other identifiers registered or used by or for the Company or any of its Subsidiaries with respect to such Social Media Account (collectively, "**Social Media Account Names**"). The Company or one of its Subsidiaries owns each Social

Media Account. All use of the Social Media Accounts complies with, and since July 1, 2022, has complied with, in all material respects (a) all terms and conditions, terms of use, terms of service and other Contracts applicable to such Social Media Accounts and (b) applicable Law.

## 2.27 <u>Capitalization of the Aggregator</u>.

- (a) <u>Authorized and Outstanding Capitalization of the Aggregator</u>. The issued and outstanding equity securities of the Aggregator consist solely of 111,604 Aggregator Units.
- (b) Aggregator Units. Section 2.27 of the Company Disclosure Schedule sets forth for each holder of Aggregator Units, (i) the number of issued and outstanding Aggregator Units held by each holder of Aggregator Units as of the Agreement Date and (ii) the date of issuance of the Aggregator Units held by such holder. No Aggregator Units are issued and outstanding as of the Agreement Date that are not set forth on Section 2.27 of the Company Disclosure Schedule. Except as set forth on Section 2.27 of the Company Disclosure Schedule, there is no outstanding (x) equity appreciation right, option, restricted equity, restricted equity unit, "phantom" equity or any similar security or right to which the Aggregator is a party that is derivative or provides any economic benefit based, directly or indirectly, on the value or price of any equity security of the Aggregator or (y) warrant, call, right, commitment, conversion privilege or preemptive or other right or Contract to which the Aggregator s is a party to purchase or otherwise acquire any Aggregator Units or any security or debt convertible into or exchangeable for Aggregator Units or obligating the Aggregator to grant, extend or enter into any such equity appreciation right, option, restricted equity, restricted equity unit, "phantom" equity, warrant, call, right, commitment, conversion privilege or preemptive or other right or Contract.
- 2.28 No Other Representations and Warranties; Non-Reliance. The Company and each of its Subsidiaries acknowledges and agrees that except, in each case, for the representations and warranties contained in Article 3 and any certificate delivered hereunder (in each case as qualified by the Parent Disclosure Schedule), and except in the case of Fraud by or on behalf of Parent or the Merger Subs, (a) none of Parent, either Merger Sub or any other Person, makes or has made any representation or warranty, express or implied, either written or oral, at law or in equity, in respect of Parent, any of its Subsidiaries, their respective business or operations or any other matter in connection with this Agreement and the transactions contemplated hereby or otherwise, including as to any information furnished or made available to Company or any of its Representatives or any other Person in connection with this Agreement, the transactions contemplated hereby, or as to the future revenue, profitability or success of Parent or any of its Subsidiaries, or any representation or warranty arising from statute or otherwise under Law, (b) neither Parent or either Merger Sub or any other Person shall have any Liability in connection with any such information provided to the Company or any of its Representatives or any other Person has relied on any information, materials, representations or warranties or other statements or omissions (including as to the accuracy or completeness thereof), whether express or implied, that may have been made, provided or made available by Parent, either Merger Sub, any of their respective Representatives or any other Person with respect to Parent or any of its Subsidiaries or their respective business or operations or otherwise in connection with or relating to the entry into this Agreement and the transactions contemplated hereby.

#### **ARTICLE 3**

#### REPRESENTATIONS AND WARRANTIES OF PARENT AND THE MERGER SUBS

Subject to Section 9.11(b), except as set forth in the Parent Disclosure Schedule or in the Parent SEC Reports (but (i) without giving effect to any amendment thereof filed with, or furnished to the SEC on or after the date hereof and (ii) other than any forward-looking disclosures contained or referenced therein under the captions "Risk Factors" or "Cautionary Note Regarding Forward-Looking Statements," and any other disclosures contained or referenced therein of information, factors, or risks that are predictive, cautionary, or forward-looking in nature), Parent and the Merger Subs, jointly and severally, represent and warrant to the Company as follows:

3.1 Organization and Good Standing. Parent is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware and has all requisite corporate power and authority to own, operate and lease its properties and to carry on its business as now conducted. Merger Sub I is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware. Merger Sub II is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware. Each of Parent and the Merger Subs is duly qualified or licensed to do business, and is in good standing, in each jurisdiction where the character or location of the assets or properties owned, leased or operated by it or the nature of its activities makes such qualification or licensing necessary, except where the failure to be so qualified or licensed would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Parent. The Charter Documents of Parent its Subsidiaries are in full force and effect with respect to Parent or any of its Subsidiaries are or have been in violation of the provisions of any current or past Charter Documents in any material respect.

### 3.2 Power, Authorization and Validity.

Power and Authority. Parent has all requisite corporate power and authority to execute and deliver this Agreement and each of the Parent Ancillary Agreements and to perform its obligations under this Agreement and each of the Parent Ancillary Agreements and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance by Parent of this Agreement and each of the Parent Ancillary Agreements and the consummation by Parent and the Merger Subs of the transactions contemplated hereby or thereby have been duly and validly approved and authorized by all necessary corporate action on the part of Parent. Merger Sub I has all requisite limited liability company power and authority to execute and deliver this Agreement and each of the Merger Sub Ancillary Agreements to which it is to be party and to perform its obligations under this Agreement and each of the Merger Sub Ancillary Agreements to which it is to be party and to consummate the transactions contemplated hereby and thereby. Merger Sub II has all requisite limited liability company power and authority to execute and deliver this Agreement and each of the Merger Sub Ancillary Agreements to which it is to be party and to perform its obligations under this Agreement and each of the Merger Sub Ancillary Agreements to which it is to be party and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance by Merger Sub I and Merger Sub II of this Agreement and each of the Merger Sub Ancillary Agreements to which it is to be party and the consummation of the transactions contemplated hereby or thereby have been duly and validly approved and authorized by all necessary limited liability company action on the part of Merger Sub I and Merger Sub II. No vote of the holders of any class or series of the Parent's capital stock or other securities is necessary for the consummation by the Parent of the transactions contemplated by this Agreement.

II. This Agreement and each of the Parent Ancillary Agreements are, or when executed by Parent shall be, assuming the due authorization, execution and delivery by the Company and the other Persons party hereto or thereto, valid and binding obligations of Parent, enforceable against Parent in accordance with their respective terms, subject to the Enforceability Exceptions. This Agreement and each of the applicable Merger Sub Ancillary Agreements to be entered into by Merger Sub I or Merger Sub II are, or when executed by Merger Sub I or Merger Sub II shall be, assuming the due authorization, execution and delivery by the Company, valid and binding obligations of Merger Sub I or Merger Sub II, enforceable against Merger Sub I or Merger Sub II in accordance with their respective terms, subject to the Enforceability Exceptions.

#### 3.3 No Conflicts; Required Consents.

- (a) The execution, delivery and performance by Parent and each Merger Sub of this Agreement and each of the Parent Ancillary Agreements and Merger Sub Ancillary Agreements, as applicable, and the consummation by Parent and each Member Sub of the transactions contemplated hereby and thereby, do not and will not: (i) conflict with or violate the Charter Documents of Parent or either Merger Sub, (ii) subject to compliance with Section 3.3(b), conflict with or violate any Laws or any judgment, decree or order to which Parent or either Merger Sub is subject or (iii) result (or would result, with or without notice or lapse of time, or both) in (A) any breach of or default under, (B) any Person's right to consent, notice, or right of termination, acceleration, cancellation, modification or amendment of, or right to any increased, additional, accelerated or guaranteed payment or performance under, or (C) any Encumbrance (other than a Permitted Encumbrance) on any properties or assets of Parent or any of its Subsidiaries pursuant to, or (D) any Parent Material Contract, except, in the case of clauses (ii) and (iii), as would not reasonably be expected to be material to Parent and its Subsidiaries, taken as a whole.
- (b) No consent, approval, order or authorization of, or registration, declaration or filing with, or notice to, any Governmental Authority is necessary or required to be made or obtained by Parent or either Merger Sub to enable Parent or either Merger Sub to lawfully execute, deliver and perform this Agreement, each of the Parent Ancillary Agreements (as to Parent) and the Merger Sub Ancillary Agreements to be entered into by the applicable Merger Sub (as to the Merger Subs) or to consummate the transactions contemplated hereby or thereby, except for (i) such consents, approvals, orders, authorizations, registrations, declarations and filings, if any, that if not made or obtained by Parent or either Merger Sub would not reasonably be expected to be material to Parent and its Subsidiaries, taken as a whole, (ii) the filing of the First Certificate of Merger or the Second Certificate of Merger with the Secretary of State of the State of Delaware, (iii) any filings required under applicable securities Laws or state "blue sky" Laws, and (iv) any filings required to comply with the rules and regulations of the New York Stock Exchange.

# 3.4 <u>Capitalization</u>.

(a) The authorized capital stock of Parent consists of: 133,333,333 shares of Parent Common Stock and 100,000,000 shares of preferred stock, \$0.0001 par value per share (the "Parent Preferred Stock"). As of June 30, 2025 (the "Parent Capitalization Date"), 12,421,476 shares of Parent Common Stock (inclusive of 10,479 shares of Parent Common Stock held in treasury) were issued and outstanding and no shares of Parent Preferred Stock were issued or outstanding.

- (b) Section 3.4(b) of the Parent Disclosure Schedule sets forth for each Parent Stock Plan and the Parent ESPP, as of the Parent Capitalization Date, (i) the number of shares of Parent Common Stock issued thereunder, (ii) the number of shares of Parent Common Stock reserved for future issuance thereunder, (iii) the aggregate number of shares of Parent Common Stock subject to issuance pursuant to outstanding Parent Stock Options under the Parent Stock Plan, (iv) the aggregate number of shares of Parent Common Stock subject to issuance pursuant to outstanding Parent RSUs under the Parent Stock Plan, and (v) the aggregate number of shares of Parent Common Stock subject to issuance pursuant to outstanding Parent PSUs under the Parent Stock Plan assuming performance at maximum.
- (c) Except (i) as set forth in this <u>Section 3.4</u> or <u>Section 3.4(b)</u> of the Parent Disclosure Schedule and for changes since the Parent Capitalization Date resulting from the exercise or settlement of Parent Warrants, Parent Stock Options or Parent RSUs or Parent PSUs outstanding on such date, the issuance of shares pursuant to the Parent ESPP and (ii) as reserved for future grants under Parent Stock Plans as of the date of this Agreement, (A) there are no equity securities of any class of the Parent, or any security exchangeable into or exercisable for such equity securities, issued, reserved for issuance or outstanding and (B) there are no options, warrants, equity securities, calls, rights or agreements to which the Parent or any of its Subsidiaries is a party or by which the Parent or any of its Subsidiaries is bound obligating the Parent or any of its Subsidiaries to issue, exchange, transfer, deliver or sell, or cause to be issued, exchanged, transferred, delivered or sold, additional shares of capital stock or other equity interests of the Parent or any security or rights convertible into or exchangeable or exercisable for any such shares or other equity interests.
- (d) All outstanding shares of Parent Common Stock are, and all shares of Parent Common Stock subject to issuance as specified in Section 3.4(b) or Section 3.4(b) of the Parent Disclosure Schedule, upon issuance on the terms and conditions specified in the instruments pursuant to which they are issuable, will be, duly authorized, validly issued, fully paid and nonassessable and not issued in violation of any preemptive rights, purchase options, rights of first refusal or similar rights under any provision of the Delaware General Corporation Law, Parents Charter Documents or any agreement to which Parent is a party or is otherwise bound.
- (e) There are no bonds, debentures, notes or other indebtedness of the Parent having the right to vote (or, other than the Senior Convertible Notes, that are convertible into, or exchangeable for, securities having the right to vote) on any matters on which holders of shares of Parent Common Stock may vote.
- 3.5 <u>Parent Shares.</u> When issued in accordance with the terms of this Agreement for the consideration expressed herein, and assuming the accuracy of the Investor Questionnaires, the Parent Shares will be duly authorized and validly issued, fully paid and nonassessable, and will be free of restrictions on transfer other than restrictions set forth herein or under any Charter Document of Parent or under the Securities Act and any other applicable Law. The issuance of the Parent Shares is not subject to any preemptive rights or rights of first refusal or similar rights applicable to Parent.
- 3.6 <u>Merger Subs</u>. Each Merger Sub was formed solely for the purpose of engaging in the transactions contemplated by this Agreement, has engaged in no other business activities and has conducted its operations only as contemplated by this Agreement.
  - 3.7 SEC Reports; Internal Control.

- Parent has timely filed or furnished all reports and other documents with the SEC required to be filed or (a) furnished by Parent since January 1, 2023 (such documents, together with any current reports filed during such period by Parent with the SEC on a voluntary basis on Form 8-K, the "Parent SEC Reports"). As of their respective filing dates, or, if amended prior to the date hereof, as of the date of (and giving effect to) the last such amendment (and in the case of registration statements and proxy statements, on the date of effectiveness and the dates of the relevant meetings, respectively), the Parent SEC Reports (a) complied in all material respects with, to the extent in effect at the time of filing, the applicable requirements of the Securities Act and the Exchange Act, as the case may be, and (b) except to the extent that information contained in such Parent SEC Reports has been revised, amended, modified or superseded (prior to the date of this Agreement) by a later filed Parent SEC Report, did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading; provided, however, that no representation is made as to the accuracy of any financial projections or forward-looking statements or the completeness of any information filed or furnished by Parent to the SEC solely for the purposes of complying with Regulation FD promulgated under the Exchange Act. Each of the financial statements (including the related notes, where applicable) of Parent included in the Parent SEC Reports was prepared in accordance with U.S. GAAP applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto, and except, in the case of unaudited statements, as permitted by the rules and regulations of the SEC) and fairly presented in all material respects the consolidated financial position of Parent and its consolidated Subsidiaries as of the respective dates thereof and the consolidated results of their operations and cash flows for the respective periods then ended, except as otherwise noted therein (subject, in the case of unaudited statements, to normal year-end adjustments and to any other adjustments described therein, including the notes thereto).
- (b) Parent is in compliance in all material respects with: (i) the applicable rules and regulations of the New York Stock Exchange and (ii) the applicable continued listing requirements of the New York Stock Exchange.
- (c) Parent is, and since January 1, 2023 has been, in compliance in all material respects with the applicable provisions of the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act") and the related rules and regulations promulgated thereunder. Parent has established and maintains disclosure controls and procedures and internal control over financial reporting (as such terms are defined in paragraphs (e) and (f), respectively, of Rule 13a-15 under the Exchange Act) as required by Rule 13a-15 under the Exchange Act. Parent's disclosure controls and procedures are reasonably designed to ensure that all material information required to be disclosed by Parent in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC, and that all such material information is accumulated and communicated to the management of Parent as appropriate to allow timely decisions regarding required disclosure and to make the certifications required pursuant to Sections 302 and 906 of the Sarbanes-Oxley Act. The management of Parent completed its assessment of the effectiveness of the Company's internal control over financial reporting in compliance with the requirements of Section 404 of the Sarbanes-Oxley Act for the fiscal year ended December 31, 2024, and such assessment concluded that as of December 31, 2024 such controls were effective. Parent's principal executive officer, principal accounting officer and its principal financial officer have disclosed, based on their most recent evaluation, to Parent's auditors and the audit committee of the board of directors of Parent (x) all significant deficiencies, if any, in the design or operation of internal control over financial reporting which are reasonably likely to materially adversely affect Parent's ability to record, process, summarize and report financial data and have identified to

such auditors any material weaknesses in internal controls and (y) to the knowledge of Parent, any fraud, whether or not material, that involves management or other employees of Parent or any of the Subsidiaries who have a significant role in Parent's internal control over financial reporting.

- (d) Since December 31, 2024, there has not been a Material Adverse Effect with respect to Parent.
- (e) To Parent's Knowledge, none of the Parent SEC Reports is the subject of ongoing SEC review and, to Parent's Knowledge, there are no inquiries or investigations by the SEC or any internal investigations, pending or threatened, of Parent or any of its Subsidiaries. Parent has not received written comments from the SEC staff regarding any of the SEC Reports that remain unresolved.
- (f) Neither Parent nor any of its Subsidiaries has any Liabilities that would reasonably be expected to be material to the business or operations of Parent and its Subsidiaries, taken as a whole, except for (i) those shown on the balance sheet included the consolidated audited balance sheet for the Company included in the Company's annual report on Form 10-K filed with the SEC for the fiscal year ended December 31, 2024 (the "Parent Balance Sheet"), (ii) those that were incurred after December 31, 2024 in the Ordinary Course of Business, (iii) any executory obligations (that are not arising out of breach of contract or breach of warranty) arising under any Contracts to which Parent or any of its Subsidiaries is a party, and (iv) expenses incurred in connection with the transactions contemplated by this Agreement. All reserves established by the Company that are set forth in or reflected in the Parent Balance Sheet have been established in accordance with U.S. GAAP. Neither Parent nor any of its Subsidiaries has any "off-balance sheet arrangement" within the meaning of Item 303 of Regulation S-K promulgated under the Securities Act. No bankruptcy, insolvency, winding up, or similar proceedings of which Parent has received written notice have occurred or are pending or, to the Knowledge of Parent, threatened against Parent or any of its Subsidiaries or Parent's or any of its Subsidiaries' properties or assets.
- 3.8 <u>Cash Resources</u>. Parent and/or its controlled Affiliates have sufficient cash resources to pay the Cash Consideration to be paid pursuant to the terms hereof.

# 3.9 <u>Compliance with Laws</u>.

- (a) Parent and each of its Subsidiaries are, and have at all times since January 1, 2023 been, in compliance in all material respects with all applicable Laws.
- (b) Since January 1, 2023, neither Parent nor any of its Subsidiaries has received any written notice or other written communication from any Governmental Authority regarding (i) any actual or possible violation of applicable Law or any Governmental Permit or any failure to comply with any term or requirement of any Governmental Permit or (ii) any actual or possible revocation, withdrawal, suspension, cancellation, termination or modification of any Governmental Permit.
- (c) Except for routine examinations conducted by the SEC, FINRA, or any other Governmental Authority in the Ordinary Course of Business of the Parent, (i) no Governmental Authority has, to the Knowledge of the Parent, initiated any administrative proceeding or investigation into the business or operations of the Parent or any of its Subsidiaries that is unresolved and (ii) to the Knowledge of the Parent, there are no unresolved material

violations or material exceptions raised by any Governmental Authority with respect to the Parent or any of its Subsidiaries.

- 3.10 <u>Litigation</u>. There is no material Action pending against Parent or any of its Subsidiaries of which Parent has received written notice, to Parent's Knowledge no such material Action is threatened in writing, there is no outstanding material judgment, decree, injunction, rule or order of any Governmental Authority, arbitrator or mediator binding on Parent or any of its Subsidiaries or any of their respective assets or properties and neither Parent nor any of its Subsidiaries has brought any Action pending against any Governmental Authority or any other Person.
- 3.11 <u>Certain Restraints</u>. There is no event, circumstance, or condition applicable to Parent or any of its Subsidiaries that restricts or limits, or to the Knowledge of Parent would reasonably be expected to (individually or in the aggregate with other events, circumstances, or conditions) restrict, limit, impede or delay Parent from (i) operating the Company Business from and after Closing or (ii) combining its existing custody business and its existing asset management business with the Company Business from and after Closing. Without limiting the generality of the foregoing, neither Parent nor any of its Subsidiaries is party to any Contract or is subject to any judgment, decree or injunction of any Governmental Authority, arbitrator or mediator that would restrict the operation of the Company Business following the Effective Time.
- No Other Representations and Warranties; Non-Reliance. Parent and each Merger Sub acknowledges and agrees that except, in each case, for the representations and warranties contained in Article 2 or any certificate delivered hereunder (in each case as qualified by the Company Disclosure Schedule), and except in the cause of Fraud by or on behalf of the Company, (a) neither the Company nor any other Person, makes or has made any representation or warranty, express or implied, either written or oral, at law or in equity, in respect of the Company, any of its Subsidiaries, their respective business or operations or any other matter in connection with this Agreement and the transactions contemplated hereby or otherwise, including as to any information furnished or made available to Parent, either Merger Sub any of their respective Representatives or any other Person in connection with this Agreement, the transactions contemplated hereby, or as to the future revenue, profitability or success of the Company or any of its Subsidiaries, or any representation or warranty arising from statute or otherwise under Law, (b) neither the Company nor any other Person shall have any Liability in connection with any such information provided to Parent, either Merger Sub, any of their respective Representatives or any other Person and (c) none of Parent, either Merger Sub, any of their respective Representatives or any other Person has relied on any information, materials, representations or warranties or other statements or omissions (including as to the accuracy or completeness thereof), whether express or implied, that may have been made, provided or made available by the Company, any of its Subsidiaries, any of their respective Representatives or any other Person with respect to the Company or any of its Subsidiaries or their respective business or operations or otherwise in connection with or relating to the entry into this Agreement and the transactions contemplated hereby.

#### **ARTICLE 4**

#### **COVENANTS**

4.1 <u>Employment Arrangements; Termination of Certain Company Benefit Arrangements.</u>

- (a) In connection with the terminated participation in the TriNet 401(k) Plan, Parent will use commercially reasonable efforts to (i) allow each Continuing Employee who is a participant in the TriNet 401(k) Plan to participate, as soon as practicable following the Closing Date, in a tax qualified plan that includes a Code Section 401(k) arrangement that is sponsored by Parent, the Surviving Company, the Surviving Entity or any of their Affiliates (the "Parent 401(k) Plan") and (ii) credit with eligibility service and vesting service for all periods of service with the Company to the extent so credited with such service under the TriNet 401(k) Plan. In addition, Parent will, or will cause the Surviving Company, the Surviving Entity or any of their respective Affiliates, to use commercially reasonable efforts to accept rollover contributions of "eligible rollover distributions" (within the meaning of Section 401(a)(31) of the Code) in the form of cash in an amount equal to the full account balance distributed to such Continuing Employee from the TriNet 401(k) Plan.
- For purposes of eligibility and vesting (and for all purposes for severance entitlements, vacation and paid time off) where length of service is relevant under any benefit plan, policy, practice or arrangement of Parent, the Surviving Company, the Surviving Entity or any of their respective Subsidiaries providing benefits to any Continuing Employee after the Effective Time (other than any defined benefit pension, post-termination welfare, change in control, deferred compensation, retention, equity, phantom equity or equity-based incentive plan, program, agreement or arrangement) (collectively, the "New Plans"), the Continuing Employees shall receive credit for service with the Company to the same extent such service credit was granted under the corresponding benefit plan, policy, practice or arrangement of the Company, except to the extent any such service credit would result in the duplication of its benefits; provided, however, no such crediting of service shall result in duplication of benefits or compensation for the same period of service. In addition and without limiting the generality of the foregoing, (i) Parent will use commercially reasonable efforts to cause each Continuing Employee to be immediately eligible to participate, without any waiting time or satisfaction of any other eligibility requirements, in any and all New Plans to the extent that (A) coverage under such New Plan replaces coverage under a Company Benefit Arrangement in which such Continuing Employee participated immediately before the Effective Time (collectively, the "Old Plans"), and (B) such Continuing Employee has satisfied all waiting time and other eligibility requirements under the Old Plan being replaced by the New Plan and (ii) for purposes of each New Plan providing medical, dental, pharmaceutical, vision or other welfare benefits to any Continuing Employee, Parent shall use commercially reasonable efforts to cause (A) all preexisting condition exclusions and actively-at-work requirements of such New Plan to be waived for such Continuing Employee and his or her covered dependents to the extent such conditions were inapplicable or waived under the comparable Old Plan and (B) any expenses incurred by any Continuing Employee and his or her covered dependents during the portion of the plan year of the Old Plan ending on the date such Continuing Employee's participation in the corresponding New Plan begins to be taken into account under such New Plan for purposes of satisfying all deductible, coinsurance and maximum out-of-pocket requirements applicable to such Continuing Employee and his or her covered dependents for the applicable plan year as if such amounts had been paid in accordance with such New Plan.
- (c) The provisions contained in this <u>Section 4.1</u> are for the sole benefit of Parent and the Company and its Subsidiaries and no current or former employee, manager, independent contractor, consultant, service provider or any other individual associated therewith shall be regarded for any purpose as a third-party beneficiary of this Agreement. Nothing in this <u>Section 4.1</u>, express or implied, shall be construed or interpreted to (i) create any right, benefit or remedy of any nature whatsoever, including any right to continued employment or service, under or by reason of this Agreement, in any other person, including any employees, former employees, any participant or any beneficiary thereof in any Company Benefit Arrangement or

employee benefit plan of Parent, the Surviving Company, the Surviving Entity or any of their Affiliates, or (ii) amend any Company Benefit Arrangement or employee benefit plan of Parent, the Surviving Company, the Surviving Entity or any of their respective Affiliates. Nothing in this Section 4.1 shall be construed or interpreted to limit the ability of Parent, the Surviving Company, the Surviving Entity or any of their respective Affiliates to amend or terminate any employee benefit plan pursuant to its terms.

### 4.2 <u>Indemnification of Company Managers and Officers.</u>

- From and after the Closing Date until the sixth anniversary thereof, Parent shall cause the Surviving Entity and each of its Subsidiaries to honor and fulfill all rights to indemnification, advancement of expenses and exculpation existing as of immediately prior to the Effective Time in favor of any person who, prior to the Effective Time, served as a director, manager or officer of the Company or any of its Subsidiaries (collectively, the "Company Indemnified Parties") as provided in the Charter Documents of the Company or any of its Subsidiaries as of the Agreement Date, the DLLCA or in any agreement set forth on Schedule 4.2(a) with respect to claims arising out of acts or omissions occurring at or prior to the Effective Time, to survive the Mergers and to continue in full force and effect until the sixth anniversary of the Closing Date; provided that all rights to indemnification and exculpation in respect of any Action arising out of or relating to matters existing or occurring at or prior to the Closing Date and asserted or made within such six-year period shall continue until the final disposition of such Action. From and after the Closing Date until the sixth anniversary thereof, the Charter Documents of the Surviving Entity and each of its Subsidiaries shall contain, and Parent shall cause such Charter Documents to so contain, provisions no less favorable with respect to indemnification, advancement of expenses and exculpation of present and former directors, managers and officers than are set forth in such Charter Documents as of immediately prior to the execution and delivery of this Agreement. However, the foregoing covenant under this Section 4.2 shall not apply to a Company Indemnified Party in its capacity as an Indemnifying Party based on a claim for indemnification made by an Indemnified Party pursuant to Article 7. For the avoidance of doubt, and notwithstanding any provision to the contrary contained in an agreement set forth on Schedule 4.2(a) of this Agreement, no Company Indemnified Party shall be entitled to coverage under any Parent director and officer insurance policy or errors and omission policy unless such Company Indemnified Party is separately eligible for coverage under such policy pursuant to Parent's policies and procedures and the terms of such insurance policy.
- (b) Parent shall, and shall cause the Surviving Entity to, maintain the Tail Policy in full force and effect, and continue to honor the obligations thereunder until the sixth anniversary of the Closing Date.
- (c) If Parent or the Surviving Entity (i) shall consolidate with or merge into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) shall transfer all or substantially all of its properties and assets to any Person, then, and in each such case, proper provisions shall be made so that the successor and assigns of Parent or the Surviving Entity, as the case may be, shall assume all of the obligations of this <u>Section 4.2</u>.
- (d) The provisions of this <u>Section 4.2</u> are intended to be in addition to the rights otherwise available to the Company Indemnified Parties by law, charter, statute, by-law or agreement. This <u>Section 4.2</u> shall survive the consummation of the Mergers, is intended to benefit each Company Indemnified Party, shall be binding on all successors and assigns of the Surviving Entity and Parent, and shall be enforceable by the Company Indemnified Parties (each

of whom is an intended third party beneficiary of this <u>Section 4.2</u>); <u>provided</u>, that recourse shall first be against the Tail Policy until it is exhausted or is otherwise unavailable before recovery against the Surviving Entity shall take place.

- 4.3 <u>Public Information</u>. With a view to making available to a Company Securityholder or any Person receiving Parent Shares the benefits of Rule 144 that permit a Company Securityholder or any Person to sell Parent Shares to the public without registration, Parent agrees, for so long as a Company Securityholder or any Person holds Parent Shares, to: (i) use commercially reasonable efforts to make and keep public information available, as those terms are understood and defined in Rule 144; and (ii) use commercially reasonable efforts to file with the SEC in a timely manner all reports and other documents required of Parent under the Securities Act and the Exchange Act so long as Parent remains subject to such requirements and the filing of such reports and other documents as may be required pursuant to the applicable provisions of Rule 144.
- 4.4 <u>Listing</u>. Parent will use commercially reasonable efforts to continue the listing and trading of Parent Common Stock on the New York Stock Exchange and, in accordance therewith, will comply with Parent's reporting, filing and other obligations under the bylaws and rules thereof. Parent shall use commercially reasonable efforts to cause to be listed, effective as promptly as practicable after the Closing, all Parent Shares on the New York Stock Exchange.

#### **ARTICLE 5**

#### ESCROW AGREEMENT AND EXCHANGE AGENT AGREEMENT

- 5.1 <u>Escrow Agreement</u>. At the Closing, each of Parent, the Securityholder Representative and the Escrow Agent shall duly execute and deliver the Escrow Agreement.
- 5.2 <u>Exchange Agent Agreement</u>. At the Closing, each of Parent and the Exchange Agent shall duly execute and deliver the Exchange Agent Agreement.

#### **ARTICLE 6**

#### **CLOSING DELIVERABLES**

- 6.1 <u>Member Deliverables</u>. At the Closing, the Company shall deliver to Parent:
  - (a) fully executed Written Consents from Company Unitholders holding 100% of the Company Units;
  - (b) fully executed Joinder Agreements from all Securityholders; and
- (c) fully executed and completed Investor Questionnaires from Company Securityholders holding 100% of the aggregate number of Company Units and Company Profits Interest Units.
- 6.2 <u>Employment Matters</u>. The employment of any employee of the Company or the services of any independent contractor of the Company who has not accepted an offer of employment contained in an offer letter or signed a consulting agreement with Parent (or an Affiliate of Parent), shall have been terminated in accordance with Law by the Company, effective no later than immediately prior to the Effective Time and that each such non-continuing

employee has delivered a general release of claims against Parent, the Company and their Affiliates, in form and substance reasonably satisfactory to Parent.

- 6.3 Other Deliverables. At the Closing, the Company shall deliver to Parent:
  - (a) <u>Closing Statement</u>. The Closing Statement in accordance with <u>Section 1.8</u>.
- (b) <u>Good Standing Certificates</u>. A certificate of good standing from the office of the Secretary of State of the State of Delaware and each other state or jurisdiction in which the Company is qualified to do business as a foreign corporation certifying, as of a date no more than five (5) Business Days prior to the Closing Date, that the Company is in good standing (or an equivalent status).
- (c) Officer's Certificate. A certificate dated as of the Closing Date, signed by an officer of the Company, certifying as to (i) an attached copy of the Charter Documents of the Company and stating that such Charter Documents have not been amended, modified, revoked or rescinded, (ii) an attached copy of the resolutions of the Governing Body of the Company evidencing the Manager Approval, and stating that such resolutions have not been amended, modified, revoked or rescinded, and (iii) an attached copy of the Written Consents received from Company Unitholders, and stating that such Written Consents constitute the Member Approval and that the resolutions set forth therein have not been amended, modified, revoked or rescinded, (iv) the results of the vote of the Company Unitholders with respect to the approval or disapproval of certain payments or benefits that, absent approval pursuant to such vote, (x) may not be deductible by reason of Section 280G of the Code or (y) may be subject to an excise tax under Section 4999 of the Code.
- (d) <u>Termination of Participation in TriNet 401(k) Plan</u>. Evidence that participation of Company employees in the TriNet 401(k) Plan has been terminated (effective as of no later than the day immediately preceding the Closing Date) pursuant to resolutions of the Governing Body of the Company.
- (e) <u>Transaction Expenses Invoice</u>. Final invoices from any third party receiving payments from the Company in respect of any Transaction Expenses.
- (f) <u>Tax Forms</u>. (i) A properly completed and executed IRS Form W-9 from each Company Securityholder (or if a Company Securityholder is a disregarded entity for income Tax purposes, its regarded owner) and any other recipient of a payment constituting Merger Consideration, Transaction Expenses, or other payment due pursuant to this Agreement and (ii) a properly executed statement, issued by the Company pursuant to Treasury Regulations Sections 1.897-2(h) and 1.1445-2(c)(3) dated no more than thirty (30) days prior to the Closing Date and signed by an officer of the Company certifying that interests in the Company do not constitute "United States real property interests" under Section 897(c) of the Code, together with the notice to the IRS in accordance with the provisions of Treasury Regulations Section 1.897-2(h)(2).
- (g) <u>Tail Policy</u>. Evidence of the Company's purchase of tail insurance coverage for the Company's and its Subsidiaries' directors, managers and officers, which shall provide such directors, managers and officers with coverage for six (6) years following the Effective Time with respect to claims arising out of acts or omissions occurring at or prior to the Effective Time (the "**Tail Policy**").

#### **ARTICLE 7**

### SURVIVAL; INDEMNIFICATION

#### 7.1 Survival.

- (a) If the First Merger is consummated, (i) the Company General Representations and the right of any Indemnified Party to bring a Company General Representation Claim, shall survive until the Expiration Date, and (ii) the Company Fundamental Representations and the right of any Indemnified Party to bring a Company Fundamental Representation Claim, shall survive until the sixth (6<sup>th</sup>) anniversary of the Closing Date; <u>provided</u>, <u>however</u>, that (x) no right to indemnification, compensation and reimbursement pursuant to this <u>Article 7</u> in respect of any Claim based upon any failure of a representation or warranty to be true and correct that is set forth in a Claim Notice delivered prior to the applicable expiration date of such representation or warranty shall be affected by the expiration of such representation or warranty and (y) no such expiration shall affect the rights of any Indemnified Party, under this <u>Article 7</u> or otherwise, to seek recovery of Losses arising out of any Fraud, which rights will survive until the later of (1) the sixth (6<sup>th</sup>) anniversary of the Closing Date, or (2) the 90th day following the expiration of the statute of limitations applicable to such Fraud.
- (b) If the First Merger is consummated, (i) the representations and warranties of Parent and the Merger Subs contained in this Agreement (other than the Parent Fundamental Representations) shall terminate at the Closing, and (ii) the Parent Fundamental Representations and the right to assert a claim in respect thereof shall survive until the Expiration Date; provided, however, that (x) no claim based upon any failure of a Parent Fundamental Representation to be true and correct that is asserted prior to the Expiration Date shall be affected by the expiration of such Parent Fundamental Representation and (y) no such expiration or termination shall affect the right to assert or maintain any claim arising out of any Fraud, which rights will survive until the later of (1) the sixth (6<sup>th</sup>) anniversary of the Closing Date, or (2) the 90th day following the expiration of the statute of limitations applicable to such Fraud.
  - (c) All covenants to be performed at or prior to the Closing shall terminate at the Closing.
- (d) Each covenant of the Company, Parent or any Merger Sub that by its terms contemplates performance after the Closing (and, for the avoidance of doubt and notwithstanding <u>Section 7.1(c)</u>, any obligation to pay or issue Merger Consideration) shall survive until the expiration of its term or the performance in full of the undertaking set forth therein.

### 7.2 <u>Indemnification of the Indemnified Parties</u>.

(a) <u>Indemnification of Parent and the Other Indemnified Parties</u>. From and after the Effective Time, each Indemnifying Party shall severally (based on each such Indemnifying Party's Pro Rata Share), and not jointly, indemnify, defend and hold harmless each of Parent, the Surviving Company, the Surviving Entity and its and their Affiliates and its and their respective directors, officers and employees (each hereinafter referred to individually as an "**Indemnified Party**" and collectively as the "**Indemnified Parties**") from and against, and compensate and reimburse each of the Indemnified Parties for, any and all Losses incurred, suffered or accrued by an Indemnified Party, directly or indirectly, and whether arising out of a third party claim or a direct claim, to the extent arising out of, related to or in connection with:

- (i) any inaccuracy in or breach of a representation or warranty made by the Company in this Agreement;
- (ii) any inaccuracy or error in or omissions from the Closing Statement, or any inaccuracy in or breach of any certification made by or on behalf of the Company pursuant to Section 1.8, except to the extent that the value of a particular item underlying the calculation of the Adjusted Cash Consideration has been finally resolved pursuant to Section 1.8;
  - (iii) any Stakeholder Claim;
- (iv) any Action pending, or threatened in writing that was received by the Company or any of its Subsidiaries, against the Company or any of its Subsidiaries as of the Effective Time;
- (v) any Pre-Closing Taxes to the extent not expressly taken into account in the calculation of Unpaid Pre-Closing Taxes;
- (vi) any Fraud by or on behalf of the Company prior to the Closing in the negotiation, execution, delivery or performance of this Agreement or any Company Ancillary Agreement, or in connection with the Mergers or any of the transactions contemplated hereby or thereby; or
  - (vii) any matter referred to in <u>Schedule 7.2(a)(vii)</u> hereto.
- 7.3 <u>Limitations</u>. The indemnification provided for in <u>Section 7.2</u> shall be subject to the following limitations:
- (a) Subject to <u>Section 7.3(d)</u>, the maximum aggregate liability of the Indemnifying Parties for all Company General Representation Claims shall be the General Representation Cap.
- (b) Subject to Section 7.3(d), in the case of any (A) Company Fundamental Representation Claim, or (B) Claim under any of clauses (ii) through (vii) of Section 7.2(a) ((A) and (B), collectively, "Special Matters"), the maximum aggregate liability of the Indemnifying Parties for all Claims for Special Matters (in the aggregate with the maximum aggregate liability for all General Representation Claims) shall be the value of the Merger Consideration actually paid to the Company Securityholders (inclusive of the Indemnity Escrow Amount, Expense Fund and any Restricted Stock Amounts, and, subject to Section 7.9, inclusive of any portion of the Earnout Amount and any portion of the Adjustment Holdback Amount that is actually paid to the Company Securityholders).
- (c) Subject to Section 7.3(d), the aggregate liability of any Indemnifying Party for all Claims under Section 7.2(a) shall be capped at an amount equal to the value of the portion of the Merger Consideration actually paid to such Indemnifying Party pursuant to Section 1.4 (inclusive of the Indemnity Escrow Amount and Expense Fund, and, subject to Section 7.9, inclusive of any portion of the Earnout Amount and any portion of the Adjustment Holdback Amount that is actually paid to such Indemnifying Party) (it being understood and agreed that with respect to Indemnifying Parties that are holders of Aggregator Units, the value of the portion of the Merger Consideration actually paid to such Indemnifying Party pursuant to Section 1.4 shall be deemed to be such Indemnifying Party's Aggregator Pro Rata Share of the portion of

the Merger Consideration distributed by the Aggregator to the holders of the Aggregator Units in accordance with the Aggregator's Acknowledgement Agreement).

- (d) Notwithstanding anything herein to the contrary, there shall be no maximum liability with respect to Fraud for any Indemnifying Party who committed, or had actual knowledge prior to the Closing of, such Fraud.
- (e) For the purposes of determining the amount of any Losses arising out of any inaccuracy or breach of a representation or warranty herein (but not for purposes of determining whether any such inaccuracy or breach exists or has occurred), the terms "material" or "materially," any clause or phrase containing "material," "materially," "material respects," "Material Adverse Effect", or any similar terms, clauses or phrases in any such representation or warranty shall be disregarded (as if such word or clause, as applicable, were deleted from such representation or warranty), other than the representations and warranties set forth in the first sentence of Section 2.11. Notwithstanding the foregoing, this Section 7.3(e) shall not change the meaning of "Company Material Contract."
- In no event shall any Indemnifying Party be responsible or liable under this Article 7 for Losses in respect of punitive or exemplary damages, except for those awarded to a third party in connection with a Third-Party Claim. The Indemnified Parties hereto acknowledge the applicability of the common law duty to mitigate damages. The amount of any Losses for which indemnification is provided under this Article 7 shall be reduced by an amount equal to the amount of any proceeds actually received by the Indemnified Party (or any of its Affiliates) from any third-party (including any insurer) for such Losses (after giving effect to any deductible or retention or increase in premium reasonably demonstrated to result therefrom and net of any costs, Taxes and reasonable out-of-pocket expenses of recovery or collection thereof). Neither Parent nor any Indemnified Party shall have any obligation to obtain insurance coverage from any third parties (other than the Tail Policy) with respect to any particular matter for which it may be entitled in connection with any Losses it incurs but will use commercially reasonable efforts to recover such Losses from Parent's existing insurance policies and from other Persons under any existing Contracts and any recovery will be only after giving effect to any deductible or retention or increase in premium reasonably demonstrated to result therefrom and net of any costs, Taxes and reasonable out-of-pocket expenses of recovery or collection thereof. If any Indemnified Party (or any of its Affiliates) receives any insurance payment or other recovery form any third party in connection with any Claim for which it has already received an indemnification payment (whether in the form of cash or shares of Parent Common Stock) under this Article 7, it shall pay to each Indemnifying Party, within ten (10) days of receiving such insurance payment or other recovery from any third party, an amount equal to such Indemnifying Party's Pro Rata Share of the excess of (x) the amount previously received by the Indemnified Party under this Article 7 with respect to such Claim plus the amount of the insurance payments from any third-party insurer (after giving effect to any deductible or retention or increase in premium reasonably demonstrated to result therefrom and net of any costs, Taxes and reasonable out-of-pocket expenses of recovery or collection thereof) or other recoveries from any third party received, over (y) the amount of Losses with respect to such Claim which the Indemnified Party has become entitled to receive under this Article 7
- (g) If an Indemnified Party's Claim under this <u>Section 7.3</u> may be properly characterized in multiple ways in accordance with this <u>Section 7.3</u> such that such Claim may or may not be subject to different caps, time limitations, and other limitations depending on such characterization, then such Indemnified Party shall have the right to characterize such matter in a manner that maximizes the recovery and time to assert claims permitted in accordance with this <u>Article 7</u>.

- (h) No Indemnified Party shall be entitled to double recovery for any indemnifiable Losses even though such Losses may be recoverable under more than one provision of <u>Section 7.2(a)</u> (and, for the avoidance of doubt, the amount of any Losses indemnifiable pursuant to this <u>Article 7</u> shall be calculated net of the amount of such Losses to the extent taken into account in the calculation Adjusted Cash Consideration set forth in the Final Closing Statement (as finally determined in accordance with <u>Section 1.8</u>)).
- (i) No Losses may be recovered under <u>Section 7.2(a)(i)</u> in respect of any Company General Representation Claim until such Losses exceed \$204,375 in the aggregate (the "**Basket Amount**"), at which time the Indemnified Parties shall be entitled to recover in accordance with this <u>Article 7</u> all such Losses, inclusive of those Losses that comprise the Basket Amount.
- Claim Notice. If Parent, acting on its own behalf or on behalf of any of the other Indemnified Parties, wishes to assert a Claim, Parent shall deliver written notice thereof, executed by a Representative of Parent (a "Claim Notice"), to the Securityholder Representative (and no Claim may be asserted except in the manner provided in this Section 7.4). The Claim Notice shall set forth: (a) that an Indemnified Party has incurred, suffered or accrued, or reasonably believes it may have to incur, suffer or accrue Losses for which such Indemnified Party is entitled to indemnification under this Article 7; (b) the actual or estimated amount of such Losses to the extent known or reasonably estimable (which, in the case of Losses not yet incurred, suffered or accrued, may be the maximum amount reasonably anticipated by Parent to be incurred, suffered or accrued or may be the amount of Losses claimed by a third party in a Third-Party Claim); and (c) a brief description, in reasonable detail (to the extent reasonably available to such Indemnified Party), of the facts, circumstances or events giving rise to the alleged Losses based on such Indemnified Party's belief thereof. A Claim Notice may be updated and amended from time to time by Parent by delivering an updated or amended Claim Notice to the Securityholder Representative, so long as such update or amendment only asserts bases for Losses reasonably related to the underlying facts and circumstances specifically set forth in such original Claim Notice. All Claims properly set forth in an original Claim Notice or any update or amendment thereto shall remain outstanding until such Claims for Losses have been finally resolved or satisfied. Notwithstanding the foregoing, no Claim Notice with respect to a breach of a representation, warranty, covenant or agreement (or any certificate with respect thereto) may be delivered following the expiration of the survival period with respect to such representation, warranty, covenant or agreement (or the right to bring a claim with respect thereto) provided for in Section 7.1.

#### 7.5 Resolution of Claim Notice.

(a) <u>Uncontested Claims</u>. If, within thirty (30) days after a Claim Notice is delivered to the Securityholder Representative (such 30-day period, the "**Objection Period**"), the Securityholder Representative either consents in writing to the Claim Notice, or does not contest all or any portion of such Claim Notice in writing to Parent as provided in <u>Section 7.5(b)</u>, then the Securityholder Representative shall be conclusively deemed to have agreed, on behalf of all Indemnifying Parties, to the recovery by each applicable Indemnified Party of the full amount of Losses (or, if a portion of the Claim Notice is contested, the uncontested portion thereof) (subject to the limitations contained in <u>Section 7.3</u>) specified in the Claim Notice, including recovery by Parent of all or a portion of the Indemnity Escrow Amount to satisfy such Losses and the recovery from the Indemnifying Parties of any amounts in excess of the Indemnity Escrow Amount, and, without further notice, to have stipulated to the entry of a final judgment for damages against the Indemnifying Parties for such amount in any court having jurisdiction over the matter where venue is proper.

- (b) <u>Contested Claims</u>. If, during the Objection Period, the Securityholder Representative delivers to Parent a written notice contesting all or any portion of a Claim Notice (the contested portion, a "Contested Claim"), then such Contested Claim shall be resolved by either (i) a written settlement agreement executed by Parent and the Securityholder Representative or (ii) in the absence of such a written settlement agreement within 30 days following receipt by Parent of the written notice of a Contested Claim (or such longer period as agreed in writing by Parent and the Securityholder Representative), in accordance with the terms and provisions of <u>Section 7.5(c)</u>.
- (c) <u>Resolution of Contested Claims</u>. If Parent and the Securityholder Representative do not enter into a written settlement agreement resolving a Contested Claim during the period contemplated by <u>Section 7.5(b)(ii)</u>, either Parent or the Securityholder Representative may bring suit in accordance with <u>Section 9.1</u> to resolve the Contested Claim. Final judgment upon any award rendered by the trial court may be entered in any court having jurisdiction. Notwithstanding the foregoing, if Parent and the Securityholder Representative mutually agree in their sole discretion, Parent and the Securityholder Representative may submit a Contested Claim to alternative dispute resolution prior to, or in lieu of, pursuing the claim in court.
- Payment of Claims. If any Losses are determined, agreed or deemed agreed to be owed to any Indemnified Party in accordance with this Section 7.5 (such amount, the "Owed Amount"), then (A) first, Parent shall be entitled to recover such amount from the Indemnity Escrow Amount with a value equal to the Owed Amount, and (B) second, if the amounts remaining in the Indemnity Escrow Amount are insufficient to cover the full Owed Amount, or if all of the Indemnity Escrow Amount has been previously released to by Parent or released to the Company Securityholders, then, subject to the limitations contained in Section 7.3, (i) if an Indemnifying Party is an Unaccredited Investor, such Indemnifying Party shall, within ten (10) Business Days following the date such Owed Amount is determined, pay an amount in cash equal to such Indemnifying Party's Pro Rata Share of the Owed Amount (or the remaining portion thereof), or (ii) if an Indemnifying Party is an Accredited Investor, such Indemnifying Party shall, within ten (10) Business Days following the date such Owed Amount is determined, pay an amount in cash and/or contribute a number of Parent Shares to Parent equal to such Indemnifying Party's Pro Rata Share of the Owed Amount (or the remaining portion thereof) (valuing such Parent Shares at the Reference Price, with the amount attributable to any fractional shares to be paid in cash), with the amount of cash paid and number of Parent Shares contributed being determined by the Indemnifying Party in its sole discretion. The Securityholder Representative hereby agrees to give notice to each Indemnifying Party of such payment obligation within three (3) Business Days of such determination, agreement or deemed agreement; provided, however, that the Securityholder Representative's obligation to notify each Indemnifying Party of any such payment obligation shall be satisfied upon transmission of email notice to the email addresses included in the Closing Statement.
- 7.6 <u>Defense and Settlement of Third-Party Claims</u>. In the event Parent receives notice of the assertion or commencement of any claim or action made or brought by a third party that Parent in good faith believes may result in a Claim by or on behalf of an Indemnified Party (a "**Third-Party Claim**"), Parent shall promptly notify the Securityholder Representative of such Third-Party Claim; <u>provided</u>, that any failure to provide such prompt notification shall not relieve the Indemnifying Parties of their indemnification obligations, except and only to the extent that the Indemnifying Parties are actually and materially prejudiced thereby. Parent shall have the right in its sole discretion to conduct the defense of and to settle or resolve such Third-Party Claim, and the Securityholder Representative shall be entitled, on behalf of the Indemnifying Parties, at their expense, to participate in, but not to determine or conduct, the

defense of such Third-Party Claim. The Securityholder Representative shall have the right to receive copies of all pleadings, notices and communications received from such third party by the Indemnified Parties with respect to the Third-Party Claim to the extent that receipt of such documents does not affect any privilege with respect to Parent or any Indemnified Party. The reasonable and documented out-of-pocket costs and expenses incurred or paid by any Indemnified Party in connection with the defense (including reasonable attorneys' fees, other professionals' and experts' fees, costs of investigation and court or arbitration costs) of any Third-Party Claim are recoverable as Losses pursuant to this <u>Article 7</u> regardless of the outcome of such Third-Party Claim (but only if indemnification is owed in respect of Losses arising from such Third-Party Claim pursuant to <u>Section 7.2(a)</u>), subject to the limitations on recovery in <u>Section 7.3</u>. Notwithstanding anything to the contrary, in the event of settlement or other resolution by Parent of any Third-Party Claim, the amount paid in such settlement or resolution (the "Settlement Amount") shall not be determinative and binding upon the Indemnifying Parties as to the existence or amount of Losses recoverable pursuant to this <u>Article 7</u> with respect thereto unless the Securityholder Representative has consented (such consent not to be unreasonably conditioned, withheld or delayed) in which case the Settlement Amount for such settlement or resolution shall be Losses for which Parent and/or the affected Indemnified Party are entitled to be indemnified, compensated and reimbursed hereunder to the extent the underlying matter is indemnifiable pursuant to <u>Section 7.2</u>, subject to the limitations in <u>Section 7.3</u>). If the Securityholder Representative has consented to any such Settlement Amount, the Indemnifying Parties shall have no power or authority to object under any provision of this <u>Article 7</u> to such Settlement Amount).

### 7.7 <u>Indemnity Escrow Arrangements</u>.

- (a) The Indemnity Escrow Amount shall be deposited with the Escrow Agent in accordance with <u>Section 1.6(b)</u>, and such deposit, together with any interest that may be earned thereon, will constitute an escrow fund to be governed by the provisions set forth herein and in the Escrow Agreement (the "**Escrow Fund**"), and the Indemnity Escrow Amount will be available to indemnify, compensate and reimburse the Indemnified Parties for any Losses for which they are entitled to recover in accordance with the terms of this <u>Article 7</u>.
- (b) No later than the second (2<sup>nd</sup>) Business Day following the Expiration Date, Parent and the Securityholder Representative will deliver a Joint Instruction to the Escrow Agent to release to the Company Securityholders all amounts then held in the Escrow Fund less the portion of the Indemnity Escrow Amount having a value equal to the amount that Parent reasonably determines may be necessary to satisfy all unresolved, unsatisfied or Contested Claims specified in any Claim Notices delivered to the Securityholder Representative before the Expiration Date (collectively, "Unresolved Claims", and such amount determined by Parent, the "Unreleased Escrow Amount"). If there are any Unresolved Claims as of the Expiration Date, then the Escrow Agent shall retain possession and custody of the Unreleased Escrow Amount; provided that (i) no later than the second (2<sup>nd</sup>) Business Day following the resolution of any Unresolved Claim, Parent and the Securityholder Representative will deliver a Joint Instruction to the Escrow Agent to release to the Company Securityholders the excess (if any) of the portion of the Unreleased Escrow Amount attributable to such Unresolved Claim over the amount (if any) required to satisfy any amount owed to the Indemnified Parties pursuant to such resolution and (ii) no later than the second (2<sup>nd</sup>) Business Day following the resolution of last remaining Unresolved Claim, Parent and the Securityholder Representative will deliver a Joint Instruction to the Escrow Agent to release all amounts then held in the Escrow Fund to the Company Securityholders less any amount owed to the Indemnified Parties pursuant to the resolution of such last remaining Unresolved Claim.

- (c) Each delivery of any portion of the Escrow Fund to Company Securityholders pursuant to <u>Section 7.7(b)</u> shall be made in proportion to the Company Securityholders' respective Pro Rata Shares of the portion of the Escrow Fund being delivered.
- (d) For purposes of this <u>Article 7</u> and <u>Article 8</u>, each share of Parent Common Stock shall be deemed to have a value equal to the Reference Price (as adjusted to appropriately reflect any stock split, reverse stock split, stock dividend, reorganization, reclassification, combination, recapitalization or other like change with respect to the shares of Parent Common Stock after the Agreement Date).
- 7.8 <u>Payment of Indemnity Escrow Amount</u>. With respect to any portion of the Indemnity Escrow Amount to be released to the Company Securityholders pursuant to <u>Section 7.7</u>, unless the Securityholder Representative provides updated payment delivery instructions, each delivery of any portion of the Indemnity Escrow Amount to a particular Company Securityholder shall be effected in accordance with the payment delivery instructions set forth in such Person's Letter of Transmittal.

# 7.9 <u>Right to Satisfy Indemnification Claims by Reducing Earnout Amounts.</u>

- (a) Subject to the applicable limitations in <u>Section 7.3</u>, Parent is expressly authorized to set off any Losses for which it is entitled to indemnification hereunder as finally determined pursuant to the final resolution of the claims set forth in any Claim Notice pursuant to <u>Section 7.4</u> (and for any portion of the Post-Closing Deficit Amount not previously satisfied pursuant to <u>Section 1.8</u>) against any Earnout Amount due and issuable, if any.
- (b) Notwithstanding Section 7.9(a), if at the time any Earnout Amount becomes due and payable, (i) there shall be any outstanding Claim Notice, and (ii) the amount of Losses that Parent reasonably anticipates to be entitled to in satisfaction of the claims set forth therein (such estimated amount for all pending claims, collectively, the "Anticipated Losses") exceeds the remaining amount of the Indemnity Escrow Amount that is then available to satisfy indemnification claims, then (A) at Parent's election, the amount of the Earnout Amount to be paid at such time may be withheld by Parent in an aggregate amount equal to the amount of the Anticipated Losses (valuing Parent Shares that comprise the Earnout Amount at the Reference Price) (the aggregate amount of all Anticipated Losses so withheld, the "Retained Loss Amount"), pending the final resolution of such unresolved indemnification claim in accordance with this Agreement, (B) Parent shall, with respect to the remaining portion of the Earnout Amount (i) instruct the Transfer Agent to issue the applicable number of shares of Parent Common Stock (for Accredited Investors) and (ii) deliver to the Exchange Agent the applicable amount of cash (for Unaccredited Investors), each in accordance with Sections 1.14(e)(ii), and 1.14(e)(iii), and (C) Parent shall promptly provide the Securityholder Representative with written notice of any Earnout Amount pursuant to this Section 7.9(b).
- (c) Following the final resolution pursuant to this Agreement of any indemnification claim for which a Retained Loss Amount was retained, (i) if Parent has been finally determined pursuant to such resolution to have suffered aggregate indemnifiable Losses under this Article 7 with respect to such indemnification claim in amount less than the Retained Loss Amount attributable to such indemnification claim after taking into account the amount of Losses to be satisfied from the Indemnity Escrow Amount and any Losses for which indemnification has already been provided under this Article 7, then Parent shall, within ten (10) calendar days of such final resolution, (i) instruct the Transfer Agent to issue a number of shares of Parent Common Stock (for Accredited Investors) and (ii) deliver an amount of cash to the Exchange Agent (for Unaccredited Investors) comprising the Earnout Amount equal to the

portion of the Retained Loss Amount that was not used to satisfy such indemnifiable Loss and that is not necessary in Parent's reasonable judgment to satisfy any other pending indemnification claim for which Parent provided the requisite notice to the Securityholder Representative in accordance with Sections 1.14(e)(i), 1.14(e)(ii), and 1.14(e)(iii) (valuing Parent Shares at the Reference Price), and (ii) if Parent has been finally determined pursuant to such resolution to have incurred, suffered or accrued aggregate indemnifiable Losses under this Article 7 with respect to such indemnification claim in an amount less than, equal to, or greater than such Retained Loss Amount (after taking into account the amount of such Losses to be satisfied from the Indemnity Escrow Amount and any such Losses for which indemnification has already been provided under this Article 7), then Parent shall be entitled to set-off and permanently retain a number of shares of Parent Common Stock comprising the Earnout Stock Consideration in an amount equal to such Retained Loss Amount attributable to such indemnification claim (or the applicable portion thereof to the extent that the finally-determined aggregate indemnifiable Losses (less any portion thereof to be satisfied from the Indemnity Escrow Amount or for which indemnification has already been provided under this Article 7) are less than such Retained Loss Amount) (valuing such Parent Shares at the Reference Price). For the avoidance of doubt, if the final amount of Losses for such indemnification claim exceeds the aggregate amount by which such Earnout Stock Consideration was set-off and permanently retained, then, subject to the applicable limitations in Section 7.3, Parent shall continue to be entitled to indemnification for the amount of such excess pursuant to the terms and conditions of this Article 7.

- (d) For the avoidance of doubt, the number of shares of Parent Common Stock Parent is entitled to set off pursuant to Section 7.9 shall be determined in accordance with Section 7.7(d).
- 7.10 <u>Tax Consequences of Indemnification Payments</u>. All payments (if any) made to an Indemnified Party pursuant to any indemnification, compensation or reimbursement obligations under this <u>Article 7</u> or pursuant to <u>Article 8</u> will be treated as adjustments to the Merger Consideration for Tax purposes and such agreed treatment will govern for purposes of this Agreement, unless otherwise required by Law.
- 7.11 No Right of Contribution. No Indemnifying Party nor the Securityholder Representative acting on his, her or its behalf shall make any claim for or be entitled to indemnification, compensation, reimbursement or contribution from Parent, the Company, any of their respective Affiliates, or any successor or assign of any of the foregoing, with respect to any Claims to the extent that any Indemnified Party is entitled to indemnification under this Article 7 for such Claim.
- 7.12 Exclusive Remedy. Following the Closing, except for (a) claims for Fraud against any Indemnifying Party who committed or had actual knowledge prior to the Closing of such Fraud and (b) claims for equitable relief, the rights to indemnification, compensation and reimbursement under this Article 7 shall be the sole and exclusive remedy of the Indemnified Parties against the Indemnifying Parties with respect to this Agreement, including for breaches of the representations, warranties, covenants and agreements set forth in this Agreement, or any of the transactions contemplated hereby. Nothing in this Section 7.12 shall be deemed to limit the rights or remedies of any Indemnified Party under any Contract, other than this Agreement, entered into by Parent or any of its Affiliates in connection with this Agreement and the transactions contemplated hereby, including any Joinder Agreement, any Investor Questionnaire, any Employment Document, any Non-Competition Agreement, or any Restricted Stock Agreement (and, notwithstanding Section 7.13 or anything else to the contrary, the Securityholder Representative shall not have any authority with respect to any such Contract).

### 7.13 <u>Appointment of Securityholder Representative</u>.

By voting in favor of the adoption of this Agreement, executing and delivering a Joinder Agreement or participating in the Mergers and receiving the benefits thereof, each Securityholder shall be deemed to have approved the designation of and hereby designates the Securityholder Representative as the representative of the Securityholders and as the attorney-in-fact and agent for and on behalf of each Securityholder with respect to Claims under this Article 7 and the taking by the Securityholder Representative of any and all actions and the making of any decisions required or permitted to be taken by the Securityholder Representative under this Agreement or the Escrow Agreement, including the exercise of the power to: (i) give and receive notices and communications (on behalf of itself or any other Securityholder) relating to this Agreement or any of the transactions and other matters contemplated hereby, (ii) authorize Parent and any other applicable Indemnified Party to be indemnified, compensated or reimbursed for Losses, including through the release to Parent of all or any portion of the Indemnity Escrow Amount or through set-off or direct recovery from Securityholders, in satisfaction of Claims by Parent or any other Indemnified Party pursuant to this Article 7, (iii) agree to, object to, negotiate, resolve, enter into settlements and compromises of, demand litigation of, and comply with orders of courts with respect to (A) Claims by Parent or any other Indemnified Party pursuant to this Article 7 or (B) any dispute between any Indemnified Party and any such Securityholder, in each case, relating to this Agreement or any of the transactions or other matters contemplated hereby, (iv) agree to, negotiate and/or comply with the determination of the Adjusted Cash Consideration pursuant to Section 1.8 and (v) take all actions necessary or appropriate in the judgment of the Securityholder Representative for the accomplishment of the foregoing. The Securityholder Representative shall have authority and power to act on behalf of each Securityholder with respect to the disposition, settlement or other handling of all Claims under this Article 7 and all rights or obligations arising under this Article 7. The Securityholders and their respective successors, heirs, estates and assigns shall be bound by all actions taken and documents executed by the Securityholder Representative in connection with this Agreement, including Article 7 and the Escrow Agreement, and Parent and the other Indemnified Parties shall be entitled to rely on any such action or decision of the Securityholder Representative. The Securityholders recognize and intend that the power of attorney granted in this Section 7.13(a) and the powers, immunities and rights to indemnification granted to the Securityholder Representative hereunder: (1) are coupled with an interest and are irrevocable (subject to Section 7.13(b)); (2) may be delegated by the Securityholder Representative; and (3) shall survive the death, incapacity, dissolution, liquidation, bankruptcy or winding up of each of the Securityholders and shall be binding on any successor thereto. Each Securityholder (x) agrees that all actions taken by the Securityholder Representative under this Agreement shall be binding upon such Securityholder and such Securityholder's successors as if expressly confirmed and ratified in writing by such Securityholder and (y) waives any and all defenses which may be available to contest, negate or disaffirm the action of the Securityholder Representative taken in good faith under this Agreement. The Securityholder Representative shall only have the duties expressly stated in this Agreement and shall have no other duty, express or implied. The Securityholder Representative may engage attorneys, accountants and other professionals and experts. The Securityholder Representative may in good faith rely conclusively upon information, reports, statements and opinions prepared or presented by such professionals, and any action taken by the Securityholder Representative based on such reliance shall be deemed conclusively to have been taken in good faith. Parent may conclusively rely, without independent verification or investigation, upon any action of the Securityholder Representative taken in connection with this Agreement, including Article 7 and the Escrow Agreement as being the binding decision or action of the Securityholders, and Parent shall not be liable to any Securityholder or any other Person for any actions taken or omitted from being taken by them or by Parent in accordance with or reliance upon any decision or action of the Securityholder

Representative taken in connection with this Agreement, including <u>Article 7</u> and the Escrow Agreement.

- (b) The Person serving as the Securityholder Representative may be replaced (including as a result of such Person's resignation) from time to time by the holders of a majority in interest of the Merger Consideration payable to the Securityholders. No bond shall be required of the Securityholder Representative, and the Securityholder Representative shall receive no compensation for his, her or its services. Notices or communications to or from the Securityholder Representative shall constitute notice to or from each of the Securityholders.
- (c) In performing the functions specified in this Agreement, the Securityholder Representative shall not be liable to any Securityholder in the absence of gross negligence or willful breach on the part of the Securityholder Representative. Each Indemnifying Party shall severally (based on each such Indemnifying Party's respective Pro Rata Share), and not jointly, indemnify and hold harmless the Securityholder Representative from and against any loss, liability or expense incurred without gross negligence or willful breach on the part of the Securityholder Representative and arising out of or in connection with the acceptance or administration of its duties hereunder, including any out-of-pocket costs and expenses and legal fees and other legal costs reasonably incurred by the Securityholder Representative (together, the "Securityholder Representative Expenses").
- The Securityholder Representative represents and warrants to Parent and the Merger Subs as of the Agreement Date and as of the Closing Date as follows: (i) the Securityholder Representative is a limited liability company, duly organized, validly existing and in good standing under the laws of the State of New York and has all requisite limited liability company power and authority to execute and deliver this Agreement and any other applicable Contract, instrument or document contemplated hereby and to perform its obligations hereunder and thereunder, (ii) the execution, delivery and performance by the Securityholder Representative of this Agreement and any other applicable Contract, instrument or document contemplated hereby have been duly and validly authorized by the Securityholder Representative and no other limited liability company act or proceeding on the part of the Securityholder Representative or its equity holders is necessary to authorize the execution, delivery or performance by the Securityholder Representative of this Agreement or any other applicable Contract, instrument or document contemplated hereby, (iii) this Agreement and any other applicable Contract, instrument or document contemplated hereby has been duly executed and delivered by the Securityholder Representative and, assuming the due authorization, execution and delivery by the other parties hereto or thereto, constitutes a valid and binding obligation of the Securityholder Representative, enforceable in accordance with its terms and (iv) neither the execution, delivery or performance of this Agreement or any other applicable Contract, instrument or document contemplated hereby by the Securityholder Representative nor the consummation of the Mergers will conflict with, or result in a termination, breach, impairment or violation of, the organizational or other governing documents of the Securityholder Representative, or, except as would not reasonably be expected to be material to the Securityholder Representative, any applicable Law or Contract to which the Securityholder Representative or its assets or properties is bound.
- (e) The Securityholder Representative shall not be paid any fee for services to be rendered hereunder but shall be reimbursed for the Securityholder Representative Expenses. The Securityholder Representative shall use the Expense Fund to pay any expenses incurred by the Securityholder Representative. If the remaining Expense Fund is insufficient to pay any Securityholder Representative Expenses, such Securityholder Representative Expenses shall be paid or reimbursed from the first proceeds from any cash portions of the Merger Consideration

otherwise available for distribution to the Securityholders. Upon the good faith determination of the Securityholder Representative that retaining any portion of the Expense Fund is no longer necessary, the Securityholder Representative shall deliver any then remaining portion of the Expense Fund to the Exchange Agent, which will (and which Parent will cause to) promptly pay to each Company Unitholder and the Aggregator its Expense Fund Sharing Percentage of any such amounts so delivered. The Securityholder Representative shall hold, invest (to the extent the Securityholder Representative elects to invest) and disburse the Expense Fund in trust for all of the Securityholders and the Expense Fund shall not be used for any other purpose. For tax purposes, the Expense Fund will be treated as having been received and voluntarily set aside by each Company Unitholder and the Aggregator at the time of Closing.

#### **ARTICLE 8**

#### TAX MATTERS

### 8.1 <u>Tax Returns</u>.

- (a) Following the Closing Date, except as set forth in Section 8.1(d), the Securityholder Representative, at the Company Securityholders' cost and expense shall prepare and timely file, or shall cause to be prepared and timely filed, all Tax Returns in respect of the Company and its Subsidiaries for all taxable periods ending on or before the Closing Date that are required to be filed (taking into account extensions) after the Closing Date for which the results of operations reflected on such Tax Returns are required to be reported in whole or in part on the Tax Returns of the Company Securityholders or any direct or indirect owner of the Company Securityholders (including, for the avoidance of doubt, IRS Form 1065 and Schedule K-1) due to the status of any such entity as a disregarded entity or partnership for income Tax purposes ("Pass-Through Tax Returns"). All such Pass-Through Tax Returns shall be prepared by treating items on such Pass-Through Tax Returns in a manner consistent with the past practices of the Company with respect to such items, except as required by this Agreement or by applicable Law. At least twenty five (25) days prior to filing any such Pass-Through Tax Returns (or as soon as reasonably practicable following Closing to the extent the due date (including extensions) for such Pass-Through Tax Return is within twenty five (25) days after Closing), the Securityholder Representative shall submit a copy of any such Pass-Through Tax Return, along with supporting work papers, to Parent. Parent shall be entitled to comment on such Pass-Through Tax Returns and the Securityholder Representative shall consider such comments in good faith. If the Securityholder Representative does not receive comments from Parent at least five (5) days prior to the filing of such Pass-Through Tax Returns, Parent shall be deemed to have no comments to such Pass-Through Tax Returns.
- (b) Following the Closing Date, Parent shall prepare and file, or cause to be prepared and filed, at Parent's cost and expense, all Tax Returns that are not Pass-Through Tax Returns required to be filed by the Company and its Subsidiaries after the Closing Date with respect to Pre-Closing Tax Periods. Parent shall permit the Securityholder Representative, at the Company Securityholders' expense, to review and comment on each such Tax Return that reflects a material amount of Tax for which the Indemnifying Parties would be liable pursuant to his Agreement at least twenty five (25) days prior to filing (or as soon as reasonably practicable following Closing to the extent the due date (including extensions) for such Tax Return is within twenty five (25) days after Closing). The Securityholder Representative shall be entitled to comment on such Tax Returns and Parent shall consider such comments in good faith. If Parent does not receive comments from the Securityholder Representative at least five (5) days prior to the filing of such Tax Returns, the Securityholder Representative shall be deemed to have no comments to such Tax Returns.

- (c) Not later than five (5) days prior to the due date of the payment of Taxes on any Tax Returns which Parent has the responsibility to cause to be filed pursuant to <u>Section 8.1(b)</u>, without duplication of, or prejudice to, the Indemnified Parties' rights to indemnification, compensation and reimbursement under <u>Section 7.2(a)</u>, Parent shall be entitled to permanently retain a portion of the Indemnity Escrow Amount with a value equal to the amount of Taxes shown as due on such Tax Returns, to the extent such amount was not expressly taken into account in the calculation of the Unpaid Pre-Closing Taxes, Adjusted Cash Consideration, otherwise in the purchase price as reasonably determined by Parent.
- (d) No later than fifteen (15) days after the Closing, the Securityholder Representative shall prepare and deliver to Parent for its review and approval a draft IRS Form 8937 completed with respect to the Mergers. The Securityholder Representative shall make any revisions to such draft IRS Form 8937 as may be reasonably requested by Parent. The Securityholder Representative shall be responsible for timely filing the IRS Form 8937 as approved by Parent and delivering copies of the IRS Form 8937 to the Company Securityholders as required by applicable Law.
- 8.2 <u>Cooperation</u>. Parent and the Securityholder Representative agree to furnish or cause to be furnished to the other, upon request, as promptly as practicable, such information and assistance relating to Taxes, including access to books and records, as is reasonably necessary for the filing of all Tax Returns by Parent or the Securityholder Representative, the making of any election relating to Taxes, the preparation for any audit by any Tax authority and the prosecution or defense of any claim, suit or proceeding relating to any Tax. Parent, the Company and the Securityholder Representative shall each retain all books and records in their possession with respect to Taxes for a period of at least seven (7) years following the Closing Date. Notwithstanding the foregoing or any other provision herein to the contrary, in no event shall the Securityholder Representative be entitled to review or otherwise have access to any Tax Return, or information related thereto, of Parent or its Affiliates (other than Tax Returns of the Company for Pre-Closing Tax Periods).

### 8.3 <u>Tax Audits</u>.

- (a) If notice of any Action or threatened Action with respect to Taxes of the Company or its Subsidiaries (a "Tax Claim") shall be received by any party for which any other party may reasonably be expected to be liable, the notified party shall notify such other party or parties in writing of such Tax Claim; provided, however, that the failure of the notified party to give any other party notice as provided herein shall not relieve such other party of its indemnification, compensation and reimbursement obligations under Article 7 or this Article 8 except to the extent that such other party is actually and materially prejudiced thereby. Notwithstanding any provision herein to the contrary, to the extent that a provision of this Section 8.3 directly conflicts with any provision of Article 7, this Section 8.3 shall govern.
- (b) The Securityholder Representative shall have the right to control the conduct of any Tax Claim with respect to any Pass-Through Tax Returns of the Company or any of its Subsidiaries (a "Pass-Through Tax Claim"). The Securityholder Representative shall (i) keep Parent reasonably informed of all material developments on a timely basis, (ii) provide to Parent copies of any and all material correspondence from any Governmental Authority related to such Pass-Through Tax Claim, (iii) provide Parent with the opportunity to attend conferences with the relevant Governmental Authority (if reasonably practical) and (iv) not settle, adjust or otherwise resolve such Pass-Through Tax Claim without the consent of Parent (such consent not to be unreasonably withheld conditioned or delayed).

- (c) Parent shall have the right to control the conduct of any Tax Claim other than a Pass-Through Tax Claim. To the extent a Tax Claim is controlled by Parent and relates to Taxes attributable to a Pre-Closing Tax Period, Parent shall (i) keep the Securityholder Representative reasonably informed of all material developments on a timely basis, (ii) provide to the Securityholder Representative copies of any and all material correspondence from any Governmental Authority related to such Tax Claim, (iii) provide the Securityholder Representative with the opportunity to attend conferences with the relevant Governmental Authority (if reasonably practical) and (iv) have the right to settle, adjust or otherwise resolve such Tax Claim without consent of the Securityholder Representative; <u>provided</u>, that, if such settlement or other compromise would give rise to a material amount of Taxes that would be (i) passed through to the Company Securityholders under applicable Law or (ii) subject to indemnification as Pre-Closing Taxes, Parent shall not settle, adjust or otherwise resolve such Tax Claim without the consent of the Securityholder Representative (such consent not to be unreasonably withheld conditioned or delayed).
- (d) Notwithstanding anything herein to the contrary, to the extent requested by Parent, the parties and the Company Securityholders shall cooperate to cause the Company to make an election under Section 6226 of the Code with respect to any applicable Tax Claim. The parties agree that, with respect to all Tax Claims relating solely to taxable periods ending on or prior to the Closing Date with respect to which the Company was classified as a partnership for U.S. federal and/or applicable state income tax purposes, the parties shall cooperate to designate a "partnership representative" within the meaning of Section 6223 of the Code for the Company as chosen by Parent.
- 8.4 <u>Transfer Taxes</u>. Any transfer, stamp, documentary, sales, use, registration, VAT and other similar Taxes (including all applicable real estate transfer Taxes) incurred in connection with this Agreement and the transactions contemplated hereby ("**Transfer Taxes**") will be borne fifty percent (50%) by the Company Securityholders and fifty percent (50%) by Parent. The Securityholder Representative and the Company Securityholders agree to file or cause to be filed in a timely manner all necessary documents (including all Tax Returns) with respect to all such amounts for which the Company Securityholders are so liable. The Securityholder Representative shall provide Parent with evidence satisfactory to Parent that such Transfer Taxes have been paid by the Company Securityholders.
- 8.5 Refunds. The Company Securityholders shall be entitled to any refunds (including any interest paid thereon) of Taxes of the Company attributable to Pre-Closing Tax Periods in excess of \$25,000 that are received within two years of the Closing Date (the "Tax Refunds"). For the avoidance of doubt, Tax Refunds shall not include: (1) Tax Refunds that arise as a result of a Tax paid by the Company after the Closing Date and that is not a Tax for which the Parent has been fully indemnified or are Taxes included in the calculations of the Merger Consideration, (2) Tax Refunds that are required to be paid to a third-party pursuant to a contract in place as of the Closing, (3) Tax Refunds that were taken into account as an asset in calculating the Merger Consideration, or (4) Tax Refunds that are attributable to a carryback of an item from a taxable period (or portion thereof, in the case of a Straddle Period) beginning after the Closing Date Parent and its Affiliates shall cooperate with the Securityholder Representative in obtaining any refund to which the Company Securityholders are entitled under this Section 8.5. Parent shall promptly forward to or reimburse the Company Securityholders for any such refunds (including any interest paid thereon) due to the Company Securityholders after receipt thereof (or credit of such refund against other Taxes). In the event Parent, the Company or any of their Affiliates is required to return to a taxing authority any such Tax Refunds the Company Securityholders received pursuant to this Section 8.5, the Company Securityholders

shall deliver and pay over to Parent by wire transfer of immediately available funds such Tax Refunds.

8.6 <u>Post-Closing Actions</u>. To the extent such action would give rise to (i) any additional Taxable income passed through to the Company Securityholders under applicable Law or (ii) any Taxes subject to indemnification as Pre-Closing Taxes, Parent shall not (and shall not cause or permit any Company or any of its Subsidiaries to) amend, re-file or otherwise modify any Tax Return or Tax election, initiate any voluntary disclosure, or agree to the waiver or any extension of the statute of limitations, in each case, for the Company or any of its Subsidiaries with respect to any Taxable period (or portion thereof) ending on or before the Closing Date, without the prior written consent of the Securityholder Representative (which consent shall not be unreasonably withheld or delayed).

### **ARTICLE 9**

#### **MISCELLANEOUS**

- 9.1 Governing Law; Jurisdiction; Venue. This Agreement shall be governed and construed in accordance with the internal Laws of the State of Delaware, irrespective of its conflicts of law principles and any other Law that would cause the application of the Laws (including the statute of limitations) of any jurisdiction other than the State of Delaware. The parties hereto hereby irrevocably submit to the exclusive jurisdiction of the Court of Chancery of the State of Delaware, New Castle County (unless the Federal courts have exclusive jurisdiction over the matter, in which case the United States District Court for the District of Delaware) for any action, suit or proceeding arising out of or relating to this Agreement and of any of the documents referred to in this Agreement, and in respect of the transactions contemplated hereby and thereby (including resolution of disputes under Section 7.5), and hereby irrevocably waive, and agree not to assert, as a defense in any action, suit or proceeding arising out of or relating to this Agreement and of any of the documents referred to in this Agreement, and in respect of the transactions contemplated hereby and thereby, that it is not subject thereto or that such action, suit or proceeding may not be brought or is not maintainable in said courts or that the venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such courts, and the parties hereto irrevocably agree that all claims with respect to such action, suit or proceeding shall be heard and determined only in the Court of Chancery of the State of Delaware, New Castle County or the United States District Court for the District of Delaware (and no such action, suit or proceeding shall be brought in any other court); provided, that a judgment rendered by such court may be enforced in any court having competent jurisdiction. The parties hereby consent to and grant any such court jurisdiction over the Person of such parties and over the subject matter of such dispute and agree that mailing of process or other papers in connection with any such action, suit or proceeding in the manner provided in <u>Section 9.8</u> or in such other manner as may be permitted by Law, shall be valid and sufficient service thereof. With respect to any particular action, suit or proceeding, venue shall lie solely in the courts of the State of Delaware in accordance with this Section 9.1, and the parties hereby agrees to waive any objection to such venue of any action, suit or proceeding arising out of or relating to this Agreement and of any of the documents referred to in this Agreement, and in respect of the transactions contemplated hereby and thereby.
- 9.2 <u>Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations under this Agreement may be assigned or delegated, in whole or in part, by operation of law or otherwise by any of the parties hereto without the prior written consent of the other parties hereto, and any such assignment without such prior written consent shall be null and void, except that Parent may, upon prior written notice to the Securityholder Representative, assign this

Agreement to any direct or indirect wholly owned Subsidiary of Parent or to any Person who acquires all or substantially all of the assets of Parent or a majority of the outstanding voting securities of Parent (whether by merger, consolidation, share purchase or otherwise) without the prior consent of any other party hereto; <u>provided</u> that no such assignment shall relieve any Person of any liability or obligation.

- 9.3 <u>Severability</u>. If any provision of this Agreement, or the application thereof, shall for any reason and to any extent be declared invalid, illegal or unenforceable, then the remainder of this Agreement shall remain in full force and effect and the application of such provision to other persons or circumstances shall be interpreted so as reasonably to effect the intent of the parties hereto. The parties further agree to replace such void or unenforceable provision of this Agreement with a valid and enforceable provision that shall achieve, to the maximum extent permitted by Law, the original economic, business and other purposes of the void or unenforceable provision.
- 9.4 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts (including via facsimile, .pdf or other electronic means), each of which shall be an original as regards any party whose signature appears thereon and all of which together shall constitute one and the same instrument. This Agreement shall become binding when one or more counterparts hereof, individually or taken together, shall bear the signatures of all parties reflected hereon as signatories.

### 9.5 Other Remedies.

- (a) Except as otherwise expressly provided herein (including Section 7.12), any and all remedies herein expressly conferred upon a party hereunder shall be deemed cumulative with and not exclusive of any other remedy conferred hereby or by Law or equity on such party, and the exercise of any one remedy shall not preclude the exercise of any other.
- (b) The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached, and that money damages would not be an adequate remedy for any such damage. The parties hereto agree that the parties shall be entitled to equitable relief by way of an injunction or injunctions, specific performance or otherwise (without posting a bond or other security) to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any State having jurisdiction, this being in addition to any other remedy to which they are entitled at law or in equity. Each of the parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that (i) the party seeking such remedy has an adequate remedy at law or (ii) an award of specific performance is not an appropriate remedy for any reason at law or equity.

#### 9.6 Amendments and Waivers.

- (a) This Agreement may be amended by the parties hereto by an instrument in writing signed on behalf of each of the parties hereto at any time before or after any approval hereof by the members of the Company and Merger Sub I.
- (b) At any time after the Effective Time, the Securityholder Representative (in the case of Parent, the Surviving Company or the Surviving Entity) or Parent (in the case of the Securityholder Representative), may, to the extent not legally prohibited, (i) extend the time for the performance of any of the obligations or other acts of the other party hereunder, (ii) waive

any inaccuracies in the representations and warranties of the other party contained herein or in any document delivered pursuant hereto and (iii) waive compliance by the other party with any of the agreements or conditions contained herein. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in a written instrument signed on behalf of the party against which such waiver or extension is to be enforced. Without limiting the generality or effect of the preceding sentence, no delay in exercising any right under this Agreement shall constitute a waiver of such right, and no waiver of any breach or default shall be deemed a waiver of any other breach or default of the same or any other provision in this Agreement. No single or partial exercise by a party hereto of its rights under this Agreement will be deemed to preclude any other or further exercise of such party's rights under this Agreement.

- 9.7 <u>Expenses</u>. Except as otherwise expressly provided herein, whether or not the Mergers are successfully consummated, each party shall bear its own respective legal, accounting, and financial advisory fees and other expenses incurred with respect to this Agreement, the Mergers and the transactions contemplated hereby, it being the intention of the parties that if the Mergers are consummated, the Unpaid Transaction Expenses be taken into account in calculating the Adjusted Cash Consideration as set forth herein and, to the extent not so taken into account, shall be Losses for which Parent is entitled to be indemnified, compensated and reimbursed for as and to the extent provided under <u>Article 7</u>. Notwithstanding anything to the contrary, Parent shall pay and be responsible for 50% of the costs and expenses owed to the Escrow Agent in connection with the transactions contemplated hereby.
- 9.8 <u>Notices</u>. All notices and other communications required or permitted under this Agreement shall be in writing and shall be either hand delivered in person, sent by electronic mail, sent by certified or registered first-class mail, postage pre-paid, or sent by nationally recognized express overnight service. Such notices and other communications shall be effective and be deemed delivered and received (a) upon receipt if hand delivered, (b) on the date of transmission (provided no "bounceback" or similar notice of non-delivery is received) if transmitted by electronic mail by 5:00 p.m. (Pacific time) on a Business Day, otherwise on the next Business Day after transmission, (c) three (3) Business Days after mailing if sent by mail, and (d) one (1) Business Day after dispatch if sent by a reputable overnight courier, to the following addresses, or such other addresses as any party may notify the other parties in accordance with this Section 9.8:

If to Parent or either Merger Sub or, following the Effective Time, the Company:

Forge Global Holdings, Inc.
4 Embarcadero Center
San Francisco, CA 94111
Attention: James Nevin; Legal Department
E-Mail: [\*\*\*]

with a copy (which shall not constitute notice) to:

Goodwin Procter LLP 601 Marshall Street Redwood City, California 94063 Attention: Stuart Ogg; Matthew B. Baudler E-Mail: [\*\*\*]

### If, prior to the Effective Time, to the Company:

Accuidity, LLC
Two Newton Executive Park, Ste 104
2227 Washington St.
Newton, MA 02462
Attn: Mark DeNatale and Vincent Gubitosi
Email: [\*\*\*]

with a copy (which shall not constitute notice) to:

Wilmer Cutler Pickering Hale and Dorr LLP 2100 Pennsylvania Avenue NW Washington DC 20037 Attention: Stephanie C. Evans E-Mail: [\*\*\*]

If to the Securityholder Representative:

Kostka LLC 223 Bayville Road Locust Valley NY 11560 Attn: Mark DeNatale and Vincent Gubitosi Email: [\*\*\*]

with a copy (which shall not constitute notice) to:

Wilmer Cutler Pickering Hale and Dorr LLP 2100 Pennsylvania Avenue NW Washington DC 20037 Attention: Stephanie C. Evans E-Mail: [\*\*\*]

9.9 WAIVER OF JURY TRIAL. EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY HERETO CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) SUCH PARTY UNDERSTANDS AND HAS CONSIDERED THE

IMPLICATIONS OF THIS WAIVER, (C) SUCH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE WAIVERS AND CERTIFICATIONS IN THIS <u>SECTION 9.9</u>.

Interpretation; Rules of Construction. The terms "hereof," "herein" and similar terms refer to this Agreement as a whole (including the Company Disclosure Schedule, Parent Disclosure Schedule and the Annexes, Exhibits and Schedules hereto), and when a reference is made in this Agreement to Annexes, Exhibits, Schedules, Sections or Articles, such reference shall be to an Annex, Exhibit or Schedule to, or Section or Article of, this Agreement, respectively, unless otherwise indicated. The words "include," "includes" and "including" when used herein shall be deemed in each case to be followed by the words "without limitation." The words "either" and "or" are not exclusive and the words "hereof," "hereto," "hereby," "herein" and "hereunder" and words of similar import when used in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement. When a reference is made to a specific law, act or statute, such reference shall include any regulations promulgated thereunder. Any agreement, instrument or statute defined or referred to herein means such agreement, instrument, or statute, in each case, as from time to time amended, modified or supplemented (in the case of agreements or instruments, if permitted under this Agreement), including in the case of statutes by succession or comparable successor statutes; provided, that any reference to any agreement or instrument on the Company Disclosure Schedule, Parent Disclosure Schedule or on any Schedule to this Agreement shall not refer to any amendment, modification or supplement thereto unless expressly set forth in the Company Disclosure Schedule, Parent Disclosure Schedule or such other Schedule. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. The terms defined herein have the meanings assigned to them in this Agreement and include plural as well as the singular. Pronouns of either gender or neuter shall include, as appropriate, the other pronoun forms. Unless stated otherwise, the terms "dollars" and "\$" shall mean United States dollars. Any action required by the terms hereof to be taken on a specific day that is not a Business Day shall instead be required to be taken on the next succeeding Business Day, and if the last day of a time period specified herein is a non-Business Day, such period shall be deemed to end on the next succeeding Business Day. Any reference to any document or information having been "made available" by the Company shall only include any such document or information that has been posted in the Virtual Data Room by 5:00 p.m. Pacific Time on the Business Day prior to the execution of this Agreement and with respect to which the Company has not terminated the availability thereto by Parent and its Representatives prior to the Closing. The parties hereto agree that they have been represented by legal counsel during the negotiation and execution of this Agreement and, therefore, waive the application of any Law or rule of construction providing that ambiguities in an agreement or other document shall be construed against the party drafting such agreement or document.

#### 9.11 Disclosure Schedule.

(a) The parties agree that any reference in a particular Section of the Company Disclosure Schedule shall only be deemed to be an exception to (or, as applicable, a disclosure for purposes of) (a) the representations and warranties of the Company that are contained in the corresponding Section of this Agreement and (b) any other representations and warranties of the Company that are contained in any other Section of Article 2, but only if the relevance of that reference as an exception to (or a disclosure for purposes of) such other representations and warranties is reasonably apparent on the face of a disclosure, without any independent knowledge of the subject matter thereof or the contents of any documents referenced therein.

- (b) The parties agree that any reference in a particular Section of the Parent Disclosure Schedule shall only be deemed to be an exception to (or, as applicable, a disclosure for purposes of) (a) the representations and warranties of Parent or the Merger Subs that are contained in the corresponding Section of this Agreement and (b) any other representations and warranties of the Parent and the Merger Subs that are contained in any other Section of Article 3, but only if the relevance of that reference as an exception to (or a disclosure for purposes of) such other representations and warranties is reasonably apparent on the face of a disclosure, without any independent knowledge of the subject matter thereof or the contents of any documents referenced therein.
- 9.12 Agreement Binding on the Parties; Third-Party Beneficiary Rights. This Agreement shall be binding upon and inure solely to the benefit of each party hereto and its successors and permitted assigns and nothing in this Agreement is intended to or shall confer upon any other Person any legal or equitable rights, benefits, or remedies of any nature whatsoever under or by reason of this Agreement; provided, that (a) the Indemnified Parties shall be third party beneficiaries of Article 7 and Article 8, (b) the Company Indemnified Parties shall be third party beneficiaries of this Agreement (provided that only the Securityholder Representative, in its capacity as the representative, agent and attorney-in-fact of the Securityholders, shall be entitled to enforce the rights granted pursuant to this clause (c)).
- 9.13 <u>Public Announcement</u>. Parent may issue a press release or other public announcements relating to the execution of this Agreement, the Mergers or the other transactions contemplated hereby; <u>provided</u>, that Parent shall consult with the Company in advance of issuing the any such press release or public announcement and will consider in good faith any comments of the Company. Notwithstanding the foregoing, neither the Company nor the Securityholder Representative shall, and each shall cause its respective Affiliates and representatives not to, issue any press releases or make any public announcements or disclosures relating to this Agreement, the Mergers or the other transactions contemplated hereby without the prior written consent of Parent.

### 9.14 Confidentiality.

- (a) The parties acknowledge that the Company and Parent previously executed the Confidentiality Agreement, which, except as provided in the proviso to this sentence, will continue in full force and effect in accordance with its terms until the Effective Time, at which time, and without further action by any party hereto, it shall terminate and be of no further force and effect; provided, that nothing in the Confidentiality Agreement shall be deemed to restrict Parent's or the Company's rights under Section 9.13.
- (b) Except as may be required by any applicable Law or otherwise required or requested by any Governmental Authority, the Securityholder Representative hereby agrees to hold this Agreement and the transactions contemplated hereby, and all confidential or proprietary information received by the Securityholder Representative in its capacity as the Securityholder Representative) with respect hereto or thereto or in connection herewith (including any confidential or proprietary information obtained with respect to any Claims), in confidence and not disclose the existence or terms hereof or any such confidential or proprietary information to any third party (other than the Securityholders or legal counsel or other advisors engaged by the Securityholder Representative, in each case solely to the extent necessary to perform its obligations hereunder and only if such Persons are subject to an obligation to keep such information confidential).

- 9.15 <u>Entire Agreement</u>. This Agreement, the Exhibits and Schedules hereto, the Company Ancillary Agreements, the Company Disclosure Schedule, the Parent Disclosure Schedule, the Parent Ancillary Agreements, the Confidentiality Agreement and the Merger Sub Ancillary Agreements constitute the entire understanding and agreement of the parties hereto with respect to the subject matter hereof and thereof and supersede all prior and contemporaneous agreements or understandings, inducements or conditions, express or implied, written or oral, between the parties with respect hereto or thereto. The express terms hereof control and supersede any course of performance or usage of the trade inconsistent with any of the terms hereof.
- 9.16 <u>Acknowledgement</u>. The Company acknowledges and agrees that, other than with respect to Losses arising out of Fraud, under no circumstances shall Parent's aggregate liability for Losses incurred by any Indemnifying Party arising out of any inaccuracies or breaches of Parent Fundamental Representations be an amount greater than the Merger Consideration that has been paid or issued to (or become payable or issuable to) Company Securityholders pursuant to <u>Section 1.4</u> at any such applicable time (valuing Parent Shares at the Reference Price).

### 9.17 <u>Waiver of Conflicts Regarding Representation; Non-Assertion of Attorney-Client Privilege.</u>

- (a) Effective as of the Effective Time, Parent hereby waives and agrees not to assert, and Parent agrees to cause the Surviving Company, the Surviving Entity and each Subsidiary of the Surviving Company or the Surviving Entity to waive and not to assert, any conflict of interest arising out of or relating to any representation after the Effective Time of the Securityholder Representative, any Securityholder, any of their respective Affiliates or any officer, employee, manager or director of the Securityholder Representative or Securityholder (any such Person, a "**Designated Person**") in any matter involving this Agreement or any agreement, certificate, instrument or other document executed or delivered pursuant to this Agreement or any transaction contemplated hereby or thereby (including any litigation, arbitration, mediation or other proceeding and including any matter regarding the negotiation, execution, performance or enforceability hereof or thereof) (each such matter, a "**Post-Closing Representation**") by Wilmer Cutler Pickering Hale and Dorr LLP ("WH"). Parent hereby acknowledges and agrees that WH has represented the Company prior to the Closing in connection with this Agreement or any agreement, certificate, instrument or other document executed or delivered pursuant to this Agreement or any transaction contemplated hereby or thereby (including the negotiation, execution or performance hereof or thereof) (the "Current Representation").
- (b) Effective as of the Effective Time, Parent hereby agrees not to control or assert, and Parent agrees to cause the Surviving Company, the Surviving Entity and each Subsidiary of the Surviving Company or any Surviving Entity not to control or assert, any attorney-client privilege, work product protection or other similar privilege or protection applicable to any communication between any legal counsel and any Designated Person or the Company or any of its Subsidiaries (including any officer, employee, manager or director of the Company or any of its Subsidiaries) during the Current Representation in connection with any Post-Closing Representation (a "Covered Communication"), and agrees not to access, use or rely on, and to cause the Surviving Company, the Surviving Entity and each Subsidiary of the Surviving Company or the Surviving Entity not to access, use or rely on, any such communications in each case in connection with any Post-Closing Representation, including in connection with a dispute with Parent, the Surviving Company, the Surviving Entity or any of their respective Affiliates (including, after the Effective Time, the Surviving Company, the Surviving Entity and each Subsidiary of the Surviving Company or the Surviving Entity), it

being the intention of the parties hereto that, notwithstanding anything to the contrary in this Agreement or the DLLCA, all rights of any Person under or with respect to such attorney-client privilege, work product protection or other similar privilege or protection, including the right to waive, assert and otherwise control such attorney-client privilege, work product protection or other similar privilege or protection, shall be (and are hereby) transferred to or retained by (as applicable), and vested solely in, such Designated Person (or the Securityholder Representative, in the case of any such rights of any Person other than a Designated Person)); provided, however, that any such privilege or protection that would be transferred to or retained by (as applicable), or vested solely in, the Company in accordance with the foregoing shall, upon the Closing, be transferred to or retained by (as applicable) and vested solely in the Securityholder Representative. Parent agrees to take the steps necessary, and to cause the Surviving Company, the Surviving Entity and each Subsidiary of the Surviving Company or the Surviving Entity to take the steps necessary, to ensure that any such attorney-client privilege, work product protection or similar privilege or protection shall survive the Closing and the Effective Time, remain in effect and be afforded the treatment described in the immediately preceding sentence. No access following the Effective Time by Parent or any of its Affiliates (including the Surviving Company, the Surviving Entity and any Subsidiary of the Surviving Company or the Surviving Entity) to any Covered Communication shall waive or otherwise alter the rights of any Designated Person with respect to any Covered Communication and none of Parent, the Surviving Company, the Surviving Entity or any Subsidiary of the Surviving Company or the Surviving Entity shall, and each shall cause its Affiliates not to, use any Covered Communication or the contents of any Covered Communication in any dispute with any Designated Person in any matter involving this Agreement or any Transaction Document or any transaction contemplated hereby or thereby (including any litigation, arbitration, mediation or other proceeding and including any matter regarding the negotiation, execution, performance or enforceability hereof or thereof).

[Signature Pages Follow]

In Witness Whereof, the parties hereto have executed this Agreement as of the date first above written.

## **PARENT**

FORGE GLOBAL HOLDINGS, INC.

By: /s/ James Nevin

Name: James Nevin

Title: Chief Financial Officer

## **MERGER SUB I**

MARGO MERGER SUB I, INC.

By: /s/ James Nevin

Name: James Nevin Title: Authorized Person

### **MERGER SUB II**

MARGO MERGER SUB II, INC.

By: /s/ James Nevin

Name: James Nevin Title: Authorized Person

[SIGNATURE PAGE TO AGREEMENT AND PLAN OF MERGER]

In Witness Whereof, the parties hereto have executed this Agreement as of the date first above written.

# **COMPANY**

ACCUIDITY, LLC

By: /s/ Mark DeNatale
Name: Mark DeNatale
Title: Member

[SIGNATURE PAGE TO AGREEMENT AND PLAN OF MERGER]

In Witness Whereof, the parties hereto have executed this Agreement as of the date first above written.

# SECURITYHOLDER REPRESENTATIVE

KOSTKA LLC

By: /s/ Vincent Gubitosi
Name: Vincent Gubitosi
Title: Co-president

[SIGNATURE PAGE TO AGREEMENT AND PLAN OF MERGER]

### **ANNEX A – CERTAIN DEFINITIONS**

For the purposes of this Agreement, the following capitalized terms shall have the meanings set forth below (which shall apply equally to both the singular and plural forms of such terms):

"Accelerator Fund GP" means DoD Accelerator Fund GP LLC, a Delaware limited liability company.

"Accredited Investor" means a Company Securityholder who either (i) completes, executes and delivers to the Company or Parent prior to the Closing an Investor Questionnaire certifying that such Company Securityholder is an "accredited investor" as set forth therein or (ii) is determined by Parent in its sole discretion prior to the Closing to be an "accredited investor" as such term is defined in Rule 501(a) of Regulation D promulgated under the Securities Act.

"Acquisition Proposal" means any agreement, offer, proposal or bona fide indication of interest for (in each case other than this Agreement or any other offer, proposal or indication of interest by Parent or any Affiliate of Parent): (i) any acquisition or purchase by any Person of any Company Securities or any merger, consolidation, business combination or similar transaction involving the Company or any of its securities, (ii) any sale, lease, exchange, transfer, license, or disposition of 20% or more of the total consolidation assets of the Company in any single transaction or series of transactions (other than the sale of products of the Company or its Subsidiaries in the ordinary course of business), (iii) any liquidation, dissolution, recapitalization or other significant corporate reorganization of the Company, or (iv) any other transaction outside of the ordinary course of business the consummation of which would impede, interfere with, prevent or delay, the consummation of the Mergers or the other transactions contemplated hereby.

"Action" means any action, demand, claim, order, suit, litigation, proceeding (including any civil, criminal, administrative or appellate proceeding), arbitration, hearing, audit, examination or investigation commenced, brought, conducted or heard by or before, any court or other Governmental Authority or any arbitrator or arbitration panel.

"Advisory Client" means any Person to which the Company provides investment advisory services pursuant to an Advisory Agreement.

"Adjusted Cash Consideration" means (i) the Cash Consideration <u>plus</u> (ii) the amount by which the Estimated Closing Cash is greater than the Cash Target Amount (it being understood that such amount shall be deemed to be "0" if the Cash Target Amount is greater than the Estimated Closing Cash), <u>plus</u> (iii) the Estimated Closing Net Working Capital Surplus, if any, <u>minus</u> (iv) the Estimated Closing Indebtedness <u>minus</u> (v) the Estimated Unpaid Transaction Expenses <u>minus</u> (vi) the Estimated Unpaid Pre-Closing Taxes, <u>minus</u> (vii) the Estimated Closing Net Working Capital Shortfall, if any.

# "Adjustment Holdback Amount" means \$[\*\*\*]

"Affiliate" means, with respect to any Person, any other Person that, directly or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, such Person, and the term "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities, by Contract or otherwise. References herein to Affiliates of Parent shall be deemed to include the Surviving Company following the Effective Time and the Surviving Entity following the Second Effective Time. Notwithstanding anything to the contrary, none of the Aggregator, the Accelerator Fund GP, or any Fund shall be deemed an Affiliate of the Company (or an Affiliate of any Subsidiary of the Company); provided that (A) each Fund shall be deemed an Affiliate of the Company for purposes of Sections 2.1, 2.6, 2.8, 2.9(d) (but only the last two sentences of Section 2.9(d)), 2.14, 2.15 (but only subsections (a) through (c), (f), (g), (j), (k) and (q) of Section 2.15), 2.17, 2.20(c), 2.21, 2.22, 2.23, 2.24, 2.25 and 7.2(a)(iv) and (B) the Accelerator Fund GP shall be deemed an Affiliate of the Company for purposes of Article 2 and Section 7.2(a)(iv).

"Aggregator" means Accuidity Aggregator, LLC, a Delaware limited liability company,

"Aggregator Pro Rata Share" means, as to each holder of an Aggregator Unit as of immediately prior to the Effective Time, the quotient obtained <u>by dividing</u> (i) the aggregate number of Aggregator Units held by such holder as of immediately prior to the Effective Time <u>by</u> (ii) the aggregate number of Aggregator Units held by all holders as of immediately prior to the Effective Time. The aggregator Pro Rata Shares of all holders of Aggregator Units shall equal 100%.

"Aggregator Unit" means a Unit (as defined in that certain Operating Agreement of the Aggregator, dated January 1, 2025).

"AI Technologies" means any and all artificial intelligence technologies and tools, including all generative artificial intelligence tools and technologies and all other deep learning, machine learning, and other artificial intelligence technologies, including (a) proprietary algorithms, software, or systems that make use of or employ neural networks, statistical learning algorithms (such as linear and logistic regression, support vector machines, random forests, or k-means clustering), or reinforcement learning, and (b) proprietary artificial intelligence embedded in related hardware or equipment.

"Balance Sheet Date" means December 31, 2024.

"Books and Records" means all files, documents, instruments, papers, books and records relating to the Company, the business, operations or condition of the Company, or the Company's properties and assets, including financial statements, payroll and human resources records, internal reports, Tax Returns and related work papers and letters from accountants, budgets, pricing guidelines, ledgers, journals, deeds, title policies, minute books, share

certificates and books, share transfer ledgers, Contracts, licenses, customer lists, computer files and programs (including data processing files and records), retrieval programs, operating data and plans, and environmental studies and plans.

"Business Day" means any day that is not a Saturday, Sunday or other day on which banks are required or authorized by Law to be closed in San Francisco, California or Wellesley, Massachusetts.

"Cash Consideration" means \$10,000,000.

"Cash Target Amount" means \$[\*\*\*]

"Charter Documents" means, with respect to an entity, the (a) articles of association, certificate of association, certificate of incorporation, certificate of formation, articles of organization, or certificate of limited partnership of such entity, (b) bylaws, limited liability company agreement, or limited partnership agreement of such entity, and (c) other equivalent or similar organizational documents of such entity.

"Claim" means a claim for indemnification, compensation or reimbursement for Losses under Article 7.

"Class B Unit Agreement" means, with respect to any Company Profits Interest Unit, the Class B Unit Agreement setting forth the terms of vesting and other terms and conditions applicable to such Company Profits Interest Unit.

"Closing Cash" means, without duplication and as of the Measurement Time, the aggregate amount of cash and cash equivalents of the Company and its Subsidiaries (on a consolidated basis); provided, that Closing Cash shall (i) be calculated net of (a) all Restricted Cash, (b) outstanding checks, draws, ACH debits and wire transfers and (c) any cash overdrafts and (ii) include any inbound wires in transit and checks and drafts deposited for the account of the Company or any of its Subsidiaries; provided, however, that Closing Cash shall be calculated by reducing amounts held outside the United States by the costs and Taxes that would be incurred in repatriating such amounts to the United States.

"Closing Company Net Working Capital" means an amount equal to (i) the current assets of the Company and its Subsidiaries (on a consolidated basis) minus (ii) the current liabilities of the Company and its Subsidiaries (on a consolidated basis), in each case calculated in accordance with the Agreed Accounting Principles as of the Measurement Time. For purposes of calculating the Net Working Capital, (x) the current assets of the Company and its Subsidiaries shall exclude all cash and cash equivalents and Tax assets, if any, and (y) the current liabilities of the Company and its Subsidiaries shall exclude all Indebtedness, all Unpaid Transaction Expenses, and all Tax Liabilities.

"Closing Employee Payment" means any cash payment triggered by or that becomes due as a result of the Closing (excluding, for the avoidance of doubt and notwithstanding

anything to the contrary, any payment contingent upon continued employment, or a change in employment status or employment conditions, following the Closing), whether due prior to, at or after the Closing owed pursuant to any management, employment, retention, bonus, change in control, paid-time off, severance, or other similar arrangement with any current or former director, officer, employee, independent contractor or any other service provider of the Company and/or its Subsidiaries and the employer portion of any associated employment, payroll or similar Tax (including any separation payment, contractual or otherwise, statutory severance payments or payments in lieu of notice, or any and all fees, costs and expenses incurred in connection with any COBRA continuation coverage, payable or related to any non-continuing employee); <u>provided</u>, <u>however</u>, that Parent Employee Obligations shall not be deemed to be Closing Employee Payments.

"Closing Indebtedness" means the total amount of the Indebtedness of the Company and its Subsidiaries (on a consolidated basis) as of the Measurement Time.

"Closing Net Working Capital Shortfall" means the amount by which the Closing Company Net Working Capital is less than the Target Company Net Working Capital; <u>provided</u>, that such amount shall equal \$0 if the Closing Company Net Working Capital is greater than the Target Company Net Working Capital.

"Closing Net Working Capital Surplus" means the amount by which the Closing Company Net Working Capital is greater than the Target Company Net Working Capital; <u>provided</u>, that such amount shall equal \$0 if the Closing Company Net Working Capital is less than the Target Company Net Working Capital.

"Closing Stock Consideration" means 1,150,000 shares of Parent Common Stock (as adjusted to appropriately reflect any stock split, reverse stock split, stock dividend, reorganization, reclassification, combination, recapitalization or other like change with respect to share of Parent Common Stock occurring after the date hereof and prior to the issuance of the Closing Stock Consideration).

"COBRA" means the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended.

"Code" means the Internal Revenue Code of 1986, as amended.

"Company AI Product" means any and all products and services of the Company previously offered, currently offered, or included in the Company's product roadmap that employ or make use of AI Technologies.

"Company Ancillary Agreement" means each agreement or document (other than this Agreement) that the Company is to enter into as a party thereto pursuant to this Agreement.

"Company Balance Sheet" means the unaudited consolidated balance sheet of the Company as of the Balance Sheet Date that is included in the Company Financial Statements.

"Company Benefit Arrangement" means each: (i) employee benefit plan within the meaning of Section 3(3) of ERISA whether or not subject to ERISA; (ii) stock option plan, stock purchase plan, other equity-based plan, bonus or incentive plan, severance pay plan, program or arrangement, deferred compensation arrangement or agreement, employment agreement, consulting agreement, compensation plan, program, agreement or arrangement, retention plan, change in control plan, program or arrangement, supplemental income arrangement, vacation or paid-time off plan, death, hospitalization, health and welfare and fringe benefit plan, retirement, postretirement or retiree welfare arrangement, educational or employee assistance plan, employee loan and all other employee benefit plans, agreements, and arrangements, not described in (i) above; and (iii) plan or arrangement providing compensation to employee and non-employee directors, in each case in which the Company or any ERISA Affiliate sponsors, contributes to, or provides benefits under or through such arrangement, or has any obligation to sponsor, contribute to or provide benefits under, for the benefit of any current or former employee, officer, director, independent contractor or other service provider of the Company (or their spouses, dependents, or beneficiaries) or with respect to which the Company has or may have any Liability or obligation.

"Company Business" means the business of the Company as presently conducted.

"Company Class A Unit" means the Class A Units of the Company as set forth in the Company LLC Agreement.

"Company Data" means all data, meta-data, or information (i) transmitted to the Company by users or customers of any Company's products or Company Web Site, or (ii) contained in any IT Systems or other databases of the Company (including any and all Proprietary Information, listings and other content displayed or distributed on or through any Company Offering or Company Software) and all other information, data and compilations thereof used by, or necessary to the business of, the Company.

"Company Disclosure Schedule" means the disclosure schedule of the Company dated as of the Agreement Date and delivered to Parent concurrently with the parties' execution of this Agreement.

"Company Financial Statements" means (i) the compiled statement of assets, liabilities and members' equity – cash basis of the Company dated as of December 31, 2024 and December 31, 2023 and the compiled statements of revenues, expenses and members' equity – cash basis of the Company for the years ended December 31, 2024 and December 31, 2023 and (ii) the statement of assets, liabilities and members' equity – cash basis of the Company dated as of May 31, 2025 and the statements of revenues, expenses and members' equity – cash basis of the Company for the five (5) month period ended May 31, 2025.

"Company Fundamental Representation Claim" means any Claim under <u>Section 7.2(a)(i)</u> with respect to any of the Company Fundamental Representations other than any such Claim involving Fraud.

"Company Fundamental Representations" means the representations and warranties of the Company set forth in Sections  $\underline{2.1}, \underline{2.2}, \underline{2.3}, \underline{2.4}, \underline{2.5(a)(i)}, \underline{2.7}$  and  $\underline{2.25}$ .

"Company General Representation Claim" means any Claim under <u>Section 7.2(a)(i)</u> with respect to any of the Company General Representations other than any such Claim involving Fraud.

"Company General Representations" means the representations and warranties of the Company set forth in <u>Article 2</u>, other than Company Fundamental Representations.

"Company Intellectual Property Right" means any Technology and Intellectual Property Right that is owned, purported to be owned, used, held for use, or practiced by, developed by or for, or exclusively licensed to, the Company, including any Intellectual Property Right incorporated into or otherwise used, held for use, practiced in connection, or developed for any Company Offering or Company Data.

"Company LLC Agreement" means that certain Third Amended and Restated Limited Liability Company Agreement of the Company, dated as of December 19, 2024.

"Company Material Contract" means (i) any Contract required to be listed on the Company Disclosure Schedule pursuant to Section 2.10 or Section 2.12(a) (whether or not so listed), (ii) any Inbound License or Outbound License and (iii) any Equity-Related Agreement.

"Company Members" means the Company Unitholders and the Company Profits Interest Unit Holders.

"Company Offering" means (i) any Work of Authorship, product or service offered, licensed, provided, sold, distributed, manufactured, marketed, imported for resale, made available or otherwise exploited by or for the Company since its inception, and any Work of Authorship, product or service under design or development (or already designed or developed) by or for the Company, including any version or release of the foregoing, together with any related documentation, materials, or information, (ii) each Company Web Site, including any platform or other Company Software used for each Company Web Site, and (iii) the Company Data.

"Company Profits Interest Unit Holder" means a holder as of immediately prior to the Effective Time of any Company Profits Interest Units.

"Company Profits Interest Units" means the Class B Units of the Company as set forth in the Company LLC Agreement.

"Company Securities" means (i) membership interests of the Company, (ii) securities of the Company convertible into or exchangeable for membership interests of the Company and (iii) options or other rights to acquire from the Company, or other obligation of the Company to

issue, any membership interests of the Company or securities convertible into or exchangeable for membership interests of the Company.

- "Company Securityholders" means, collectively, the Company Unitholders and Company Profits Interest Unit Holders.
- "Company Software" means all Software owned by or developed by or for the Company.
- "Company Technology" means any and all Technology owned or purported to be owned, used, held for use, practiced by or developed by or for the Company, including any Technology incorporated into or otherwise used, held for use or practiced in connection with, or developed for any Company Offering.
  - "Company Units" means the Company Class A Units.
  - "Company Unitholder" means a holder as of immediately prior to the Effective Time of Company Units.
- "Company Web Site" means any public or private web site owned, maintained, or operated at any time by or on behalf of the Company, including the web site at www. accuidity.com and any online service made available by the Company.
- "Confidentiality Agreement" means that certain Mutual Nondisclosure Agreement, by and between Parent and the Company, dated as of September 4, 2024.
- "Contaminant" means any "back door," "drop dead device," "time bomb," "Trojan horse," "virus," "corruptant," "worm," "malware," "spyware," "ransomware," or "trackware" (as such terms are commonly understood in the software industry) or any other code designed, intended to, or that does have any of the following functions: (i) disrupting, disabling, harming or otherwise impeding in any manner the operation of, or providing unauthorized access to, any computer, tablet computer, handheld device or other device, or (ii) damaging or destroying any data or file without a user's consent.
- "Continuing Employee" means each employee of the Company as of the Closing Date who continues employment with Parent or any of Parent's Affiliates immediately following the Closing Date.
- "Contract" means any contract, agreement, instrument, arrangement, commitment, understanding or undertaking (including leases, licenses, mortgages, notes, guarantees, sublicenses, subcontracts, purchase orders and sale orders), whether written or oral, in each case that is legally binding.
- "Copyleft License" means any license of Technology that provides, as a condition to the use, modification, or distribution of such licensed Technology, that such licensed Technology or

any other Technology that is incorporated into, derived from, based on, linked to, or used or distributed or made available with such licensed Technology, be licensed, distributed, or otherwise made available (i) in a form other than binary or object code (e.g., in source code form), (ii) under terms that permit redistribution, reverse engineering or creation of derivative works or other modification or (iii) without a license fee. "Copyleft License" includes the GNU General Public License, the GNU Library General Public License, the GNU Lesser General Public License, the Mozilla Public License, the Common Development and Distribution License, the Eclipse Public License, and any Creative Commons "sharealike" license.

"Copyright" means all rights associated with works of authorship, including exclusive exploitation rights, copyrights, copyrights registrations and applications therefor and all other rights corresponding thereto, including moral rights.

"Credentials" means any logins, passwords, IDs, user IDs, account IDs, tokens, entitlements, certificates, authorization codes or any other assigned data or code for access to or use of any Third-Party Platform.

"Encumbrance" means, with respect to any asset, any mortgage, deed of trust, lien, Tax lien, pledge, charge, security interest, title retention device, collateral assignment, adverse claim of title, exclusive license, option to obtain an exclusive license, or other encumbrance in respect of such asset (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, any restriction on the receipt of any legally permissible income or proceeds derived from any asset, any restriction on the legally permissible use of any asset and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).

"Enforceability Exceptions" means, with respect to any Contract, limitations on enforcement and other remedies imposed by or arising under or in connection with (i) applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws now or hereafter in effect relating to or affecting rights of creditors generally, and (ii) rules of law and general principles of equity, including those governing specific performance, injunctive relief and other equitable remedies.

"Environmental Law" means any Law relating to pollution or protection of the environment (including ambient air, surface water, ground water, land surface or subsurface strata), including any Law or regulation relating to any emission, discharge, release or threatened release of Materials of Environmental Concern or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Materials of Environmental Concern.

"Equity-Related Agreement" means any agreement to which the Company is a party or is bound relating to the voting (including voting trusts and proxies) or purchase, sale or transfer of any Company Security, including the Company LLC Agreement.

Annex	Α	_	8

- "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- "ERISA Affiliate" means each Subsidiary of the Company and any entity that is, or at any applicable time was, a member of a group that includes the Company described in Section 414(b), (c), (m) or (o) of the Code or been considered a single employer with the Company under Section 4001(b)(1) of ERISA.
  - "Escrow Agent" means Truist Bank.
  - "Escrow Agreement" means the Escrow Agreement in the form attached hereto as Exhibit H.
  - "Exchange Act" means the Securities Exchange Act of 1934, as amended.
  - "Expense Fund" means an amount in cash equal to \$100,000.
- "Expense Fund Sharing Percentage" means, with respect to each Company Unitholder and the Aggregator, the percentage set forth opposite such Person's name on Schedule A.
  - "Expiration Date" means the date that is fifteen (15) months following the Closing Date.
  - "FINRA" means the Financial Industry Regulatory Authority, Inc.
- "Foreign Government Official" means any officer or employee of a Governmental Authority or an Affiliate thereof (including any sovereign wealth fund) or of a public international organization, or any Person acting in an official capacity for or on behalf of any such Governmental Authority or an Affiliate thereof, or for or on behalf of any such public international organization, or any political party, party official, or candidate thereof.
  - "Founder Group" means the [\*\*\*].
  - "Founders" means [\*\*\*].
- "Fraud" means common law fraud (including the element of scienter) under Delaware law (but excluding constructive fraud or equitable fraud).
  - "GAAP" means United States generally accepted accounting principles as in effect for the applicable period or date.
  - "General Representation Cap" means \$2,000,000.
- "Governing Body" means the board of directors, board of managers (or management committee), or similar body (or concept) of any entity.
- "Governmental Authority" means any (i) multinational or supranational body exercising legislative, judicial, taxing or regulatory powers, (ii) nation, state, commonwealth,

province, territory, county, municipality, district or other jurisdiction of any nature, (iii) federal, state, local, municipal, foreign or other government or (iv) governmental or quasi-governmental authority of any nature (including any governmental division, department, agency, bureau, commission, instrumentality, official, organization, unit, body, subdivision, court, arbitrator or other tribunal, in each case exercising legislative, judicial, taxing or regulatory powers), and includes, for the avoidance of doubt, FINRA.

"Immediate Family" means, with respect to any individual, such individual's spouse or domestic partner, parents, grandparents, children, grandchildren, and siblings, including adoptive relationships and relationships through marriage, or any other relative of such individual that shares such individual's home.

"Indebtedness" means, without duplication and as determined for the Company and its Subsidiaries on a consolidated basis, (i) all obligations (including the principal amount thereof and the amount of accrued and unpaid interest thereon) of the Company and its Subsidiaries, whether or not represented by bonds, debentures, notes or other securities, for the repayment of money borrowed, whether owing to banks, financial institutions, on equipment leases or otherwise, (ii) all deferred indebtedness of the Company and its Subsidiaries for the payment of the purchase price of property or assets purchased (other than accounts payable incurred in the Ordinary Course of Business), (iii) all obligations of the Company or any of its Subsidiaries to pay rent or other payment amounts under a lease which is required to be classified as a capital lease or a liability on the face of a balance sheet prepared in accordance with the Agreed Accounting Principles, (iv) all outstanding reimbursement obligations of the Company or any of its Subsidiaries with respect to letters of credit, bankers' acceptances or similar facilities, (v) all obligations of the Company or any of its Subsidiaries under any interest rate swap agreement, forward rate agreement, interest rate cap or collar agreement or other financial agreement or arrangement entered into for the purpose of limiting or managing interest rate risks, (vi) all obligations secured by any Encumbrance (other than a Permitted Encumbrance) existing on property owned by the Company or any of its Subsidiaries, (vii) all unpaid Wage Obligations, (viii) all deferred revenue of the Company and its Subsidiaries (which deferred revenue shall exclude management fees paid in advance), (ix) all premiums, penalties, fees, expenses, breakage costs and change of control payments required to be paid in respect of any of the foregoing on payment or prepayment (regardless if any of such are actually paid), as a result of the consummation of the Mergers or any of the other transactions contemplated hereby, and (x) all guaranties, endorsements, assumptions and other contingent obligations of the Company or any of its Subsidiaries in respect of, or to purchase or to otherwise acquire, any of the obligations owed by third parties of the type described in any of the clauses (i) through (ix). For the avoidance of doubt and notwithstanding anything to the contrary, it is understood that this definition shall not include any Transaction Expenses. For the avoidance of doubt, Indebtedness shall be calculated in accordance with the Agreed Accounting Principles, consistent with the Closing Statement attached hereto as Exhibit G.

"Indemnifying Parties" means each Company Securityholder and each holder of Aggregator Units as of immediately prior to the Effective Time.

"Indemnity Escrow Amount" means \$2,250,000.

"Intellectual Property License" means any license, sublicense, right, covenant, non-assertion or similar covenant, permission, immunity, consent, release or waiver under or with respect to any Intellectual Property Rights or Technology.

"Intellectual Property Right" means any or all of the following and all rights in, arising out of, or associated therewith, in each case, throughout the world: (i) Patent, (ii) Copyright, (iii) other right with respect to Software, including any registration of such right or any application to register such right, (iv) industrial design right or registration of such right and any application to register such right, (v) right with respect to any Mark, and any registration for any Mark and any application to register any Mark, along with all goodwill associated with each of the foregoing, (vi) right with respect to all World Wide Web addresses, social media identifiers, uniform resource locators ("URLs") and any Domain Name, including any registration for any Domain Name, along with all goodwill associated with each of the foregoing, (vii) right with respect to any Proprietary Information, including any right to limit the use or disclosure of Proprietary Information by any Person, (viii) right with respect to any Database, including any registration of such right and any application to register such right, (ix) right of publicity and personality, including any right with respect to use of a Person's name, signature, likeness, image, photograph, voice, identity, personality, and biographical and personal information and materials, (x) moral right, (xi) renewal, reissue, reversion, reexamination, or extension of any of the foregoing, and (xii) any right equivalent or similar to any of the foregoing.

"IRS" means the United States Internal Revenue Service.

"IT Systems" means any information technology and computer system (including Software, information technology and telecommunication hardware, network and other equipment) relating to the transmission, storage, maintenance, organization, presentation, generation, processing or analysis of data and information and any support, disaster recovery and online service whether or not in electronic format, used in or necessary to the conduct of the Company Business.

"Knowledge" means, with respect to any Indemnifying Party that is an entity, the actual knowledge of the executives of such Person directly responsible for the investment in the Company.

"Knowledge of the Company" or "Company's Knowledge" (or similar term) means the knowledge of a particular fact, circumstance, event or other matter in question of any of the Persons listed on Schedule 1.1(b) (collectively, the "Company Entity Representatives"). Any such Company Entity Representative will be deemed to have knowledge of a particular fact, circumstance, event or other matter if such Company Entity Representative would reasonably be expected to have knowledge thereof after (i) reasonable inquiry of such Person's direct reports who would reasonably be expected to have knowledge of the subject matter in question and (ii) conducting a due and diligent inquiry of such Person's books, records and email accounts.

"Knowledge of Parent" or "Parent's Knowledge" (or similar term) means the knowledge of a particular fact, circumstance, event or other matter in question of any of the Persons listed on Schedule 1.1(c) (collectively, the "Parent Entity Representatives"). Any such Parent Entity Representative will be deemed to have knowledge of a particular fact, circumstance, event or other matter if such Parent Entity Representative would reasonably be expected to have knowledge thereof after (i) reasonable inquiry of such Person's direct reports who would reasonably be expected to have knowledge of the subject matter in question and (ii) conducting a due and diligent inquiry of such Person's books, records and email accounts.

"Law" means any foreign, federal, state, local or municipal law, statute, ordinance, directive, edict, regulation, standard, or rule, any order, ruling, writ, injunction, award, judgment or decree (and any regulations promulgated thereunder), and any other legislative measure or decision having the force of law, treaty, convention or other agreement between states, or between states and supranational bodies, rule of common law, and equity and any civil or other code of any Governmental Authority.

"Liability" means any debt, duty, Tax, obligation or liability of any kind or nature (including any unknown, undisclosed, unmatured, unaccrued, unasserted, contingent, indirect, conditional, implied, vicarious, derivative, joint, several or secondary liability), regardless of whether such debt, duty, Tax, obligation or liability would be required to be disclosed on a balance sheet prepared in accordance with U.S. GAAP and regardless of whether such debt, duty, liability, Tax or obligation is immediately due and payable.

"Losses" means any and all liabilities, judgments, settlements, assessments, losses, damages, interest, fines, penalties, costs, Taxes, and expenses of any kind or nature, including reasonable and documented out-of-pocket legal, accounting and other costs and expenses of professionals incurred in connection with investigating, defending, settling or otherwise satisfying any of the foregoing, and in seeking and enforcing rights to indemnification, compensation and reimbursement hereunder.

"Mark" means all trade names, logos, common law trademarks and service marks, trademark and service mark registrations and applications therefor and all goodwill associated therewith.

"Material Adverse Effect" when used in connection with an entity means any change, event, circumstance, condition or effect that is, or would reasonably be expected to be, individually or in the aggregate with other changes, events, circumstances, conditions or effects, materially adverse to (a) the financial condition, assets or liabilities (such assets and liabilities being taken together), business, operations or results of operations of such entity and its Subsidiaries, taken as a whole or (b) the ability of such entity to enter into this Agreement or to timely consummate the transactions contemplated hereby or thereby; provided, however, that, in the case of clause (a), none of the following shall be deemed to constitute, or be taken into account in determining whether there has been, or would reasonably expected to be, a Material Adverse Effect: (i) general economic conditions (or changes in such conditions) in the United

States or any other jurisdiction, or conditions (or change in such conditions) in the securities markets, credit markets, currency markets or other financial markets in the United States or any other country or region in the world, including (A) changes in interest rates in the United States or any other country or region in the world and changes in exchange rates for the currencies of any countries and (B) any suspension of trading in securities (whether equity, debt, derivative or hybrid securities) generally on any securities exchange or over-the-counter market operating in the United States or any other country or region in the world, (ii) conditions (or changes in conditions) affecting the industry generally in which such entity and its Subsidiaries operate, (iii) the existence, occurrence or continuation of any earthquakes, floods, hurricanes, tropical storms, wildfires, other natural disasters, or any pandemic, epidemic or disease outbreak, (iv) political conditions (or changes in such conditions) in the United States or any other jurisdiction, the outbreak or escalation of war, hostilities or terrorist activities, either in the United States or any other jurisdiction, (v) changes in Law or GAAP, or the interpretation thereof. (vi) any change in, or failure by the Company and its Subsidiaries to meet internal estimates, predictions, projections or forecasts, including as provided to another party or its Representatives (provided, that the underlying causes giving rise or contributing to any such failure shall not be excluded if not otherwise excluded by the foregoing clauses (i) through (v) or the following clauses (vii) or (viii)), (vii) any action taken or failure to take action, in each case which Parent (when this definition is used in connection with the Company or any of its Subsidiaries) or the Company (when this definition is used in connection with Parent or any of its Subsidiaries) has expressly requested or (viii) the pendency or announcement of the transactions contemplated by this Agreement or that certain Term Sheet, dated April 16, 2025, among Parent, the Company and the Founders, including actions of clients or vendors or losses of employees, except, in the case of each of the foregoing clauses (i) through (v), to the extent such changes disproportionately adversely affect such entity and its Subsidiaries, taken as a whole, as compared to other Persons or businesses that operate in the industry in which such entity and its Subsidiaries operate.

"Materials of Environmental Concern" means any chemical, pollutant, contaminant, waste, toxic substance, petroleum or petroleum product or any other substance that is currently regulated by an Environmental Law or that is otherwise a danger to health, reproduction or the environment.

"Measurement Time" means 12:01 a.m. local time in New York on the Closing Date; <u>provided</u>, that for Tax purposes the Measurement Time shall be the end of the day on the Closing Date.

"Member Approval" means Company Members holding more than 66.67% of the total outstanding Class A Units.

"Merger Consideration" means the consideration that is payable and/or issuable to Company Securityholders in respect of units of Company Class A Units and Company Profits Interest Units pursuant to Section 1.4.

"Merger Sub Ancillary Agreements" means, with respect to a Merger Sub, each agreement or document (other than this Agreement) that such Merger Subs is to enter into as a party thereto pursuant to this Agreement.

"Merger Sub I Units" means the units of the Merger Sub I.

"Multiemployer Plan" has the meaning set forth in Section 3(37) of ERISA.

"Non-Negotiated Vendor Contract" means a Contract that meets all of the following conditions: (i) such Contract grants to the Company a non-exclusive license to download or use generally commercially available, non-customized Software or a non-exclusive right to access and use generally commercially available non-customized functionality or services on a hosted or "software-as-a-service" basis (and does not include any other Intellectual Property Licenses), (ii) such Contract is a non-negotiable "shrink-wrap" or "click-through" Contract, (iii) such Contract does not grant any right by the Company that survives termination or expiration of such Contract, (iv) the Software is not included, incorporated or embedded in, linked to, combined, distributed or made available with, any Company Software that is distributed to or made available for use by customers, clients, or end users of any Company Offering, (v) such Contract does not require the Company to pay any license fee, subscription fee, service fee or other amount except for a one-time license fee of no more than \$10,000 or ongoing subscription or service fees of no more than \$10,000 per year, and (vi) such Contract is not a license for Open Source Software.

"OpCo" means Forge Global, Inc., a Delaware corporation and direct wholly owned subsidiary of Parent.

"Open Source Software" means any Software or other Technology that is distributed as "open source software" or "free software" or is otherwise publicly distributed or made generally available in source code or equivalent form under terms that permit modification and redistribution of such software or Technology. Open Source Software includes any software under a similar licensing or distribution model (including but not limited to the GNU General Public License (GPL), GNU Lesser General Public License (LGPL), Mozilla Public License (MPL), BSD licenses, the Artistic License, the Netscape Public License, the Sun Community Source License (SCSL), the Sun Industry Standards License (SISL) and the Apache License).

"Ordinary Course of Business" means a course of business that is in the ordinary course of business and consistent with its past practices, including with respect to frequency and amounts.

"Parent Ancillary Agreements" means each agreement or document (other than this Agreement) that Parent is to enter into as a party thereto pursuant to this Agreement.

"Parent Common Stock" means the common stock of Parent, par value \$0.0001 per share.

"Parent Disclosure Schedule" means the disclosure schedule of Parent and the Merger Subs as of the Agreement Date and delivered to the Company concurrently with the parties' execution of this Agreement.

"Parent Employee Obligation" means (i) any payment, benefit or other obligation to be made by Parent (or any of its Affiliates) to any Continuing Employee pursuant to the terms of an Employment Agreement or offer letter entered into by and between Parent (or any of its Affiliates) and such Continuing Employee, (ii) fifty (50%) percent of any cash severance or separation payment that becomes due to any non-continuing employee or independent contractor as a result of such employee or independent contractor not receiving an offer of employment from or consulting agreement with Parent (or one of its Affiliates) and (iii) the employer portion of any employment, payroll or similar Tax associated with the amounts described in the preceding clauses (i) or (ii).

"Parent ESPP" means Parent's 2022 Employee Stock Purchase Plan.

"Parent Fundamental Representations" means the representations and warranties of Parent and the Merger Subs set forth in <u>Sections 3.1</u> (excluding the last two sentences thereof), <u>3.2</u>, and <u>3.5</u>.

"Parent General Representations" means the representations and warranties of Parent and the Merger Subs set forth in Article 3, other than the Parent Fundamental Representations.

"Parent Material Contract" means each Contract filed as an exhibit or incorporated by reference to the Parent SEC Report.

"Parent PSU" means any restricted stock unit subject to performance-based vesting with respect to any shares of Parent Common Stock granted under any Parent Stock Plan or otherwise.

"Parent RSU" means any restricted stock unit subject solely to time-based vesting with respect to any shares of Parent Common Stock granted under any Parent Stock Plan or otherwise.

"Parent Shares" means shares of Parent Common Stock issuable pursuant to this Agreement.

"Parent Stock Option" means an option to purchase shares of Parent Common Stock granted under any Parent Stock Plan or otherwise.

"Parent Stock Plan" means any stock incentive or equity-related plan of Parent.

"Parent Warrant" means any warrant to purchase shares of capital stock of Parent issued by Parent.

"Patent" means all patents and utility models and applications therefor and all reissues, divisions, re-examinations, renewals, extensions, provisionals, continuations and continuations-

in-part thereof, and equivalent or similar rights anywhere in the world in inventions and discoveries including invention disclosures.

"Permitted Encumbrance" means (i) any statutory lien for Taxes (a) not yet due or delinquent or (b) the validity or amount of which is being contested in good faith by appropriate proceedings; provided, that in the case of clause (b), adequate reserves in accordance with the Agreed Accounting Principles have been established therefor on a basis consistent with prior periods and are reflected on the Company Financial Statements; (ii) any mechanics', carriers', workers', repairers' or other similar Encumbrance; (iii) any pledge, deposit or other Encumbrance securing the performance of bids, trade contracts, leases or statutory obligations; (iv) any Encumbrance arising under workers' compensation, unemployment insurance or other social security, retirement or similar legislation; (v) with respect to any real property leased by the Company (a) any Encumbrance on leases, subleases, easements, licenses, rights of use, rights to access and rights of way arising therefrom or benefiting or created by any superior estate, right or interest, (b) any Encumbrance that would be set forth in any title policies, endorsements, title commitments, title certificates and/or title reports and any zoning, entitlement, conservation restriction and other land use and environmental regulations by Governmental Authorities, and (c) any minor encroachment; provided, however, that none of the foregoing Encumbrances or encroachments described in clause (v) could, individually or in the aggregate, impair, in any material respect, the continued use and operation of the property to which they relate in the Company Business; and (vi) any non-exclusive licenses of Intellectual Property Rights in the Ordinary Course of Business.

"Person" means any individual, corporation, company, limited liability company, partnership, limited partnership, limited liability partnership, trust, estate, proprietorship, joint venture, association, organization, or other entity of any kind or nature or any Governmental Authority.

"Personal Information" means, in addition to all information defined or described by the Company as "personal data," "personal information," "personally identifiable information," "PII," "PHI" (as that term is defined under HIPAA) or any similar term in the Company's privacy policies or other public-facing statement, any information that is subject to any Privacy Law or regarding or capable of being associated with an individual consumer or device, including: (i) information that identifies, could be used to identify (alone or in combination with other information) or is otherwise identifiable with an individual or a device, including name, physical address, telephone number, email address, financial account number, government-issued identifier (including Social Security number and driver's license number), medical, health or insurance information, gender, date of birth, educational or employment information, any religious or political view or affiliation, marital or other status, photograph, face geometry, or biometric information, and any other data used or intended to be used to identify, contact or precisely locate an individual, (ii) any data regarding any activity of an individual online or on a mobile device or other application (e.g., any search conducted, web page or content visited or viewed), whether or not such information is associated with an identifiable individual, and (iii) any Internet Protocol address or other persistent identifier. Personal Information may relate to

any individual, including any user of any Internet or device application who views or interacts with any Company Offering or Company Web Site, or a current, prospective or former customer, employee or vendor of any Person. Personal Information includes information in any form, including paper, electronic and other forms.

"PHI" means "protected health information" as defined in 45 C.F.R. § 160.103.

"Pre-Closing Tax" means (i) any Tax imposed on the Company and its Subsidiaries in respect of any Pre-Closing Tax Period, including Taxes that would have been payable during the Pre-Closing Tax Period but for any deferrals granted under the CARES act (or any corresponding or similar provision of state, local or foreign Tax Law) calculated by including (1) any Taxes of the Company or Company Securityholders imposed as a result of any adjustment under Section 481 of the Code (or any similar or corresponding provision of state, local or non-U.S. Law) as a result of any change in accounting method of the Company during or with respect to a Pre-Closing Tax Period (regardless of when such income is recognized for tax purposes), (2) Taxes due with respect to income that was, or would be, recognized in a Pre-Closing Tax Period (or portion of a Straddle Period ending on or prior to the Closing Date) with respect to the Company under Sections 951 or 951A of the Code under a "closing of the books" if the Closing Date were the last day of the taxable year for the Company, and (3) any deferred payments to be made in future taxable periods pursuant to Section 965(h) of the Code as though such amounts were due as of the Closing Date, (ii) any Tax of any Company Unitholder or any of their Affiliates for which any of the Company or any of its Subsidiaries is liable, whether by reason of any requirement to withhold or otherwise, (A) for any Pre-Closing Tax Period, or (B) incurred in connection with the Merger or this Agreement for any Tax period, (iii) any Tax for which the Company or any of its Subsidiaries is held liable under Treasury Regulations Section 1.1502-6 (or any corresponding or similar provision of state, local or foreign Tax Law) by reason of the Company or any of its Subsidiaries being included in any consolidated, affiliated, combined or unitary group in any Pre-Closing Tax Period, (iv) any Tax of another Person for which the Company or any of its Subsidiaries is held liable as a result of being a successor or transferee of such Person on or prior to the Closing Date or as a result of any express or implied obligation existing on or prior to the Closing Date to indemnify, compensate or reimburse any such Person, by Contract or otherwise, and (v) any liabilities for any Transfer Taxes for which the Company Securityholders are liable pursuant to Section 8.4. For all purposes of this Agreement, in the case of a Straddle Period, the amount of any Tax based on or measured by income or receipts or imposed in connection with any transaction that is allocable to the portion of a Straddle Period ending on the Closing Date shall be determined based on an interim closing of the books as of the close of business on the Closing Date (and for such purpose, the Tax period of any partnership or other pass-through entity in which the Company holds a beneficial interest shall be deemed to terminate at such time), and the amount of any other Tax of the Company that is allocable to the portion of a Straddle Period ending on the Closing Date shall be deemed to be the amount of such Tax for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of days in the portion of the Straddle Period ending on and including the Closing Date, and the denominator of which is the total number of days in the entire Straddle

Period. Notwithstanding the foregoing, any Taxes included in Indebtedness or Transaction Expenses shall not be deemed to be Pre-Closing Taxes.

"Pre-Closing Tax Period" means any Tax period ending on or before the Closing Date and that portion of any Straddle Period ending on and including the Closing Date.

"Privacy Law" means any Law that governs the receipt, collection, compilation, use, storage, processing, sharing, safeguarding, security, disposal, destruction, disclosure or transfer of Personal Information and any such Law governing breach notification, any penalties and compliance with any order, including the Telephone Consumer Protection Act, the Gramm-Leach-Bliley Act, all U.S. state privacy laws including the California Consumer Privacy Act ("CCPA"), the Video Privacy Protection Act, the Communications Decency Act, the CAN-SPAM Act, the Telephone Consumer Protection Act, Canada's Personal Information Protected and Electronic Documents Act, Canada's Anti-Spam Legislation, the UK Data Protection Act 2018, Regulation (EU) 2016/679 of the European Parliament and of the Council of 27 April 2016 (General Data Protection Regulation or "GDPR"), EU Directive 2002/58/EC and any Law or regulation implementing either or both of the GDPR and EU Directive 2002/58/EC (each as amended from time to time) and all analogous Laws in all foreign jurisdictions as are applicable to the Company, as well as all applicable industry standards.

# "Pro Rata Share" means:

- a) As to each Indemnifying Party or Company Securityholder, with respect to (i) any distribution to Parent or the Company Securityholders from the Adjustment Holdback Amount or the Escrow Fund, (ii) any distribution of the Post-Closing Excess Amount to the Company Securityholders, or (iii) any distribution of an Earnout Amount to the Company Stockholders, the quotient obtained by dividing (x) the aggregate number of Company Class A Units and Company Profits Interest Units held by such Indemnifying Party as of immediately prior to the Effective Time by (y) the aggregate number of Company Class A Units and Company Profits Interest Units held by all Indemnifying Parties as of immediately prior to the Effective Time. To the extent that an Indemnifying Party does not hold any Company Class A Units or Company Profits Interests Units as of immediately prior to the Effective Time, such Indemnifying Party's Pro Rata Share shall be "0" for purposes of this calculation.
- b) As to each Indemnifying Party, with respect to any other circumstance:
  - a. Immediately following the Effective Time and prior to the time that the Aggregator has distributed any portion of the Merger Consideration that has been paid or issued to the Aggregator pursuant to Section 1.4 to the holder of any Aggregator Units, the quotient obtained by dividing (i) the aggregate number of Company Class A Units and Company Profits Interest Units held by such Indemnifying Party as of immediately prior to the Effective Time by (ii) the aggregate number of Company Class A

Units and Company Profits Interest Units held by all Indemnifying Parties as of immediately prior to the Effective Time. To the extent an Indemnifying Party does not hold any Company Class A Units or Company Profits Interests Units as of immediately prior to the Effective Time, such Indemnifying Party's Pro Rata Share shall be "0" for purposes of this calculation.

- b. After the time that the Aggregator has distributed any portion of the Merger Consideration that has been paid or issued to the Aggregator pursuant to Section 1.4 to the holders of the Aggregator Units and prior to the time that the Aggregator has distributed all of the Merger Consideration that has been paid or issued to the Aggregator pursuant to Section 1.4 to the holders of the Aggregator Units, the quotient obtained by dividing (i) the sum of (x) the aggregate number of Company Class A Units and Company Profits Interest Units held by such Indemnifying Party as of immediately prior to the Effective Time plus (y) such Indemnifying Party's Allocated Aggregator Units by (ii) the sum of (A) the aggregate number of Company Class A Units and Company Profits Interest Units held by all Indemnifying Parties as of immediately prior to the Effective Time plus (B) the number of Aggregator Units outstanding immediately prior to the Effective Time; provided that for purposes of the calculation contemplated by this paragraph (b), the Aggregator shall be deemed not to hold any Company Profits Interest Units. For purposes of this paragraph (b), the number of "Allocated Aggregator Units" shall mean, (X) in the case of any Indemnifying Party other than the Aggregator, the product of (AA) the number of Aggregator Units outstanding immediately prior to the Effective Time multiplied by (BB) the quotient obtained by dividing (1) the aggregate amount of Merger Consideration all Indemnifying Parties have received from the Aggregator through a distribution of Merger Consideration by the Aggregator to holders of Aggregator Units in accordance with the terms of the Aggregator's Acknowledgement Agreement (valuing each share of Parent Common Stock at the Reference Price) by (2) the aggregate amount of Merger Consideration the Aggregator has been paid or issued pursuant to Section 1.4 (valuing each share of Parent Common Stock at the Reference Price) multiplied by (CC) such Indemnifying Party's Aggregator Pro Rata Share, and, (Y) in the case of the Aggregator, the number of Aggregator Units outstanding immediately prior to the Effective Time less the number of Allocated Aggregator Units allocated to Indemnifying Parties pursuant to the preceding clause (X).
- c. After the time that the Aggregator has distributed *all* of the Merger Consideration that has been paid or issued to the Aggregator pursuant to Section 1.4 to the holders of the Aggregator Units, the quotient obtained

- <u>by dividing</u> (i) the aggregate number of Company Class A Units, Company Profits Interest Units and Aggregator Units held by such Indemnifying Party as of immediately prior to the Effective Time <u>by</u> (ii) the aggregate number of Company Class A Units, Company Profits Interest Units and Aggregator Units held by all Indemnifying Parties as of immediately prior to the Effective Time. For purposes of the calculation contemplated by this clause (c), the Aggregator shall be deemed not to hold any Company Profits Interest Units.
- d. If (i) the Aggregator has distributed any of the Merger Consideration that has been paid or issued to the Aggregator pursuant to Section 1.4 to the holders of the Aggregator Units, (ii) the Indemnifying Parties have satisfied their indemnification obligations to the Indemnified Parties such that the Indemnified Parties have recovered from the Indemnifying Parties 100% of the Merger Consideration that has been paid or issued to the Indemnifying Parties pursuant to Section 1.4, and (iii) subsequently, the Indemnifying Parties receive pursuant to Section 1.4 a post-Closing payment or issuance in respect of the Post-Closing Excess Amount or Earnout Stock Consideration, and as of such time, the Aggregator has not distributed any portion of such post-Closing Merger Consideration that has been paid or issued to the Aggregator to the holder of any Aggregator Units, the quotient obtained by dividing (x) the aggregate number of Company Class A Units and Company Profits Interest Units held by such Indemnifying Party as of immediately prior to the Effective Time by (y) the aggregate number of Company Class A Units and Company Profits Interest Units held by all Indemnifying Parties as of immediately prior to the Effective Time. To the extent an Indemnifying Party does not hold any Company Class A Units or Company Profits Interests as of immediately prior to the Effective Time, such Indemnifying Party's Pro Rata Share shall be "0" for purposes of this calculation.
- c) The aggregate Pro Rata Shares of all Indemnifying Parties shall equal 100%.

"Process," "Processed" or "Processing" means any operation or set of operations, with respect to data or information, whether or not by automated means, such as the use, collection, processing, storage, recording, organization, adaption, alteration, transfer, retrieval, consultation, disclosure, dissemination, combination, erasure, or destruction of such data, or any other operation that is otherwise considered "processing" or similar term under applicable Privacy Requirements.

"**Proprietary Information**" means any information or material not generally known to the public, including any trade secret and other rights in know-how and confidential or proprietary information.

"Reference Price" means \$15.00 (as adjusted and appropriate to reflect any stock splits, stock dividends, reverse stock splits, combinations, reorganizations, reclassifications, or similar events affecting Parent Common Stock).

"Registered Company Intellectual Property Right" means (i) any issued Patent, pending Patent application, Mark registration, application for Mark registration, Copyright registration for Copyright registration and Domain Name registration owned, purported to be owned, filed or applied for by or on behalf of or exclusively licensed to the Company, and (ii) any other application, registration, recording and filing filed by or on behalf of the Company (or otherwise authorized by or in the name of the Company) with respect to any Company Intellectual Property Right.

"Related Party" means (i) any Affiliate (other than a Subsidiary) of the Company, or any director, executive officer, general partner or managing member of such Affiliate, (ii) any officer, director or manager of the Company or any Subsidiary of the Company, (iii) any Immediate Family member of a Person described in clause (ii), or (iv) any other Person who holds, individually or together with any Affiliate of such other Person and any member(s) of such Person's Immediate Family, more than five percent (5%) of the outstanding Company Securities.

"Representatives" means, with respect to any Person, such Person's officers, directors, managers, employees, agents, advisors (including any attorneys, financial advisors, investment bankers or accountants) and other representatives.

"Restricted Cash" means any cash or cash equivalents of the Company that is not freely usable by Company immediately following the Closing because it is subject to restrictions or limitations on use or distribution imposed by applicable Law or Contract, including, in each case to the extent imposed by such restrictions or limitations, (i) restrictions on withdrawals or dividends, (ii) cash that has been historically classified as restricted cash by the Company, (iii) cash that is necessary to meet any minimum cash, deposit or equity balances, (iv) cash that is represented by real estate lease deposits, (v) cash that is held in reserve accounts or third-party escrow accounts, (vi) cash collateralizing any obligation, (vii) cash subject to a dominion, control or similar agreement (other than those that will be terminated at or prior to the Closing), (viii) cash that is held in third party accounts (excluding, for the avoidance of doubt, the Company's bank accounts), and (ix) any cash received in connection with insurance claim(s) to the extent the underlying damage, destruction or loss giving rise thereto has not been cured or remediated.

"SEC" means the U.S. Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended.

"Securityholder" means each Company Securityholder and each holder of any Aggregator Units as of immediately prior to the Effective Time.

"Software" means any (i) computer program, including any API or SDK, software implementation of any algorithm, model or methodology, whether in source code, object or executable code, or other form, (ii) Database, (iii) descriptions, flow-charts and other work product used to design, plan, organize and develop any of the foregoing, screens, subroutines, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons and (iv) documentation, including user manuals and other training documentation, related to any of the foregoing.

"Stakeholder Claim" means any claim asserted by (i) any current, former or purported holder of Company Securities or any Person who has any right or claim with respect to ownership of any Company Securities (whether against the Company, Parent, any Affiliate of the Company or Parent, or any officer, director, employee, agent or representative of any of the foregoing) (A) with respect to this Agreement, the Mergers or any of the other transactions contemplated hereby (other than a claim seeking to enforce a right under this Agreement or any other Transaction Document), (B) alleging any ownership of or interest in any Company Securities that is not specifically disclosed in the Closing Statement, (C) relating to any rights under the Charter Documents related to such Person's current, former or purported ownership of Company Securities, (D) that such Person's securities were wrongfully issued or repurchased by the Company, or (E) relating to any actual or alleged breach of fiduciary duties; or (ii) any current or former officer, manager or director of the Company or any of its Subsidiaries for indemnification agreement in respect of claims made against them arising out of or in connection with their service as a director, manager or officer of the Company or any of its Subsidiaries at or prior to the Effective Time, whether or not asserted prior to the Effective Time.

"Straddle Period" means any Tax period beginning before or on the Closing Date and ending after the Closing Date.

"Subsidiary" means, with respect to any Person, any entity (whether or not incorporated) of which (i) such Person or any other Subsidiary of such Person is a general partner (excluding partnerships, the general partnership interests of which held by such Person or any Subsidiary of such Person do not have a majority of the voting interest in such partnership) or (ii) at least a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the board of directors or others performing similar functions with respect to such entity or a majority of the profit interests in such entity is directly or indirectly owned or controlled by such Person or by any one or more of its Subsidiaries, or by such Person and one or more of its Subsidiaries. Notwithstanding anything to the contrary, none of the Aggregator, DoD Accelerator Fund GP LLC, a Delaware limited liability company, or any Fund shall be deemed a Subsidiary of the Company (or a Subsidiary of any Subsidiary of the Company); provided that (A) each Fund shall be deemed a Subsidiary of the Company for purposes of Sections 2.1, 2.6, 2.8, 2.9(d) (but only the last two sentence of Section 2.9(d)), 2.14, 2.15 (but only subsections (a) through (c), (f), (g), (j), (k) and (q) of Section 2.15), 2.17, 2.20(c), 2.21, 2.22, 2.23, 2.24, 2.25

and  $\underline{7.2(a)(iv)}$  and (B) the Accelerator Fund GP shall be deemed a Subsidiary of the Company for purposes of <u>Article 2</u> and <u>Section 7.2(a)(iv)</u>.

"Target Company Net Working Capital" means an amount equal to \$0.

"Tax" (and, with correlative meaning, "Taxes") means (i) any federal, state, local or foreign income, alternative or add-on minimum, gross income, gross receipts, sales, use, VAT, transfer, franchise, profits, license, withholding, payroll, employment, excise, severance, stamp, occupation, premium, property, environmental, unclaimed property, estimated or windfall profit tax, custom duty, national insurance tax, health tax or other tax or other like assessment or charge in the nature of a tax, together with any interest or any penalty, addition to tax or additional amount imposed by any Governmental Authority responsible for the imposition of any such tax (domestic or foreign), whether disputed or not, (ii) any Liability for the payment of any amount of the type described in clause (i) as a result of being a member of an affiliated, consolidated, combined, unitary or aggregate group for any Tax period, and (iii) any Liability for the payment of any of the type described in clause (i) or (ii) as a result of being a transferee of or successor to any Person or as a result of any express or implied obligation to indemnify any other Person, by Contract or otherwise.

"Tax Return" means any return, amended return, election declaration, report, voluntary disclosure, claim for refund, information return or statement filed or required to be filed in respect of Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Technology" means any or all of the following: (i) technology, formulae, algorithm, diagram, formulae, procedure, process, method, technique, idea, know-how, creations, inventions, discoveries and improvement (whether patentable or unpatentable and whether or not reduced to practice); (ii) technical, engineering, manufacturing, product, marketing, servicing, business, financial, supplier, personnel or other information and materials; (iii) customer list, customer contact and registration information, customer correspondence and customer purchasing history; (iv) specification, design, industrial design, model, device, prototype, schematic, configuration and development tool; (v) Software, computer programs, whether in source code or in executable code form, subroutines, user interfaces, architecture, schematics, configuration, website, content, image, logo, graphic, text, photographs, artwork, audiovisual works, sound recording, graph, drawing, reports, analysis, writing, of any other work of authorship and copyrightable subject matter ("Work of Authorship"); (vi) database or other compilation or collection of data or information ("Database"); (vii) mask work, layout, topography or other design feature with respect to any integrated circuit ("Mask Work"); (viii) Mark; (ix) domain name, uniform resource locator or other name or locator associated with the Internet, and applications and registrations therefor ("Domain Name") or social media identifier; and (x) other forms of technology (whether embodied in any tangible forms and including all tangible embodiments of any of the foregoing, in any form or media whether or not specifically listed in this definition).

"Third-Party AI Product" means any product or service of any other Person that employs or makes use of AI Technologies.

"Third-Party Platform" means any other Person's device, platform, server, application, operating system, website, networked physical object (including Internet of Things (IoT)), software as a service, platform as a service, infrastructure as a service, cloud service or similar service.

"Transaction Expense" means, in each case only to the extent incurred or accrued prior to the Measurement Time or becoming payable solely as a result of the consummation of the Closing, any costs or expenses of any kind or nature incurred by or on behalf of, paid by, or to be paid by the Company or any of its Subsidiaries in connection with the Mergers and this Agreement and the transactions contemplated by this Agreement, as well as any related sale or financing process, including, without duplication, (i) all legal, accounting, financial advisory, consulting and all other fees and expenses of third parties owed by the Company or any of its Subsidiaries in connection with the negotiation, preparation and execution of this Agreement and the consummation of the transactions contemplated hereby (ii) any premium or related cost for any directors' and officers' liability insurance purchased by the Company or any of its Subsidiaries in connection with the transactions contemplated by this Agreement, including any brokerage or finders' fee, or agents' commission, (iii) any employer-level employment Tax based on any payment pursuant to this Agreement (for the avoidance of doubt, excluding any Parent Employee Obligations) or attributable to a Closing Employee Payment, (iv) any Closing Employee Payment, (v) any Transfer Taxes payable by the Company Securityholders pursuant to Section 8.4, (vi) 50% of the costs and expenses owed to the Escrow Agent in connection with the transactions contemplated hereby, and (vii) in the event that the termination of participation in the TriNet 401(k) Plan would reasonably be anticipated to trigger liquidation charges, surrender charges, other fees or any other Liabilities, then such charges, fees and/or Liabilities, in each case to the extent actually incurred, shall be deemed to be Transaction Expenses; provided, however, that [\*\*\*].

"Transfer Agent" means Continental Stock Transfer & Trust Company.

"Treasury Regulations" means the United States Treasury Regulations promulgated under the Code.

"TriNet 401(k) Plan" shall mean the TriNet Select 401(k) Plan as adopted by Accuidity, LLC, with an effective date of January 1, 2024.

"Unaccredited Investor" means any Company Securityholder that is not an Accredited Investor.

"Unpaid Pre-Closing Taxes" means the aggregate amount, which may not be less than zero, of all unpaid Pre-Closing Taxes, calculated through the end of the Closing Date and, with respect to a Straddle Period, in accordance with the second to last sentence of the definition of Pre-Closing Tax, in each case that are unpaid as of the Measurement Time.

"Unpaid Transaction Expenses" means all Transaction Expenses, in each case whether due prior to, on or after the Closing, that are unpaid as of the Measurement Time (provided, for the avoidance of doubt and notwithstanding anything to the contrary, the reduction contemplated by clause (B) of the definition of "Transaction Expenses" shall be applied in calculating the Unpaid Transaction Expenses, as a result of which the Unpaid Transaction Expenses may be a negative number, in which case the absolute value of the Unpaid Transaction Expenses as so calculated shall be added to the Cash Consideration in calculating the Adjusted Cash Consideration pursuant to clause (v) of the definition of "Adjusted Cash Consideration").

"VAT" means any ad valorem, value added, goods and services or similar Tax.

"Virtual Data Room" means the electronic data site jointly maintained by the Company and Parent at sharepoint.com in connection with the Merger.

"WARN Act" means the Worker Adjustment and Retraining Notification Act, as amended, and all state and local statutory equivalents.

"Wage Obligations" means all remuneration required to be paid by the Company or any of its Subsidiaries, or obligations therefor incurred by the Company or any of its Subsidiaries, on or before the Closing or contingent upon the occurrence of the Closing, to any current or former director, officer, employee or independent contractor of the Company or any of its Subsidiaries, and any salary, wages, accrued but unpaid vacation, paid time-off or other unpaid vacation benefits, accrued but unpaid bonus or other bonus obligations, commission payments, any unpaid severance liabilities, deferred compensation Liabilities, termination indemnity, defined benefit pension, multiemployer pension and other employee benefits, any post-retirement payments or benefits, and the employer portion of any associated employment, payroll or similar Tax, in each case with respect to any service provided to the Company or any of its Affiliates by any such Person on or prior to the Closing, including any such remuneration arising out of the termination of any such service; provided, that any obligations of the Company to any Person arising under a Company Benefit Arrangement that remain unpaid as of the Closing, including obligations arising under the Company's PPI or Co-Investment Plan, shall be deemed to be Wage Obligations; provided, however, that Parent Employee Obligations and Pre-Closing Taxes shall not be deemed to be Wage Obligations.

Other capitalized terms used in the Agreement and not defined in this <u>Annex A</u> shall have the meanings assigned to such terms in this Agreement in the sections referenced below.

Defined Term	Section
2026 Earnout Period	1.14(a)(i)
2027 Earnout Period	1.14(a)(ii)
2026 Earnout Target	1.14(a)(iii)
2027 Earnout Target	1.14(a)(iv)
2026 Earnout Threshold Ceiling	1.14(a)(v)
2026 Earnout Threshold Floor	1.14(a)(vi)

2027 Earnout Threshold Ceiling	1.14(a)(vii)
2027 Earnout Threshold Floor	1.14(a)(viii)
Adjustment Resolution Period	1.8(f)
Adjustment Review Period	1.8(e)
Advisers Act	2.15(h)
Advisory Agreement	2.15(k)
Aggregator Acknowledgement Agreement	Recitals
Aggregator Restricted Stock Amount	Recitals
Agreed Accounting Principles	1.8(a)
Agreement	Preamble
Agreement Date	Preamble
Anticipated Loss Amount	7.9(b)
Basket Amount	7.3(i)
Bribery Act	2.22(a)(vi)
Budget Schedule	1.14(a)(ix)
Cancelled Units	1.4(b)
Change in Control Transaction	1.14(a)(x)
Claim Notice	7.4
Closing	1.1
Closing Date	1.1
Closing Statement	1.8(a)
Company	Preamble
Company Indemnified Parties	4.2(a)
Company Service Providers	2.14(a)
Contested Claim	7.5(b)
Covered Communication	9.17(b)
Current Representation	9.17(b)
Designated Person	9.17(a)
Determination Date	1.8(g)
DLLCA	Recitals
DMCA	2.13(o)
Earnout Amount	1.14(a)(xi)
Earnout Business	1.14(a)(xii)
Earnout Calculation Statement	1.14(c)(i)
Earnout Default Event	1.14(a)(xiii)
Earnout Dispute Deadline	1.14(c)(ii)
Earnout Dispute Notice	1.14(c)(ii)
Earnout Period	1.14(a)(xiv)
Earnout Stock Consideration	1.14(a)(xv)
Earnout Target	1.14(a)(xvi)

Effective Time	1.2
Employment Agreement	Recitals
Employment Documents	Recitals
Escrow Fund	7.7(a)
Estimated Closing Cash	Schedule 1.8
Estimated Closing Indebtedness	Schedule 1.8
Estimated Closing Net Working Capital Shortfall	Schedule 1.8
Estimated Closing Net Working Capital Surplus	Schedule 1.8
Estimated Unpaid Pre-Closing Taxes	Schedule 1.8
Estimated Unpaid Transaction Expenses	Schedule 1.8
Exchange Agent	1.7(a)
Exchange Agent Agreement	1.7(b)
Expense Cap	1.14(a)(xviii)
Export Approvals	2.23(b)
Export Control Laws	2.23(a)
FCPA	2.22(a)(vi)
Filings	2.15(i)
Final Closing Statement	1.8(g)
First Certificate of Merger	1.2
First Earnout Amount	1.14(a)(xviii)
First Earnout Stock Consideration	1.14(a)(xix)
First Earnout VWAP	1.14(a)(xix)
First Merger	Recitals
Founder Restricted Stock Agreement	Recitals
Founder Restricted Stock Amount	Recitals
Fractional Share Consideration	1.9
Fund	2.15(r)
Fund Investor	2.20(a)
Governmental Permits	2.15(b)
Inbound License	2.12(a)(iv)
Indemnified Party	7.2(a)
Indemnifying Party	7.3
Independent Accountant	1.8(f)
Information Security Review	2.14(f)
Insurance Policies	2.18
Intended Tax Treatment	1.13(b)
Invention Assignment Agreement	2.13(1)
Investment Company Act	2.15(1)
Investor Questionnaires	Recitals
Joinder Agreement	Recitals

Letter of Transmittal	1.7(c)
Licensed IP	2.13(c)
Manager Approval	2.3(c)
Mergers	Recitals
Merger Sub I	Preamble
Merger Sub II	Preamble
Merger Subs	Preamble
New Limited Partner or New Equityholder	1.14(a)(xx)
New Plans	4.1(b)
Non-Competition Agreement	Recitals
Notice of Adjustment Disagreement	1.8(e)
Objection Period	7.5(a)
Old Plans	4.1(b)
Outbound Investment Rules	2.24
Outbound License	2.12(a)(iv)
Owed Amount	7.5(d)
Owned Company IP	2.13(c)
Parent	Preamble
Parent 401(k) Plan	4.1(a)
Parent Balance Sheet	3.9(f)
Parent Capitalization Date	3.4(a)
Parent Preferred Stock	3.4(a)
Parent SEC Reports	3.7(a)
Pass-Through Tax Claim	8.3(b)
Pass-Through Tax Return	8.1(a)
PICA	Recitals
Post-Closing Deficit Amount	1.8(h)
Post-Closing Excess Amount	1.8(i)
Post-Closing Representation	9.17(a)
Post-Closing Statement	1.8(c)
Privacy Requirements	2.14(a)
Recurring Net Revenue	1.14(a)(xxi)
Resolved Matters	1.8(f)
Restricted Stock Agreements	Recitals
Restricted Stock Amounts	Recitals
Retained Loss Amount	7.9(b)
Sales Taxes	2.7(a)(vii)
Sarbanes-Oxley Act	3.7(c)
SEC Approval Earnout Period	1.14(a)(xxii)
Second Certificate of Merger	1.2

Second Earnout Amount	1.14(a)(xxiii)
Second Earnout Stock Consideration	1.14(a)(xxiv)
Second Earnout VWAP	1.14(a)(xxiv)
Second Effective Time	1.2
Second Merger	Recitals
Securityholder Representative	Preamble
Securityholder Representative Expenses	7.13(b)
Settlement Amount	7.6
Significant Fund Investor	2.20(a)
Significant Supplier	2.20(b)
Signing Members	Recitals
Social Media Accounts	2.26
Social Media Account Names	2.26
Special Matters	7.3(b)
Specified Person	2.4(f)
Standard EULAs	2.12(a)(ii)
Standard Form Agreements	2.13(g)
Surviving Company	1.2
Surviving Entity	1.2
Tail Policy	6.3(g)
Tax Claim	8.3(a)
Tax Refunds	8.5
Third Earnout Amount	1.14(a)(xxv)
Second Earnout Stock Consideration	1.14(a)(xxvi)
Second Earnout VWAP	1.14(a)(xxvi)
Third-Party Claim	7.6
Transfer Taxes	8.4
Triggering Event	1.14(a)(xxvii)
Triggering Event Consideration Period	1.14(b)(iii)
Triggering Event Dispute Notice	1.14(b)(iii)
Triggering Event Notice	1.14(b)(i)
Unreleased Escrow Amount	7.7(b)
Unresolved Claims	7.7(b)
Unresolved Matters	1.8(f)
WH	9.17(a)
Written Consent	Recitals

# FORGE GLOBAL HOLDINGS, INC. CONSULTING AGREEMENT

This Consulting Agreement (the "Agreement") is hereby made and entered into as of April 25, 2025 (the "Effective Date"), by and between **Forge Global Holdings, Inc. and its Affiliates**, a Delaware corporation ("Forge" or the "Company"), having a place of business at 4 Embarcadero Center, Floor 15, San Francisco, CA 94111 and **Larry Leibowitz** ("Consultant").

- 1. <u>Engagement of Services</u>. During the period of this Agreement, Consultant hereby agrees to provide strategic guidance and other services as may be requested or assigned from time to time (the "Services"), to Forge.
- 2. <u>Compensation; Timing.</u> In consideration for the consulting services bargained for in this Agreement, Forge will compensate the Consultant with the following:
- 2.1 <u>Consulting Fee.</u> Subject to the approval of the Board of Directors of the Company (the "Board"), Consultant shall be granted an equity award (the "Award") that is subject to Forge's 2022 Stock Option and Incentive Plan (the "Plan") and equal to \$110,000 divided by the Value. Unless otherwise determined by the Board, the Award will consist of restricted stock units, each representing the contingent right to receive one share of Forge's common stock. 4/12<sup>th</sup> of the Award shall vest on April 31, 2025, and the remaining 8/12<sup>th</sup> of the Award shall vest in equal monthly installments on the last day of each month thereafter, provided Consultant continues to have a Service Relationship (as defined in the Plan) with Forge through each such vesting date. For purposes hereof, "Value" means the grant date fair value of the Award determined in accordance with the reasonable assumptions and methodologies employed by the Company for calculating the fair value of equity awards under ASC Topic 718.
- 2.2 <u>Expense Reimbursement</u>. Forge will reimburse Consultant's reasonable travel and other out-of-pocket expenses incurred by Consultant from time to time at Forge's request. Any expenses in excess of one-hundred dollars (\$100) must be pre-approved by Forge (by email or other written manner). Travel required to perform Services for the Company will adhere to Forge's applicable travel expense policies.
- 3. <u>Independent Contractor Relationship.</u> Consultant's relationship pursuant to this Agreement with Forge is that of an independent contractor, and nothing in this Agreement is intended to, or should be construed to, create a partnership, agency, joint venture or employment relationship. Consultant will not be entitled to any of the benefits which Forge may make available to its employees, including, but not limited to, group health or life insurance, profit-sharing or retirement benefits. Consultant is not authorized to make any representation, contract or commitment on behalf of Forge unless specifically requested or authorized in writing to do so by Forge's Chief Executive Officer. Consultant is solely responsible for, and will file, on a timely basis, all tax returns and payments required to be filed with, or made to, any federal, state or local tax authority with respect to the performance of services and receipt of fees under this Agreement. Consultant is solely responsible for, and must maintain adequate records of, expenses incurred in the course of performing services under this Agreement. No part of Consultant's compensation will be subject to withholding by Forge for the payment of any social

security, federal, state or any other employee payroll taxes. Forge will regularly report amounts paid to Consultant by filing Form 1099-MISC with the Internal Revenue Service as required by law.

## 4. <u>Intellectual Property Rights</u>.

## 4.1 <u>Disclosure and Assignment of Innovations</u>:

(a) <u>Innovations</u>; <u>Forge Innovations</u>. "Innovations" includes processes, machines, compositions of matter, improvements, inventions (whether or not protectable under patent laws), works of authorship, information fixed in any tangible medium of expression (whether or not protectable under copyright laws), moral rights, mask works, trademarks, trade names, trade dress, trade secrets, know-how, ideas (whether or not protectable under trade secret laws), and all other subject matter protectable under patent, copyright, moral right, mask work, trademark, trade secret or other laws, and includes, without limitation, all new or useful art, combinations, discoveries, formulae, manufacturing techniques, technical developments, artwork, software, and designs. "Forge Innovations" are Innovations that Consultant, solely or jointly with others, conceives, reduces to practice, creates, derives, develops or makes within the scope of Consultant's work for Forge under this Agreement.

(b) <u>Disclosure of Ownership of Forge Innovations</u>. Consultant agrees to make and maintain adequate and current records of all Forge Innovations, which records shall be and remain the property of Forge. Consultant agrees to promptly disclose to Forge every Forge Innovation. Consultant hereby does and will assign to Forge or Forge's designee, Consultant's entire worldwide right, title and interest in and to all Forge Innovations and all associated records and intellectual property rights.

(c) <u>Assistance</u>. Consultant agrees to execute, upon Forge's request, a signed transfer of Forge Innovations to Forge for each of Forge's Innovations, including, but not limited to, computer programs, notes, sketches, drawings and reports. Consultant agrees to assist Forge in any reasonable manner to obtain, perfect and enforce, for Forge's benefit, Forge's rights, title and interest in any and all countries, in and to all patents, copyrights, moral rights, mask works, trade secrets, and other property rights in each of Forge's Innovations. Consultant agrees to execute, when requested, for each of Forge's Innovations (including derivative works, improvements, renewals, extensions, continuations, divisionals, continuations in part, or continuing patent applications thereof), (i) patent, copyright, mask work or similar applications related to such Forge Innovation, (ii) documentation (including, without limitation, assignments) to permit Forge to obtain, perfect and enforce Forge's right, title and interest in and to such Forge Innovation, and (iii) any other lawful documents deemed necessary by Forge to carry out the purpose of this Agreement. If called upon to render assistance under this paragraph, Consultant will be entitled to a fair and reasonable fee in addition to reimbursement of authorized expenses incurred at the prior written request of Forge. In the event that Forge is unable for any reason to secure Consultant's signature to any document is required to execute under this Paragraph 4.1(c), Consultant hereby irrevocably designates and appoints Forge and Forge's duly authorized officers and agents as Consultant's agents and attorneys-in-fact to act for and in Consultant's behalf and instead of Consultant, to execute such document with the same legal force and effect as if executed by Consultant.

# 4.2 Confidential Information.

(a) <u>Definition of Confidential Information</u>. "Confidential Information" as used in this Agreement shall mean any and all technical and non-technical information including patent, copyright, trade secret, and proprietary information, techniques, sketches, drawings, models, inventions, know-how, processes, apparatus, equipment, algorithms, software programs, software source documents,

and formulae related to the current, future and proposed product and services of Forge, Forge's suppliers and customers, and includes, without limitation, Forge Innovations, Forge Property, and Forge's information concerning research, experimental work, development, design details and specifications, engineering, financial information, procurement requirements, purchasing manufacturing, customer lists, business forecasts, sales and merchandising and marketing plans and information.

- (b) <u>Nondisclosure and Nonuse Obligations</u>. Except as permitted in this paragraph, Consultant shall neither use nor disclose the Confidential Information. Consultant may use the Confidential Information solely to perform the consulting services bargained in this Agreement for the benefit of Forge. Consultant agrees that Consultant shall treat all Confidential Information of Forge with the same degree of care as Consultant accords to Consultant's own Confidential Information, but in no case less than reasonable care. Consultant agrees not to communicate any information to Forge in violation of the proprietary rights of any third party. Consultant will immediately give notice to Forge of any unauthorized use or disclosure of the Confidential Information. Consultant agrees to assist Forge in remedying any such unauthorized use or disclosure of the Confidential Information.
- (c) Exclusion from Nondisclosure and Nonuse Obligations. Consultant's obligations under Paragraph 4.2(b) ("Nondisclosure and Nonuse Obligations") with respect to any portion of the Confidential Information shall not apply to any such portion which Consultant can demonstrate: (a) was in the public domain at or subsequent to the time such portion was communicated to Consultant by Forge through no fault of Consultant; (b) was rightfully in Consultant's possession free of any obligation of confidence at or subsequent to the time such portion was communicated to Consultant by Forge; or (c) was developed by Consultant independently of and without reference to any information communicated to Consultant by Forge. A disclosure of Confidential Information by Consultant, either (a) in response to a valid order by a court or other governmental body, (b) otherwise required by law, or (c) necessary to establish the rights of either party under this Agreement, shall not be considered to be a breach of this Agreement or a waiver of confidentiality for other purposes; provided, however, that Consultant shall provide prompt prior written notice thereof to Forge to enable Forge to seek a protective order or otherwise prevent such disclosure.
- 4.3 Ownership and Return of Forge Property. All materials (including, without limitation, documents, drawings, models, apparatus, sketches, designs, lists, and all other tangible media of expression) furnished to Consultant by Forge, whether delivered to Consultant by Forge or made by Consultant in the course of performing the consulting services bargained for in this Agreement (collectively, the "Forge Property") are the sole and exclusive property of Forge or Forge's suppliers or customers, and Consultant hereby does and will assign to Forge all rights, title and interest Consultant may have or acquire in Forge's Property. Consultant agrees to keep all Forge Property at Consultant's premises unless otherwise permitted in writing by Forge. At Forge's request, and no later than five (5) days after such request, Consultant shall destroy or deliver to Forge, at Forge's option, (a) all Forge Property, (b) all tangible media of expression in Consultant's possession or control which incorporate or in which are fixed any Confidential Information, and (c) written certification of Consultant's compliance with Consultant's obligations under this sentence.
- 4.4 <u>Observance of Forge Rules</u>. At all times while on Forge's premises, Consultant will observe Forge's rules and regulations with respect to conduct, health and safety and protection of persons and property.
- 4.5 <u>Protected Disclosures and Other Protected Actions; Defend Trade Secrets Act Notice</u>. Nothing contained in this Agreement limits Consultant's ability to file a charge or complaint with

any federal, state or local governmental agency or commission (a "Government Agency"). In addition, nothing contained in this Agreement limits Consultant's ability to communicate with any Government Agency or otherwise participate in any investigation or proceeding that may be conducted by any Government Agency, including Consultant's ability to provide documents or other information, without notice to Company, nor does anything contained in this Agreement apply to truthful testimony in litigation. If Consultant files any charge or complaint with any Government Agency and if the Government Agency pursues any claim on Consultant's behalf, or if any other third party pursues any claim on Consultant's behalf, Consultant waives any right to monetary or other individualized relief (either individually, or as part of any collective of class action); provided that nothing in this Agreement limits any right Consultant may have to receive a whistleblower award or bounty for information provided to the Securities and Exchange Commission. Consultant understands that pursuant to the Defend Trade Secrets Act of 2016, Consultant shall not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a trade secret that (A) is made (i) in confidence to a federal, state, or local government official, either directly or indirectly, or to an attorney, and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal.

## 5. Term and Termination.

- 5.1 <u>Term.</u> This Agreement is effective as of the Effective Date set forth above and will terminate on December 31, 2025, unless earlier terminated in accordance with the Agreement's terms. The Agreement may be extended by mutual agreement in writing.
- 5.2 <u>Termination by Forge</u>. Forge may terminate this Agreement without cause at any time, with termination effective thirty (30) days after Forge's delivery to Consultant of written notice of termination. Forge also may terminate this Agreement (i) immediately upon Consultant's breach of Paragraph 4 ("Intellectual Property Rights"), or (ii) ten (10) days after Forge's delivery to Consultant of written notice of Consultant's material breach of any other provision or obligation owed by Consultant under this Agreement that is not cured within such fifteen (15) day period.
- 5.3 <u>Termination by Consultant</u>. Consultant may terminate this Agreement without cause at any time, with termination effective thirty (30) days after Consultant's delivery to Forge of written notice of termination. Consultant also may terminate this Agreement for material breach by Forge if Forge has not cured the breach within fifteen (15) days of receiving written notice from Consultant.
- 5.4 <u>Survival</u>. The definitions contained in this Agreement and the rights and obligations contained in Paragraphs 4 ("Intellectual Property Rights"), 5.4 ("Survival"), and 6 ("General Provisions") will survive any termination or expiration of this Agreement.

#### 6. General Provisions.

- 6.1 <u>Successors and Assigns</u>. Consultant may not subcontract or otherwise delegate Consultant's obligations under this Agreement without Forge's prior written consent. Subject to the foregoing, this Agreement will be for the benefit of Forge's successors and assigns, and will be binding on Consultant's assignees.
- 6.2 <u>Notices</u>. Any notice required or permitted by this Agreement shall be in writing and shall be delivered as follows, with notice deemed given as indicated: (a) by personal delivery, when delivered personally; (b) by overnight courier, upon written verification of receipt; (c) by email with

acknowledgment of read receipt of electronic transmission; or (d) by certified or registered mail, return receipt requested, upon verification of receipt. Notice shall be sent to the addresses set forth above or to such other address as either party may specify in writing.

- 6.3 <u>Governing Law.</u> This Agreement shall be governed in all respects by the laws of the United States of America and by the laws of the State of New York. The state or federal courts of New York County, New York, will have exclusive jurisdiction over any dispute arising out of or relating to this Agreement and the parties hereby irrevocably waive any objection to that jurisdiction, whether based upon *forum non conveniens* or similar doctrine.
- 6.4 <u>Severability</u>. If any provision of this Agreement is held by a court of law to be illegal, invalid or unenforceable, (i) that provision shall be deemed amended to achieve as nearly as possible the same economic effect as the original provisions, and (ii) the legality, validity and enforceability of the remaining provisions of this Agreement shall not be affected or impaired thereby.
- 6.5 <u>Waiver; Amendment; Modification</u>. No term or provision hereof will be considered waived by Forge, and no breach excused by Forge, unless such waiver or consent is in writing signed by Forge. The waiver by Forge of, or consent by Forge to, a breach of any provision of this Agreement by Consultant shall not operate or be construed as a waiver of, consent to, or excuse of any other or subsequent breach by Consultant. This Agreement may be amended or modified only by mutual agreement of authorized representatives of the parties in writing.
- 6.6 <u>Injunctive Relief for Breach</u>. Consultant's obligations under this Agreement are of a unique character that gives them particular value; Consultant's breach of any of such obligations will result in irreparable and continuing damage to Forge for which there will be no adequate remedy at law; and, in the event of such breach, Forge will be entitled to injunctive relief and/or a decree for specific performance, and such other or further relief as may be proper (including monetary damages if appropriate).
- 6.7 <u>Entire Agreement</u>. This Agreement constitutes the entire agreement between the parties relating to this subject matter and supersedes all prior or contemporaneous oral or written agreements concerning such subject matter. The terms of this Agreement will govern all services undertaken by Consultant for Forge.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

# FORGE GLOBAL HOLDINGS, INC.

# **CONSULTANT**

By: <u>/s/ Kelly Rodriques</u>
Name: Kelly Rodriques
Title: Chief Executive Officer

By: <u>/s/ Larry Leibowitz</u> Name: Larry Leibowitz Title: Consultant

### SEPARATION AND GENERAL RELEASE AGREEMENT

THIS SEPARATION AND GENERAL RELEASE AGREEMENT (this "Agreement" or "Separation Agreement") is entered into by and between Forge Global Holdings, Inc. (the "Company"), and Jennifer Phillips ("Executive").

#### WITNESSETH

**WHEREAS**, Executive has entered into an Amended and Restated Employment Agreement dated as of March 26, 2024 with the Company (the "Employment Agreement") and has been employed by the Company in the position of Chief Revenue and Growth Officer;

WHEREAS, as of the Separation Date (as defined below), Executive is expected to hold a total of (i) 75,365 shares underlying stock options (the "Options"), all of which are vested but unexercised as of the Separation Date (as defined below) and (ii) 25,686 unvested restricted stock units subject to time-based vesting (including performance-based units for which actual performance achievement has already been certified) (the "Time-Based RSUs"), in each case, pursuant to the terms and conditions of the Company's 2018 Equity Incentive Plan and/or the Company's 2022 Stock Option and Incentive Plan (each, as amended from time to time, the "Plan") and the applicable Plan award agreements (collectively, the "Award Agreements");

**WHEREAS**, unless otherwise approved by the Company, Executive's last day of employment with the Company will be June 26, 2025 (the "Separation Date"), upon which Executive's employment and/or service relationship with the Company and its affiliates shall terminate for all purposes;

**WHEREAS**, in exchange for Executive executing and complying with the terms of this Agreement, including without limitation Executive's execution and non-revocation of the general release of claims herein, the Company has agreed to provide Executive the below-described benefits and arrangements;

WHEREAS, it was previously contemplated that Executive would be eligible to receive an award of restricted stock units in 2025; however, the Company and Executive have agreed that such award will not be granted to Executive or, to the extent such award is made prior to the execution of this document, such restricted stock units will be forfeited in their entirety upon the Separation Date (the "2025 RSUs"); and

**WHEREAS**, voluntarily and of Executive's own free will, Executive desires to accept the Company's offer in exchange for executing and complying with the terms of this Agreement.

**NOW, THEREFORE**, in consideration of the mutual promises and covenants set forth herein, the Company and Executive hereby agree as follows:

# 1. Separation of Employment; Transition Period.

- (a) Separation Date. Executive confirms and understands that unless otherwise expressly set forth herein, if the Company executes this Agreement and if Executive executes and does not revoke Executive's acceptance to this Agreement, Executive's employment and/or service relationship with the Company and any of its affiliates shall terminate and cease for all purposes effective as of the Separation Date, including for purposes of all salary, wages, compensation, benefits, bonuses, equity, vesting, and any other interests or benefits unless otherwise expressly set forth herein. Executive shall be paid Executive's (i) base salary earned through the Separation Date, unpaid expense reimbursements and unused paid time off that accrued through the Separation Date on or before the time required by law and (ii) any vested benefits the Executive may have under any employee benefit plan of the Company through the Separation Date, which vested benefits shall be paid and/or provided in accordance with the terms of such employee benefit plans, including but not limited to all said benefits set forth in Section 4.2(a)(3) of the Employment Agreement.
- **(b) Transition Period**. Effective through the Separation Date (the "Transition Period"), Executive (i) shall assist in the smooth and orderly transition of Executive's duties and responsibilities, (ii) shall continue to serve as an employee in the role(s) Executive serves/served immediately prior to the Separation Date, and (iii) shall otherwise provide reasonable assistance to the Company in accordance with the reasonable directives of the Company.
- (c) Consulting Agreement. Effective on or about on June 27, 2025 or such other date as the parties may mutually agree, the Company shall engage Executive as an independent consultant upon the terms set forth in the Consulting Agreement attached hereto as Exhibit B (the "Consulting Agreement"). Notwithstanding anything to the contrary set forth in the Plan or the Award Agreements, the Company and Executive hereby acknowledge and agree that Executive's transition from an employee to an independent consultant under the Consulting Agreement shall constitute an interruption of Executive's Service Relationship (as defined in the Company's 2022 Stock Option and Incentive Plan) with regard to continued vesting of Executive's Options and Time-Based RSUs (collectively, the "Equity Awards"), and, other than as set forth in Section 2(b), below, no additional vesting will occur with respect to such Equity Awards following the Separation Date.
- 2. Separation Benefits. In consideration of Executive's promises and covenants set forth herein and further conditioned upon: (a) Executive executing, dating and delivering this Agreement to the Company in accordance with the terms provided in the paragraph titled "Older Worker Benefit Protection Act Disclosure"; (b) Executive not revoking this Agreement as provided for in the section titled "Older Worker Benefit Protection Act Disclosure"; (c) Executive executing, dating and delivering the Affirmation of General Release of Claims attached hereto as Exhibit A to the Company no earlier than the Separation Date and no later than three (3) business days after the Separation Date; (d) Executive's compliance with all the terms of this Agreement, as well as Executive's continued compliance with that certain Proprietary Information and Additional Covenants Agreement, dated May 8, 2020 (the "PICA"), and the restrictive covenants contained in the Employment Agreement (it being understood and

agreed that the "Restricted Period" as defined therein shall begin to run upon the Separation Date), Company shall pay or provide to Executive the following (collectively, the "Separation Benefits"):

- (a) Severance Payment. The Company agrees to continue to pay Executive her current base salary of \$425,000 for a period of twelve (12) months following Executive's Separation Date (the "Severance Period") in substantially equal installments in accordance with the Company's payroll practice during the Severance Period (the "Base Salary Amount"). In addition, the Company agrees to pay Executive the lump sum total of \$459,941 (the "Bonus Amount") within ten days after delivery of the Affirmation of General Release of Claims.
- **(b)** Accelerated Vesting of RSUs. Executive shall be entitled to receive accelerated vesting with respect to 19,963 unvested Time-Based RSUs. Executive will be treated for all purposes with respect to Options and Time-Based RSUs in accordance with Section 4.2(a)(3) of the Employment Agreement, and as if Executive had remained employed by the Company until December 26, 2026 so that the time period over which Executive has the right to exercise Executive's Options shall be the same as if there had been no termination of employment until December 26, 2026. For the avoidance of doubt, any unvested Equity Awards not described herein, as well as the 2025 RSUs and any rights thereto, will terminate immediately upon the Separation Date.
- **(c) COBRA Payments**. If the Executive was participating in the Company's group health plan immediately prior to the Separation Date and properly elects to continue health coverage under the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA"), then the Company shall pay the monthly employer COBRA premium for the same level of group health coverage as in effect for Executive on the Separation Date through the duration of the Severance Period.

The payments, benefits, and arrangements described in this Section 2 are collectively referred to herein as the "Separation Benefits."

- (d) The Company agrees to reimburse Executive for up to \$15,000 in legal fees incurred in connection with Executive's separation from the Company, within 14 days of presentation of an invoice for the amount of services actually rendered to Executive.
- 3. Full Payment and Satisfaction. Executive acknowledges and agrees that (i) subject to receipt of the Executive's receipt of the Separation Benefits described in Paragraph 2 above, Executive has received full and timely payment of all compensation, salary, wages, bonuses, commissions, distributions, vacation pay, paid time off, severance, incentive pay, equity, profits, phantom stock payments, change in control bonuses, or other payments or benefits owed or which ever have been owed by the Company or any of its affiliates through the Separation Date, including without limitation pursuant to the Award Agreements or the Plan and (ii) the Separation Benefits provided herein, individually or in the aggregate, are in exchange for Executive's execution and non-revocation of this Agreement, as well as Executive's compliance with the terms hereof; provided, however, that Executive shall continue to be entitled to the

rights and benefits set forth in the Consulting Agreement, subject to the additional terms set forth therein. Other than as set forth in Section 2(b), above, no Equity Awards will vest following the Separation Date. Separation Benefits will generally be paid during the Severance Period.

- 4. General Release of Claims. In exchange for the consideration set forth herein, including the Separation Benefits, Executive, for Executive, Executive's agents, attorneys, heirs, beneficiaries, administrators, executors, successors, assigns, and other representatives, and anyone acting or claiming on Executive's or their joint or several behalf, hereby releases, waives, and forever discharges the Company, and all of its past, present or future parents, subsidiaries or affiliates, and any of its or their past, present or future employees, officers, directors, managers, trustees, board members, owners, equity holders, agents, attorneys, insurers or benefit plans and any of its or their predecessors, successors, assigns, and other representatives, and anyone acting on their joint or several behalf (the "Releasees"), from any and all known and unknown, suspected or unsuspected, claims, causes of action, demands, damages, costs, expenses, attorneys' fees, liabilities, or other losses, from the beginning of time through the date this Separation Agreement is executed, including, but not limited to, those that in any way arise from, grow out of, or are related to Executive's employment and/or service relationship with the Company or any of its affiliates or subsidiaries or the termination thereof or any contracts, agreements, awards, policies, plans, programs or practices with or of any Releasee. By way of example only and without limiting the immediately preceding sentence, Executive agrees that Executive is releasing, waiving, and discharging
- (a) any and all claims against the Company and the Releasees under (i) any federal, state, or local employment law or statute, including, but not limited to Title VII of the Civil Rights Act(s) of 1964 and 1991, the Age Discrimination in Employment Act, as amended, the Family and Medical Leave Act, the Worker Adjustment and Retraining Notification (WARN) Act and any state or local equivalent law(s), the Older Workers Benefit Protection Act, the Fair Labor Standards Act, Section 806 of the Corporate and Criminal Fraud Accountability Act of 2002, the Rehabilitation Act of 1973, the Fair Credit Reporting Act, the Employee Retirement Income Security Act of 1974, and the Uniformed Services Employment and Reemployment Rights Act; (ii) applicable state or local civil rights law(s), including without limitation the California Family Rights Act, the California Labor Code, the California Workers' Compensation Act, the California Fair Employment and Housing Act, each as amended; and (iii) any federal, state or municipal law, statute, ordinance or common law doctrine;
- (b) any claim to benefits under any plan, or under the federal Employee Retirement Income Security Act of 1974, as amended, except for Executive's right to any benefits to which Executive is entitled under any retirement plan of the Company that is intended to be qualified under Section 401(a) of the Internal Revenue Code of 1986, as amended, or Executive's rights, if any, under COBRA;
- (c) any and all claims under common law, including any claim for tort, breach of contract (express or implied, written or oral), quasi contract, wrongful or constructive discharge, defamation, intentional infliction of emotional distress, misrepresentation, fraud, or negligence; (d) any and all claims arising under or in connection with any agreement, understanding,

contract, promise, or arrangement, written or oral, between Executive and the Company and/or any of the Releasees, including but not limited to any claims arising under or in connection with the Award Agreements or the Plan; (e) any and all claims for or relating to any compensation, distributions, salary, wages, bonuses, commissions, income, carried interest, contributions, distributions, fees, income, contributions, overtime pay, incentive compensation, equity-based compensation, stock options, shares, equity, or any other actual or prospective stock, equity, option, profits, awards or grants; and (f) any other claims Executive may now have against the Releasees, whether known or unknown to Executive. Notwithstanding any releases granted in the foregoing, the following shall not be released, altered, or limited in any way as a result of the foregoing: (i) Executive's right to be held harmless and Executive's right to indemnification from the Company, as reflected in that certain indemnification agreement, dated April 12, 2023 (the "Indemnification Agreement"), or under the Company's by-laws or certificate of incorporation, (ii) Executive's right to indemnification pursuant to Section 5 of the Consulting Agreement, and (iii) Executive's right to coverage pursuant to the terms of any applicable Company directors' and officers' liability insurance policy.

In addition, in exchange for Executive's release herein and Executive's compliance with this Agreement, the Company, on its behalf as well as any agents, attorneys, successors, assigns, Board of Directors and other representatives and anyone acting or claiming on the Company's or their joint or several behalf, hereby releases, waives, and forever discharges Executive and all of Executive's heirs, executors and/or affiliates, agents, attorneys, insurers and any of its or their predecessors, successors, assigns, and other representatives, and anyone acting on their joint or several behalf from any and all known and unknown, suspected or unsuspected, claims, causes of action, demands, damages, costs, expenses, attorneys' fees, liabilities, or other losses, from the beginning of time through the date this Agreement is executed, including, but not limited to, those that in any way arise from, grow out of, or are related to Executive's employment and/or service relationship with the Company or any of its affiliates or subsidiaries or the termination thereof or any contracts, agreements, awards, policies, plans, programs or practices.

Nothing contained herein shall be construed to prohibit Executive from filing a charge with the Equal Employment Opportunity Commission (the "EEOC"), the National Labor Relations Board, the Securities and Exchange Commission (the "SEC"), or other similar governmental agency (federal, state, or local), or participating in investigations by such government entity. However, Executive acknowledges that the release Executive executes herein waives Executive's right to seek individual remedies in any such action or to seek or accept individual remedies or monetary damages in any such action or lawsuit arising from such charges or investigations, including but not limited to, back pay, front pay, or reinstatement; however, Executive may accept any monetary award offered by the SEC pursuant to Section 21F of the Securities and Exchange Act of 1934. Executive further agrees that if any person, organization, or other entity should bring a claim against the Releasees involving any matter covered by this Agreement, Executive will not accept any personal relief in any such action, including damages, attorneys' fees, costs, and all other legal or equitable relief. Executive also does not release: (i) any rights or claims that may arise after Executive signs this Agreement or which arise under this Agreement; (ii) any rights or claims for unemployment compensation or

workers' compensation benefits; (iii) any rights under the Company's 401(k) plan, if any; or (iv) any other rights that cannot be waived by private agreement or operation of law.

**5.** California Civil Code Section 1542. Executive acknowledges that Executive has been advised to consult with legal counsel and is familiar with the provisions of California Civil Code Section 1542, a statute that otherwise prohibits the release of unknown claims, which provides as follows:

A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY.

Executive, being aware of said code section, agrees to expressly waive any rights Executive may have thereunder, as well as under any other statute or common law principles of similar effect.

- **6.** No Claims Filed. Executive affirms that, as of the date of execution of this Separation Agreement, Executive has filed no lawsuit, charge, claim or complaint with any governmental agency or in any court against the Company or the Releasees.
- 7. Affirmation of Post-Employment Obligations. In executing this Separation Agreement, Executive hereby reaffirms the post-employment obligations set forth in the PICA. It is expressly understood that the Company may (i) be entitled to seek repayment of all or a portion of the payments under Section 2 in the event that Executive breaches any of Executive's obligations under this Separation Agreement or the PICA or (ii) be entitled to all Erroneously Award Compensation as defined under the Company's Compensation Recovery Policy.
- 8. Return of Company Property. Executive agrees that Executive will, unless otherwise agreed to in writing by the Company or as contemplated by the Consulting Agreement, return to the Company all property and equipment of the Company or its affiliates in Executive's possession; provided, that Executive shall retain the Company-issued laptop computer (and anything contained therein and any records related to matters upon which Executive is consulting) and Company-issued e-mail address through the end of the Term (as defined in the Consulting Agreement), subject to any removal and/or modifications as reasonably requested by the Company. Otherwise, all records, paper and electronic files, documents, software programs, and copies thereof, pertaining to the business of the Company or its affiliates, including without limitation any records, files, documents and programs which may constitute trade secrets and proprietary information belonging to the Company or its affiliates, in each case, shall be returned to the Company by or upon the Separation Date. Executive may not retain copies of any such records, files, documents or programs, and hereby relinquishes and assigns to the Company any and all rights, if any, that Executive may have in any such records, files, documents or programs. Executive represents that Executive has not retained any information about the Company or its affiliates on any personal computer, personal phone, portable data storage device or otherwise.

# 9. Non-Disclosure; Confidentiality.

- (a) Confidential Information. Subject to Section 11 below, Executive agrees that the circumstances of Executive's separation and/or transition from the Company are strictly confidential, except that Executive may disclose the terms and conditions to Executive's attorneys, accountants, tax consultants, spouse, state and federal tax authorities or as may otherwise be required by law (provided such parties are instructed to comply with this section).
- (b) Defend Trade Secrets Act of 2016 Notice. Pursuant to 18 U.S.C. § 1833(b), Executive will not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a trade secret of the Company or its affiliates that (i) is made (x) in confidence to a federal, state, or local government official, either directly or indirectly, or to Executive's attorney and (y) solely for the purpose of reporting or investigating a suspected violation of law; or (ii) is made in a complaint or other document that is filed under seal in a lawsuit or other proceeding. If Executive files a lawsuit for retaliation by the Company for reporting a suspected violation of law, Executive may disclose the trade secret to Executive's attorney and use the trade secret information in the court proceeding, if Executive files any document containing the trade secret under seal, and does not disclose the trade secret, except pursuant to court order. Nothing in this Separation Agreement or any other agreement between the parties or any other policies of the Company or its affiliates is intended to conflict with 18 U.S.C. § 1833(b) or create liability for disclosures of trade secrets that are expressly allowed by such section.
- 10. Non-Disparagement. Subject to Section 11 below, Executive agrees that she will not make or cause to be made any negative, defamatory, or disparaging statements about the Company, any of the Releasees, and/or any of their respective businesses or reputations, in any manner, form, forum or media (including, without limitation, orally or in writing, in, to or via the press, any media, social media or forum, on-line, e-mail, text message, blog, posting, or otherwise) to any person, entity or third party, including without limitation to (a) any future, current or former Company employees, (b) the general public, (c) investors, clients, or customers of the Company, or (d) persons or entities with which the Company or other Releasees has or have or is or are known to be seeking business relationships. This obligation shall not in any way affect Executive's obligation to testify truthfully in any legal proceeding or any rights Executive may have as described in Section 11 below. In addition, the Company shall not, through its directors and officers or, if willfully directed by the Company, through any other employee or agent of the Company, make or cause to be made any negative, defamatory, or disparaging statements about him in any manner, form, forum or media (including, without limitation, orally or in writing, in, to or via the press, any media, social media or forum, on-line, e-mail, text message, blog, posting, or otherwise) to any person, entity or third party.
- 11. Permitted Disclosures. Nothing in this Separation Agreement or any other agreement between the parties or any other policies of the Company or its affiliates shall prohibit or restrict Executive or Executive's attorneys from: (a) making any disclosure of relevant and necessary information or documents in any action, investigation, or proceeding relating to this Separation Agreement, or as required by law or legal process, including with respect to possible

violations of law; (b) participating, cooperating, or testifying in any action, investigation, or proceeding with, or providing information to, any governmental agency or legislative body, any self-regulatory organization, and/or pursuant to the Sarbanes-Oxley Act; or (c) accepting any U.S. Securities and Exchange Commission awards.

- 12. Future Cooperation. Executive agrees to cooperate with the Company as may be reasonably requested by the Company (a) in the defense or prosecution of any claims or actions now in existence or which may be brought in the future against, by or on behalf of the Company or any of its affiliates; and (b) in connection with any inquiry regarding, investigation of or by, or legal action or complaint against the Company and/or any other Releasees that relates to events or occurrences that transpired while Executive was employed by the Company or about which Executive may otherwise have knowledge or information. The Company will reimburse Executive for reasonable business travel expenses and, if such cooperation occurs following the Term of the Consulting Agreement, will pay fees for time incurred by Executive in connection with providing cooperation under sub-parts (a) and (b) of this section, at a rate of \$500 per hour subject to receipt of appropriate documentation of such expenses.
- 13. No Admission of Wrongful Conduct. Executive hereby acknowledges and agrees that, by the Company providing the consideration described above and entering into this Separation Agreement, neither the Company nor any of the Releasees is admitting any unlawful or otherwise wrongful conduct or liability to Executive or Executive's heirs, executors, administrators, assigns, agents, or other representatives. The Company and Releasees expressly deny any liability or alleged violation of any law, contract or applicable policy.
- 14. Older Worker Benefit Protection Act Disclosure. Executive recognizes that as part of Executive's agreement to release any and all claims against the Releasees, Executive is releasing claims for age discrimination under the Age Discrimination in Employment Act. Accordingly, Executive has been afforded more than twenty-one (21) days from the date Executive received this Agreement to review this Agreement (the "Review Period") and Executive has an additional period of seven (7) days after executing this Agreement to revoke it in writing to the Company under the terms of the Older Worker Benefit Protection Act. Any negotiated changes to this Agreement shall not restart the Review Period or extend any deadline for Executive to execute and return the Agreement to the Company. If timely signed, dated and returned to the Company and not revoked prior to the end of the seven (7) day revocation period, this Agreement shall be effective upon the expiration of the seven (7) day revocation period (the "Effective Date"). By Executive's signature below, Executive represents and warrants that Executive has been advised to consult with an attorney of Executive's own choosing, that Executive has been given a reasonable amount of time to consider this Agreement, and that Executive is voluntarily and knowingly waiving any right she could possibly claim to review the Agreement for an additional period of time.
- 15. Taxes. The Company may withhold from any amounts payable under this Separation Agreement all federal, state, city or other taxes as the Company is required to withhold pursuant to any applicable law, regulation or ruling. Notwithstanding any other provision of this Separation Agreement (including the Consulting Agreement attached hereto), the Company shall

not be obligated to guarantee any particular tax result for Executive with respect to any payment provided to Executive hereunder (or the Consulting Agreement), the Company makes no representation or warranty regarding the tax treatment and/or tax consequences of any payment described herein (including the Consulting Agreement), and Executive shall be responsible for any taxes imposed on Executive with respect to any such payment. The foregoing shall not, however, be interpreted to limit the application of Section 5 of the Consulting Agreement, if applicable.

- 16. Section 409A. As applicable, the parties intend that this Agreement will be administered in accordance with Section 409A of the Internal Revenue Code of 1986, as amended (the "Code"). To the extent that any provision of this Agreement is ambiguous as to its compliance with Section 409A of the Code, the provision shall be read in such a manner so that all payments hereunder comply with Section 409A of the Code. Each payment pursuant to this Agreement is intended to constitute a separate payment for purposes of Treasury Regulation Section 1.409A2(b)(2). The parties agree that this Agreement may be amended, as reasonably requested by either party, and as may be necessary to fully comply with Section 409A of the Code and all related rules and regulations in order to preserve the payments and benefits provided hereunder without additional cost to either party. The Company makes no representation or warranty and shall have no liability to Executive or any other person if any provisions of this Agreement are determined to constitute deferred compensation subject to Section 409A of the Code but do not satisfy an exemption from, or the conditions of, such Section.
- 17. Governing Law. This Separation Agreement will be governed by and construed in accordance with the laws of the State of New York, without regard to principles of conflict of laws.
- 18. Statement Regarding Executive's Transition. Executive and the Company agree to coordinate in good faith on the terms of an internal statement with respect to Executive's separation from the Company and Executive's related engagement as an independent consultant.
- 19. Entire Agreement. Except as expressly set forth herein, this Separation Agreement (including Exhibit A and Exhibit B hereto), constitutes the entire agreement between the Company and Executive with respect to the subject matter of this Separation Agreement and supersedes and replaces any prior understandings, agreements or representations, written or oral, concerning the subject matter of this Agreement. The terms of Exhibit A and Exhibit B to this Agreement, as well as the Indemnification Agreement, the PICA, and the restrictive covenants set forth in the Employment Agreement, are incorporated herein by reference and are part of this Agreement as if set forth fully herein. This is an integrated document. By signing below, Executive acknowledges that she is not relying on any promises or representations by the Company or the agents, representatives, or attorneys of any of the entities within the definition of Company regarding any subject matter addressed in this Agreement.
- **20.** Severability. If any provision of this Agreement is held invalid, such invalidation shall not affect other provisions or applications of the Agreement which can be given effect

without the invalid provision or application, and to this end the provisions of this Agreement are declared to be severable. If, moreover, any one or more of the provisions (or portions thereof) contained in this Agreement shall for any reason be determined by a court of competent jurisdiction to be unenforceable because it is excessively broad as to duration or scope of prohibited activities, it shall be (to the extent permitted by law) construed by limiting and reducing it, so as to extend and be enforced only over the maximum duration and scope of activities as to which it may be enforceable under applicable law.

- **21. Modification.** This Agreement may be modified or amended only by a written instrument duly signed by each of the parties hereto or their respective successors or assigns. The parties' rights and obligations hereunder shall survive this Agreement becoming effective and Executive's termination of employment in accordance with their respective terms.
- 22. Waiver. No waiver of any breach of any term or provision of this Agreement shall be construed to be, nor shall be, a waiver of any other breach of this Agreement. No waiver shall be binding unless in writing and signed by the party waiving the breach.
- **23. Acceptance of Agreement.** Executive agrees that if Executive timely revokes this Agreement prior to the end of the seven (7) day revocation period, the provisions of this Agreement shall be void and of no further force and effect, and Executive will not be eligible to serve as an independent consultant of the Company or receive any of the compensation described in the Consulting Agreement.
- **24. Successors and Assigns**; Third-Party Beneficiaries. Executive may not assign this Separation Agreement. Executive hereby consents to the assignment by the Company of all of its rights and obligations hereunder to any successor to the Company by merger or consolidation or purchase of all or substantially all of the Company's assets. The Releasees are expressly intended to be third-party beneficiaries of this Separation Agreement and it may be enforced by each of them.
- 25. Knowing and Voluntary Agreement. By Executive's signature below, Executive represents and warrants that Executive (i) has been given a reasonable amount of time of at least twenty-one (21) days to review and consider this Agreement, (ii) is hereby advised in writing to consult with independent legal counsel in connection with this Agreement; (iii) is represented by counsel of Executive's choosing in connection with this matter; (iv) has read and reviewed this Agreement thoroughly and fully understands its terms and conditions and their significance and has discussed them with Executive's independent legal counsel, or has had a reasonable opportunity to have done so; and (v) agrees to all the terms and conditions of this Agreement and is signing this Agreement voluntarily and of Executive's own free will, with the full understanding of its terms, conditions and legal consequences, and with the intent to be bound hereby.
- **26. Counterparts.** This Separation Agreement may be executed in counterparts, each of which will be deemed an original but all of which together will constitute one and the same instrument. An originally executed version of this Separation Agreement that is scanned as an

image file (e.g., Adobe PDF, TIF, etc.) and then delivered by one party to the other party via electronic mail as evidence of signature, shall, for all purposes hereof, be deemed an original signature. In addition, an originally executed version of this Separation Agreement that is delivered via facsimile by one party to the other party as evidence of signature shall, for all purposes hereof, be deemed an original.					
[Signature Page Follows]					

**IN WITNESS WHEREOF**, Executive and a duly authorized representative of the Company hereby certify that they have read this Separation Agreement in its entirety and voluntarily executed it, as of the date(s) set forth under their respective signatures.

**EXECUTIVE** 

FORGE GLOBAL HOLDINGS, INC.

By: /s/ Jennifer Phillips

Jennifer Phillips

4/24/2025 **Date**  By: /s/ Kelly Rodriques
Name: Kelly Rodriques

**Title: Chief Executive Officer** 

4/24/2025 **Date** 

# **EXHIBIT A**

# AFFIRMATION OF GENERAL RELEASE OF CLAIMS

[To be signed and dated **no earlier than** the Separation Date and **no later than** three (3) business days after the Separation Date]

Pursuant to Section 2 of the attached Separation and General Release Agreement (the "Separation Agreement"), the undersigned, Jennifer Phillips, hereby re-affirms the provisions of Sections 4 and 14 of the Separation Agreement, including the general release of all claims, causes of action, demands, damages, costs, expenses, attorneys' fees, liabilities, or other losses set forth therein from the beginning of time through the date this Affirmation of General Release of Claims is executed.

JENNIFER PHILLIPS

Signature: /s/ Jennifer Phillips

Dated: 06/26/2025

## EXHIBIT B

## **CONSULTING AGREEMENT**

THIS CONSULTING AGREEMENT (the "Consulting Agreement") is made and entered by and between Jennifer Phillips ("Consultant") and Forge Global Holdings, Inc. (the "Company") (each a "Party" and collectively the "Parties").

**WHEREAS**, reference is hereby made to that certain Separation and General Release Agreement executed by the Parties and dated April 24, 2025 (the "Separation Agreement");

WHEREAS, this Consulting Agreement is conditioned upon the execution and non-revocation of the Separation Agreement pursuant to the terms and conditions of the Separation Agreement and execution of this Consulting Agreement before Separation Date, and if the Separation Agreement is not executed by Consultant or is revoked by Consultant in accordance with the terms and conditions of the Separation Agreement or this Consulting Agreement is executed after the Separation Date, this Consulting Agreement shall be void ab initio;

WHEREAS, capitalized terms not otherwise defined herein shall have the meaning ascribed to them in the Separation Agreement; and

WHEREAS, in consideration for Consultant executing the Separation Agreement, the Company wishes to engage Consultant as an independent consultant pursuant to this Consulting Agreement, and Consultant agrees to accept the Company's offer in regard to the same.

**NOW, THEREFORE**, in consideration of the mutual promises and covenants contained herein, Consultant and the Company agree as follows:

- 1. Term. The term of this Consulting Agreement and Consultant's Services hereunder shall commence effective as of June 27, 2025 and shall end on December 27, 2025, unless earlier terminated as provided in this Section 1 (the "Term"). The Term shall terminate prior to December 27, 2025 upon (a) Consultant's death or Disability; (b) Consultant's conviction of (A) a felony or (B) any misdemeanor involving moral turpitude, deceit, dishonesty or fraud; (c) Consultant's willful failure to perform Consultant's duties as defined in Section 2 for a period of ten (10) days after written notice from the Company; (d) Executive's material violation of the PICA that is not cured within ten (10) days of written notice to Consultant; or (e) the thirtieth (30th) day after the Company's receipt of Consultant's written notice of termination for any or no reason. As used herein, "Disability" means any physical or mental illness, injury, or other incapacitating condition that results in Consultant being unable to perform essential functions of Consultant's required services, with or without a reasonable accommodation, for a period of sixty (60) consecutive calendar days in any consecutive 12-month period.
- **2. Services.** In accordance with the terms and conditions of this Consulting Agreement, the Company hereby engages Consultant as an independent contractor to provide strategic advice as may be requested during the Term from time to time by the Company (the "Services") for a

maximum of (i) ten hours during the first two months of the Term and (ii) five hours for each month thereafter, at such times as Consultant may choose so as not to interfere with Consultant's other activities. Unused time in any given month may not be carried forward to a subsequent month. The Services may include, without limitation, providing strategic guidance, consultation, and advice to Company's executive leadership team.

Consultant may perform the Services at a location other than the offices of the Company. Unless otherwise expressly authorized in writing by the Company, Consultant may not delegate, assign or sub-contract any of the Services to any other individual or entity, and it is further agreed that the Company may limit the Services at any time, provided that it may not terminate this Consulting Agreement (other than as expressly provided elsewhere herein).

**3. Fees for Services.** In consideration for Consultant providing the Services during the Term, Consultant shall receive a cash payment in the aggregate amount equal to \$472,470.50, on a Form 1099 which shall be payable in equal monthly installments starting on the first day of July through December 2025 during the Term, provided, that upon the Consultant's death or Disability, any unpaid Consulting Fees would be immediately paid to the Consultant or her estate, as applicable (the "Consulting Fees").

# 4. Independent Contractor/Non-Employee Status.

- (a) Consultant and the Company hereby acknowledge, understand and agree that Consultant is an independent contractor. Nothing herein, explicitly or implicitly, shall be deemed or construed to create a joint venture, joint employer, partnership, agency, affiliate or employee/employer relationship between Consultant and the Company (or any of its parents, subsidiaries, affiliates or related entities) for any purpose including, but not limited to, taxes, compensation, employee benefits, political contributions or other charitable contributions. Consultant understands and agrees that during the Term and in connection with the performance of the Services hereunder and the payment of any and all Consulting Fees, Consultant will be solely responsible for: (i) complying with all federal, state and local laws, ordinances, regulations and orders with respect to the performance of the Services under this Consulting Agreement; and (ii) paying all federal, state and local taxes (including income tax, FICA, FUTA and other taxes that may be due) as a result of any and all Consulting Fees Consultant receives or is deemed to receive pursuant to this Consulting Agreement.
- (b) As an independent contractor, Consultant understands and agrees that Consultant will not accrue any benefits under, or in any way be covered by, benefit plans of the Company or any of its parents, subsidiaries, affiliates, or related entities.
- **5. Indemnification; Hold Harmless**. Consultant shall not be liable for (x) any act, omission, interpretation, construction or determination made in good faith in connection with the Services, or (y) any challenge to Consultant's independent contractor status, and Consultant shall be entitled in all cases to indemnification and reimbursement by the Company in respect of any claim, loss, damage or expense (including, without limitation, reasonable attorneys' fees) arising or resulting therefrom to the fullest extent permitted by law and/or under the Company's by-

laws, certificate of incorporation or any directors' and officers' liability insurance coverage which may be in effect from time to time and/or any indemnification agreement between such individual and the Company.

**6.** No Authority to Bind the Company. Consultant is not a representative, agent or affiliate of the Company and Consultant does not have the right or authority to assume or create any obligation of any kind, express or implied, on behalf of the Company (or any of its affiliates or related entities) or to bind the Company (or any of its affiliates or related entities) in any respect whatsoever, unless the Company gives express written authorization in advance to Consultant to act with such agency and authority.

## 7. Miscellaneous.

- (a) This Consulting Agreement, together with the Separation Agreement and exhibits thereto, contains the entire understanding of the Parties with respect to the matters contained herein, and supersedes all proposals and agreements, written or oral, and all other communications between the Parties relating to the subject matter of this Consulting Agreement. Notwithstanding anything herein to the contrary, all confidentiality, non-disclosure, non-disparagement, return of Company property obligations or covenants described in the Separation Agreement, or any documents incorporated by reference therein are hereby incorporated into this Consulting Agreement, are not superseded by this Consulting Agreement, and remain and shall remain in full force and effect through and following the Term hereof. In signing this Consulting Agreement, Consultant is not relying on any promise or representation of the Company except as expressly set forth herein.
- (b) Neither this Consulting Agreement nor any right or obligation hereunder or interest herein may be assigned or transferred by Consultant without the express written consent of the Company. The Company may assign this Consulting Agreement to its affiliates, successors and assigns, and Consultant expressly consents to such assignment.
- (c) This Consulting Agreement shall be governed by and construed in accordance with the laws of the State of New York without regard to conflict of laws or choice of law rules. Employee consents to personal and exclusive jurisdiction in New York County, New York.
- (d) This Consulting Agreement may not be modified or amended except in writing signed or executed by Consultant and the Company.
- (e) In case any provisions (or portions thereof) contained in this Consulting Agreement will, for any reason, be held invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability will not affect the other provisions of this Consulting Agreement, and this Consulting Agreement will be construed as if such invalid, illegal or unenforceable provision had never been contained herein. If, moreover, any one or more of the provisions contained in this Consulting Agreement shall for any reason be held to be excessively broad as to duration, geographical scope, activity or subject, it will be construed by limiting and reducing it, so as to be enforceable to the extent compatible with the applicable law as it will then appear.



**IN WITNESS WHEREOF**, the Parties to this Consulting Agreement, intending to be legally bound, have caused this Consulting Agreement to be executed as of the date indicated below.

		Acknowledged, Accepted and Agreed to:
Date:	By:	Name: Jennifer Phillips
		Forge Global Holdings, Inc.
Date:	Ву:	Name: Kelly Rodriques Title: Chief Executive Officer

#### CONSULTING AGREEMENT

THIS CONSULTING AGREEMENT (the "Consulting Agreement") is made and entered by and between Jennifer Phillips ("Consultant") and Forge Global Holdings, Inc. (the "Company") (each a "Party" and collectively the "Parties").

**WHEREAS**, reference is hereby made to that certain Separation and General Release Agreement executed by the Parties and dated April 24, 2025 (the "Separation Agreement");

WHEREAS, this Consulting Agreement is conditioned upon the execution and non-revocation of the Separation Agreement pursuant to the terms and conditions of the Separation Agreement and execution of this Consulting Agreement before Separation Date, and if the Separation Agreement is not executed by Consultant or is revoked by Consultant in accordance with the terms and conditions of the Separation Agreement or this Consulting Agreement is executed after the Separation Date, this Consulting Agreement shall be void ab initio;

WHEREAS, capitalized terms not otherwise defined herein shall have the meaning ascribed to them in the Separation Agreement; and

WHEREAS, in consideration for Consultant executing the Separation Agreement, the Company wishes to engage Consultant as an independent consultant pursuant to this Consulting Agreement, and Consultant agrees to accept the Company's offer in regard to the same.

**NOW, THEREFORE**, in consideration of the mutual promises and covenants contained herein, Consultant and the Company agree as follows:**NOW, THEREFORE**, in consideration of the mutual promises and covenants contained herein,

- 1. Term. The term of this Consulting Agreement and Consultant's Services hereunder shall commence effective as of June 27, 2025 and shall end on December 27, 2025, unless earlier terminated as provided in this Section 1 (the "Term"). The Term shall terminate prior to December 27, 2025 upon (a) Consultant's death or Disability; (b) Consultant's conviction of (A) a felony or (B) any misdemeanor involving moral turpitude, deceit, dishonesty or fraud; (c) Consultant's willful failure to perform Consultant's duties as defined in Section 2 for a period of ten (10) days after written notice from the Company; (d) Executive's material violation of the PICA that is not cured within ten (10) days of written notice to Consultant; or (e) the thirtieth (30th) day after the Company's receipt of Consultant's written notice of termination for any or no reason. As used herein, "Disability" means any physical or mental illness, injury, or other incapacitating condition that results in Consultant being unable to perform essential functions of Consultant's required services, with or without a reasonable accommodation, for a period of sixty (60) consecutive calendar days in any consecutive 12-month period.
- **2. Services.** In accordance with the terms and conditions of this Consulting Agreement, the Company hereby engages Consultant as an independent contractor to provide strategic advice

as may be requested during the Term from time to time by the Company (the "Services") for a maximum of (i) ten hours during the first two months of the Term and (ii) five hours for each month thereafter, at such times as Consultant may choose so as not to interfere with Consultant's other activities. Unused time in any given month may not be carried forward to a subsequent month. The Services may include, without limitation, providing strategic guidance, consultation, and advice to Company's executive leadership team.

Consultant may perform the Services at a location other than the offices of the Company. Unless otherwise expressly authorized in writing by the Company, Consultant may not delegate, assign or sub-contract any of the Services to any other individual or entity, and it is further agreed that the Company may limit the Services at any time, provided that it may not terminate this Consulting Agreement (other than as expressly provided elsewhere herein).

**3. Fees for Services.** In consideration for Consultant providing the Services during the Term, Consultant shall receive a cash payment in the aggregate amount equal to \$472,470.50, on a Form 1099 which shall be payable in equal monthly installments starting on the first day of July through December 2025 during the Term, provided, that upon the Consultant's death or Disability, any unpaid Consulting Fees would be immediately paid to the Consultant or her estate, as applicable (the "Consulting Fees").

# 4. Independent Contractor/Non-Employee Status.

- (a) Consultant and the Company hereby acknowledge, understand and agree that Consultant is an independent contractor. Nothing herein, explicitly or implicitly, shall be deemed or construed to create a joint venture, joint employer, partnership, agency, affiliate or employee/employer relationship between Consultant and the Company (or any of its parents, subsidiaries, affiliates or related entities) for any purpose including, but not limited to, taxes, compensation, employee benefits, political contributions or other charitable contributions. Consultant understands and agrees that during the Term and in connection with the performance of the Services hereunder and the payment of any and all Consulting Fees, Consultant will be solely responsible for: (i) complying with all federal, state and local laws, ordinances, regulations and orders with respect to the performance of the Services under this Consulting Agreement; and (ii) paying all federal, state and local taxes (including income tax, FICA, FUTA and other taxes that may be due) as a result of any and all Consulting Fees Consultant receives or is deemed to receive pursuant to this Consulting Agreement.
- (b) As an independent contractor, Consultant understands and agrees that Consultant will not accrue any benefits under, or in any way be covered by, benefit plans of the Company or any of its parents, subsidiaries, affiliates, or related entities.
- **5. Indemnification; Hold Harmless.** Consultant shall not be liable for (x) any act, omission, interpretation, construction or determination made in good faith in connection with the Services, or (y) any challenge to Consultant's independent contractor status, and Consultant shall be entitled in all cases to indemnification and reimbursement by the Company in respect of any claim, loss, damage or expense (including, without limitation, reasonable attorneys' fees) arising

or resulting therefrom to the fullest extent permitted by law and/or under the Company's by-laws, certificate of incorporation or any directors' and officers' liability insurance coverage which may be in effect from time to time and/or any indemnification agreement between such individual and the Company.

**6.** No Authority to Bind the Company. Consultant is not a representative, agent or affiliate of the Company and Consultant does not have the right or authority to assume or create any obligation of any kind, express or implied, on behalf of the Company (or any of its affiliates or related entities) or to bind the Company (or any of its affiliates or related entities) in any respect whatsoever, unless the Company gives express written authorization in advance to Consultant to act with such agency and authority.

#### 7. Miscellaneous.

- (a) This Consulting Agreement, together with the Separation Agreement and exhibits thereto, contains the entire understanding of the Parties with respect to the matters contained herein, and supersedes all proposals and agreements, written or oral, and all other communications between the Parties relating to the subject matter of this Consulting Agreement. Notwithstanding anything herein to the contrary, all confidentiality, non-disclosure, non-disparagement, return of Company property obligations or covenants described in the Separation Agreement, or any documents incorporated by reference therein are hereby incorporated into this Consulting Agreement, are not superseded by this Consulting Agreement, and remain and shall remain in full force and effect through and following the Term hereof. In signing this Consulting Agreement, Consultant is not relying on any promise or representation of the Company except as expressly set forth herein.
- (b) Neither this Consulting Agreement nor any right or obligation hereunder or interest herein may be assigned or transferred by Consultant without the express written consent of the Company. The Company may assign this Consulting Agreement to its affiliates, successors and assigns, and Consultant expressly consents to such assignment.
- (c) This Consulting Agreement shall be governed by and construed in accordance with the laws of the State of New York without regard to conflict of laws or choice of law rules. Employee consents to personal and exclusive jurisdiction in New York County, New York.
- (d) This Consulting Agreement may not be modified or amended except in writing signed or executed by Consultant and the Company.
- (e) In case any provisions (or portions thereof) contained in this Consulting Agreement will, for any reason, be held invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability will not affect the other provisions of this Consulting Agreement, and this Consulting Agreement will be construed as if such invalid, illegal or unenforceable provision had never been contained herein. If, moreover, any one or more of the provisions contained in this Consulting Agreement shall for any reason be held to be excessively broad as to

[Remainder of page intentionally left blank. Signature page(s) immediately follow.]						

**IN WITNESS WHEREOF**, the Parties to this Consulting Agreement, intending to be legally bound, have caused this Consulting Agreement to be executed as of the date indicated below.

Acknowledged, Accepted and Agreed to:

Date: <u>06/10/2025</u> By: <u>/s/ Jennifer Phillips</u>

Name: Jennifer Phillips

Forge Global Holdings, Inc.

Date: <u>06/10/2025</u> By: <u>/s/ Kelly Rodriques</u>

Name: Kelly Rodriques

Title: Chief Executive Officer

## FORGE GLOBAL HOLDINGS, INC.

## OUTSIDE DIRECTOR COMPENSATION POLICY

Forge Global Holdings, Inc. (the "Company") believes that providing cash and equity compensation to members of its Board of Directors (the "Board," and members of the Board, the "Directors") represents an effective tool to attract, retain, and reward Directors who are not employees of the Company (the "Outside Directors"). This Outside Director Compensation Policy (the "Policy") formalizes the Company's policy regarding cash compensation and grants of equity awards to its Outside Directors. Unless otherwise defined herein, capitalized terms used in this Policy will have the meaning given such term in the Company's 2022 Stock Option and Incentive Plan, as amended from time to time (the "Plan"), or if the Plan is no longer in use at the time of an equity award, the meaning given such term or any similar term in the equity plan then in place under which such equity award is granted. Each Outside Director will be solely responsible for any tax obligations incurred by such Outside Director as a result of the equity and cash payments such Outside Director receives under this Policy.

## 1. CASH COMPENSATION

Annual Cash Retainer

Each Outside Director will be paid an annual cash retainer of \$35,000. There are no per-meeting attendance fees for attending Board meetings.

Committee Annual Cash Retainer

Each Outside Director who serves as the Chairperson of the Board, the Lead Independent Director, or the Chair or a member of a committee of the Board will be eligible to earn additional annual fees (paid quarterly in arrears on a prorated basis) as follows:

Chairperson of the Board	\$30,000
Lead Independent Director	\$16,500
Chair of Audit Committee	\$20,000
Member of Audit Committee	\$10,000
Chair of Compensation Committee	\$12,000
Member of Compensation Committee	\$6,000
Chair of Nominating and Corporate Governance Committee	\$8,000
Member of Nominating and Corporate Governance Committee	\$4,000
Chair of Risk Committee	\$8,000
Member of Risk Committee	\$4,000

For clarity, each Outside Director who serves as the Chair of a committee will receive only the annual fee as the Chair of the committee and will not also receive the additional annual fee as a member of the committee. In addition to the above standing committees, the Board may also establish various ad hoc committees from time to time to accomplish a specific goal or purpose. Effective as of the initial effective date of this Policy (the "Effective Date"), Outside Directors who serve on such ad hoc committees may be compensated up to 1) \$1,500 per committee meeting attended, or such lesser amount as determined by the Board, or 2) additional annual fees (paid quarterly in arrears on a prorated basis) of up to \$10,000 for each ad hoc committee on which they serve, or such lesser amount as determined by the Board.

Outside Directors may also serve on the board of directors (or equivalent governing body) of any subsidiary of the Company from time to time (each, a "Subsidiary Board"). Effective as of the Effective Date, Outside Directors are eligible to earn additional annual fees (paid quarterly in arrears on a prorated basis) of up to \$20,000 for each Subsidiary Board on which they serve, or such lesser amount as determined by the Board.

# Payment

A quarterly portion of each annual cash retainer under this Policy will be paid quarterly in arrears on a prorated basis to each Outside Director who has served in the relevant capacity at any point during the immediately preceding fiscal quarter, and such payment shall be made no later than 30 days following the end of such immediately preceding fiscal quarter. For purposes of clarification, an Outside Director who has served as an Outside Director, or as a member of an applicable committee (or chair thereof) or Subsidiary Board during only a portion of the relevant Company fiscal quarter will receive a prorated payment of the quarterly payment of the applicable annual cash retainer(s), calculated based on the number of days during such fiscal quarter such Outside Director has served in the relevant capacities. For purposes of clarification, an Outside Director who has served as an Outside Director, or as a member of an applicable committee (or chair thereof) or Subsidiary Board, as applicable, during only a portion of the relevant Company fiscal quarter will receive a prorated payment of the quarterly payment of the applicable annual cash retainer(s), calculated based on the number of days during such fiscal quarter such Outside Director has served in the relevant capacities.

# 2. EQUITY COMPENSATION

Outside Directors will be eligible to receive all types of Awards (except Incentive Stock Options) under the Plan (or the applicable equity plan in place at the time of grant), including discretionary Awards not covered under this Policy. All grants of Awards to Outside Directors pursuant to Section 2 of this Policy will be made in accordance with the following provisions, except as otherwise provided herein:

<u>a. No Discretion.</u> No person will have any discretion to select which Outside Directors will be granted any Awards under this Policy or to determine the number of shares of Stock ("Shares") to be covered by such Awards.

b. Initial Award. Each individual who first becomes an Outside Director beginning from and following the Amendment Date will be granted a one-time Award of Restricted Stock Units covering Shares that have a grant date Fair Market Value equal to \$240,000 (each, an "Initial Award"). Each Initial Award will be granted as soon as practicable by the Board after the date on which such individual first becomes an Outside Director (the "Start Date"), whether through election by the stockholders of the Company or appointment by the Board to fill a vacancy. If an individual was a member of the Board and also an employee, becoming an Outside Director due to termination of employment will not entitle the Outside Director to an Initial Award. Subject to Section 3 of this Policy, 1/12th of the Shares subject to each Initial Award will vest and settle beginning on the first Quarterly Vesting Date occurring after the applicable Outside Director's Start Date and each Quarterly Vesting Date thereafter, until the Initial Award is fully vested, in each case subject to the Outside Director maintaining a Service Relationship through the applicable vesting date. For the purposes of this Policy, Quarterly Vesting Dates are defined as March 1, June 1, September 1, and December 1 of a given year.

c. Annual Award. On the date of each annual meeting of the Company's stockholders (each, an "Annual Meeting"), each Outside Director will be granted Restricted Stock Units (each, an "Annual Award") covering Shares that have a grant date Fair Market Value equal to \$170,000. Subject to Section 3 of this Policy, each Annual Award will vest and settle as to 1/4th of the Shares subject to the Annual Award beginning on the first Quarterly Vesting Date occurring after the date of the applicable Annual Meeting and each Quarterly Vesting Date thereafter, until the Annual Award is fully vested, in each case subject to the Outside Director maintaining a Service Relationship through the applicable vesting date.

d. Prorated Award. Effective as of the Effective Date, any Outside Director whose Start Date is not at an Annual Meeting is eligible to be granted Restricted Stock Units (each, a "Prorated Award") covering Shares that have a grant date Fair Market Value equal to the product of \$170,000 multiplied by a fraction, (i) the numerator of which is the number of fully completed days between the applicable Start Date and the date of the first Annual Meeting to occur after such individual first becomes an Outside Director, and (ii) the denominator of which is 365, with any resulting fraction rounded down to the nearest whole Share. An Outside Director who is first appointed to the Board at an Annual Meeting is entitled to receive an Initial Award and Annual Award, but is not eligible to receive a Prorated Award. For the avoidance of doubt, the Board may determine not to grant a Prorated Award to an eligible Outside Director or grant a Prorated Award of a lesser value, with such determination to be in the Board's sole and absolute discretion. Subject to Section 3 of this Policy, all Shares subject to each Prorated Award will vest and settle beginning on the first Quarterly Vesting Date occurring after the grant date of the Prorated Award, in each case subject to the Outside Director maintaining a Service Relationship through the applicable vesting date.

e. Fair Market Value. For purposes herein, "Fair Market Value" shall have the meaning set forth in the Plan, provided that, in lieu of referencing the Company's closing price on such date, "Fair Market Value" shall be determined on the basis of the trailing 20 trading day period ending on the last day immediately prior to the grant date, unless determined by the Board at or prior to the time of grant.

## 3. SALE EVENT

In the event of a Sale Event, each Outside Director will fully vest in his or her outstanding Company equity awards, including any Initial Awards or Annual Awards, and all restrictions on Restricted Stock Units will lapse provided that the Outside Director continues to be an Outside Director through such date.

## 4. ANNUAL COMPENSATION LIMIT

No Outside Director may be paid, issued, or granted, in any fiscal year of the Company ("Fiscal Year"), cash compensation and equity awards (including any Awards issued under this Plan) with an aggregate value greater than \$1,000,000 for such Outside Director's first year of service or \$750,000 in any subsequent year (with the value of each equity award based on its grant date fair value (determined in accordance with U.S. generally accepted accounting principles)). Any cash compensation paid or Awards granted to an individual for his or her services as a Company employee, or for his or her services as a Consultant (other than as an Outside Director), will not count for purposes of the limitation under this Section 4.

## 5. TRAVEL EXPENSES

Each Outside Director's reasonable, customary, and documented travel expenses to Board meetings will be reimbursed by the Company.

## 6. ADDITIONAL PROVISIONS

All provisions of the Plan not inconsistent with this Policy will apply to Awards granted to Outside Directors.

## 7. ADJUSTMENTS

In the event that any dividend or other distribution (whether in the form of cash, Shares, other securities, or other property), recapitalization, stock split, reverse stock split, reorganization, merger, consolidation, split-up, spin-off, combination, repurchase, or exchange of Shares or other securities of the Company, or other change in the corporate structure of the Company affecting the Shares occurs, the Administrator, in order to prevent diminution or enlargement of the benefits or potential benefits intended to be made available under this Policy, will adjust the number of Shares issuable pursuant to Awards granted under this Policy.

# 8. SECTION 409A

In no event will cash compensation or expense reimbursement payments under this Policy be paid after the later of (i) the 15th day of the 3rd month following the end of the Fiscal Year in which the compensation is earned or expenses are incurred, as applicable, or (ii) the 15th day of the 3rd month following the end of the calendar year in which the compensation is earned

or expenses are incurred, as applicable, in compliance with the "short-term deferral" exception under Section 409A of the Internal Revenue Code of 1986, as amended, and the final regulations and guidance thereunder, as may be amended from time to time (together, "Section 409A"). It is the intent of this Policy that this Policy and all payments hereunder be exempt from or otherwise comply with the requirements of Section 409A so that none of the compensation to be provided hereunder will be subject to the additional tax imposed under Section 409A, and any ambiguities or ambiguous terms herein will be interpreted to be so exempt or comply. In no event will the Company reimburse an Outside Director for any taxes imposed or other costs incurred as a result of Section 409A.

## 9. REVISIONS

The Board may amend, alter, suspend, or terminate this Policy at any time and for any reason. No amendment, alteration, suspension, or termination of this Policy will materially impair the rights of an Outside Director with respect to compensation that already has been paid or awarded, unless otherwise mutually agreed between the Outside Director and the Company. Termination of this Policy will not affect the Board's or the Compensation Committee's ability to exercise the powers granted to it under the Plan with respect to Awards granted under the Plan pursuant to this Policy prior to the date of such termination.

Unless otherwise required by applicable law, this Policy shall not be subject to approval by the Company's stockholders, including, for the avoidance of doubt, as a result of or in connection with an action taken with respect to this Policy as contemplated in Section 9 hereof.

This Policy was initially adopted June 1, 2022, with an initial effective date of March 21, 2022. It was then amended on December 10, 2024 and June 19, 2025.

## CERTIFICATION OF PERIODIC REPORT UNDER SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

## I, Kelly Rodriques, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Forge Global Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 7, 2025

By: /s/ Kelly Rodriques
Kelly Rodriques

Chief Executive Officer (Principal Executive Officer)

## CERTIFICATION OF PERIODIC REPORT UNDER SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

## I, James Nevin, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Forge Global Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 7, 2025 By: /s/ James Nevin

James Nevin

Chief Financial Officer (Principal Financial Officer)

# CERTIFICATIONS OF CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Kelly Rodriques, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report on Form 10-Q of Forge Global Holdings, Inc. for the fiscal quarter ended June 30, 2025 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended, and that the information contained in such Quarterly Report on Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of Forge Global Holdings, Inc.

Date: August 7, 2025

By: /s/ Kelly Rodriques

Kelly Rodriques

Chief Executive Officer (Principal Executive Officer)

I, James Nevin, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report on Form 10-Q of Forge Global Holdings, Inc. for the fiscal quarter ended June 30, 2025 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended, and that the information contained in such Quarterly Report on Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of Forge Global Holdings, Inc.

Date: August 7, 2025 By: /s/ James Nevin

James Nevin

Chief Financial Officer (Principal Financial Officer)